## Southampton

## University of Southampton Research Repository

Copyright © and Moral Rights for this thesis and, where applicable, any accompanying data are retained by the author and/or other copyright owners. A copy can be downloaded for personal non-commercial research or study, without prior permission or charge. This thesis and the accompanying data cannot be reproduced or quoted extensively from without first obtaining permission in writing from the copyright holder/s. The content of the thesis and accompanying research data (where applicable) must not be changed in any way or sold commercially in any format or medium without the formal permission of the copyright holder/s.

When referring to this thesis and any accompanying data, full bibliographic details must be given, e.g.

Thesis: Davide Bufalini (2023) "Black Holes in String Theory: Beyond Supergravity.", University of Southampton, Faculty of Social Sciences, Department of Applied Mathematics, PhD Thesis.

Data: Davide Bufalini (2023) Black Holes in String Theory: Beyond Supergravity.

## **UNIVERSITY OF SOUTHAMPTON**

Faculty of Social Sciences Department of Applied Mathematics

## Black Holes in String Theory: Beyond Supergravity.

by

## Davide Bufalini

ORCiD: 0000-0001-8700-4394

A thesis for the degree of Doctor of Philosophy

September 2023

#### University of Southampton

Abstract

Faculty of Social Sciences Department of Applied Mathematics

Doctor of Philosophy

## Black Holes in String Theory: Beyond Supergravity.

by Davide Bufalini

The aim of this thesis is to demonstrate the effectiveness of worldsheet string theory in the context of black hole physics, going beyond the supergravity approximation. For a certain class of microstates, which in a particular limit admit an  $\alpha'$ -exact description in terms of gauged Wess-Zumino-Witten (gWZW) models, we will derive consistency conditions, spectrum of massless excitations, and compute an extensive set of correlation functions. In particular, we will show how the consistency of the worldsheet conformal field theory (CFT) is in one-to-one correspondence with the absence of closed timelike curves, singularities, and horizon of the background. These conditions give rise to a set of quantisation constraints on the gauging parameters that will allow us to prove that the family of microstates we consider is the most general solution of our gWZW model. In addition, we will construct the physical states in the full gauged model and in its three-dimensional anti-de Sitter  $(AdS_3)$  limit. We will also derive a formula that expresses heavy-light correlators involving an arbitrary number of massless fields in the AdS<sub>3</sub> limit of the coset model in terms of AdS<sub>3</sub>  $\times$  S<sup>3</sup> vacuum correlator. Finally, we prove a conjecture on three-point functions in the  $SL(2, \mathbb{R})$  WZW model involving spectrally flowed vertex operators. These results constitute an important contribution for the understanding of black hole microstates beyond the usual supergravity approximation, and for AdS<sub>3</sub>/CFT<sub>2</sub> holography at finite 't Hooft coupling.

## Contents

Li	st of ]	Figures	۶ ۱	ix
Li	st of [	Tables		xi
D	eclara	tion of	Authorship	xiii
A	cknov	vledge	ments	xv
1	Intr	oductio	)n	1
	1.1	Black	holes and paradoxes	2
		1.1.1	Black hole thermodynamics	4
		1.1.2	The Entropy Puzzle	5
		1.1.3	The Information Paradox	5
	1.2	Outlir	ne of the thesis	10
2	Sup	ergravi	ity, D1-D5 CFT, and worldsheet string theory	13
	2.1	Super	gravity and compactifications	13
		2.1.1	Type IIB Supergravity	13
		2.1.2	Type IIB on $\mathbb{T}^5$ : maximal 5D $\mathcal{N} = 8$ Supergravity $\ldots \ldots \ldots$	15
		2.1.3	Type IIB on $\mathbb{T}^4$ : maximal 6D $\mathcal{N} = (2, 2)$ Supergravity	16
		2.1.4	Attracted supergravity moduli space	19
		2.1.5	Half-maximal and quarter-maximal 6D Sugra	21
	2.2	Cvetic	e-Youm solution and its limits	22
		2.2.1	Cvetic-Youm black hole	23
		2.2.2	The extremal rotating D1-D5-P black hole	24
		2.2.3	The BMPV black hole	24
	2.3	The D	1-D5(-P) black hole	25
		2.3.1	The naive D1-D5 small black hole	26
		2.3.2	The NS5-F1 system	30
		2.3.3	Comparing the D1-D5 and the NS5-F1 system and their moduli .	31
		2.3.4	Bulk microstates of the 2-charge system	32
		2.3.5	The D1-D5-P large black hole	34
	2.4	From	the brane system to the D1-D5 CFT	35
		2.4.1	The instanton description	36
		2.4.2	The gauge theory description	37
		2.4.3	The D1-D5 superconformal field theory	41
		2.4.4	Attracted moduli space and moduli space of instantons	42
	2.5	The h	olographic CFT	44

		2.5.1	Chiral primaries in the D1D5 CFT	46
	2.6	Supers	string theory on $\mathrm{AdS}_3 \times \mathbb{S}^3 \times \mathbb{T}^4$	49
		2.6.1	Bosonic WZW model for $SL(2,\mathbb{R})$	49
		2.6.2	Bosonic WZW model for SU(2)	53
		2.6.3	Superstrings in $AdS_3 \times S^3 \times T^4$	55
		2.6.4	Vertex operators and two-point functions	57
			2.6.4.1 NS sector	58
			2.6.4.2 Ramond sector	62
			2.6.4.3 Holographic dictionary for light chiral primaries	65
	2.7	Null-g	auged WZW models	68
		2.7.1	Null-gauged sigma models	69
		2.7.2	Null-gauged WZW models	70
		2.7.3	The models we study	71
		2.7.4	Supergravity fields	75
•	<b>D1</b>			-
3	Blac	k hole	microstates from the worldsheet	79
	3.1	Overv	1ew of the JMak1 cosets	/9
	3.2		Stency of the worldsneet spectrum	82
		3.2.1	Superstring Theory in hull-gauged models	83
	2.2	3.2.2 A ra a lar	Physical spectrum and consistency conditions	00
	3.3		Sis of the supergravity backgrounds	92
		3.3.1	Absence of beginning	92
		3.3.2	Absence of nonzons	93
		3.3.3	Smoothness and quantisation	94
	2.4	S.S.4	ing to MoDT and two charge limits	90
	5.4		MaRT matrix and <i>R</i> field	97
		3.4.1	The dilaton	97
		3.4.Z		100
		3.4.5	Two charge limits and novel non BPS NS5 P solutions	101
		3.4.4	AdSe limit and holography	102
		316	From AdS back to the linear dilaton background	104
	35	Discus	ssion	105
	0.0	Discut	551011	100
4	Wor	ldshee	t Correlators in Black Hole Microstates	109
	4.1	Null-g	auged description and worldsheet spectrum	113
		4.1.1	BRST quantization	113
		4.1.2	The unflowed NS sector	115
		4.1.3	The unflowed R sector	118
		4.1.4	Flowed/winding sectors	123
	4.2	Novel	heavy-light correlators from the worldsheet	125
		4.2.1	Light states in the $AdS_3$ regime $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	125
		4.2.2	Identifying the spacetime modes	126
		4.2.3	HLLH correlators: first example	128
		4.2.4	Non-BPS HLLH correlators for $h = \frac{1}{2}$	132
	4.3	More	general heavy-light correlators	134
		4.3.1	HLLH correlators for general $h$	134

		4.3.2	Matching between worldsheet and symmetric product orbifold .	137
		4.3.3	Higher-point heavy-light correlators	139
		4.3.4	An HLLLH correlator in worldsheet and holographic CFT	143
		4.3.5	Hawking radiation from the worldsheet	145
	4.4	Discu	ssion	148
5	A pi	roof for	string three-point functions in AdS <sub>3</sub>	153
	5.1	Conve	entions and brief review of the conjecture	155
		5.1.1	Vertex operators	155
		5.1.2	Recursion relations among correlators	157
		5.1.3	The conjectured solution	158
	5.2	The p	roof for <i>y</i> -basis three-point functions	161
		5.2.1	Holomorphic covering maps and odd parity correlators	162
		5.2.2	Series identification and even parity correlators	165
		5.2.3	Edge cases	169
			5.2.3.1 Even edge cases	171
			5.2.3.2 Odd edge cases	173
		5.2.4	Three-point functions with unflowed insertions	174
		5.2.5	Normalization	175
	5.3	Discu	ssion	176
6	Con	clusior	15	179
References				

## **List of Figures**

2.1	Summary of the relation between the various moduli spaces. The mod-	
	uli space $\mathcal{M}_{Sugra}$ and $\mathcal{M}_{Gauge}$ flow to the same 'attracted' moduli space	
	$\mathcal{M}^*_{Sugra} \equiv_{loc} \mathcal{M}_{SCFT}$ , where the identification is understood to be lo-	
	cal. The latter is related to the Hyperkähler structure of the (non-trivial	
	part of the) Higgs branch $\mathcal{M}_{Inst}$ , which is seen as a compact smooth Hy-	
	perkähler resolution of the orbifold $\text{Symm}_{n_1n_5}(\tilde{\mathbb{T}}^4)$ .	43
2.2	A very qualitative cartoon of the relation between various points in mod-	
	uli space, according to the AdS/CFT duality. The figure emphasizes that,	
	despite having $\mathcal{M}^*_{Sugra} \equiv_{loc} \mathcal{M}_{SCFT}$ , quantities computed in regime of	
	low spacetime curvature can be compared to those obtained at orbifold	
	point only if protected, and viceversa. Note that, in a more accurate	
	representation, the marked points may actually reside at cusps or at the	
	boundary of the moduli space.	43
	•	

## **List of Tables**

2.1	The re-arrangement of the 27 $U(1)$ charges in Type IIB Sugra on $\mathbb{T}^4$ is	
	shown. In the first column, we report the transformation properties	
	of the fields under Lorentz transformations for an observer in the non-	
	compact directions. In the fourth column, the letter <i>m</i> in $B_{u5}^{(m)}$ , $C_{u5}^{(m)}$ stands	
	for "magnetic". In the last column, the mass scaling with $\ell_s$ is shown	
	(in the decoupling limit). The non-compact and circle indices run over	
	$\mu = 0, \dots, 5$ . The fifth direction is that of the distinguished $\mathbb{S}^1_{\nu}$ . The torus	
	directions are denoted by $i = 6, \ldots, 9$ .	18
2.2	The 20 moduli parametrising the attracted moduli space of $\mathbb{T}^4$ compact-	
	ifications of IIB supergravity. Note that all the moduli fields are singlet	
	under $SU(2)_{L,R}$ .	20
2.3	Dictionary between worldsheet vertex operators and chiral primaries of	
	the holographically dual CFT. Here $j_{\omega} = j + \frac{n_5}{2}\omega, j = 1, \frac{3}{2}, \dots, \frac{n_5}{2}$ , and	
	$\omega = 0, 1, \dots, \dots,$	66

## **Declaration of Authorship**

I declare that this thesis and the work presented in it is my own and has been generated by me as the result of my own original research.

I confirm that:

- 1. This work was done wholly or mainly while in candidature for a research degree at this University;
- 2. Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated;
- 3. Where I have consulted the published work of others, this is always clearly attributed;
- 4. Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work;
- 5. I have acknowledged all main sources of help;
- 6. Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself;
- 7. Parts of this work have been published as: [1, 2, 3, 4]

Signed:..... Date:.....

## Acknowledgements

This acknowledgements section offers an opportunity for me to try in expressing my profound gratitude to the remarkable people I've had the privilege to encounter throughout the past four years of my life. I genuinely hope for my words to aptly reflect my emotions.

I hold immense gratitude towards my supervisor, David Turton, starting from the unforgettable day he offered me the opportunity to join his research, all the way to the final moments of my PhD journey. He guided me with patience, intelligence, and wisdom, allowing me to learn from his technical expertise and profound physical intuitions. Furthermore, I can assert without a doubt that David stands as the most understanding PhD advisor I know. There is much to learn and admire: from his impeccable conduct to his invaluable suggestions, from his meticulous attention to detail to his unwavering support during the concluding months of my PhD. I am sure that future students will consider themselves as fortunate as I do. Thank you very much, David.

I want to express my heartfelt gratitude to Nicolas Kovensky for being a good friend in difficult times, the most exceptional collaborator I could have ever envisioned, and an unofficial 'second supervisor'. My admiration and gratitude are beyond words. I will never forget his astounding technical abilities, his incredible patience and optimism in solving very hard problems. His dedication to research and his genuine passion are truly inspirational. The knowledge I gained from him has significantly enriched my thesis; without his guidance, it would be only half its current length and much less interesting. Nico, it has been an immense pleasure to work together at any time of the day in Southampton or Paris, to brainstorm and stare at a blackboard, to do hard computations, to get frustrated and then satisfied; in other words, it has been a pleasure to do good research with you. I would also like to extend my gratitude to Sergio Iguri, who has proven to be an exceptional collaborator and an extraordinary individual. Every interaction has been characterized by humor, comfort, and informality, making each meeting a moment to look forward to. Thank you Sergio, I will never forget when you tried to help me in shaping my future.

A great journey has always an essential ingredient: an amazing company. These years, I have had the immense fortune to share great memories with Federico Capone, Michele Santagata, and Sami Rawash, my 'academic older brother'. I will never forget the music, the conversations, the fun, and all the excellent questions on string theory and general physics. You all made me realise what it means to really 'understand' something. I also want to thank Stephanie Baines for her friendship, trust, support, understanding, and remarkably insightful advice across various facets of life. I also want to thank Luigi Bobbio, Alex Davey, Daniel Hasenbichler, Filip Landgren, Tristan Madeleine, David Sola Gil, Enrico Parisini, Savvas Pitsinigkos.

Grazie mille to the people in Rome: Massimo Bianchi, my 'academic father' for whom my admiration will always be infinite (and non-renormalisable), Dario Consoli, Alfredo Grillo, Salvo Mancani, Giancarlo Pradisi, Fabio Riccioni, Raffaele Savelli. In Southampton, my gratitude goes to Oscar Campos Dias, Kostas Skenderis, Marika Taylor, and Benjamin Whiters.

I extend my gratitude to those researchers with whom I've had the privilege of sharing delightful moments and engaging in stimulating discussions about physics: Ivano Basile, Iosif Bena, Federico Carta, Nejc Čeplak, Andrea Dei, Lorenz Eberhardt, Francesco Galvagno, Bogdan Ganchev, Stefano Giusto, Bin Guo, Shaun Hampton, Matthew Heydman, Anthony Houppe, Bob Knighton, Yixuan Li, Carlo Maccaferri, Emil Martinec, Stefano Massai, Samir Mathur, Prahar Mitra, Alessandro Sfondrini, Chiara Toldo, Mritunjay Verma, Jakub Vošmera, and Nick Warner.

I hold a special gratitude for my grandmother, whose daily actions have instilled in me the values of diligence, hard work, and dedication. She is, and always will be, a role model for the rest of my life. I am grateful to my mother, for being always patient with me, and dealing with some aspects of my character. Finally, I want to thank Nina. Her presence has been a guiding light, providing comfort and understanding during very challenging times, while being a source of joy and unconditional love for nearly three years. Her patience and empathy have motivated me while experiencing moments of frustration and self-doubt, especially in these last months. I also want to express my gratitude to her for patiently listening to me during countless late evenings, as we discussed topics ranging from black holes and holography, to string theory. Her willingness to engage in these conversations has allowed me to share my enthusiasm freely. Thank you as well for being an essential part of my academic journey. The fact that you are familiar with the names mentioned in these acknowledgments highlights the depth of your care and involvement in this significant aspect of my life. Thank you for being such a truly wonderful gift.

Thank you to each and every one of you, from the bottom of my heart.

"The fact that we humans, who are ourselves mere collections of fundamental particles of nature, have been able to come this close to an understanding of the laws governing us in our universe, is a great triumph.

I want to share my excitement and enthusiasm about this quest.

So, remember to look at the stars and not down at your feet. Try to make sense of what you see and wonder about what makes the universe exists.

#### Be curious.

And however difficult life may seem, there is always something you can do, and succeed at. It matters that you don't just give up. There are no limits to the human spirit: I believe what makes us unique is trascending our limits.

We are here together and we need to live together with tolerance and respect. We must become global citizens.

I have been enormously privileged through my work to be able to contribute to our understanding of the universe, but it would be an empty universe indeed if it were not for the people I love and who love me.

We are all time travelers, journeying together into the future but let us work together to make that future a place we want to visit.

Be brave, be determined, overcome the odds!

#### It can be done."

Stephen William Hawking. 08/01/1942 - 14/03/2018

## Chapter 1

## Introduction

The hypothesis of the existence of a 'theory of everything' is undoubtedly one of the most fascinating concepts in science: a unique and coherent theoretical framework capable of explaining all the measurable phenomena in our universe. Despite its status of hypothesis, expectations of its existence based on past experimental and theoretical achievements are supported by Maxwell's discovery of the electromagnetic theory and that of the electroweak model made possible by the combined efforts of many theoretical physicists, including the 1979 Nobel prize winners S. Glashow, A. Salam and S. Weinberg.

Once the strong nuclear force is taken into account, our best model for the description of non-gravitational interactions is nowadays the Standard Model of Particle Physics. Despite its incredible successes, it is well known and accepted in the scientific community that a better model should be formulated: in addition to not considering gravity, the Standard Model does not explain neutrino masses, does not take into account the matter-antimatter asymmetry, and does not provide any information about the nature of dark matter and dark energy.

On the other hand, gravitational phenomena are incredibly well described by Einstein's theory of general relativity. Indeed, LIGO's first direct observation of gravitational waves [5] followed a few years later by the publication of the first image of a black hole by the Event Horizon Telescope Collaboration (EHT) [6] represent incredible milestones for the validation of Einstein's theory of gravity. Thanks to LIGO and the EHT collaborations, we have experimental evidence of the theoretical prediction of general relativity, within its regime of validity. However, also in this case, there are reasons to believe that this theory should not be the ultimate description of gravitational interactions when short distance physics or quantum mechanics is taken into account. Notorious problems in such instances include the puzzling description of cosmological and black holes' curvature singularities, and the paradoxes associated to black holes.

The current understanding of our universe is unsatisfactory and humbling. It would thus be desirable to formulate a framework in which all the known fundamental forces can be unified in a consistent fashion, while resolving the aforementioned problems. String theory is so far the only candidate for such a remarkable endeavor. String theory is a consistent theoretical framework in which gravity and (non-)abelian gauge theories naturally emerge from the quantisation of the theory on the worldsheet of closed and open strings, respectively. In addition to fundamental strings, D-branes and NS5branes are essential constituents: black holes in string theory are low-energy descriptions of bound states of (possibly excited) strings and branes at strong coupling. It is precisely because of this large set of degrees of freedom, together with the extended nature of strings, that the paradoxes and puzzles arising in quantum descriptions of black holes can be tackled and, possibly, solved.

#### **1.1** Black holes and paradoxes

Classically, the 'No Hair Theorem'<sup>1</sup> holds [8]: every four-dimensional, asymptotically flat, black hole is described *uniquely* in terms of three parameters, namely the mass M, the angular momentum J and the charge<sup>2</sup> Q. Once we specify these three numbers, the black hole is completely determined in a unique way. Using this fact we can immediately note that the entropy associated to such a system is  $S = k_B \ln(1) = 0$ , because for an observer at asymptotic infinity the number of accessible microstates  $\Omega$  possessing those numbers is exactly 1. In addition, classical black holes have no entropy S = 0and their temperature is zero<sup>3</sup> T = 0. In 1973, J. Bekenstein noticed [9] that the laws of thermodynamics are completely analogous to the laws that govern black holes: as an example, the area of the horizon is never decreasing precisely as the entropy of a thermodynamical system. This means that we can *formally* associate, even in the classical case, an entropy and a temperature to a black hole. Quoting Bekenstein:

"It is then natural to introduce the concept of black-hole entropy as the measure of the inaccessibility of information (to an exterior observer) as to which particular internal configuration of the black hole is actually realized [...]."

Unfortunately, classically we can not have a non-trivial entropy and temperature for a black hole, hence it seemed that this analogy was simply formal. However, taking into account quantum effects one can solve this puzzle: in 1974 Hawking [10, 11] was able to show that a black hole can emit a thermal radiation<sup>4</sup>, thus implying that we

<sup>&</sup>lt;sup>1</sup>More precisely, the "hairs" are fields associated to a stationary black hole and, today, we should speak about the "No Scalar-Hair Conjecture" [7].

<sup>&</sup>lt;sup>2</sup>More generally, multiple electric and magnetic charges are allowed.

<sup>&</sup>lt;sup>3</sup>This is often stated as "classical black holes are really black", they do not emit radiation.

<sup>&</sup>lt;sup>4</sup>The radiation is thermal but not Planckian since it is modified by the absorption cross section.

can associate a temperature to a black hole. In addition, the entropy is proportional to the area of the black hole horizon, in Planck units. Hawking's original calculation [11] showed that the average number of outgoing particles, in a fixed frequency mode  $\omega$ , is distributed in accordance with a thermal spectrum

$$\langle N \rangle \sim \frac{1}{e^{\frac{\hbar\omega}{k_BT}} - 1}$$
 (1.1)

In addition, later it was shown [12, 13] that the full probability distribution (and not just the average number) is that of a thermal radiation.

The particle emission from the black hole is in agreement with the 'Generalised Second Law' of thermodynamics, which states that:

$$\frac{d}{dt}\left(S_{BH}+S_{rad}\right)\geq0\,,\tag{1.2}$$

where  $S_{rad}$  is the entropy of the emitted radiation and  $S_{BH}$  is the **B**ekenstein-Hawking entropy of a **B**lack Hole

$$S_{BH} = \frac{k_B c^3}{\hbar G_N} \frac{A_H}{4} = \frac{A_H}{4},$$
 (1.3)

where  $A_{\mathcal{H}}$  is the surface area of the horizon,  $k_B$  is the Boltzmann constant, c is the speed of light,  $\hbar$  is the reduced Planck constant and  $G_N$  is the Newton constant. In the second equality we have set  $\hbar = c = k_B = G_N = 1$ , a convention we will adopt for the rest of the thesis unless otherwise stated. The fact that the entropy is proportional to the area of the black hole, and not proportional to its volume, is remarkable. This led to the idea of the Holographic Principle [14, 15], and one explicit realization of holography is the celebrated AdS/CFT duality [16, 17, 18]. Note that the above discussion implies that the dimension of the Hilbert space associated to the microstates of the black hole (within the full Quantum Gravity theory) is

$$\dim(\mathcal{H}_{\rm micro}) \sim e^{S} \,, \tag{1.4}$$

with possible subleading corrections omitted. This is the expected number of microstates for a black hole with entropy *S*. As an example, the Bekenstein-Hawking entropy of a Schwarzschild black hole of mass *M* is

$$S_{
m Schw} \sim 10^{76} \left(rac{M}{M_{
m sun}}
ight)$$
 , (1.5)

which is very large. The discrepancy between the classical and the quantum prediction is enormous. The Hawking temperature of a Schwarzschild black hole is

$$T_H = \frac{\hbar c^3}{k_B} \frac{1}{8\pi G_N M} \sim 6 \times 10^{-8} \left(\frac{M_{\rm sun}}{M}\right) K.$$
 (1.6)

The bigger the black hole, the lower its temperature. More generally the temperature is given in terms of the surface gravity  $\kappa$  as

$$T_H = \frac{\hbar c^3}{G_N k_B} \frac{\kappa}{2\pi} = \frac{\kappa}{2\pi} , \qquad (1.7)$$

where the surface gravity  $\kappa$  is defined by

$$\kappa k^{\mu} = k^{\nu} \nabla_{\nu} k^{\mu} \bigg|_{\mathcal{H}}, \qquad (1.8)$$

where  $k^{\mu}$  is the Killing vector associated with the Killing horizon. As an example, the surface gravity for a four-dimensional Kerr-Newmann black hole is

$$\kappa = \frac{r_{+} - r_{-}}{2(r_{+}^{2} + a^{2})} = \frac{\sqrt{M^{2} - Q^{2} - \frac{J^{2}}{M^{2}}}}{2M^{2} - Q^{2} + 2M\sqrt{M^{2} - Q^{2} - \frac{J^{2}}{M^{2}}}},$$
(1.9)

where

$$r_{\pm} = M \pm \sqrt{M^2 - Q^2 - a^2}$$
,  $a = \frac{J}{M}$ . (1.10)

#### **1.1.1 Black hole thermodynamics**

Black hole mechanics is governed by the following four laws, similar to those of thermodynamics:

- 0. The 'Zeroth Law' states that the surface gravity  $\kappa$  is constant over the horizon. This is analogous to thermal equilibrium.
- 1. The 'First Law' states energy conservation: the change in mass *M* of the black hole is related to the change of its horizon area  $A_{\mathcal{H}}$ , angular momentum *J* and charge *Q* by

$$\delta M = \frac{\kappa}{8\pi} \delta A_{\mathcal{H}} + \Omega \delta J + \Phi \delta Q , \qquad (1.11)$$

where  $\Omega$  is the angular velocity and  $\Phi$  is the electrostatic potential.

2. The 'Second Law' states that the total entropy of a system consisting of a black hole and matter never decreases

$$dS_{\rm Tot} = dS_{BH} + dS_{\rm matter} \ge 0.$$
(1.12)

3. The 'Third Law' states that it is impossible to reduce the surface gravity  $\kappa$  to zero by a finite sequence of operations. It is analogous to Nernst's Law.

#### **1.1.2** The Entropy Puzzle

Imagine the following *Gedankenexperiment*: consider a box with some gas (thus with a non-trivial entropy) and throw it into an asymptotically flat black hole. After a sufficiently long time, an asymptotic observer is not able to resolve the presence of the highly red-shifted matter next to the black hole horizon. Since the gas, and hence his entropy, is not accessible anymore we have decreased the entropy of the observable universe, naïvely violating the second law of thermodynamics. The solution of this problem is to associate an entropy to a black hole (1.3), following the intuition of Bekenstein and Hawking. In turn, this give rise to another profound question: what are the *microstates* of the black hole? This is the so-called Entropy Puzzle.

In a seminal paper, Strominger and Vafa [19] provided, within the string theory framework, the first microphysical calculation of a black hole entropy in terms of strings. They gave the first microscopical explanation for the Bekenstein-Hawking formula, and they showed that the entropy of a black hole is precisely a quarter of the area of the event horizon (in Planck units), as expected. This computation is perceived as one of the major successes of string theory. For an introductory review and for a conceptual analysis of the Strominger-Vafa results see also [20].

#### 1.1.3 The Information Paradox

We have learned that black holes have an entropy, a temperature, and they radiate following a thermal distribution, but what is the physical origin of the Hawking radiation? An explanation, often given to respond to the above answer in a more intuitive fashion, is the following: quantum fluctuations of the vacuum create particle-antiparticle pairs and, in the presence of the gravitational field sourced by the black hole, one of the two particles (just inside the horizon) falls towards the singularity, while the other particle (just outside the horizon) can escape towards a far away observer. The infalling particle has negative energy<sup>5</sup> and subtracts mass to the black hole, which thus evaporates. These escaping particles constitutes the Hawking radiation. The radiation which emerges is *not* in a pure quantum state: indeed, the emitted quanta are in a mixed state. Before discussing the information paradox, let us recall some basic definitions.

**Definition 1.1.** A state  $\rho$  of a quantum system is a self-adjoint operator that is

1. Trace-class<sup>6</sup>

$$\operatorname{Tr}(\rho) = \sum_{e_n} \langle e_n, \rho \, e_n \rangle < \infty \,, \tag{1.13}$$

where  $\{e_n\}_n$  are elements of *any* orthonormal basis for a Hilbert space  $\mathcal{H}$ .

<sup>&</sup>lt;sup>5</sup>With respect to the observer, i.e. with respect to the timelike Killing vector defined at infinity.

<sup>&</sup>lt;sup>6</sup>Here we are remarking the fact that not every self-adjoint operator on an infinite-dimensional vector space has a trace defined. If this happens, it is called trace class.

2. Unit trace

$$\operatorname{Tr}(\rho) = 1, \qquad (1.14)$$

3. Non-negative

$$\langle \varphi, \rho | \varphi \rangle \ge 0$$
,  $\forall \varphi \in \mathcal{H}$ . (1.15)

*Remark* 1.2. The second and the first points in the above definition above can be combined by noting that, when well defined, the trace is independent of the orthonormal basis chosen.

**Definition 1.3.** Often, the above  $\rho$  is referred to as density matrix. We can thus equivalently say that a density matrix is a non-negative unit-trace self-adjoint operator.

**Definition 1.4.** Given a Hilbert space  $\mathcal{H}$ , given two states  $\rho_1, \rho_2 \in \mathcal{H}$  and  $c_1, c_2 \ge 0$  such that  $c_1 + c_2 = 1$ , a state  $\Psi \in \mathcal{H}$  is called convex combination of  $\rho_1, \rho_2$  if it is written as

$$\Psi = c_1 \rho_1 + c_2 \rho_2 = c_1 \rho_1 + (1 - c_1) \rho_2.$$
(1.16)

**Definition 1.5.** A state  $\Psi$  is called pure if it can not be written as a convex combination of  $\rho_1, \rho_2 \neq \Psi$ , unless trivially  $c_1 = 1, c_2 = 0$  or  $c_1 = 0, c_2 = 1$ .

*Remark* 1.6. Pictorially, pure states are the extreme points of a convex set of states.

**Definition 1.7.** An equivalent formulation of a pure state can be given in terms of the density matrix  $\rho$ , by looking at its properties. Given<sup>7</sup>  $|\psi\rangle \in \mathcal{H}$  a state is called a pure state if the density matrix can be written as

$$\rho = \frac{\langle \psi, \cdot \rangle \psi}{\langle \psi, \psi \rangle} = \frac{|\psi\rangle \otimes \langle \psi|}{||\psi||^2}, \qquad (1.17)$$

Equivalently,  $Tr(\rho^2) = Tr(\rho) = 1$ .

**Definition 1.8.** All the other states, for which such  $|\psi\rangle \in \mathcal{H}$  does not exist, are called mixed states. The density matrix is written, as opposed to the one above, as

$$\rho = \sum_{\ell \ge 1} c_{\ell} |\psi_{\ell}\rangle \otimes \langle \psi_{\ell}| , \qquad c_{\ell} \ge 0, \quad \sum_{\ell \ge 1} c_{\ell} = 1.$$
(1.18)

*Remark* 1.9. Note that for a pure state we have  $c_1 = 1$ ,  $c_{\ell} = 0$ ,  $\ell > 1$  and the above decomposition is trivial. Equivalently, a state is mixed if  $\text{Tr}(\rho^2) < 1$ .

*Remark* 1.10. Let us denote the density matrix associated to  $\psi_{\ell}$  as  $\rho_{\psi_{\ell}}$ . One should note that, given  $c \in (0, 1)$ ,

$$c\rho_{\psi_{\ell}} + (1-c)\rho_{\psi_{k}} \neq \rho_{c\psi_{\ell} + (1-c)\psi_{k}},$$
(1.19)

since the left hand side describes a mixed state (convex combination of density matrices) while the right hand side is a pure state. Indeed, using (1.17), one can easily check that the right hand side satisfies  $Tr(\rho^2) = Tr(\rho) = 1$ .

<sup>&</sup>lt;sup>7</sup>Note that we are considering also the case when  $|\psi\rangle$  is not normalized.

**Definition 1.11.** Consider two systems *A*, *B* with associated Hilbert spaces  $\mathcal{H}_A$ ,  $\mathcal{H}_B$  and basis  $\{|\psi_i\rangle_A\}_i, \{|\psi_j\rangle_B\}_j$  respectively. The most general state  $\Psi$  on  $\mathcal{H} = \mathcal{H}_A \otimes \mathcal{H}_B$  is given by

$$|\Psi\rangle = \sum_{i,j} c_{ij} |\psi_i\rangle_A \otimes |\psi_j\rangle_B . \qquad (1.20)$$

The state  $|\Psi\rangle$  is called **separable** if  $c_{ij} = c_i^A c_j^B$ , so that

$$|\Psi\rangle = |\psi_A\rangle \otimes |\psi_B\rangle$$
,  $|\psi_{A,B}\rangle = \sum_k c_k^{A,B} |\psi_k\rangle_{A,B}$ . (1.21)

Otherwise, the state is called entangled (or inseparable).

Mixed states inevitably arise from pure states when, for a composite quantum system defined on  $\mathcal{H}_1 \otimes \mathcal{H}_2$  with an entangled state constructed with states living in their own Hilbert space, the part  $\mathcal{H}_2$  is inaccessible to an observer. The state is thus expressed with a partial trace over  $\mathcal{H}_2$ . It is important to note that, if we have a pure state and then we perform a partial trace, we lose information about the system we are tracing over. This does not imply that the original system was not in a pure state, but rather that we have an effective description of the system, and we are left with a mixed state. Nevertheless, in principle, we should be able to recover the fine-grained description and hence have access to the properties of the original state.

The information paradox does not solely rely on the thermal nature of the emitted radiation. It is a consequence of the combined properties of the radiation *and* entanglement. Let's explain this carefully, following [21]. According to Hawking's arguments, one can show that the state of the emitted radiation is of the form

$$\begin{split} |\psi\rangle_{\text{emit}} &= C\left(|0\rangle_{b}\otimes|0\rangle_{c} + \gamma \,\hat{b}^{\dagger} \,|0\rangle_{b}\otimes\hat{c}^{\dagger} \,|0\rangle_{c} + \frac{1}{2}\gamma^{2} \,\hat{b}^{\dagger}\hat{b}^{\dagger} \,|0\rangle_{b}\otimes\hat{c}^{\dagger}\hat{c}^{\dagger} \,|0\rangle_{c} + \dots\right) \\ &= C\left(|0\rangle_{b}\otimes|0\rangle_{c} + \gamma \,|1\rangle_{b}\otimes|1\rangle_{c} + \gamma^{2} \,|2\rangle_{b}\otimes|2\rangle_{c} + \dots\right) , \end{split}$$
(1.22)

where  $|0\rangle_b$ ,  $|0\rangle_c$  are two different vacua of two different locations in the curved spacetime, a crucial feature of quantum field theory on a curved background. Without entering in details about the origin of this state, we can immediately note that the above state is not factorised. This implies that the subsystems are entangled. Suppose now that the quanta created by the operator  $\hat{c}^{\dagger}$  fall towards the singularity while the quanta created by the operator  $\hat{b}^{\dagger}$  propagate to infinity. The two are entangled. If the black hole evaporates, the outgoing radiation is entangled with nothing. We know that the emitted quanta cannot be described by a pure state, but now we cannot write a mixed state either since we have nothing to mix the radiation with! The only way to describe the system *B* is by using a reduced density matrix  $\rho_B$ . This description is statistical rather than quantum-mechanical. It is thus impossible to reconstruct the whole system (i.e. obtain a pure state) and the information about the original pure system is lost, violating unitarity. **Proposition 1.12** (Information Paradox). *The information paradox states the inability to reconstruct the original pure state from the thermal radiation collected by an observer at infinity due to the disappearance of entangled quanta during the black hole evaporation process. Hence, for an observer at infinity, the original pure state has undergone a non-unitary transformation, in contrast to quantum mechanics' unitarity postulate.* 

For this reason, a modification in the formulation of quantum mechanics was originally deemed necessary by Hawking. However, as we will discuss below, using string theory's description of black holes (instead of quantum field theory on curved spacetime) it is possible to preserve the postulates of quantum mechanics *and* potentially solve the information paradox.

How can we solve this paradox? One possibility would be to consider *small, local* and subtle correlations between the emitted quanta. In other words, one could imagine that perturbative corrections to Hawking's computation could lead to a successful unitary picture of black hole evaporation. Under certain physically-reasonable assumptions, one can show however that this is not possible. Mathur [22] proved the following "Small correction" no-go theorem that we recall without proof:

**Theorem 1.13** (Mathur's small correction no-go theorem). *Consider a black hole solution* of mass M and Schwarzschild radius  $R_S$ . Define  $\ell_P, m_P$  to be the Planck length and mass respectively. Consider the following hypothesis (not all independent):

- 1. Small curvatures: the quantum state is defined on a spacelike slice whose intrinsic and extrinsic (when embedded in a 4D spacetime) curvatures are small: <sup>(3)</sup> $R \ll \frac{1}{\ell_P}$ ,  $K \ll \frac{1}{\ell_P}$ . Furthermore, the spacetime curvature in a neighbourhood of the spacelike slice is small as well: <sup>(4)</sup> $R \ll \frac{1}{\ell_P}$ . We assume this to be true at sufficiently large times, as long as the mass of the evaporating black hole is  $M \gg m_P$ .
- 2. Energy conditions: we assume that the matter fields obey the usual energy conditions (say, the dominant energy condition).
- 3. Local Hamiltonian Evolution. The quantum state of the matter evolves according to a local Hamiltonian.
- 4. Information-free horizon (traditional black hole): we assume that at any point in the neighbourhood of the event horizon, the evolution of field modes with wavelength  $\ell_P \ll \lambda \lesssim R_S$  is given by the semi-classical evolution of quantum fields on empty curved space-time, up to terms suppressed as  $m_P/M$ . In other words, the No-Hair theorem holds.

*Under the above assumptions, the formation and evaporation of the black hole will lead to mixed states or remnants.* 

Roughly put, the theorem is telling us that, under physically-reasonable assumptions, small and local corrections are not enough to restore unitarity: we would be left with a mixed state or remnants. To evade this no-go theorem, at least one of the assumptions must be violated.

Recall that remnants [23] are long-lived objects with bounded mass and size but unbounded entanglement with a distant external system. Their formation is conjectured to have place when quantum gravity stabilising effects take over during the final stages of the evaporation process of a black hole, whose size approaches the Planck scale. The system comprised of a remnant and Hawking radiation would be in a pure state. However, their existence is nowadays deemed unlikely due to their conflict with Bekenstein's entropy bound and the generalised second law of thermodynamics [24, 25]. For other arguments against their existence see also [26, 27]. From now on we assume that the unique final stage of the evaporation process under the assumptions of Mathur's theorem is a thermal radiation.

A possible way to evade the above claim is to consider some sort of non-local interactions between the interior of the black hole and the asymptotic observer, thus relaxing the assumption (3) of the theorem. This idea is part of the so-called 'island paradigm' [28, 29, 30, 31], for a review see [32]. In this approach, most successfully applied to 2D gravity, the formula for the entanglement entropy of the radiation is computed via a gravitational path integral. The authors assume the validity in flat space-time of the so-called Quantum Extremal Surface proposal [33], which states that the fine-grained (i.e. Von Neumann) entropy of the radiation is given by an extremisation problem over appropriate spacetime regions. The solution to this extremisation problem is called (quantum) extremal surface. Their work suggests that, in order to account for the entanglement entropy of the Hawking radiation, there must exist a disconnected spacetime region called "islands" located in the interior of the black hole. By considering Hawking modes in this interior region together with the outgoing radiation one is able to recover the Page curve, nowadays considered a necessary condition for solving the information paradox. However, the description of the final state of the evaporation process is still problematic: the radiation at infinity collected by an observer is still mixed and is unclear how to recover the full pure state from it, especially in the context of four-dimensional gravity. Furthermore, one assumes some non-local relation between the interior of the black hole spacetime and the collected modes that seem not justified a priori.

More criticisms have been moved towards this set of semi-classical ideas. In particular, the authors of [34] highlight the contrasts with the so-called 'Fuzzball proposal' [22], which instead is based on the knowledge of the microscopic constituents of black hole and utilises fundamental ingredients from string theory. For an enlightening review see [35]. The main claim of this proposal can be summarised as follows:

**Proposition 1.14** (The Fuzzball Proposal). Consider a low-energy effective theory of string theory, with a black hole solution. The Fuzzball proposal claims that quantum gravity effects are non-negligible on the scale of the would-be horizon.

The point of the above claim is that quantum gravity effects cease to be negligible already at scales of the order of the *would-be* black hole horizon [36], violating the assumption (4) of Mathur's small correction theorem. The fuzzballs in string theory are the microstates of the black hole, the latter being a low-curvature solution of the lowenergy effective theory of string theory. Hence, the black hole represents a *coarse-grained* description of the underlying stringy system. In other words, such effective description of the stringy bound state has the characteristic features of the black hole like, for instance, its highly absorptive nature. It is widely expected that only a subset of all the possible Fuzzballs will be well-described by solutions of the low-energy theory (for instance, supergravity) and thus we expect the existence of some "stringy" fuzzballs, not well-described in the supergravity approximation. Out of these stringy solutions, one can recognize coherent-like states that well approximate classical behaviors: for black hole microstates, we expect them to be described by smooth and horizonless solutions of the supergravity equations of motion.

### **1.2** Outline of the thesis

The aim of this thesis is to present research results published in [1, 2, 3, 4], which constitute a set of original contributions for the understanding of the role of string theory in the resolution of black hole paradoxes and their internal structure. Most of the thesis is devoted to the study of a family of non-supersymmetric microstates called JMaRT (after the authors of [37]) that, in a special regime, admit an  $\alpha'$ -exact worldsheet description in terms of gauged Wess-Zumino-Witten (gWZW) models. The thesis is organised as follows.

In Chapter 2 we provide the necessary technical ingredients for the understanding of the subsequent chapters. In particular, we consider compactifications of Type IIB supergravity and the D1-D5-P black hole. After reviewing the essential properties of this well-studied system, we show how its weak coupling description provides a natural and elegant explanation for the existence of the dual D1-D5 conformal field theory (CFT). We review the marvelous relation between the attracted moduli space of the D1-D5-P system, the dual CFT moduli space and the instanton moduli space. We then proceed in setting up the notation for the D1-D5 CFT system at the orbifold point. From there, and for the rest of the thesis, we focus on worldsheet techniques. We review the Ramond-Neveu-Schwarz (R-NS) formalism for superstrings in  $AdS_3 \times S^3 \times T^4$  with

pure NS-NS fluxes and, finally, we present the null-gauged Wess-Zumino-Witten models that encode the dynamics of a closed string in the NS5-decoupling limit of the JMaRT background.

In Chapter 3 we present the results contained in [1]. We show that the consistency of the spectrum of the worldsheet CFT implies a set of quantisation conditions and parity restrictions on the gauging parameters. We also derive these constraints from an independent geometrical analysis of smoothness, absence of horizons and absence of closed timelike curves. This allows us to prove that the complete set of consistent backgrounds in this class of models is precisely the general family of (NS5-decoupled) JMaRT solutions, together with their various (BPS and non-BPS) limits. We clarify several aspects of these backgrounds by expressing their six-dimensional solutions explicitly in terms of five non-negative integers and a single length-scale. Finally we study non-trivial two-charge limits, and exhibit a novel set of non-BPS supergravity solutions describing bound states of NS5 branes carrying momentum charge.

Chapter 4 is an exposition of the PRL publication [2] and its extension [3]. Here we compute a large collection of string worldsheet correlators describing light probes interacting with heavy black hole microstates. We construct physical vertex operators in these cosets, including all massless fluctuations. We compute a large class of novel heavy-light-light-heavy correlators in the AdS<sub>3</sub> limit, where the light operators include those dual to chiral primaries of the holographically dual CFT. We compare a subset of these correlators to the holographic CFT at the symmetric product orbifold point, and find precise agreement in all cases, including for light operators in twisted sectors of the orbifold CFT. The agreement is highly non-trivial, and includes amplitudes that describe the analogue of Hawking radiation for these microstates. We further derive a formula for worldsheet correlators consisting of n light insertions on these backgrounds, and discuss which subset of these correlators are likely to be protected. As a test, we compute a heavy-light five-point function, obtaining precisely the same result both from the worldsheet and the symmetric orbifold CFT.

In Chapter 5 we present the results contained in [4]. AdS<sub>3</sub> correlators are essential building blocks for the correlators in the JMaRT microstate backgrounds. However, correlation functions of the SL(2, $\mathbb{R}$ )-WZW model involving spectrally flowed vertex operators are notoriously difficult to compute. An explicit integral expression for the corresponding three-point functions was recently conjectured in [38]. In this chapter, we provide a proof for this conjecture. For this, we extend the methods of [39] based on the so-called SL(2, $\mathbb{R}$ ) series identifications, which relate vertex operators belonging to different spectral flow sectors. We also highlight the role of holomorphic covering maps in this context. These results constitute an important milestone for proving this instance of the AdS<sub>3</sub>/CFT<sub>2</sub> holographic duality at finite 't Hooft coupling.

## **Chapter 2**

# Supergravity, D1-D5 CFT, and worldsheet string theory

In this chapter we provide the technical tools necessary to understand the original results published in [1, 2, 3, 4] and presented in the remaining chapters of the thesis. We start by reviewing the relevant supergravity compactifications and some important solutions in six-dimensions. We then focus on the D1-D5(-P) system and recall the derivation of the D1-D5 CFT in both the strong coupling and weak coupling regimes, and review the description of the dual CFT at the orbifold point. In doing so, we review the incredibly fascinating, albeit subtle, relation between the attracted supergravity moduli space, the dual CFT moduli space and the instanton moduli space in the context of the D1-D5-P system. We follow with an extensive review of the R-NS formulation of superstring theory in  $AdS_3 \times S^3 \times T^4$  with pure NS-NS fluxes and in a particular family of coset models which play a crucial role for the understanding of string theory in a specific family of black hole microstates.

## 2.1 Supergravity and compactifications

Here we recall basic facts about Type IIB supergravity and some compactifications that will play a prominent role for the rest of the thesis.

#### 2.1.1 Type IIB Supergravity

We start by considering Type IIB Supergravity, the massless truncation of Type IIB Superstring theory. The fields are given by the following representations of  $\mathfrak{so}(8)$ :

• NS-NS: Dilaton, B-field, Graviton  $\mathbf{1} \oplus \mathbf{28} \oplus \mathbf{35}_v = \{\Phi, B^{(2)}, G\}$ 

- NS-R: Dilatino, Gravitino  $\mathbf{8}_c \oplus \mathbf{56}_c = \{\lambda_{\alpha}^{(1)}, \chi_{\alpha}^{(1),\mu}\}$
- R-NS: Dilatino, Gravitino  $\mathbf{8}_c \oplus \mathbf{56}_c = \{\lambda_{\alpha}^{(2)}, \chi_{\alpha}^{(2),\mu}\}$
- R-R: 0-form (Axion), 2-form, 4-form  $\mathbf{1} \oplus \mathbf{28} \oplus \mathbf{35}_{s}^{(+)} = \{C^{(0)}, C^{(2)}, C^{(4),SD}\}$

where v, s, c label the three different representations of  $\mathfrak{so}(8)$  related by triality, and the (SD) in the RR 4-form the denotes the self duality of its field-strength. The two gravitini have the same chirality. The theory has  $128_B$  bosonic and  $128_F$  fermionic (on-shell) degrees of freedom, as supersymmetry requires. Indeed, the IIB supergravity theory is a chiral  $\mathcal{N} = (2,0)$  theory in 10 dimensions with 32 supercharges<sup>8</sup>. The IIB theory is also obtained from superstring theory, by choosing the Ramond vacua to have the same eigenvalue under the worldsheet fermion number  $(-1)^F = (-1)^{\overline{F}}$  when performing the GSO projection.

The string-frame action for the bosonic fields of Type IIB Supergravity is given by

$$S_{IIB}^{(Str)} = \frac{1}{2\kappa^2} \left\{ \int d^{10}x \sqrt{-G} \, e^{-2\Phi} \, R - \frac{1}{2} \int \left[ e^{-2\Phi} \left( -8 \, d\Phi \wedge \star d\Phi + H^{(3)} \wedge \star H^{(3)} \right) \right. \\ \left. + F^{(1)} \wedge \star F^{(1)} + \tilde{F}^{(3)} \wedge \star \tilde{F}^{(3)} + \frac{1}{2} \tilde{F}^{(5)} \wedge \star \tilde{F}^{(5)} + C^{(4)} \wedge H^{(3)} \wedge F^{(3)} \right] \right\},$$
(2.1)

where  $2\kappa^2 = 16\pi G_N^{10D} = (2\pi)^7 g_s^2 \ell_s^8$  and

$$G_N^d = \frac{G_N^{10D}}{(2\pi)^{10-d} V_{10-d}}, \qquad G_N^{10D} = 8\pi^6 g_s^2 \ell_s^8.$$
(2.2)

We have used  $H^{(3)} = dB^{(2)}$  and

$$\tilde{F}^{(3)} = F^{(3)} - C^{(0)} \wedge H^{(3)} , \qquad \tilde{F}^{(5)} = F^{(5)} - \frac{1}{2}C^{(2)} \wedge H^{(3)} + \frac{1}{2}B^{(2)} \wedge F^{(3)} , \qquad (2.3)$$

with  $F^{(n)} = dC^{(n-1)}$ , and the self-duality of  $\tilde{F}^{(5)}$  must be imposed on-shell. In Einstein frame<sup>9</sup> the same action becomes

$$S_{IIB}^{(Ein)} = \frac{1}{2\kappa^2} \left\{ \int d^{10}x \sqrt{-G} R - \frac{1}{2} \int \left[ d\Phi \wedge \star d\Phi + e^{-\Phi} H^{(3)} \wedge \star H^{(3)} + e^{2\Phi} F^{(1)} \wedge \star F^{(1)} + e^{\Phi} \tilde{F}^{(3)} \wedge \star \tilde{F}^{(3)} + \frac{1}{2} \tilde{F}^{(5)} \wedge \star \tilde{F}^{(5)} - C^{(4)} \wedge H^{(3)} \wedge F^{(3)} \right] \right\}.$$
(2.4)

The action in Einstein frame can be rewritten in a manifestly  $SL(2, \mathbb{R})$ -invariant fashion, hence the symmetry group is classically  $SL(2, \mathbb{R})$ . However, in the full string theory it is broken to a discrete  $SL(2,\mathbb{Z})$  subgroup which includes T- and S-duality transformations.

<sup>&</sup>lt;sup>8</sup>The spacetime supercharges are the integrals of the *worldsheet* gravitini's vertex operators. <sup>9</sup>The string and Einstein frame metric are related by  $G_{MN,Ein} = e^{-\Phi/2} G_{MN,Str}$ , where  $\Phi$  is the Dilaton.

The RR-forms  $C^{(p+1)}$  couple to Dp-branes. The string frame action of a stack of *N* coincident Dp-brane is the sum of the Dirac-Born-Infeld (DBI) and a Wess-Zumino (WZ) term, namely

$$S_{Dp}^{(Str)} = -T_{Dp} \int_{\mathcal{M}_{p+1}} d^{p+1} x \ e^{-\Phi} \operatorname{Tr} \left( \sqrt{-\det\left(G_{\alpha\beta} + 2\pi\alpha' \,\mathcal{F}_{\alpha\beta}\right)} \right) + \mu_p \int_{\mathcal{M}_{p+1}} \sum_q C^{(q)} \wedge \operatorname{ch}\left(\mathcal{F}\right) \wedge \sqrt{\frac{\widehat{A}(R_{T\mathcal{M}_{p+1}})}{\widehat{A}(R_{N\mathcal{M}_{p+1}})}},$$
(2.5)

where the brane tension is given by

$$T_{Dp} = \frac{1}{g_s(2\pi)^p \left(\alpha'\right)^{\frac{p+1}{2}}} = \frac{1}{g_s \ell_s(2\pi\ell_s)^p},$$
(2.6)

and  $\mu_p = \pm T_{Dp}$  is the RR charge for the (anti-)branes. In the DBI term, the pull-back on the brane worldvolume  $\mathcal{M}_{p+1}$  of the metric tensor and of the gauge invariant field strength is denoted by  $G_{\alpha\beta}$  and  $2\pi\alpha' \mathcal{F}_{\alpha\beta} = 2\pi\alpha' F_{\alpha\beta}^{(2)} + B_{\alpha\beta}^{(2)} \mathbb{1}$ ,  $\alpha, \beta = 0, \ldots, p$  respectively. Here  $F^{(2)}$  (not to be confused with the above RR field strengths) is the curvature of the connection  $A^{(1)}$  of the U(N) bundle over  $\mathcal{M}_{p+1}$ , while the Kalb-Ramond field  $B^{(2)}$  is a singlet under the U(N) group (and only couples to  $\mathrm{Tr}(F)$ ). In the WZ term, the formal sum is over the pull-back RR-forms  $C^{(q)}$  and is understood that in the wedge products one has to consider only (p + 1)-forms. The Chern class is ch  $(\mathcal{F}) =$  $\mathrm{Tr}\left(e^{2\pi\alpha'\mathcal{F}}\right)$  and  $\widehat{A}(R)$  is the Dirac (A-roof) polynomial of the tangent or normal bundle, which can be written in terms of Pontryagin classes as  $\widehat{A}(R) = p_0 - \frac{1}{24}p_1 + \frac{1}{5670}(7p_1^2 - 4p_2) + \ldots$ , where  $p_0 = 1$ ,  $p_1 = -\frac{1}{2}\frac{1}{(2\pi)^2} \mathrm{Tr}(R^2)$ ,  $p_2 = \frac{1}{8}\frac{1}{(2\pi)^4} \left[(\mathrm{Tr} R^2)^2 - 2 \mathrm{Tr} R^4\right]$  where R is the curvature 2-form of the bundle and  $R^n = \wedge^n R$ . The CS term does not require a metric and is independent on the choice of the connection of the bundle. For future reference, note that

$$S_{CS} = \mu_p \int_{\mathcal{M}_{p+1}} \left\{ N C^{(p+1)} + 2\pi \alpha' C^{(p-1)} \wedge \operatorname{Tr} F^{(2)} + \frac{(2\pi \alpha')^2}{2} C^{(p-3)} \wedge \left[ \operatorname{Tr}(F^{(2)} \wedge F^{(2)}) + \frac{N}{48} \left( \operatorname{Tr}(R_T^{(2)} \wedge R_T^{(2)}) - \operatorname{Tr}(R_N^{(2)} \wedge R_N^{(2)}) \right) \right] \right\},$$
(2.7)

with N the number of branes, showing the appearance of lower-dimensional branes inside the worldvolume of the Dp-brane [40].

#### **2.1.2** Type IIB on $\mathbb{T}^5$ : maximal 5D $\mathcal{N} = 8$ Supergravity

Consider Type IIB Supergravity compactified on  $\mathbb{T}^5$ . Toroidal compactifications preserve the original amount of supersymmetries, hence the number of supercharges in five dimensions is still 32. The resulting theory is the maximally supersymmetric fivedimensional  $\mathcal{N} = 8$  supergravity. By compactifying Type IIB supergravity to five dimensions on circles, the U-duality group<sup>10</sup> is given by  $E_{6,(6)}$ . In the full string theory, the duality group is broken by quantum effects to a discrete subgroup  $E_{6,(6)}(\mathbb{Z})$ . This group contains both the T-duality group  $O(5,5;\mathbb{Z})$  and the S-duality group  $SL(2,\mathbb{Z})$ [43]. When compactifying on the  $\mathbb{T}^5$ , the massless fields of Type IIB Supergravity give rise to 27 U(1) gauge fields and 42 scalars. The 27 vectors, transforming in the fundamental representation **27** of  $E_{6,(6)}$ , are given by:

$$\{(5) G_{\mu i}, (5) B_{\mu i}, (5) C_{\mu i}, (10) C_{\mu i j k}, (1) A^{1}_{\mu}, (1) A^{2}_{\mu}\}, \qquad \mu = 0, \dots 4, \ i = 5, \dots 9.$$
(2.8)

The two (magnetic) vectors  $A_{\mu}^{1,2}$  are obtained as follows. Consider the NS B-field  $B_{\mu\nu}^{(2)}$ and the RR 2-form  $C_{\mu\nu}^{(2)}$ . Their exterior derivate give rise to two field strength 3-forms, namely  $H_{\mu\nu\rho}$  and  $F_{\mu\nu\rho}$ . Using Hodge duality in five dimensions, these can be dualized into two (magnetic) field strengths  $F_{\mu\nu}^1$  and  $F_{\mu\nu}^2$  respectively. We have  $F^{1,2} = dA^{1,2}$ . The above vectors define 27 abelian charges, namely 5 KK momenta, 5 F1 winding modes, 5 D1 winding modes, 10 D3 winding modes, 1 NS5 winding mode for wrapping the  $\mathbb{T}^5$ , 1 D5 winding mode for wrapping the  $\mathbb{T}^5$ . Note that the spectrum of charges is compatible with *S*-duality.

The 42 scalars parametrise the moduli space of the supergravity theory  $E_{6,(6)}/Sp(4)$ , where Sp(4) = USp(8), which is  $9 \times 8/2 = 36$  dimensional, is the maximally compact subgroup of  $E_{6,(6)}$  [43, 44]. The scalars are given by

$$\{(15) G_{ij}, (10) B_{ij}, (10) C_{ij}, (5) C_{ijkl}, (1) \Phi, (1) C^0\}, \qquad i = 5, \dots 9.$$
(2.9)

Taking into account discrete identifications, the moduli space is given by

$$E_{6(6)}(\mathbb{Z}) \setminus E_{6(6)} / USp(8).$$
 (2.10)

### **2.1.3** Type IIB on $\mathbb{T}^4$ : maximal 6D $\mathcal{N} = (2, 2)$ Supergravity

In this subsection we briefly review the structure of the moduli space of vacua for the Type IIB Supergravity theory on  $\mathbb{T}^4$ .

Consider the ten-dimensional spacetime to be of the form  $\mathbb{R}^{1,5} \times \mathbb{T}^4$ , and compactify the  $\mathbb{T}^4$ . One obtains the maximal  $\mathcal{N} = (2,2)$  Supergravity in six dimensions. This

<sup>&</sup>lt;sup>10</sup>The maximally non-compact (split) real form of a Lie group of rank *n* has precisely *n* more noncompact generators than compact ones, see for example [41]. For instance, the number of non-compact generators of  $\mathfrak{e}_{8,(8)},\mathfrak{e}_{7,(7)}$  and  $\mathfrak{e}_{6,(6)}$  are respectively 128, 70, 42. Recall that the maximally non-compact exceptional (classical) global symmetry groups  $E_{n,(n)}$  arise after  $\mathbb{T}^n$  compactifications of M-theory or  $\mathbb{T}^{n-1}$ compactifications of Type II string theory. Note that one can continue the exceptional sequence as  $\mathfrak{e}_{5,(5)} \equiv$  $\mathfrak{so}(5,5), \mathfrak{e}_{4,(4)} \equiv \mathfrak{sl}(5,\mathbb{R}), \mathfrak{e}_{3,(3)} \equiv \mathfrak{sl}(3,\mathbb{R}) \oplus \mathfrak{sl}(2,\mathbb{R}), \mathfrak{e}_{2,(2)} \equiv \mathfrak{sl}(2,\mathbb{R}) \oplus \mathfrak{sl}(2,\mathbb{R}), \mathfrak{e}_{1,(1)} \equiv \mathfrak{sl}(2,\mathbb{R})$ , see page 314 in [42].
is equivalent to the  $\mathbb{T}^5$  reduction of 11D Supergravity, yielding an  $E_{5,(5)} = SO(5,5)$ global symmetry in 6-dimensions, which can be seen as the unbroken maximally noncompact subgroup of the exceptional group  $E_{6,(6)}$ , after one considers an  $\mathbb{S}^1 \subset \mathbb{T}^5$  to be much bigger than the other compact one-cycles. Similarly, the U-duality group of the full string theory  $E_{6(6)}(\mathbb{Z})$  gets broken to the subgroup  $E_{5(5)}(\mathbb{Z}) = SO(5,5;\mathbb{Z})$ . Consequently, the **27** of  $E_{6,(6)}$  gets broken to the **10** (vector), **16** (spinor), **1** (singlet) representation of SO(5,5).

The bosonic content is given by 1 graviton, 5 tensor gauge fields, 16 vectors, 25 scalars. The five 2-forms should be thought in terms of their self-dual and anti-self-dual parts

five 2-forms 
$$= \left( B_{\mu\nu}^{+I}, B_{\mu\nu}^{-I} \right)$$
,  $I = 1, \dots, 5$ , (2.11)

which transform in the **10** of SO(5,5). The vectors, on the other hand, transform in the spinor **16** of SO(5,5). The R-Symmetry group<sup>11</sup> is  $USp(4) \times USp(4) \simeq SO(5) \times SO(5)$ . The 25 scalar parametrize the Teichmuller space<sup>12</sup>

$$\mathcal{K} = \frac{SO(5,5)}{SO(5) \times SO(5)} \,. \tag{2.12}$$

To obtain the fermionic content of the 6D theory we need to reduce the two dilatini and the two gravitini from 10 to 6 dimensions. We also recall that in 10 dimension the minimal spinor representation is that of a Majorana-Weyl spinor of dimension 16, while in 6 dimensions the minimal spinor representation is that of a symplectic-Weyl (or symplectic-Majorana) spinor and has dimension 8. Making use results present in App. B of [45] Eq. (B.1.43) and (B.1.44), the fermionic content of the maximal  $\mathcal{N} = (2, 2)$ Supergravity theory in 6D is given by (4 + 4) gravitini and (20 + 20) spin 1/2 fermions. To summarise, the matter content of the maximal  $\mathcal{N} = (2, 2)$  Supergravity theory in 6D is given by 1 graviton, 5 tensor gauge fields, 16 vectors, 25 scalars (4+4) gravitini, and (20+20) spin 1/2 fermions.

Following the presentation in [46], we organise the charges and the gauge fields according to their transformation properties and masses:

<sup>&</sup>lt;sup>11</sup>Sometimes one finds the notation  $Sp(n_L) \times Sp(n_R)$ , where the total number of chiral (anti-chiral) Weyl spinors in 6D is denoted by  $2n_L$  ( $2n_R$ ).

<sup>&</sup>lt;sup>12</sup>We refer to the Teichmuller space as the covering space of the moduli space. The latter is obtained from the former by discrete (global) identifications. Note that  $\mathcal{K}$  is also a Grassmannian.

The 27 $U(1)$ charges in IIB Sugra on $\mathbb{T}^4$							
<i>SO</i> (1,5) rep	<i>SO</i> (5,5) rep	charges	gauge field	mass			
tensor	vector 10	$\{n_5, d_5, f_1, d_1, D^{5ij}\}$	$\{B_{\mu5}^{(m)}, C_{\mu5}^{(m)}, B_{\mu5}, C_{\mu5}, C_{\mu5ij}\}$	$\ell_s^{-2}$			
vector	spinor 16	$\{w^{i}_{F1}, w^{i}_{D1}, D_{ijk}, p^{i}\}$	$\{B_{\mu i}, C_{\mu i}, C_{\mu i j k}, G_{\mu i}\}$	$\ell_s^{-1}$			
scalar	scalar 1	n <sub>p</sub>	$G_{\mu 5}$	$\ell_s^0$			

TABLE 2.1: The re-arrangement of the 27 U(1) charges in Type IIB Sugra on  $\mathbb{T}^4$  is shown. In the first column, we report the transformation properties of the fields under Lorentz transformations for an observer in the non-compact directions. In the fourth column, the letter *m* in  $B_{\mu 5}^{(m)}$ ,  $C_{\mu 5}^{(m)}$  stands for "magnetic". In the last column, the mass scaling with  $\ell_s$  is shown (in the decoupling limit). The non-compact and circle indices run over  $\mu = 0, \ldots, 5$ . The fifth direction is that of the distinguished  $\mathbb{S}_y^1$ . The torus directions are denoted by  $i = 6, \ldots, 9$ .

By looking at the first row of the above Table 2.1, one can see that the masses of the NS5, D5, F1, the D1 and the 6 D3 branes (all winding the  $S_y^1$  circle) scale as  $\ell_s^{-2}$ . In the decoupling limit  $R_y^4 \gg V_{\mathbb{T}^4}$ , they are infinitely massive and will constitute the background for the other fields [44]. The 16 branes that do not wrap the circle have masses scaling as  $\ell_s^{-1}$ .

For applications present in this thesis, we will not turn on all the fields listed above. This will have important consequences for the properties of the moduli space, as we will see in due course. Preliminarily, note that the coset in Eq. (2.12) is a particular case of that obtained in the pure supergravity multiplet coupled to  $n_t$  tensor multiplets [47, 48]. The more general coset is the Grassmannian

$$\frac{SO(5,n_t)}{SO(5) \times SO(n_t)}.$$
(2.13)

One finds five self-dual tensor gauge fields and  $n_t$  anti-self-dual tensor gauge fields transforming in the vector of  $SO(5, n_t)$ . The moduli space of vacua is the given by a further quotient with the discrete duality group  $SO(5, n_t; \mathbb{Z})$ . For compactifications on K3, anomaly cancellation is readily achieved since in this case  $n_t = 21$ , which follows from the Hodge diamond structure of K3.

#### A technical detour on the D1-D5 system

Consider now the D1-D5 system, where the D5 wraps  $\mathbb{T}^4$  or *K*3. Following [49] and the presentation in [50], one can describe the brane bound state by a primitive<sup>13</sup> (Mukai) charge vector  $v \in \Gamma^{5,n_t}$ , where  $\Gamma^{5,n_t}$  is an even unimodular lattice. For the D1-D5 system,

<sup>&</sup>lt;sup>13</sup>We consider the charge vector to be not a multiple of another vector. If v is not primitive the brane system is singular: the system breaks into sub-systems with no cost in energy for *all* values of the moduli.

the lattice decomposes as  $\Gamma^{5,n_t} = \Gamma^{1,1} \oplus \Gamma^{4,n_t-1}$ , where the additional  $\Gamma^{1,1}$  parametrizes F1 and NS5 charges, while  $\Gamma^{4,n_t-1}$  parametrizes the RR charges. The two  $\Gamma^{1,1}$  lattices in  $\Gamma^{1,1} \oplus \Gamma^{4,n_t-1}$  are exchanged by S-duality that acts as a  $\mathbb{Z}_2$  group. In one of them, the D1 and D5 branes are parametrized by multiples of the vectors (1,0) and (0,1) respectively. Hence, for the 'pure' D1-D5 system we consider a Mukai charge vector  $(d_1, d_5)$ , where  $d_1$  and  $d_5$  are taken to be coprime in order to ensure that the vector is primitive. In the pure D1-D5 system, the charge vector can be shown to have norm  $v^2 = 2N = 2d_1d_5$  and to saturate the BPS bound. Because of this, the system has no binding energy and the branes can separate with no cost in energy. This is reflected on the structure of the moduli space of the theory, as discussed below.

For our black hole applications, we will focus on the D1-D5-P system or, most of the time, on the S-dual F1-NS5-P system. In the latter frame, a specific class of microstates of the NS5-F1-P black hole can be described with an exact null-gauged WZW model [51, 52, 53, 54, 1] since only NSNS fluxes are turned on. With this worldsheet model, one can probe the physics of these microstates (perturbatively) to all orders in  $\alpha'$  and is the main focus of this thesis.

#### 2.1.4 Attracted supergravity moduli space

We have seen above that the 25 scalars parametrise a 25-dimensional moduli space. However, only a 20-dimensional subspace survives the near horizon limit [49]. In other words, 5 scalars get fixed near the horizon and perturbations around their VEV are massive. This is a consequence of the attractor mechanism [55]. As an example, consider  $n_1$  D1-branes wrapping the distinguished  $S_y^1$  circle of radius  $R_y$  and  $n_5$  D5-branes wrapping the circle and the four-torus  $V_{\mathbb{T}^4}$ . This defines a five-dimensional black hole, when a reduction on the  $S_y^1$  is performed. In six-dimensions, the system is a black string with  $AdS_3 \times S^3$  near horizon geometry. In this case, the fixed scalars are the torus volume (proportional to  $G_{ii}$ ), a linear combination of  $C^{(0)}$  and  $C^{(4)}_{6789}$ , and three anti-self-dual components of the B-field are set to zero (forcing the B-field to be selfdual on the torus) [44]. Thus, in the D1-D5 frame, the remaining 20 moduli are given as follows: 9 from the traceless  $G_{ij}$ , 6 from  $C_{ij}$ , 1 from  $C^{(0)}$ , 3 from  $B_{ij}^{SD}$  and 1 from the dilaton  $\Phi$ . It is important to note that for the NS5-F1 system (the S-dual of the D1-D5) the 20 moduli are given by: 9 from traceless  $G_{ij}$ , 6 from  $B_{ij}$ , 1 from  $C^{(0)}$ , 3 from  $C_{ij}^{SD}$  and 1 from the dilaton  $\Phi$ . Clearly, the role of  $B^{(2)}$  and  $C^{(2)}$  has been exchanged.

These near-horizon moduli parametrize the 'attracted' coset (Grassmannian) [49]

$$\mathcal{K}^* = \frac{SO(4,5)}{SO(4) \times SO(5)} \subset \mathcal{K} = \frac{SO(5,5)}{SO(5) \times SO(5)}.$$
(2.14)

Global identifications are implemented by those elements in  $\mathcal{H}_v \subseteq SO(5,5;\mathbb{Z})$  that preserve the background charge vector  $(d_1, d_5)$ , namely its stabiliser. The elements contained in the U-duality group  $SO(5,5;\mathbb{Z})$  but not in  $\mathcal{H}_v$ , do not preserve the conditions that fix the moduli at the horizon [44]. One has to quotient the above coset by  $\mathcal{H}_v$  to obtain the fundamental domain  $\mathcal{F}$ , made out of copies of the form<sup>14</sup>

$$\mathcal{M}_{\text{Sugra}}^* = \mathcal{H}_v \setminus SO(4,5) / SO(4) \times SO(5) \,. \tag{2.15}$$

The charge vector is nevertheless covariant under the action of  $\mathcal{H}_v$ , and a pair of charge  $(d_1, d_5)$  can be mapped to the so-called canonical vector  $(1, d_1d_5)$ . As an example, this is achieved by an  $SO(2, 2; \mathbb{Z}) = SL(2, \mathbb{Z})_L \times SL(2, \mathbb{Z})_R$  subgroup, which contains the  $SL(2, \mathbb{Z})$  symmetry of Type IIB string theory in ten dimensions which acts via S- and T-duality on the moduli. In particular,  $SL(2, \mathbb{Z})_R$  acts on the axio-dilaton  $\tau = C^{(0)} + i/g_s$  while  $SL(2, \mathbb{Z})_L$  acts on the complexified modulus  $\tilde{\tau} = C_{6789}^{(4)} + iv/g_s$ , with v the (fixed) volume of  $\mathbb{T}^4$ .

It is useful to consider a basis for this 20-dimensional moduli space. The obvious choice is given by the field component that remain massless, up to a minor modification: instead of  $C^{(0)}$  alone, we consider a linear combination of  $\Xi = C^{(0)} + aC_{6789}^{(4)}$ , for an appropriate value of *a* that we do not specify here. Note that the latter is not the same combination that gets attracted at the horizon and is indeed still massless. These fields have specific transformation properties under the  $SO(4)_E \simeq SU(2)_L \times SU(2)_R$  isometry group of S<sup>3</sup> and the  $SO(4) = SU(2)_1 \times SU(2)_2$  symmetry group of the tangent space of  $\mathbb{T}^4$ . In particular, the moduli are all singlets under the former and some of them transform non-trivially under the latter, as reported in the following table for the case of the D1-D5 system [46]:

The 20 near-horizon moduli							
Field	$SO(4)_E \simeq SU(2)_L \times SU(2)_R$	$SO(4) = SU(2)_1 \times SU(2)_2$	DOF				
$G_{ij} - \frac{1}{4}\delta_{ij}G_{kk}$	(1,1)	(3,3)	9				
$B_{ij}^{SD}$	(1,1)	(3,1)	3				
$C_{ij}^{(2)}$	(1,1)	<b>(3,1)</b> ⊕ <b>(1,3)</b>	6				
$\Xi = C^{(0)} + a C^{(4)}_{6789}$	(1,1)	(1,1)	1				
Φ	(1,1)	(1,1)	1				

TABLE 2.2: The 20 moduli parametrising the attracted moduli space of  $\mathbb{T}^4$  compactifications of IIB supergravity. Note that all the moduli fields are singlet under  $SU(2)_{L,R}$ .

<sup>&</sup>lt;sup>14</sup>For an arbitrary number of tensor multiplets, the attracted moduli space is made of copies of the form  $\mathcal{M}^*_{Sugra} = \mathcal{H}_v \setminus SO(4, n_t) / SO(4) \times SO(n_t)$ .

A co-dimension 4 subspace of the 20-dimensional moduli space is actually singular [56, 57, 50]. In the D1-D5 frame, this subspace is defined by  $\Xi = B_{ij}^{SD} = 0$ . Note that, as a consequence, the full B-field on the torus  $B_{ij}^{(2)}$  is set to zero. In this region, the D1-D5 system becomes unstable under fragmentation and the brane can nucleate at no cost in energy and is exactly the point where the charge vector v has norm  $v^2 = 2d_1d_5$ , as discussed above. Indeed, one can show that when  $B_{ij}^{SD} \neq 0$ , the branes have a non-zero binding energy, see also [58]. In the NS5-F1 frame all works analogously, with the B-field dualised to the RR two-form and the singular subspace is defined by  $\Xi = C_{ij}^{(2),SD} = 0$ . Note that a pure NS worldsheet description necessarily lies in the singular locus of the moduli space.

## 2.1.5 Half-maximal and quarter-maximal 6D Sugra

There are two half-maximal Supergravity theories in six dimensions: the non-chiral  $\mathcal{N} = (1,1)$  theory and the chiral  $\mathcal{N} = (2,0)$  theory. The latter is of interest for us, since we will focus on the near-horizon geometry of D1-D5 (or F1-NS5) branes present in the Type IIB theory. There are two ways of obtaining this theory: the first is to compactify IIB Supergravity on K3, being a half-BPS Calabi-Yau space. This implies the holonomy constraint on the IIB Killing spinors which, upon the condition  $\Gamma^{0...9}\epsilon = \epsilon$ , now must satisfy

$$\Gamma^{6789}\epsilon = \epsilon \quad \Rightarrow \quad \Gamma^{012345}\epsilon = \epsilon \,.$$
 (2.16)

The second way is to take the *chiral* compactification of IIB Supergravity which consists in imposing the same condition above on the  $\mathcal{N} = (2, 2)$  theory. We thus remain with 16 supersymmetries and the R-symmetry reduces to USp(4) only. Note that the near horizon of the D1-D5 system is  $AdS_3 \times S^3 \times \mathbb{T}^4$  and this space preserves 16 supercharges, exhibiting an enhancement of supersymmetries.

Finally, we conclude this section by briefly recalling the field content of the quartermaximal 6D  $\mathcal{N} = (1,0)$  Supergravity theory, which will be useful for later applications. We now list the (1,0) multiplets:

- (1,0) Graviton multiplet "**G**": 1 graviton, 1 self-dual tensor gauge field, two lefthanded gravitini
- (1,0) Gravitino (L) multiplet "**g1**": 4 self-dual tensor gauge field, two left-handed gravitini
- (1,0) Gravitino (R) multiplet "g2": 4 vector fields, two right-handed gravitini, 2 left-handed spinors
- (1,0) Tensor multiplet "T": 1 anti-self-dual tensor gauge field, 2 right-handed spinors (tensorini), 1 scalar

- (1,0) Vector multiplet "V": 1 vector field, 2 left-handed spinors
- (1,0) Hypermultiplet "H": 4 scalars, 2 right-handed spinors

*Remark* 2.1. We can reconstruct the (2, 2) and (2, 0) multiplets as follows: (2, 2) Graviton multiplet =  $\mathbf{G} + \mathbf{g1} + 5 \times (\mathbf{T} + \mathbf{H}) + 2 \times (\mathbf{g2} + 4 \times \mathbf{V})$ ; (2, 0) Graviton multiplet =  $\mathbf{G} + \mathbf{g1}$ ; (2, 0) Tensor multiplet =  $\mathbf{T} + \mathbf{H}$ .

In most of the present thesis, we will study the properties a non-supersymmetric solitonic solution of the 6D  $\mathcal{N} = (1,0)$  Sugra theory with  $n_t = 1$  tensor multiplets, using worldsheet techniques. Thus, the multiplets of our interests will be the (1,0) Graviton multiplet and one (1,0) tensor multiplet. The particular Supergravity fields we are interested in define what is known as JMaRT solution [37]. The latter is found as a particular case of the Cvetic-Youm family [59, 60], which we now discuss.

# 2.2 Cvetic-Youm solution and its limits

We consider the most general rotating three-charge D1-D5-P solutions of maximal 6D supergravity presented in [59, 60]. We will focus on the interpretation of the fivedimensional black hole solution in terms of the rotating six-dimensional black string. The reasons for this are two: the solution (although still involved) is more simple if written as a six-dimensional black string and, furthermore, it will be useful when discussing the 6D JMaRT family in the rest of this thesis. We partly follow the notation and discussion of [61].

The string frame metric and dilaton of the so-called Cvetic-Youm solution [60] are

$$ds_{6D,S}^{2} = \frac{1}{\sqrt{H_{1}H_{5}}} \left[ -\left(1 - \frac{2mf_{D}}{r^{2}}\right) d\tilde{t}^{2} + d\tilde{y}^{2} + H_{1}H_{5}f_{D}^{-1}\frac{r^{4}}{(r^{2} + l_{1}^{2})(r^{2} + l_{2}^{2}) - 2mr^{2}} dr^{2} - \frac{4mf_{D}}{r^{2}}\cosh\alpha_{1}\cosh\alpha_{5}(l_{2}\cos^{2}\theta d\psi + l_{1}\sin^{2}\theta d\phi)d\tilde{t} - \frac{4mf_{D}}{r^{2}}\sinh\alpha_{1}\sinh\alpha_{5}(l_{1}\cos^{2}\theta d\psi + l_{2}\sin^{2}\theta d\phi)d\tilde{y} + \left(\left(1 + \frac{l_{2}^{2}}{r^{2}}\right)H_{1}H_{5}r^{2} + (l_{1}^{2} - l_{2}^{2})\cos^{2}\theta\left(\frac{2mf_{D}}{r^{2}}\right)^{2}\sinh^{2}\alpha_{1}\sinh^{2}\alpha_{5}\right)\cos^{2}\theta d\psi^{2} + \left(\left(1 + \frac{l_{1}^{2}}{r^{2}}\right)H_{1}H_{5}r^{2} + (l_{2}^{2} - l_{1}^{2})\sin^{2}\theta\left(\frac{2mf_{D}}{r^{2}}\right)^{2}\sinh^{2}\alpha_{1}\sinh^{2}\alpha_{5}\right)\sin^{2}\theta d\phi^{2} + \frac{2mf_{D}}{r^{2}}\left(l_{2}\cos^{2}\theta d\psi + l_{1}\sin^{2}\theta d\phi\right)^{2} + H_{1}H_{5}f_{D}^{-1}r^{2}d\theta^{2}\right],$$

$$(2.17)$$

$$e^{2\Phi} = g_s^2 \frac{H_1}{H_5}, \qquad (2.18)$$

where

$$d\tilde{t} = \cosh \alpha_p dt - \sinh \alpha_p dy , \qquad d\tilde{y} = -\sinh \alpha_p dt + \cosh \alpha_p dy , \qquad (2.19)$$

$$H_{1,5} = 1 + \frac{2mf_D}{r^2} \sinh \alpha_{1,5} , \qquad f_D^{-1} = 1 + \frac{l_1^2 \cos^2 \theta + l_2^2 \sin^2 \theta}{r^2} , \qquad (2.20)$$

and  $y \sim y + 2\pi R_y$ . The parameters *m*, *l*<sub>1</sub>, *l*<sub>2</sub>,  $\alpha_{1,5,p}$  are related to mass, charges angular momenta as

$$M = m \sum_{i=1,5,p} \cosh 2\alpha_i$$
,  $Q_i = m \sinh 2\alpha_i$ ,  $i = 1, 5, p$ , (2.21a)

$$J_{L/R} = \frac{1}{2} (J_{\phi} \mp J_{\psi}) = \frac{\pi}{4G_N^{5D}} m(l_1 \mp l_2) \left( \prod_{i=1,5,p} \cosh \alpha_i \pm \prod_{i=1,5,p} \sinh \alpha_i \right) , \qquad (2.21b)$$

and the 5D Newton constant is related to the six and ten dimensional one as  $(2\pi)^4 V (2\pi R_y) G_N^{5D} = (2\pi)^4 V G_N^{6D} = G_N^{10D} = 8\pi^6 g_s^2 \alpha'^4$ . The brane charges are quantised and are given by

$$Q_1 = c_1 n_1$$
,  $c_1 = \frac{g_s \alpha'^3}{V}$ ,  $n_1 = \frac{mV}{g_s \alpha'^3} \sinh 2\alpha_1$ , (2.22a)

$$Q_5 = c_5 n_5$$
,  $c_5 = g_s \alpha'$ ,  $n_5 = \frac{m}{g_s \alpha'} \sinh 2\alpha_5$ , (2.22b)

$$Q_p = c_p n_p$$
,  $c_p = \frac{g s^2 {\alpha'}^4}{R_y^2 V}$ ,  $n_p = \frac{m R_y^2 V}{g_s^2 {\alpha'}^4} \sinh 2\alpha_p$ , (2.22c)

where  $n_{1,5,p} \in \mathbb{Z}$ . The angular momenta are quantised as well:

$$J_{\phi} = J_L + J_R = \frac{n_{\phi}}{2}$$
,  $J_{\phi} = -J_L + J_R = \frac{n_{\psi}}{2}$ , (2.23)

with  $n_{\phi,\psi} \in \mathbb{Z}$ . It is very important to note that the parameter space of the above solution contains black holes, smooth solitons, conical defects or naked singularities. We now temporarily focus on the black hole region in this parameter space.

#### 2.2.1 Cvetic-Youm black hole

The Cvetic-Youm black hole is defined as the dimensional reduction on  $S_y^1$  of supergravity fields in Eq. (2.17) with the additional important constraint

$$m \ge \frac{(|l_1| + |l_2|)^2}{2}. \tag{2.24}$$

The Cvetic-Youm black hole is also known as the non-extremal five-dimensional rotating D1-D5-P black hole. As particular cases, it includes the BMPV black hole and the celebrated non-rotating extremal Strominger-Vafa black hole, as we will discuss below. The horizons of the five-dimensional Cvetic-Youm black hole are located at

$$r_{\pm}^{2} = m - \frac{l_{1}^{2} + l_{2}^{2}}{2} \pm \sqrt{\left(m - \frac{l_{1}^{2} + l_{2}^{2}}{2}\right)^{2} - l_{1}^{2} l_{2}^{2}},$$
(2.25)

and the Bekenstein-Hawking entropy is given by

$$S_{BH} = \frac{\pi^2 m}{2G_5} \left[ \left( \prod_{i=1,5,p} \cosh 2\alpha_i + \prod_{i=1,5,p} \sinh 2\alpha_i \right) \sqrt{2m - (l_1 - l_2)^2} + \left( \prod_{i=1,5,p} \cosh 2\alpha_i - \prod_{i=1,5,p} \sinh 2\alpha_i \right) \sqrt{2m - (l_1 + l_2)^2} \right].$$
 (2.26)

Understanding the microscopic origin of the above entropy from a bulk perspective is a major fundamental and open question. We can consider this to be one of the most important (and difficult) aims of string theorists working in the field of black hole microstates.

## 2.2.2 The extremal rotating D1-D5-P black hole

We have seen that Eq. (2.24) is the necessary and sufficient condition for having a black hole. In the particular case when<sup>15</sup>

$$m = \frac{(l_1 + l_2)^2}{2}, \qquad (2.27)$$

then the outer and inner horizon coincide, and we recover the three-charge extremal rotating D1-D5-P black hole solution. The horizon is at  $r_H = \sqrt{l_1 l_2}$ . The Bekenstein-Hawking entropy reduces to

$$S_{BH} = 2\pi \sqrt{\left(\frac{\pi}{4G_N^{5D}}\right)^2 Q_1 Q_5 Q_p - J_L^2 + J_R^2} = 2\pi \sqrt{n_1 n_5 n_p + \frac{n_\phi n_\psi}{4}}.$$
 (2.28)

In this formula, the black hole attractor mechanism is manifest in a simple manner.

## 2.2.3 The BMPV black hole

Note that the above solution is extremal but not supersymmetric. If we further require that the black hole is BPS saturated, we obtain the three-charge BMPV black hole [62]. The BPS bound reads

$$M \ge Q_1 + Q_5 + Q_p \,. \tag{2.29}$$

<sup>&</sup>lt;sup>15</sup>We are assuming  $l_{1,2}$  positive for simplicity.

The BPS limit must be taken carefully, by appropriately rescaling certain quantities. Let's define

$$e^{\alpha_i} = \frac{\eta_i}{\sqrt{m}}$$
,  $l_{1,2} = \sqrt{m} j_{1,2}$ , (2.30)

then the BPS limit is defined by sending  $m \to 0$ ,  $\alpha_i \to \infty$  while keeping  $\eta_i$ ,  $j_{1,2}$ ,  $G_N^{10D}$ , V,  $R_y$  fixed. In this way, we saturate the BPS bound Eq. (2.29) and the line element becomes simpler:

$$ds_{6D,S}^{2} = \frac{1}{\sqrt{H_{1}H_{5}}} \left[ -dt^{2} + dy^{2} + H_{p}(dt - dy)^{2} + H_{1}H_{5}(dr^{2} + r^{2}d\Omega_{S^{3}}^{2}) - \frac{8G_{N}^{5D}J_{L}}{\pi r^{2}}(\sin^{2}\theta d\phi - \cos^{2}\theta d\psi)(dt - dy) \right], \qquad (2.31)$$

where this time the "1" has been dropped from  $H_p$ , namely  $H_p = Q_p/r^2$ , while  $H_i = 1 + Q_i/r^2$  for i = 1, 5. Note that the right angular momentum is zero  $J_R = 0$  due to the minus sign in the last parenthesis in the definition of  $J_R$  in Eq. (2.21b).

# 2.3 The D1-D5(-P) black hole

By further setting  $l_1 = l_2$  (namely  $j_1 = j_2$ ) in the BMPV black hole we have  $J_L = 0$  as well. In this case we obtain the non-rotating three-charge D1-D5-P black hole, also known as<sup>16</sup> the Strominger-Vafa black hole. The latter has proven to be crucial for the first microscopic counting of microstates in string theory [19], and the first instance of what would have become the AdS/CFT correspondence [16, 17, 18]. Because of its importance, we review its brane construction and features.

Before diving into the details of the three-charge black hole, we first explain the construction and properties for the two-charge D1-D5 black hole. The reason for this are two-fold. First, its simplicity allows us to emphasize the important details shared also by the D1-D5-P black hole. Second, the bulk microstates reproducing the entropy of the D1-D5 black hole have been found in [64, 65, 66, 67]. These microstates admit a supergravity description, and an appropriate quantisation of the moduli space shows that they account for the entropy of the two-charge D1-D5 black hole [68].

In the following, we will denote by  $d_1$ ,  $d_5$  the number of D1 and D5-branes, while we will use  $n_1$ ,  $n_5$  to the denote the number of fundamental strings and NS5-branes. This echoes the conventions of [44].

<sup>&</sup>lt;sup>16</sup>Actually, Strominger and Vafa used more general configurations in Type II and Heterotic string theory. The D1-D5-P solution was explicitly considered in [63].

## 2.3.1 The naive D1-D5 small black hole

Let's consider Type IIB String Theory compactified on  $S_y^1 \times \mathbb{T}^4$ , namely on a distinguished circle  $S_y^1$  of radius  $R_y$  and a 4-torus  $\mathbb{T}^4$ . The D-brane setup that leads to the (naive) two-charge extremal D1-D5 black hole is the following:

IIB	0	1	2	3	4	5	6	7	8	9
$d_1$ D1	×					×	$\sim$	$\sim$	$\sim$	~
<i>d</i> <sub>5</sub> D5	×					×	×	×	×	×

where ~ means that the D1-branes have been smeared along the given direction. The coordinates  $x_1, \ldots, x_4$  parametrise the non-compact  $\mathbb{R}^4$  directions not wrapped by the brane system. The  $\mathbb{S}^1_y$  circle, parametrised by y, is the fifth coordinate and is shared by both types of branes. The last four coordinates parametrise the torus. The tendimensional Lorentz group is broken to  $SO(1,1) \times SO(4)_E \times SO(4)_I$ , where "E" stands for external and "I" for internal<sup>17</sup>. The Killing spinor equations lead to

$$\Gamma^{056789}\epsilon_L = \epsilon_R , \qquad \Gamma^{05}\epsilon_L = \epsilon_R , \qquad (2.32)$$

which shows that the brane configuration is 1/4-BPS<sup>18</sup>, hence the total number of supercharges for the asymptotically flat solution is 8. The supergravity fields sourced by this brane configuration are given by

$$d\tilde{s}^{2} = \frac{1}{(H_{1}H_{5})^{\frac{1}{2}}} (-dt^{2} + dy^{2}) + (H_{1}H_{5})^{\frac{1}{2}} (dr^{2} + r^{2}d\Omega_{S^{3}}^{2}) + \left(\frac{H_{1}}{H_{5}}\right)^{\frac{1}{2}} ds_{\mathbb{T}^{4}}^{2},$$
  

$$F^{(3)} = -\frac{1}{\tilde{g}_{s}} dH_{1}^{-1} \wedge dt \wedge dy + 2\alpha' d_{5} \Omega_{S^{3}},$$
  

$$e^{2\tilde{\Phi}} = \tilde{g}_{s}^{2} \frac{H_{1}}{H_{5}}, \qquad H_{1,5} = 1 + \frac{Q_{1,5}}{r^{2}},$$
(2.33)

where we have denoted by a tilde the quantities in the D1-D5 frame, to distinguish them from the NS5-F1 one that we will use throughout this thesis. The metric is given in string frame, and  $\tilde{g}_s$  is the coupling constant at infinity<sup>19</sup>. Note that the *r* dependence of the harmonic function  $H_1$  takes into account the smearing of the D1 branes. The coordinate *r* is the transverse radial distance from the brane sources in the non-compact directions. The charges  $Q_1, Q_5$  are quantised and, when parametrising the volume of the (square)  $\mathbb{T}^4$  in terms of its asymptotic volume *v* as measured from infinity,  $V_{\mathbb{T}^4}$  =

<sup>&</sup>lt;sup>17</sup>Properly the  $SO(4)_I$  group is broken to the  $\mathbb{T}^4$  to  $U(1)^4$ . We still consider the  $SO(4)_I$  acting on the tangent bundle of the torus, which will be useful for organising the matter fields.

<sup>&</sup>lt;sup>18</sup>Recall that each spinor  $\epsilon_{L,R}$  has 16 real independent components.

<sup>&</sup>lt;sup>19</sup>It can be seen as a boundary condition, since the dilaton has a profile that changes with *r*.

 $(2\pi\ell_s)^4 \tilde{v}$ , their expressions are given by

$$Q_1 = \frac{\tilde{g}_s d_1 \alpha'}{\tilde{v}}, \qquad Q_5 = \tilde{g}_s d_5 \alpha'.$$
(2.34)

In what follows, we will sometimes consider the  $\mathbb{T}^4$  to be very small. In practice, this means we consider  $\tilde{v} \sim \mathcal{O}(1)$ , so that  $V_{\mathbb{T}^4} \sim (2\pi\ell_s)^4$ . As a consequence, the charges are now very similar  $Q_i = \tilde{g}_s d_i \ell_s^2$ , clearly exhibiting a symmetry under the exchange  $d_1 \leftrightarrow d_5$ . Note that the  $S_y^1$  circle is macroscopically large compared to the torus, since  $R_y^4 \gg V_{\mathbb{T}^4}$ . This means that the solution in Eq. (2.33) describes a *black string* in six dimensions. To obtain a black hole solution, one must dimensionally reduce on the circle.

We can consider the near horizon limit of (2.33) for the metric and dilaton. Roughly, this means neglecting the "1" in the harmonic functions<sup>20</sup>. We obtain

$$d\tilde{s}_{S}^{2} = \left(\frac{Q_{1}}{Q_{5}}\right)^{\frac{1}{4}} d\tilde{s}_{E}^{2} = \frac{r^{2}}{\sqrt{Q_{1}Q_{5}}} (-dt^{2} + dy^{2}) + \sqrt{Q_{1}Q_{5}} \frac{dr^{2}}{r^{2}} + \sqrt{Q_{1}Q_{5}} d\Omega_{S^{3}}^{2} + \sqrt{\frac{Q_{1}}{Q_{5}}} ds_{\mathbb{T}^{4}}^{2}$$

$$e^{2\tilde{\Phi}_{hor}} = \tilde{g}_{s,hor}^{2} = \tilde{g}_{s}^{2} \frac{Q_{1}}{Q_{5}} = \frac{\tilde{g}_{s}^{2}}{\tilde{v}} \frac{d_{1}}{d_{5}}.$$
(2.35)

where "S" and "E" stands for string and Einstein frame respectively. The near-horizon ten-dimensional string and gravitational coupling constants are

$$\tilde{g}_{s,\text{hor}}^2 = \frac{\tilde{g}_s^2}{\tilde{v}} \frac{d_1}{d_5}, \qquad 2\kappa_{\text{hor}}^2 = (2\pi)^7 \alpha'^4 \, \tilde{g}_{s,\text{hor}}^2.$$
(2.36)

Notably, the near horizon geometry is  $\operatorname{AdS}_3 \times \mathbb{S}^3 \times \mathbb{T}^4$ , as can be clearly seen by defining  $z = \sqrt[2]{Q_1Q_5}/r$ . The isometries are thus  $(SL(2,\mathbb{R}) \times SU(2))_L \times (SL(2,\mathbb{R}) \times SU(2))_R \times U(1)^4$ . More precisely, in presence of supersymmetry we should consider the isometry supergroup  $\operatorname{PSU}(1,1|2)_L \times \operatorname{PSU}(1,1|2)_R$ . The  $\operatorname{AdS}_3$  and  $\mathbb{S}^3$  part have the same size fixed in terms of the charges of the brane system as  $L_{\operatorname{AdS}} = R_{\operatorname{S}^3} = \sqrt[4]{Q_1Q_5}$ . In the near-horizon region, the number of supersymmetries gets enhanced from 8 to 16.

The scalar curvatures in this regime are given by

$$R = 0, \qquad R_{\mu\nu\rho\sigma}R^{\mu\nu\rho\sigma} = R_{\mu\nu}R^{\mu\nu} = \frac{24}{Q_1Q_5} = \frac{\tilde{v}}{\tilde{g}_s^2 d_1 d_5} \frac{24}{\alpha'^2}.$$
 (2.37)

In order to have a reliable Supergravity description, we require the above quantities to be small or, more precisely,  $L_{AdS} \gg \ell_s$  and  $R_{\mu\nu\rho\sigma}R^{\mu\nu\rho\sigma} = R_{\mu\nu}R^{\mu\nu} = \frac{24}{Q_1Q_5} \ll \frac{1}{\ell_s^4}$ . This means that stringy  $\alpha'$  corrections are negligible and that the background is gently curved. In addition, in order to suppress string loops, we require the near-horizon string coupling to be small  $e^{2\tilde{\Phi}_{hor}} = \tilde{g}_s^2 \frac{Q_1}{Q_5} \ll 1$ . Keeping  $\tilde{v}$  generic for this analysis, we

<sup>&</sup>lt;sup>20</sup>The Maldacena decoupling limit is properly performing by considering  $\alpha' \to 0$  while keeping  $U = r/\alpha'$  and  $\tilde{v}$  fixed. Note that  $\tilde{v}$  is a fixed scalar in the D1-D5 frame.

need to assume

$$L_{\rm AdS}^4 = Q_1 Q_5 \gg \alpha'^2$$
,  $\tilde{g}_s^2 \frac{Q_1}{Q_5} \ll 1$ . (2.38)

From (2.34) we learn that

$$Q_1 Q_5 = \frac{\tilde{g}_s^2 {\alpha'}^2}{\tilde{v}} d_1 d_5 , \qquad \frac{Q_1}{Q_5} = \frac{d_1}{d_5 \, \tilde{v}} , \qquad (2.39)$$

hence the above requirements become

$$\frac{\tilde{g}_s^2}{\tilde{v}} \gg \frac{1}{d_1 d_5} , \qquad \frac{\tilde{g}_s^2}{\tilde{v}} \ll \frac{d_5}{d_1} , \qquad (2.40)$$

where the first condition is to suppress stringy corrections while the second is to suppress loops. We thus need

$$\frac{1}{d_1 d_5} \ll \frac{\tilde{g}_s^2}{\tilde{v}} \ll \frac{d_5}{d_1}.$$
(2.41)

Note that the first inequality is consistent with what is expected for the supergravity regime to hold. At this point, it is natural and convenient to define the six-dimensional coupling constant

$$\tilde{g}_6^2 = \frac{\tilde{g}_s^2}{\tilde{v}}, \qquad (2.42)$$

which will appear in later discussions. Note that,  $\tilde{g}_6$  is a modulus in the D1-D5 frame since it explicitly depends on the asymptotic value  $\tilde{g}_s$  of the dilaton. On the other hand, the attractor mechanism [55] fixes  $\tilde{v}$  via the "fixed volume condition" in [44] to  $\tilde{v} = d_1/d_5$ . This can be seen by requiring that the volume of the torus in the near horizon regime is unit normalised  $\sqrt{\frac{Q_1}{Q_5}} = 1$ . Note that the above inequalities can also be recast as

$$\frac{d_1}{d_5} \ll \tilde{g}_6^{-2} \ll d_1 d_5 \,, \tag{2.43}$$

where the first inequality suppresses loops and the second suppresses stringy corrections.

The horizon of the black string solution is topologically  $S_y^1 \times S^3$ . We now want to calculate its horizon area and the macroscopic entropy. To achieve this, we must write the metric in the Einstein frame, which reads

$$ds_{6D,E}^{2} = H_{1}^{-\frac{3}{4}}H_{5}^{-\frac{1}{4}}\left(-dt^{2} + dy^{2}\right) + H_{1}^{\frac{1}{4}}H_{5}^{\frac{3}{4}}\left(dr^{2} + r^{2}d\Omega_{S^{3}}^{2}\right).$$
(2.44)

The near horizon geometry is

$$ds_{6D,E}^{2,r\to0} = \frac{r^2}{Q_1^{3/4}Q_5^{1/4}} \left(-dt^2 + dy^2\right) + \frac{Q_1^{1/4}Q_5^{3/4}}{r^2} dr^2 + Q_1^{1/4}Q_5^{3/4} d\Omega_{S^3}.$$
 (2.45)

The area of the horizon is thus (naively)

$$A_{\mathcal{H}} = \int_{\mathbb{S}^{1} \times \mathbb{S}^{3}} \sqrt{g_{\mathcal{H}}} \bigg|_{r=0} = 2\pi^{2} (2\pi) \sqrt{r^{2} Q_{1}^{-3/4} Q_{5}^{-1/4} \left(Q_{1}^{1/4} Q_{5}^{3/4}\right)^{3}} \bigg|_{r=0} = 0.$$
(2.46)

Since the horizon area vanishes, the entropy is trivial. We thus conclude that the *naive* two-charge black hole solution has vanishing entropy. In addition, the Ricci scalar behaves near r = 0 as  $R \sim r^{-2/3}$ , and we find a naked singularity. Some observations are in order:

*Remark* 2.2. The reason why the horizon of six-dimensional black string solution has zero area is associated to the fact that the  $S_y^1$  shrinks to zero size. This is due to the tension of the branes wrapping the circle.

*Remark* 2.3. The fact that, naively, the two-charge six-dimensional black string solution has zero entropy implies that the two-charge five-dimensional black hole has zero entropy as well. This can be seen after performing a dimensional reduction along the  $S_{y}^{1}$ , that leads to the following five-dimensional metric in Einstein frame:

$$ds_{5D,Ein}^2 = -(H_1H_5)^{-2/3}dt^2 + (H_1H_5)^{1/3}(dr^2 + r^2d\Omega_{S^3}^2).$$
(2.47)

Repeating the above steps for the computation of the horizon area, we obtain

$$A_{\mathcal{H}} = \int_{\mathcal{H}=S^3} \sqrt{g_{\mathcal{H}}} = 2\pi^2 \left[ \sqrt[6]{\frac{Q_1}{r^2} \frac{Q_5}{r^2}} r \right]^3 \bigg|_{r=0} = 2\pi^2 \sqrt{Q_1 Q_5} r \bigg|_{r=0} = 0, \quad (2.48)$$

as claimed.

There is an important subtlety about the entropy of the two-charge naive solution. By looking at the above results, we would be tempted to conclude that the small black hole has zero macroscopic entropy. However, by a sequence of T and S dualities, one can map the D1-D5 system to a fundamental string with momentum, whose microscopic entropy is given by

$$S_{F1-P} = 2\pi \sqrt{2n_1 n_p} \,, \tag{2.49}$$

which clearly differs from our macroscopic expectation. Sen [69] argued that in the D1-D5 frame, the entropy should be obtained by counting solutions with no horizons, while in the U-dual NS1-P frame the entropy should be obtained in a different manner, using Wald's entropy formula for geometries with horizons. This could lead to the idea that the computation of the entropy somehow frame (gauge) dependent. However, in [70] it is argued that this is *not* the case: the microstate are smooth and horizonless (or, more generally, fuzzballs) in any frame.

## 2.3.2 The NS5-F1 system

The D1-D5 system can be mapped, via S-duality, to the NS5-F1 system. The advantage of the latter is the presence of only NS-NS fluxes. Indeed, a closed string propagating in those backgrounds can be described via a class of  $\alpha'$ -exact cosets involving SL(2,  $\mathbb{R}$ ) × SU(2) Wess-Zumino-Witten models. For this reason, it is important to study the NS5-F1 system in great detail and understand its differences with respect to its D1-D5 cousin.

The Supergravity fields sourced by the NS5-F1 system are

$$ds^{2} = \frac{1}{H_{1}} (-dt^{2} + dy^{2}) + H_{5} (dr^{2} + r^{2} d\Omega_{S^{3}}^{2}) + ds_{\mathbb{T}^{4}}^{2},$$
  

$$H^{(3)} = -\frac{1}{g_{s}} dH_{1}^{-1} \wedge dt \wedge dy + 2\alpha' n_{5} \Omega_{S^{3}},$$
  

$$e^{2\Phi} = g_{s}^{2} \frac{H_{5}}{H_{1}}, \qquad H_{1,5} = 1 + \frac{Q_{1,5}}{r^{2}},$$
(2.50)

with

$$Q_1 = \frac{g_s^2 n_1 \alpha'}{v}$$
,  $Q_5 = n_5 \alpha'$ , (2.51)

where v enters the torus volume as  $V_{\mathbb{T}^4} = (2\pi \ell_s)^4 v$ . Note that when  $g_s \to 0$ , the above solution coincides with the CHS background [71] since the NS5-branes become much heavier than the background fundamental strings.

The near-horizon fields are

$$ds^{2} = \frac{r^{2}}{Q_{1}}(-dt^{2} + dy^{2}) + Q_{5}\frac{dr^{2}}{r^{2}} + Q_{5}d\Omega_{S^{3}}^{2} + ds_{\mathbb{T}^{4}}^{2},$$
  

$$H^{(3)} = 2\alpha' n_{5} (\operatorname{vol}_{AdS_{3}} + \operatorname{vol}_{S^{3}}),$$
  

$$e^{2\Phi_{hor}} = g_{s,hor}^{2} = g_{s}^{2}\frac{Q_{5}}{Q_{1}} = v \frac{n_{5}}{n_{1}}.$$
(2.52)

The near-horizon geometry is again  $AdS_3 \times S^3 \times T^4$  as one can see with the same change of coordinates  $z = \sqrt[2]{Q_1Q_5}/r$ . However, this time the curvature scale is set by  $Q_5$  only. Indeed, in this regime, the curvature invariants read

$$R = 0, \qquad R_{\mu\nu\rho\sigma}R^{\mu\nu\rho\sigma} = R_{\mu\nu}R^{\mu\nu} = \frac{24}{Q_5^2} = \frac{24}{n_5^2{\alpha'}^2}.$$
 (2.53)

To suppress curvature and loop effects respectively, one demands

$$1 \ll n_5 \ll \frac{n_1}{v}$$
, (2.54)

and can equivalently recast the above bounds as

$$\frac{v}{n_5} \ll v \ll \frac{n_1}{n_5}$$
, (2.55)

where v is a modulus for the NS5-F1 system. Indeed, in the NS5-F1 frame the string coupling constant at the horizon is a fixed scalar independent of the asymptotic value  $g_s$  of the dilaton. As a consequence, the six-dimensional coupling constant is attracted to the value [44]

$$g_6^2 = \frac{g_{s,\text{hor}}^2}{v} = \frac{n_5}{n_1},$$
 (2.56)

and is important to note that is independent of the modulus v. Note that T-duality transformations can exchange  $v \mapsto \frac{1}{v}$ , hence we can restrict ourselves to  $v \ge 1$ . When considering the  $\alpha'$ -exact WZW models, the condition ensuring absence of stringy corrections can be relaxed. In other words, when dealing with worldsheet string theory, it is not necessary to consider  $n_5 \gg 1$ , which is only required for the validity of the supergravity solution. However, we will continue to suppress string loops and thus we will always work in a regime where  $n_1 \gg n_5$ .

## 2.3.3 Comparing the D1-D5 and the NS5-F1 system and their moduli

We now remark similarities and differences of the two S-dual systems. Recall that the tilde quantities refer to the D1-D5 system.

*Remark* 2.4. In the D1-D5 system, the six-dimensional coupling constant  $\tilde{g}_6^2$  is a modulus while the torus volume parametrised by  $\tilde{v}$  is a fixed scalar. In the NS5-F1 system, v is a modulus while the six-dimensional coupling constant  $g_6^2$  is a fixed scalar.

We now show how S-duality relates the two modulus and the two fixed scalars. We start with the moduli. Recall that

$$\tilde{g}_{s,\text{hor}}^2 = \frac{\tilde{g}_s^2}{\tilde{v}} \frac{d_1}{d_5}, \qquad g_{s,\text{hor}}^2 = v \frac{n_5}{n_1}.$$
(2.57)

However, because of S-duality  $\tilde{g}_{s,\text{hor}}^2 = g_{s,\text{hor}}^{-2}$ , hence we get

$$\frac{\tilde{g}_s^2}{\tilde{v}}\frac{d_1}{d_5} = \frac{n_1}{vn_5},$$
(2.58)

and since under S-duality the number of branes do not change  $d_i = n_i$ , we finally get

$$\frac{\tilde{g}_s^2}{\tilde{v}} = \tilde{g}_6^2 = \frac{1}{v} \,. \tag{2.59}$$

This shows that in the NS5-F1 frame v behaves like the six-dimensional coupling constant. It is very easy to show how to relate the fixed moduli. Recall their fixed value:

$$g_6^{-2} = \frac{n_1}{n_5}, \qquad \tilde{v} = \frac{d_1}{d_5},$$
 (2.60)

but since under S-duality  $n_i = d_i$ , it immediately follows that

$$g_6^{-2} = \tilde{v}$$
. (2.61)

*Remark* 2.5. Recall that the unit normalisation of the torus volume in the D1-D5 system required  $\tilde{v} = \frac{d_1}{d_5}$ , which gives the fixed scalar condition for  $\tilde{v}$ . At the same time, from the previous remark we have

$$\tilde{g}_{s,\text{hor}}^2 = \frac{\tilde{v}}{v} = \frac{n_1}{v \, n_5} \,.$$
 (2.62)

where we have used the fact that  $d_i = n_i$ .

*Remark* 2.6. The two pictures are related via the non-perturbative S-duality and hence we do not expect both tree-level supergravity to be valid at the same time. Allowing general stringy corrections, let's consider the various strong coupling regimes. Recall that

$$\tilde{g}_{s,\text{hor}}^2 = \frac{\tilde{g}_s^2}{\tilde{v}} \frac{d_1}{d_5} = \frac{n_1}{v n_5} = g_{s,\text{hor}}^{-2} \,. \tag{2.63}$$

Requiring that the D-brane picture is weakly coupled demands

$$\frac{\tilde{g}_s^2}{\tilde{v}}\frac{d_1}{d_5} = \frac{n_1}{vn_5} \ll 1 \tag{2.64}$$

which is clearly a *strong coupling* requirement for the NS5-F1 picture. We thus conclude that both pictures cannot be valid at the same time if we demand the coupling to be small.

## 2.3.4 Bulk microstates of the 2-charge system

We now recall, in two different U-dual frames, the supergravity solutions corresponding to microstates of the two-charge black hole. As explained in [67], in order to properly account for the full entropy of the two-charge system [68], one needs to consider microstates with both bosonic and fermionic [67] excitations. In [65], the bosonic excitations were purely external along the transverse  $\mathbb{R}^4$ , while in [64, 72] excitations along the internal manifold were considered. For simplicity, we focus on those with external bosonic excitations, thus a subset of microstates collectively known as Lunin-Mathur solutions [64, 65].

#### F1-P frame

Consider the same compactification manifolds discussed above for the two-charge black hole. Let's parametrise the distinguished circle with  $0 \le y < L$ . Let's consider a fundamental string (F1) with  $n_w$  units of winding and  $n_p$  units of momentum, both on the circle  $S_y^1$ . The strings wounds  $n_w$  times the physical circle, but in the covering space of the  $\mathbb{S}_y^1$  it looks single wounded with length  $L_T = n_w L$ . Let's parametrise this covering space with the coordinate  $0 \le \hat{y} < L_T$ , and let's define  $\hat{v} = t - \hat{y}$ . In the classical supergravity limit of large  $n_w n_p$ , coherent (micro-)states are well described by a continuous<sup>21</sup> periodic vibration profile function  $\mathbf{F}(\hat{v}) = \mathbf{F}(\hat{v} + L_T)$  which we take to be a function in  $\mathbb{R}^4$ , ignoring the four torus for simplicity.

The F1-P supergravity solutions are given by

$$ds^{2} = \frac{1}{H} \left( -dudv + K dv^{2} + 2A_{i}dx^{i}dv \right) + ds_{\mathbb{R}^{4}}^{2} + ds_{\mathbb{T}^{4}}^{2}$$
(2.65a)

$$B_{uv} = -\frac{1}{2}(H^{-1} - 1)$$
,  $B_{vi} = H^{-1}A_i$  (2.65b)

$$e^{2\Phi} = H^{-1}$$
, (2.65c)

where

$$H = 1 + \frac{Q_1}{L_T} \int_0^{L_T} \frac{d\hat{v}}{|\mathbf{x} - \mathbf{F}(\hat{v})|^2} , \qquad A_i = -\frac{Q_1}{L_T} \int_0^{L_T} \frac{\dot{\mathbf{F}}(\hat{v})d\hat{v}}{|\mathbf{x} - \mathbf{F}(\hat{v})|^2} , K = \frac{Q_1}{L_T} \int_0^{L_T} \frac{|\dot{\mathbf{F}}(\hat{v})|^2 d\hat{v}}{|\mathbf{x} - \mathbf{F}(\hat{v})|^2} .$$
(2.66)

These solutions are horizonless, as microstates are expected to be. They are also smooth, up to a physical (thus allowed) singularity associated to the presence of the string source. The same properties continue to hold for solutions with fermionic and internal excitations.

#### D1-D5 frame

After a sequence of T and S dualities, we can map Eq. (2.65) to the D1-D5 frame. The sequence of dualities that leads to the D1-D5 system is given by  $S T_{\mathbb{T}^4} S T_{\mathbb{S}^1_{\nu}} S$ .

The D1-D5 supergravity solutions, in the more compact notation of [68], are then given by

$$ds_{S}^{2} = \frac{1}{f_{1}f_{5}} \left[ -\left(dt + A_{i}dx^{i}\right)^{2} + \left(dy + B_{i}dx^{i}\right)^{2} \right] + \sqrt{f_{1}f_{5}} ds_{\mathbb{R}^{4}}^{2} + \sqrt{\frac{f_{1}}{f_{5}}} ds_{\mathbb{T}^{4}}^{2} ,$$

$$C^{(2)} = \frac{1}{f_{1}}(dt + A) \wedge (dy + B) + \mathcal{C} ,$$

$$e^{2\Phi} = \frac{f_{1}}{f_{5}} ,$$
(2.67)

<sup>&</sup>lt;sup>21</sup>In other words, we are smearing over the strands of the string in the base space, assuming a continuous distribution.

where the harmonic functions and forms are

$$f_{1} = 1 + \frac{Q_{5}}{L} \int_{0}^{L} \frac{|\mathbf{F}'(s)|^{2}}{|\mathbf{x} - \mathbf{F}(s)|^{2}} ds , \qquad f_{5} = 1 + \frac{Q_{5}}{L} \int_{0}^{L} \frac{1}{|\mathbf{x} - \mathbf{F}(s)|^{2}} ds A^{(1)} = \frac{Q_{5}}{L} \int_{0}^{L} \frac{F_{i}(s)}{|\mathbf{x} - \mathbf{F}(s)|^{2}} ds dx^{i} , \qquad dB^{(1)} = \star_{\mathbb{R}^{4}} dA^{(1)} , \qquad d\mathcal{C}^{(2)} = -\star_{\mathbb{R}^{4}} df_{5} .$$
(2.68)

The charges are related as [73]

$$Q_1 = \frac{Q_5}{L_T} \int_0^L |\mathbf{F}'(s)|^2 ds \,. \tag{2.69}$$

Again, these solutions are parametrized by the closed curve

$$x_i = F_i(s)$$
,  $s \in (0, L)$ ,  $\forall i = 1, ..., 4$ , (2.70)

and in order to ensure regularity of the fields, it is important that the curve is smooth and non-selfintersecting. These solutions are horizonless and, as opposed to the previous case, completely smooth.

One can immediately notice that if we take

$$f_{1,5} \to 1 + \frac{Q_{1,5}}{|x|^2} = H_{1,5} , \qquad A_i, B_i \to 0 ,$$
 (2.71)

one recovers the *naive* D1-D5 metric Eq. (2.33). This shows how at large distances the above microstates resemble the naive black hole, even though they start to differ at small distances from r = 0.

It has been shown in [68] that an appropriate quantisation of the moduli space of the Lunin-Mathur solution, together with the more general solutions of [67], gives the correct counting of the entropy for the D1-D5 black hole. This has a remarkable conceptual consequence: for the small black hole, all the microstates are described within supergravity and the metric field describes smooth and horizonless geometries that asymptotically look like the naive solution. This is a striking confirmation, although so far limited to the two charge case, of the Fuzzball Proposal. It should be noted, however, that in the case of three charge black holes it is expected that stringy physics may actually be important in recovering the full entropy of a macroscopic black hole.

## 2.3.5 The D1-D5-P large black hole

We now turn to the extremal non-rotating D1-D5-P black hole [63]. The setup is the same as that of the D1-D5 solution, but now  $n_p$  units of momentum wave are added along the  $S_y^1$  circle, and smeared along the  $\mathbb{T}^4$ . We add this wave holomorphically, in

order to preserve some supersymmetries. In this case, the asymptotically flat solution preserves 1/8 of the original 32 supercharges.

The line element in Einstein frame of the five-dimensional black hole reads

$$ds_{5D,E}^{2} = -\lambda^{-2/3} dt^{2} + \lambda^{1/3} \left( dr^{2} + r^{2} d\Omega_{S^{3}}^{2} \right) , \qquad (2.72a)$$

$$\lambda = \prod_{i=1,5,p} \left( 1 + \frac{Q_i}{r^2} \right) , \qquad (2.72b)$$

$$Q_1 = \frac{g_s n_1 \ell_s^6}{V}$$
,  $Q_5 = g_s n_5 \ell_s^2$ ,  $Q_P = \frac{g_s^2 n_p \alpha'^2}{R_y^2 V}$ . (2.72c)

This time the area of the horizon is finite, giving

$$A_{\mathcal{H}} = 2\pi^2 \sqrt{Q_1 Q_5 Q_P} = 2\pi^2 \frac{g_s^2 \alpha'^4}{R_y V} \sqrt{n_1 n_5 n_p}, \qquad (2.73)$$

and thus the Bekenstein-Hawking entropy is given by

$$S_{BH} = \frac{A_{\mathcal{H}}}{4G_N} = 2\pi \sqrt{n_1 n_5 n_p},$$
 (2.74)

where the expression of the five-dimensional Newton constant has been used.

As expected, the moduli do not appear in the expression of the entropy and the Uduality symmetry is manifest. More generally, the five-dimensional expression for the entropy is given in terms of the Surd cubic invariant  $\triangle$  as  $S = 2\pi\sqrt{\triangle}$ . This is a consequence of the non-compact  $E_{6,(6)}$  exceptional symmetry that characterises M or string theory toroidal compactifications.

The microscopic counting of the above entropy was performed for the first time in [19] by Strominger and Vafa. This was achieved by counting the degeneracy of massless open string states stretched between the D1 and the D5 branes. The result, obtained at weak coupling, was then successfully extrapolated at strong coupling, being the elliptic genus protected by supersymmetry. Strominger and Vafa showed how String Theory was able to account for the microscopic degrees of freedom of the D1-D5-P black hole. Their work was then generalised, in various dimensions, to different non-extremal and rotating black holes. See for instance the non-exhaustive list [74, 75, 62, 76]

# 2.4 From the brane system to the D1-D5 CFT

In this section we review the definition of the D1-D5 CFT starting from the brane setup in string theory. We will see that the D1-D5 CFT naturally arises from the low-energy dynamics of the sigma model on the moduli space of  $n_1$  instantons of a  $U(n_5)$  gauge theory on  $\mathbb{T}^4$ .

## 2.4.1 The instanton description

Consider the pure D1-D5 system, where both branes are wrapped on the circle and the  $n_5$  D5-branes also wrap the torus. The theory on the D5-branes is a 5 + 1-dimensional theory, whose support is  $\mathbb{R}_t \times \mathbb{S}_y^1 \times \mathbb{T}^4$ . The  $n_1$  D-strings are extended along  $\mathbb{R}_t \times \mathbb{S}_y^1$  and are thus localised with respect to the torus directions. From the DBI action, one notices that the low-energy theory living on the worldvolume of the D5-branes is the six-dimensional SYM theory. Considering the normal and tangent bundles to be flat and  $B^{(2)} = 0$ , expanding the Wess-Zumino part of the D5-brane action as in Eq. (2.7) one obtaines a term of the form

$$\int_{\mathcal{M}_6} C^{(2)} \wedge \operatorname{Tr}\left(F^{(2)} \wedge F^{(2)}\right) \propto \int_{\mathcal{M}_6} C^{(2)} \wedge \operatorname{ch}_2(F) \,. \tag{2.75}$$

We note that Yang-Mills instantons (contributing to the second Chern class) carry a RR two-form charge. In particular, the D1-branes are instantons for the Euclidean fourdimensional theory on  $\mathbb{T}^4$  inside the D5-branes. To see this, one has to consider gauge field configurations which are independent of  $\mathbb{R}_t \times S_y^1$  and whose field strength is selfdual with respect to the  $\mathbb{T}^4$ . Noting that, when integrating the (t, y)-independent connections on  $\mathbb{T}^4$ , we find  $ch_2(F) = n_1$ . This is the instanton number of the gauge group, and hence we learn that in the  $n_1$  instanton sector of the gauge theory we have  $n_1$  D1brane charge units<sup>22</sup> [77]. Note that for the pure D1-D5 system, the first Chern-class vanishes as indicated by the absence of a coupling of the form  $C^{(4)} \wedge \text{Tr}(F)$ .

As explained in [49], in the limit where the  $\mathbb{T}^4$  is small, the theory on  $\mathbb{R}_t \times \mathbb{S}_y^1 \times \mathbb{T}^4$ effectively reduces to a 2D sigma model with base space  $\mathbb{R}_t \times \mathbb{S}_y^1$  with target space the moduli space of  $U(n_5)$  instantons on  $\mathbb{T}^4$ . The moduli space of  $n_1$  instantons of a  $U(n_5)$  gauge theory on  $\mathbb{T}^4$  is the the smooth manifold  $\mathcal{M}_{\text{Inst}}$  which is a resolution of the symmetric product orbifold  $\text{Sym}_{n_1n_5}(\tilde{\mathbb{T}}^4) = (\tilde{\mathbb{T}}^4)^{\otimes n_1n_5}/S_{n_1n_5}$ , namely

$$\mathcal{M}_{\text{Inst}} \xrightarrow{\pi} \text{Sym}_{n_1 n_5}(\tilde{\mathbb{T}}^4),$$
 (2.76)

where  $S_N$  is the permutation group of N elements,  $\tilde{\mathbb{T}}^4$  is in general different from the compactification torus  $\mathbb{T}^4$  [78] and  $\pi$  is a suitable projection map. The space  $\mathcal{M}_{\text{Inst}}$  is a simply-connected Hyperkähler space<sup>23</sup> and has the same cohomology as  $\text{Sym}_{n_1n_5}(\tilde{\mathbb{T}}^4)$ . If the charge vector v is taken to be primitive, then this space is also smooth [49].

<sup>&</sup>lt;sup>22</sup>The case of K3 has a correction due to the non-vanishing first Pontryagin class [77].

<sup>&</sup>lt;sup>23</sup>A manifold (*M*, *I*, *J*, *K*, *g*) is Hyperkähler if *I*, *J*, *K*  $\in$  End(*TM*) satisfy Hamilton's quaternionic relations, are integrable complex structures (the Nijenhuis tensor vanishes by Newlander–Nirenberg theorem) and the Riemannian metric *g* is Kähler with respect to *I*, *J*, *K*. Assuming a Levi-Civita connection  $\nabla$ , one can show that *I*, *J*, *K* are parallel  $\nabla I = \nabla J = \nabla K = 0$  and hence the holonomy is contained in the group Sp(n). The three Kähler forms are  $\omega_I(\cdot, \cdot) = g(\cdot, I \cdot)$  and similarly for *J*, *K*.

As discussed in previous sections, our 'pure' brane system is marginally unstable and suffers from fragmentation at no cost in energy. How can we see this feature in the instanton description? The intuitive explanation is that singularities of the moduli space are associated to a D1-brane instanton that can become small and thus separate from the D1-D5 system. These are the so-called 'long strings' [50], whose signature will be present also in the pure NS worldsheet description in terms of the so-called 'missing chiral primaries'. To dive into the details of this process, we first turn to the gauge theory description of the D1-D5 system.

#### 2.4.2 The gauge theory description

Given that the number of supercharges preserved by the system is 8, from the discussion above we conclude that the low-energy dynamics of the pure D1-D5 brane system is described by a 2D small<sup>24</sup>  $\mathcal{N} = (4, 4)$  sigma model from  $\mathbb{R}_t \times \mathbb{S}_y^1$  to  $\mathcal{M}_{\text{Inst}}$ . Performing an S-duality, one gets analogous conclusions for the NS5-F1 system. As we will discuss below, this sigma model flows in the IR to a 2D  $\mathcal{N} = (4, 4)$  Superconformal field theory known as the D1-D5 CFT. However, in this case the number of fermionic symmetries is enhanced from 8 to 16, where the new 8 symmetries are given by fermionic generators of the superconformal algebra (usually denoted by *S*). This is a similar phenomenon that is found in the case of the 4d  $\mathcal{N} = 4$  Super Yang-Mills theory, where the fermionic symmetries are 32: 16 supercharges and 16 fermionic generators of the superconformal algebra.

As we will discuss in due course, the above sigma model has a moduli space which coincides with Eq. (2.15) [78, 49]. One thus concludes that the D1-D5 system is on the moduli space of this sigma model [50], which legitimates the study of the dynamics of low-energy open string excitations.

The gauge theory description, less general then the instanton description, is used when  $1 \ll n_1, n_5 \ll 1/g_s$ , since the backreaction of the branes is negligible and thus gravity effects can be ignored. The branes are boundary conditions for the open strings stretched between them. Let's denote the  $\mathbb{R}_t$  and  $\mathbb{S}_y^1$  directions as  $x_0, x_5$  respectively. The non-compact direction are  $x_1, \ldots, x_4$  while the  $\mathbb{T}^4$  is parametrised by  $x_6, \ldots, x_9$ . The various massless strings are (1, 1), (1, 5), (5, 1), (5, 5) strings. They give rise to the following matter content:

#### • (1,1) Strings

These strings have NN boundary conditions along  $x_0$ ,  $x_5$  and DD along the others. Along the first two directions, the massless excitations give two components of

<sup>&</sup>lt;sup>24</sup>There are two different  $\mathcal{N} = 4$  superconformal algebras in two dimensions. The case of the D1-D5 CFT is referred as the small  $\mathcal{N} = 4$  superalgebra, where only one  $SO(4)_R$  is realised as R-symmetry instead of the usual  $SO(\mathcal{N}_+) \times SO(\mathcal{N}_-)$  R-symmetry.

a vector,  $A_{x_0}$ ,  $A_{x_5}$ . The vector transforms in the adjoint of  $U(n_1)$  and is a  $n_1 \times n_1$  matrix. Excitations transverse to the D1-branes are scalars. We have 8 transverse directions and thus 8 adjoint scalars. Fermions are obtained imposing supersymmetry. Arranging the fields in multiplets of  $\mathcal{N} = 2$  in 4D, we can note that the fields give rise to one Hypermultiplet (it consists of 4 real scalars + 2 Weyl fermions) and a Vector multiplet (one vector and two reals scalars + 2 Weyl fermions).

How can we decide whether four of the eight scalars should be part of the Hyper or part of the Vector multiplet? The answer is R-Symmetry! We know that inside the<sup>25</sup>  $SO(4)_{Int} \simeq SO(3) \times SO(3) \simeq SU(2) \times SU(2)$  one of the  $SU(2)_R$  acts as an R-Symmetry group, rotating the Hypers. The Hypermultiplet is thus formed by those scalars that are charged under R-Symmetry and hence they correspond to the internal coordinates  $x_6, \ldots, x_9$ . In conclusion, we have (at fixed  $n_1, n_5$ )

- One Hypermultiplet whose four scalars are denoted by

$$Y_{x_6}^{(1,1)} \qquad Y_{x_7}^{(1,1)} \qquad Y_{x_8}^{(1,1)} \qquad Y_{x_9}^{(1,1)} \tag{2.77}$$

These real scalars can be recast in a complex doublet transforming in the **2** of  $SU(2)_R$ . The total number of scalars in the hypermultiplets is  $4n_1^2$ .

- One Vector multiplet with one vector and two scalars:

$$A_{x_0}^{(1,1)} \qquad A_{x_5}^{(1,1)} \qquad Y_{x_1}^{(1,1)} \qquad Y_{x_2}^{(1,1)} \qquad Y_{x_3}^{(1,1)} \qquad Y_{x_4}^{(1,1)}$$
(2.78)

All these fields transforms in the adjoint of  $U(n_1)$  since the strings have endpoints on the same stack of D1-branes.

#### • (5,5) Strings

Since the torus is of the order of the string scale, the winding and the momentum modes of the string goes like  $\sim 1/\ell_s$ . This means that if we study the system at energies  $E \ll 1/\ell_s$ , we are allowed to neglect these modes. The  $x_5$  coordinate is taken to be compactified on a circle of radius R which is much bigger than the string length. In this respect, we can treat the  $x_5$  coordinate to be non-compact  $x_5 \in \mathbb{R}$  instead of  $x_5 \in \mathbb{S}^1_y$ . The theory on the D5-branes is seen at low energies as (1+1)-dimensional. The field content is thus analogous to that of the (1,1) strings (at fixed  $n_1, n_5$ ):

- One Hypermultiplet

$$Y_{x_6}^{(5,5)} \qquad Y_{x_7}^{(5,5)} \qquad Y_{x_8}^{(5,5)} \qquad Y_{x_9}^{(5,5)} \tag{2.79}$$

<sup>&</sup>lt;sup>25</sup>There is a subtlety: since we are wrapping the D5 branes on a  $\mathbb{T}^4$  it is not completely correct to claim that the symmetry group for the internal coordinates is SO(4). This is *locally* correct and the SO(4) is a useful approximate symmetry.

These real scalars can also be recast in a complex doublet transforming in the **2** of  $SU(2)_R$ . The total number of scalars in the hypermultiplets is  $4n_5^2$ .

- One Vector multiplet:

$$A_{x_0}^{(5,5)} \qquad A_{x_5}^{(5,5)} \qquad Y_{x_1}^{(5,5)} \qquad Y_{x_2}^{(5,5)} \qquad Y_{x_3}^{(5,5)} \qquad Y_{x_4}^{(5,5)} \tag{2.80}$$

All the fields are in the adjoint of  $U(n_5)$ .

#### • (1,5) and (5,1) Strings

The quantisation of these strings is slightly more subtle and requires some care and is not often spelled out in great detail in the literature [79]. First of all, recall that worldsheet boson for the open string are integer moded when the boundary conditions are NN or DD. When the latter are of the type DN or ND, the bosons are half-integer moded. The same is true the fermions in the Ramond sector. However, fermions in the NS sector are integer moded for DN and ND boundary conditions and are half-integer moded for NN and DD boundary condition<sup>26</sup>. The moding affects the zero point energies "E" as follows:

$$E_{X,\mathbb{Z}} = -E_{\psi,\mathbb{Z}} = -\frac{1}{24}$$
,  $E_{X,\frac{\mathbb{Z}}{2}} = -E_{\psi,\frac{\mathbb{Z}}{2}} = \frac{1}{48}$ , (2.81)

where "*X*,  $\psi$ " stand for boson and fermion while  $\mathbb{Z}$  and  $\frac{\mathbb{Z}}{2}$  stands for integer and half-integer moding. Let's denote by  $\nu$  the number of DN or ND directions. Taking into account all the possible boundary conditions in *d* space-time dimensions, the total vacuum energy for a boson is

$$E_X^{\text{tot}} = (d - 2 - \nu) E_{X,\mathbb{Z}} + \nu E_{X,\frac{\mathbb{Z}}{2}}.$$
 (2.82)

Note that the vacuum energy for the Ramond sector is exactly the opposite  $E_X^{\text{tot}} = -E_{\psi_R}^{\text{tot}}$  and the vacuum energies exactly cancel. On the other hand, the total vacuum energy for the fermions in the NS sector is

$$E_{\psi_{NS}}^{\text{tot}} = (d - 2 - \nu) E_{\psi_{NS}, \frac{\mathbb{Z}}{2}} + \nu E_{\psi_{NS}, \mathbb{Z}}.$$
(2.83)

Note that for the critical superstring, we get

$$E_X^{\text{tot}} + E_{\psi_{NS}}^{\text{tot}} = -\frac{1}{2} + \frac{\nu}{8}.$$
 (2.84)

For the D1-D5 system,  $\nu = 4$  and thus the bosonic and fermionic vacuum energies cancel also in the NS sector. Let's discuss now the (1,5) or (5,1) strings. There are four periodic  $\psi_{NS}^i$  fermions in the torus directions i = 6,7,8,9 and they give  $2^{4/2} = 4$  degenerate ground states. Labelling their spins in the (i, j) plane with  $s_{ij} = \pm \frac{1}{2}$ , we get four states  $|s_{67}, s_{89}\rangle$  and they behave as two SU(2) doublets  $\mathbf{2} \oplus$ 

 $<sup>^{26}</sup>$ A reference for this can be [80], Eq. (7.70).

**2'** under the internal  $SO(4)_I$ , while being singlets under the external  $SO(1,1) \times SO(4)_E$  group. We now need to impose the GSO projection<sup>27</sup> with the operator exp  $(i\pi F)$ ,  $F = \sum_a s_a$ . Taking into account the extra sign coming from the ghosts, the GSO projection demands that exp  $(i\pi F) = -1$ , which implies  $s_{67} = s_{89} = \frac{1}{2}$ . Only the state with same spins survives, and transforms as a doublet **2**. This immediately implies that there are only two bosonic degrees of freedom, and not four.

One can ask what happens for the Ramond sector, which we expect would give 2 fermionic degress of freedom by supersymmetry. In the R sector, there are four transverse periodic fermions  $\psi_R^m$ , m = 1, ..., 4, along the non-compact directions external to both type of branes which thus cannot give fermionic contributions. In addition we have four internal anti-periodic fermions  $\psi_R^i$ : the R fermions along the ND and DN directions are half-integer moded and give fermionic states. The R vacuum give rise to  $2^{4/2} = 4 SO(4)_I$ -singlet ground states  $|s_{12}, s_{34}\rangle$ , out of which only two survive the GSO projection (which has no extra sign in the R sector of the ghosts) which requires  $s_1 = -s_2$ . These space-time fermions transform as a doublet **2'** under the little group  $SO(4)_E$ . We have thus concluded that the number of bosons coming from the (p, p') strings is only 2 and come from the NS sector. The additional 2 bosonic degrees of freedom come from the (p', p) strings in the NS sector. By supersymmetry, the R sector gives 2 fermionic degrees of freedom for each orientation.

It is straightforward to apply these results to our D1-D5 system. From the (1, 5) and (5, 1) strings we obtain  $2n_1n_5 + 2n_1n_5 = 4n_1n_5$  scalars from the NS sector, which constitute another type of Hypermultiplet, different from those considered for the (1,1) and (5,5) strings. The scalars can be recast in a complex doublet  $\chi = (A, B^{\dagger}), A, B \in \mathbb{C}$  and  $\chi$  is a chiral spinor satisfying  $\Gamma_{6789}\chi = \chi$ . These strings are bifundamental fields: for instance, the (1,5) strings transform in the fundamental of  $U(n_1)$  and in the anti-fundamental of  $U(n_5)$ . Note that is the presence of these strings that breaks the supersymmetry from 16 to 8, in agreement with the fact that the D1-D5 system is 1/4 BPS. From a 6D viewpoint, the theory on the D5-branes is a  $\mathcal{N} = 1$  theory with R-symmetry  $SU(2)_R$ . Note that the fermionic superpartners of the bifundamental scalars are singlets of  $SO(4)_I$  but are charged under the  $SU(2)_{1,2}$  in  $SO(4)_E \simeq SU(2)_1 \times SU(2)_2$ .

From the above discussions, and from Section 2.4.1, we conclude that the gauge theory of the D1D5 system is a (1+1)-dimensional  $\mathcal{N} = (4, 4)$  sigma model with gauge group  $U(n_1) \times U(n_5)$  and target space Eq. (2.76).

<sup>&</sup>lt;sup>27</sup>A reference for this is [45], Eq. (10.2.24) and Eq. (13.5.19).

#### 2.4.3 The D1-D5 superconformal field theory

The classical moduli space of vacua of the 2D sigma model is found by imposing the D-flatness equations, modulo the  $U(n_1) \times U(n_5)$  gauge group. Details can be found in [58]. Depending on the VEV of the scalars in the various multiplets, these can parametrise the Coulomb or the Higgs branch<sup>28</sup> of the theory. The Coulomb branch is spanned by the VEVs of the scalars in the vector multiplets, while the scalars in the hypermultiplets are set to zero. In this case, the branes can separate from each other and generically the gauge group is completely broken to  $U(1)^n$ . The Higgs branch is defined by the VEVs of the scalars in the hypermultiplets, having set to zero the scalars in the vector multiplet. The two branches meet at the origin (where all the scalars are zero) and it has been shown in [56] that the two remain distinct also quantum mechanically. To see this, one can consider the conformal field theory limit of the 2D sigma model. In fact, the latter flows to *two* distinct CFTs in the IR, one corresponding to the Coulomb branch and one corresponding to the Higgs branch of the theory [81]. Indeed, the two conformal field theories have different R-symmetry and usually different central charges [56]: the Coulomb branch has  $c = 6n_5$  while the Higgs branch has  $c = 6n_1n_5$  (or  $6(n_1n_5 + 1)$  for the K3 compactification). As a consequence, the moduli space of instantons has dimension  $\dim(\mathcal{M}_{Inst}) = 4n_1n_5$  (or  $4(n_1n_5+1)$  for the K3 compactification). Note that, for black hole applications, the Higgs branch is the most entropic sector of the theory<sup>29</sup> and the CFT limit corresponds to the near-horizon limit of the D1-D5 supergravity solution. The theory on the worldvolume of the effective 6d D-strings sees the attractor mechanism as its RG-flow. It is thus expected that the SCFT will be sensible to the 20-dimensional *attracted* moduli space  $\mathcal{M}^*_{Sugra}$ , so that the moduli space of the SCFT is  $\mathcal{M}_{SCFT} = \mathcal{M}_{Sugra}^*$  [49]. Indeed, this is know from the works of [82, 83] that locally the moduli space of any  $\mathcal{N} = (4, 4)$  SCFT is of the form  $SO(4, n)/SO(4) \times SO(n)$ , where n = 5,21 for  $\mathbb{T}^4$  and K3 respectively. This holds to be true also at first order in conformal perturbation theory [84]. Note that this is consistent with the idea found in usual AdS/CFT dualities where the radial space-time coordinate corresponds to the energy of the dual CFT theory.

We conclude that the low energy description of the D1-D5 system is a 2D  $\mathcal{N} = (4, 4)$ SCFT with moduli space  $\mathcal{M}_{SCFT} = \mathcal{M}_{Sugra}^*$  and with target space

$$\mathbb{R}^4 \times \left( \mathbb{T}^4 \times \mathcal{M}_{\text{Inst}} \right) , \qquad (2.85)$$

where the first factor describes the Coulomb branch and the last two the Higgs branch. The  $\mathbb{T}^4$  is a free theory describing the center of mass motion of the branes inside the torus.

<sup>&</sup>lt;sup>28</sup>There can be mixed branches, which we ignore for simplicity of discussion.

<sup>&</sup>lt;sup>29</sup>This can be seen by using the Cardy formula for the entropy, where the central charge contributes to the entropy.

The origin of the two branches is related to the small instanton singularity of the gauge theory, corresponding to the emission of long strings. Quantum mechanically, the Coulomb branch metric  $ds_{\mathbb{R}^4}^2 = dr^2 + r^2 d\Omega_{S^3}^2$  gets deformed to a tube-like metric of the form  $r^{-2} (dr^2 + r^2 d\Omega_{S^3}^2)$ . This shows that now the origin r = 0 is at infinite distance. The Higgs branch, being Hyperkähler, is protected from quantum corrections. However, using the correct variables near the singularities of the space, one can show that the physics near the origin is described in terms of a dynamical Liouville field which induces a strong coupling singularity [50]. Again, there is a tube-like behaviour at the quantum level. This means that, starting from the Higgs branch of the D1-D5 system, an instanton can travel through the infinitely long tube and 'come out' of the Coulomb branch. This emission of long strings has been shown to be a peculiarity only of the co-dimension four singular space [85]. By turning on the fields parametrising the additional four directions (for instance,  $B_{ij}^{SD} \neq 0$ ), the singularity gets resolved and the long string emission stops.

#### 2.4.4 Attracted moduli space and moduli space of instantons

It remains to understand what is the relation between the (attracted) moduli space  $\mathcal{M}_{SCFT} = \mathcal{M}_{Sugra}^*$  and the (non-trivial, Higgs branch) target space  $\mathcal{M}_{Inst}$ . To understand this, we follow [49]. The instanton moduli space has a natural hyperkähler metric induced<sup>30</sup> by the compactification manifold  $X = \mathbb{T}^4$ , K3, in terms of the  $L^2$ -metric on (adjoint-valued) one-forms on X. Additionally, one can turn on a (self-dual) B-field on X, which modifies the hyperkähler moment map used in the definition of  $\mathcal{M}_{Inst}$ . In particular, the B-field enters in the definition of the Kähler form while the complex structure of  $\mathcal{M}_{Inst}$  is determined by the complex structure of  $\mathcal{M}_{SCFT} = \mathcal{M}_{Sugra}^*$ . We thus conclude [49] that the 20-dimensional moduli space of the 2D  $\mathcal{N} = (4, 4)$  SCFT  $\mathcal{M}_{SCFT} = \mathcal{M}_{Sugra}^*$  should be matched with the hyperkähler structure of the instanton moduli space  $\mathcal{M}_{Inst}$ , which constitute the (non-trivial) part of the Higgs branch of the theory. Importantly, in the D1-D5 frame, when the B-field is set to zero, the hyperkähler metric is of the sigma model.

We now summarise the relation between the moduli spaces with the following diagram:

<sup>&</sup>lt;sup>30</sup>In general, not unique.



FIGURE 2.1: Summary of the relation between the various moduli spaces. The moduli space  $\mathcal{M}_{Sugra}$  and  $\mathcal{M}_{Gauge}$  flow to the same 'attracted' moduli space  $\mathcal{M}_{Sugra}^* \equiv_{loc} \mathcal{M}_{SCFT}$ , where the identification is understood to be local. The latter is related to the Hyperkähler structure of the (non-trivial part of the) Higgs branch  $\mathcal{M}_{Inst}$ , which is seen as a compact smooth Hyperkähler resolution of the orbifold Symm<sub>n ns</sub>( $\tilde{\mathbb{T}}^4$ ).

Below we compare the *bulk* physics (be it supergravity or the more general string theory description) to the *boundary* physics at various points of the moduli space.



FIGURE 2.2: A very qualitative cartoon of the relation between various points in moduli space, according to the AdS/CFT duality. The figure emphasizes that, despite having  $\mathcal{M}^*_{Sugra} \equiv_{loc} \mathcal{M}_{SCFT}$ , quantities computed in regime of low spacetime curvature can be compared to those obtained at orbifold point only if protected, and viceversa. Note that, in a more accurate representation, the marked points may actually reside at cusps or at the boundary of the moduli space.

A more precise representation and discussion about the moduli space of the D1-D5 system can be found in [44]. In particular, parametrizing a two-dimensional subspace of the moduli space with  $\tau = C^{(0)} + ig_s^{-1}$  at fixed  $N = n_1n_5$ , the free symmetric orbifold

point is found at  $i\infty$ . Deformations along the  $C^{(0)}$  direction (and  $C_{ij}^{(3),5D}$ ) would generically break the orbifold structure, and stop the long string emission [85]. On the other hand, changes along the imaginary axis would still lead to the presence of long strings in the spectrum. Notably, the bulk dual of the symmetric product orbifold theory at  $i\infty$  does not have a continuous spectrum given to the shortening of the representations [86].

One may thus wonder what is the relation between the D1D5 CFT at the orbifold point and the supergravity description of the D1-D5 system. These two points are different and only protected quantities can be compared between the two points. In addition, the supergravity solution admits a continuum of states that are not present in the discrete spectrum of the dual D1D5 CFT at the orbifold point, as mentioned above. For the NS5-F1 system with one unit of NS flux  $n_5 = 1$ , at a point of the moduli space  $\mathcal{M}^*_{Sugra} \simeq$  $\mathcal{M}_{SCFT}$  one finds that the Hyperkähler structure of the Higgs branch  $\mathcal{M}_{Inst}$  degenerates to that of an orbifold space and the holographic CFT becomes a particularly simple theory in terms of free bosons and fermions. Indeed, it has been recently shown [87, 88] that the bulk dual is the so-called tensionless string, described by a PSU(1,1|2) WZW model whose affine algebra  $\widehat{\mathfrak{psu}}(1,1|2)_1$  has level  $n_5 = 1$ . The level, coinciding with the number of five-branes, is identified with the size of AdS<sub>3</sub> in string units (which is the same as that of  $S^3$  for critical strings) and the tensionless regime describes the particular situation of having a string-sized target space  $L^2_{AdS}/\alpha' = 1$ . At the orbifold point, the relation between the string coupling and N is correct only at large values of the latter. As explained in [89], one must compare the genus expansion of the worldsheet theory with that of covering space of the dual orbifold theory: at a given genus g, there is an infinite series of subleading terms in powers of *N*.

The supergravity regime is reliable when the size of AdS is much larger to that of the string length. In case of only NS-NS fluxes (as an example, the NS5-F1-P system), some supergravity backgrounds are (partly) described by  $SL(2, \mathbb{R})$  Wess-Zumino-Witten model whose affine Lie algebra  $\widehat{\mathfrak{sl}(2, \mathbb{R})}_k$  has level  $k \in \mathbb{Z}_{>1}$ . For the case of the bosonic string in AdS<sub>3</sub> (a particular case of the more general model we consider in this thesis), the holographic CFT has been recently proposed [90, 91] to be a deformation of  $\operatorname{Sym}_N(\mathbb{R}_Q \times \mathbb{T}^4)$ , where  $\mathbb{R}_Q$  is a linear dilaton theory of slope  $Q = \frac{k-3}{\sqrt{k-2}}$ , where this time k is the bosonic affine level. The deformation is parametrised by the operator  $\Phi = e^{-\sqrt{k-2}\phi}\sigma_2$ , where  $\phi$  is the linear dilaton field and  $\sigma_2$  is a twist-2 operator of the symmetric orbifold theory.

## 2.5 The holographic CFT

As discussed, in the AdS<sub>3</sub> limit of the NS5-F1 system there is a locus in moduli space at which the holographically dual CFT is an  $\mathcal{N} = (4, 4)$  symmetric product orbifold CFT with target space  $(\mathbb{T}^4)^N / S_N$ , where  $N = n_1 n_5$ . To make the presentation selfcontained, we review some aspects of this theory and set the notation.

Recall that we work in Type IIB compactified on  $S^1 \times \mathbb{T}^4$ , with  $n_5$  NS5 branes wrapped on  $S^1 \times \mathbb{T}^4$ ,  $n_1$  units of F1 winding on  $S^1$ , and  $n_P$  units of momentum charge along  $S^1$ . The moduli space is 20-dimensional and the symmetric product orbifold CFT is conjectured to lie at a particular locus of this moduli space [44], see also [92]. The configuration breaks the SO(1,9) Lorentz group to SO(1,1) × SO(4)<sub>*E*</sub> × U(1)<sup>4</sup>, where the external R-symmetry SO(4)<sub>*E*</sub>  $\simeq$  SU(2)<sub>*L*</sub> × SU(2)<sub>*R*</sub> corresponds to rotations in the spatial  $\mathbb{R}^4$  transverse to the branes (in the IR limit, rotations of the S<sup>3</sup>). It is customary to introduce an approximate internal SO(4)<sub>*I*</sub>  $\simeq$  SU(2)<sub>1</sub> × SU(2)<sub>2</sub>, which is broken to U(1)<sup>4</sup> by the compactification, but which is useful for classifying states and organizing fields [58, 46].

In the symmetric product orbifold theory, for each copy of  $\mathbb{T}^4$  there are four free bosons, together with their left and right-moving fermionic superpartners. Indices  $\alpha$ ,  $\dot{\alpha}$ , A,  $\dot{A}$  correspond respectively to  $SU(2)_L$ ,  $SU(2)_R$ ,  $SU(2)_1$ ,  $SU(2)_2$ . The free fields are denoted as (we use the conventions of [93])

$$X_{A\dot{A}(r)}(z,\bar{z})$$
,  $\psi^{\alpha A}_{(r)}(z)$ ,  $\bar{\psi}^{\dot{\alpha} A}_{(r)}(\bar{z})$ , (2.86)

where the subscript (r) denotes the r-th copy of the seed  $\mathbb{T}^4$  theory. Omitting this copy subscript and focusing on the holomorphic sector, the energy-momentum tensor  $\mathbb{T}(z)$ , the supercurrents  $\mathbb{G}^{\alpha A}(z)$  and the SU(2)<sub>L</sub> currents  $\mathbb{J}^a$  generate the small (4,4) superconformal algebra. We denote holographic CFT conformal weights by h and Rsymmetry quantum numbers by  $(\hat{j}, m')$  and  $(\hat{j}, \bar{m}')$ , respectively. Thanks to the quotient with the symmetric group  $S_N$ , different copies of the theory can be combined together into strands of various lengths. Thus, in the orbifold theory we can distinguish two different sectors, namely the untwisted sector, where the theory is thought as a collection of N strands of length equal to one, and the twisted sector, where multiple strands are joined together and have length greater than one. The operator that implements such transformation is the twist operator  $\sigma_n := \sigma_{(12...n)}$ . More precisely, given a permutation (12...n), n < N, the operator  $\sigma_n$  implements the following boundary conditions on the fields:  $X_{(1)}^i \to X_{(2)}^i \to \cdots \to X_{(n)}^i \to X_{(1)}^i$ , while the remaining strands from n + 1 to N are left untouched. Another important feature of the orbifold theory is the presence of an automorphism of the su(2) subalgebra of the superconformal algebra, called holographic spectral flow<sup>31</sup>. This automorphism implements a change in the weight and R-charge of the states given by

$$h' = h + \alpha j + \frac{c \alpha^2}{24}$$
,  $m' = m + \frac{c \alpha}{12}$ ,  $\alpha \in \mathbb{R}$ . (2.87)

<sup>&</sup>lt;sup>31</sup>For the rest of the thesis, the reader should be careful in distinguishing the holographic spectral denoted by  $\alpha$  flow from the worldsheet spectral flow denoted by w, which instead represents an asymptotic winding of the strings in AdS<sub>3</sub>.

The heavy states we are interested in are obtained by fractional spectral flow [94], see also [95, 96]. We start from the state dual to  $\mathbb{Z}_k$ -orbifolded global  $AdS_3 \times \mathbb{S}^3 \times \mathbb{T}^4$ , which is a tensor product of N/k identical states of twist k, each of which is the lowest dimension state in the k-twisted sector. This state has conformal dimension

$$h = \bar{h} = \frac{c}{24} \left[ 1 - \frac{1}{k^2} \right],$$
 (2.88)

where the central charge c = 6N. The R-charges of this state are zero. Because all the component states (i.e. the "strands" mentioned above) have twist k, there is an enhancement of the usual spectral flow, such that one can perform spectral flow with fractional parameters,

$$\alpha = \frac{m+n}{k} = \frac{2s+1}{k}, \qquad \bar{\alpha} = \frac{m-n}{k} = \frac{2\bar{s}+1}{k},$$
(2.89)

where  $s, \bar{s} \in \mathbb{Z}$ , and we note that the range  $m > n \ge 0$  is the range  $s \ge \bar{s} \ge 0$ . This generates a new state with quantum numbers

$$h = \frac{c}{24} \left[ 1 - \frac{1}{k^2} + \alpha^2 \right] , \qquad m' = \frac{\alpha c}{12} ,$$
  
$$\bar{h} = \frac{c}{24} \left[ 1 - \frac{1}{k^2} + \bar{\alpha}^2 \right] , \qquad \bar{m}' = \frac{\bar{\alpha} c}{12} . \qquad (2.90)$$

These states are "heavy" in the sense that their conformal dimensions scale linearly with the large central charge c. By constrast, the "light" perturbative string states probing these backgrounds will correspond to holographic CFT states with conformal dimensions that are independent of c.

## 2.5.1 Chiral primaries in the D1D5 CFT

We briefly review the construction of chiral primary operators in the symmetric orbifold CFT [97]. We focus primarily on the holomorphic sector in the following; the antiholomorphic sector is entirely analogous. In the untwisted sector, on each copy of the seed  $\mathbb{T}^4$  theory, the chiral primary operators correspond to the states (suppressing the copy (*r*) label)

$$|0_{\rm NS}\rangle$$
,  $\psi_{-\frac{1}{2}}^{+\dot{A}}|0_{\rm NS}\rangle$ ,  $\mathbb{J}_{-1}^{+}|0_{\rm NS}\rangle = \psi_{-\frac{1}{2}}^{+\dot{1}}\psi_{-\frac{1}{2}}^{+\dot{2}}|0_{\rm NS}\rangle$ , (2.91)

where  $|0_{\text{NS}}\rangle$  is the NS vacuum. The corresponding weights and R-charges are  $h = m' = 0, \frac{1}{2}, 1$ , respectively. Physical configurations in the orbifold theory are obtained by symmetrizing the states in (2.91) over the different copies of the seed theory.

By including the antiholomorphic sector we can obtain, for instance, the dimension  $(\frac{1}{2}, \frac{1}{2})$  operator (see e.g. [98])

$$O^{++} = \sum_{r=1}^{N} O^{++}_{(r)} = \frac{-i}{\sqrt{2}} \sum_{r=1}^{N} \psi^{+\dot{A}}_{(r)} \varepsilon_{\dot{A}\dot{B}} \bar{\psi}^{+\dot{B}}_{(r)} , \qquad (O^{++})^{\dagger} = O^{--} .$$
(2.92)

We will use this operator in an explicit example later in the thesis.

In order to construct more general chiral primaries, one needs to consider the twisted sectors of the theory. Consider the 'bare' twist operators  $\sigma_n$ , defined on the cylinder, that impose the following boundary conditions corresponding to a single-cycle permutation,

$$X_{(1)} \to X_{(2)} \to \dots \to X_{(n)} \to X_{(1)},$$
  

$$\psi_{(1)} \to \psi_{(2)} \to \dots \to \psi_{(n)} \to -\psi_{(1)},$$
(2.93)

and likewise for the antiholomorphic fermions. The bare twist operators are defined to be the lowest-dimension twist operators that impose the above boundary conditions; they have dimension  $h = \bar{h} = \frac{1}{4} \left( n - \frac{1}{n} \right)$  and zero R-charge. Chiral operators are obtained by exciting the bare twist operators operators to add R-charge. The lowest-dimension chiral operators have  $h = m' = \frac{n-1}{2}$ . For *n* odd, these operators are obtained by acting with modes of the SU(2) currents, which are bilinears in the free fermions. Due to the twist operator, the SU(2) currents are fractional-moded in units of 1/n. The relation between these modes and those of free fermions on the *n* copies of the seed theory can be found in [97]. To construct the chiral operators, one acts with the currents  $J_{-l/n}^+$  for which *l* is odd and l < n,

*n* odd : 
$$\sigma_n^- = \prod_{p=1}^{(n-1)/2} \mathbb{J}_{-\frac{2p-1}{n}}^+ \sigma_n = \mathbb{J}_{-\frac{n-2}{n}}^+ \cdots \mathbb{J}_{-\frac{3}{n}}^+ \mathbb{J}_{-\frac{1}{n}}^+ \sigma_n.$$
 (2.94)

For *n* even, one first acts with a spin field  $S_n^+$ , which has weight  $\frac{1}{4n}$  and charge  $\frac{1}{2}$ , putting the fermions into the Ramond sector (i.e. their boundary conditions are similar to Eq. (2.93) but with the final sign being  $+\psi_{(1)}$ ). One then acts with the currents  $\mathbb{J}_{-l/n}^+$  for which *l* is even and l < n,

*n* even: 
$$\sigma_n^- = \prod_{p=1}^{(n-2)/2} \mathbb{J}_{-\frac{2p}{n}}^+ S_n^+ \sigma_n = \mathbb{J}_{-\frac{n-2}{n}}^+ \cdots \mathbb{J}_{-\frac{4}{n}}^+ \mathbb{J}_{-\frac{2}{n}}^+ S_n^+ \sigma_n.$$
 (2.95)

As in the untwisted case, for both odd and even *n* we can act with  $\psi_{-\frac{1}{2}}^{+\dot{A}} \equiv \sum_{r=1}^{n} \psi_{-\frac{1}{2}(r)}^{+\dot{A}}$  to obtain a chiral operator  $\psi_{-\frac{1}{2}}^{+\dot{A}}\sigma_{n}^{-}$  which has  $h = m' = \frac{n}{2}$ . Similarly we can act with  $\mathbb{J}_{-1}^{+}$  to obtain a chiral operator  $\mathbb{J}_{-1}^{+}\sigma_{n}^{-}$  which has  $h = m' = \frac{n+1}{2}$ . Together with the analogous antiholomorphic operators, this exhausts the single-cycle chiral operators. Indeed, by making use of anti-commutators of the supercurrent modes  $\mathbb{G}_{-m/n}^{\pm A}$  in the

corresponding twisted sectors, one can show that chiral weights are bounded by [46]

$$\frac{n-1}{2} \le h \le \frac{n+1}{2}.$$
 (2.96)

In the twisted sectors, it is often convenient to work in a basis that diagonalizes the twisted boundary conditions. We shall make use of this basis in later chapters. One defines

$$\tilde{\psi}_{\rho}^{\alpha \dot{A}} = \frac{1}{\sqrt{n}} \sum_{r=1}^{n} e^{\alpha \frac{2\pi i r \rho}{n}} \psi_{(r)}^{\alpha \dot{A}} , \qquad \rho = 0, \dots, n-1 , \qquad (2.97)$$

where the index  $\alpha = \pm$  should not be confused with the spectral flow parameter in Eq. (2.89). These are mutually orthogonal, and diagonalize the twisted boundary conditions as

$$\tilde{\psi}^{\alpha\dot{A}}_{\rho}(e^{2\pi i}z) = e^{-\alpha \frac{2\pi i\rho}{n}} \tilde{\psi}^{\alpha\dot{A}}_{\rho}(z) \,. \tag{2.98}$$

These fermions can be bosonized to construct an explicit expression for the spin fields mentioned above. Note that the fields  $\tilde{\psi}_{\rho=0}^{\alpha \dot{A}}$  are invariant under the twisting. For further discussion, see [99].

We now combine the above holomorphic construction with its antiholomorphic counterpart and define the complete list of scalar left-right chiral primaries we will be interested in:

$$O_n^{--} = \sigma_n^{--} , \qquad O_n^{\dot{A}\dot{B}} = \tilde{\psi}_{\rho=0}^{+\dot{A}} \tilde{\psi}_{\rho=0}^{+\dot{B}} \sigma_n^{--} , \qquad O_n^{++} = \tilde{\psi}_{\rho=0}^{+\dot{1}} \tilde{\psi}_{\rho=0}^{+2} \tilde{\psi}_{\rho=0}^{+\dot{1}} \tilde{\psi}_{\rho=0}^{+2} \sigma_n^{--} ,$$
(2.99)

where  $\sigma_n^{--}$  is defined similarly to Eqs. (2.94), (2.95), with the same construction in the antiholomorphic sector. The operators in (2.99) are normalized such that they have unit two-point functions.

For later reference, we note that in each case the respective weights and twist numbers can be written in terms of  $j = \frac{n+1}{2}$  as

$$h\left[O_{n}^{--}\right] = j - 1$$
,  $h\left[O_{n}^{\dot{A}\dot{B}}\right] = j - \frac{1}{2}$ ,  $h\left[O_{n}^{++}\right] = j$ . (2.100)

An analogous list of anti-chiral primaries (which have h = -m') is obtained by acting on the bare twist fields with current and fermion modes with opposite charge, i.e.  $\mathbb{J}_{-l/n}^$ and  $\psi^{-\dot{A}}$ . As we will shortly review, and up to a shift related to worldsheet spectral flow, this *j* will be identified with the principal quantum number of the bosonic (global) SL(2, $\mathbb{R}$ ) algebra of the worldsheet theory, to which we now turn.

# **2.6** Superstring theory on $AdS_3 \times S^3 \times T^4$

We now review the basics of superstring theory on  $AdS_3 \times S^3 \times T^4$  using the RNS formalism with BRST quantization. We first discuss the bosonic SL(2,  $\mathbb{R}$ ) and SU(2) WZW models and then present their supersymmetric counterparts. We present the current algebra and review the spectrum, including states arising from worldsheet spectral flow. Finally, we review the dictionary between holographic CFT operators and their counterparts in the worldsheet theory, following [99, 100].

#### **2.6.1** Bosonic WZW model for $SL(2,\mathbb{R})$

The SL(2, $\mathbb{R}$ ) WZW model was studied in detail in [101, 102, 103]. Here we will mostly follow the notation of [99, 100, 38], and normal ordering will be implicitly assumed.

The holomorphic SL(2,  $\mathbb{R}$ ) currents will be denoted  $j^a(z)$ . They satisfy the OPEs

$$j^{a}(z)j^{b}(w) \sim \frac{k}{2} \frac{\eta^{ab}}{(z-w)^{2}} + \frac{f^{ab}_{\ c} j^{c}(w)}{z-w},$$
 (2.101)

where *k* is the level of the affine algebra, and where

$$-2\eta^{33} = \eta^{+-} = 2, \qquad f^{+-}_{\ 3} = -2, \quad f^{3+}_{\ +} = -f^{3-}_{\ -} = 1.$$
 (2.102)

The holomorphic stress tensor and the central charge follow from the Sugawara construction, and are given by (likewise for the antiholomorphic sector)

$$T_{\rm sl}(z) = \frac{1}{k-2} \left[ -j^3(z)j^3(z) + \frac{1}{2}j^+(z)j^-(z) + \frac{1}{2}j^-(z)j^+(z) \right], \qquad c_{\rm sl} = \frac{3k}{k-2}.$$
 (2.103)

We denote bosonic SL(2,  $\mathbb{R}$ ) primary vertex operators by  $V_{j,m,\bar{m}}(z,\bar{z})$ . Their zero-mode wavefunctions do not factorize between holomorphic and antiholomorphic sectors, however as is often done we shall work primarily with the holomorphic sector, and suppress the  $\bar{m}$  and  $\bar{z}$  dependence. The relevant representations of the holomorphic zero-mode algebra are as follows. The principal series discrete representations of lowest (highest) weight are spanned by

$$\mathcal{D}_{i}^{\pm} = \{ |j,m\rangle , m = \pm j, \pm j \pm 1, \pm j \pm 2, \cdots \},$$
 (2.104)

respectively, where  $j_0^3|j,m\rangle = m|j,m\rangle$ . These are unitary representations for any positive real *j*, and one is the charge conjugate of the other (we will restrict the range of *j* momentarily). There are also the principal continuous series representations, spanned

by

$$C_{j}^{\hat{\alpha}} = \{ |j, \hat{\alpha}, m \rangle , 0 \le \hat{\alpha} < 1 , j = \frac{1}{2} + is , s \in \mathbb{R} , m = \hat{\alpha}, \hat{\alpha} \pm 1, \hat{\alpha} \pm 2, \cdots \}.$$
 (2.105)

The particular case  $\hat{\alpha} = 1/2 = j$  is actually reducible. It was shown in [101] that the spectrum of the model is built out of continuous and lowest weight representations with

$$\frac{1}{2} < j < \frac{k-1}{2}$$
, (2.106)

together with their spectrally flowed images, to be introduced below. The allowed range (2.106) follows from  $L^2$  normalization conditions, no-ghost theorems and spectral flow considerations.

Before considering worldsheet spectral flow (we refer to this as the "unflowed" sector), the action of the currents on the primary states is given by

$$j_0^3|j,m\rangle = m|j,m\rangle, \qquad (2.107a)$$

$$j_0^{\pm}|j,m\rangle = \begin{cases} (m \mp (j-1))|j,m \pm 1\rangle & \text{if } m \neq \mp j \\ 0 & \text{if } m = \mp j , \end{cases}$$
(2.107b)

$$j_n^a |j,m\rangle = 0 \qquad \forall n > 0.$$
(2.107c)

These vertex operators can be obtained from those of the Euclidean counterpart of the model, namely the  $H_3^+$  WZW model [104, 105, 106] (see also [107]), as follows. One introduces a set of operators depending on a complex label x, written as  $V_j(x|z)$ , and having conformal weight

$$\Delta = -\frac{j(j-1)}{k-2}.$$
 (2.108)

The action of the currents on  $V_i(x, z)$  is given by

$$j^{a}(z)V_{j}(x,w) \sim \frac{D_{j}^{a}V_{j}(x,w)}{(z-w)},$$
 (2.109)

where

$$D_j^+ = \partial_x$$
,  $D_j^3 = x\partial_x + j$ ,  $D_j^- = x^2\partial_x + 2jx$ . (2.110)

The two-point function is given by [106]

$$\langle V_{j_1}(x_1, z_1) V_{j_2}(x_2, z_2) \rangle = \frac{1}{|z_{12}|^{4\Delta_1}} \left[ \delta^2(x_1 - x_2) \delta(j_1 + j_2 - 1) + \frac{B(j_1)}{|x_{12}|^{4j_1}} \delta(j_1 - j_2) \right],$$
(2.111)

with

$$B(j) = \frac{2j-1}{\pi} \frac{\Gamma[1-b^2(2j-1)]}{\Gamma[1+b^2(2j-1)]} \nu^{1-2j} , \qquad \nu = \frac{\Gamma[1-b^2]}{\Gamma[1+b^2]} , \ b^2 = (k-2)^{-1} .$$
 (2.112)

The operators  $V_{j,m}(z)$  are related to  $V_j(x|z)$  by means of the following Mellin-like transform:

$$V_{j,m}(z) = \int_{\mathbb{C}} d^2 x \, x^{j-m-1} \bar{x}^{j-\bar{m}-1} V_j(x,z) \,. \tag{2.113}$$

In the Euclidean  $H_3^+$  model, *j* takes values j = 1/2 + is. To obtain the unflowed  $V_{j,m}$  for Lorentzian AdS<sub>3</sub>, one assumes a well-defined analytic continuation to real values of *j*. This procedure was discussed in [103], which identified the physical origin of the different divergences arising in correlation functions. For related work, see [108]. The two-point functions in the *m*-basis then follow from (2.111), (2.113). Using the shorthand  $V_i \equiv V_{j_i,m_i}$ , one finds

$$\langle V_1 V_2 \rangle = \frac{\delta^2(m_1 + m_2)}{|z_{12}|^{4\Delta_1}} \left[ \delta(j_1 + j_2 - 1) + \delta(j_1 - j_2) \frac{\pi B(j_1)}{\gamma(2j_1)} \frac{\gamma(j_1 + m_1)}{\gamma(1 - j_1 + m_1)} \right], \quad (2.114)$$

where  $\gamma(x) = \Gamma(x)/\Gamma(1-\bar{x})$ , and where  $\delta^2(m)$  is a Dirac delta in  $m + \bar{m}$  times a Kroenecker delta in  $m - \bar{m}$ .

At first sight, the complex variable *x* may appear simply as an SL(2, $\mathbb{R}$ ) version of the isospin variables defined for SU(2) in [109]. However, given that the integrated zero modes of the currents realize the spacetime Virasoro modes  $L_0$  and  $L_{\pm 1}$ , and by examining the expressions of the associated differential operators (2.110), one is led to interpret *x* as the local coordinate on the boundary theory [78]. In other words, the *x* variable corresponds to the insertion point of the holographic dual operator at the boundary of AdS<sub>3</sub>, and computing worldsheet correlation function in the *x*-basis is equivalent (after an appropriate integration over the worldsheet coordinates *z*,  $\bar{z}$ ) to computing correlators of the holographic CFT. Indeed, as we will see in more detail later on, according to (2.111), in the bosonic theory a *z*-integrated vertex operator  $V_j(x)$  is identified with a local operator on the boundary theory with weight *j*. Conversely, the corresponding boundary modes are given by the *m*-basis operators. Indeed, for states in the discrete sector, the transform in Eq. (2.113) can be inverted, giving

$$V_{j}(x,z) = \sum_{m=j+n, n \in \mathbb{N}_{0}} x^{m-j} \bar{x}^{\bar{m}-j} V_{j,m\bar{m}}(z) .$$
(2.115)

The vertex  $V_j(x, z)$  is realized via Eq. (2.115) as  $V_{j,j}(z)$  translated from the origin to x. Poles in the integrand of (2.113) coming from the expansion around x = 0 ( $x = \infty$ ) are associated to states in the  $\mathcal{D}_i^+$  ( $\mathcal{D}_i^-$ ) representations [88, 38].

Spectral flow automorphisms of the current algebra (2.101) are defined as

$$j^{\pm}(z) \to \tilde{j}^{\pm}(z) = z^{\pm w} j^{\pm}(z) , \qquad j^{3}(z) \to \tilde{j}^{3}(z) = j^{3}(z) - \frac{k\omega}{2} \frac{1}{z} ,$$
 (2.116)

where the so-called spectral flow charge  $\omega$  is an integer. Analogous formulas hold for the antiholomorphic sector. We work with the universal cover of SL(2,  $\mathbb{R}$ ), which imposes that the holomorphic and antiholomorphic spectral flow parameters must be equal,  $\bar{\omega} = \omega$ . The action of (2.116) on the above representations defines in general inequivalent representations that must be considered in order to generate a consistent spectrum. This holds up to the so-called series identifications due to the fact that the affine modules  $\hat{D}_{j}^{+,w}$  and  $\hat{D}_{k/2-j}^{-,w+1}$  are isomorphic. Thus, as mentioned above, the discrete series spectrum is constructed solely upon lowest weight representations with *j* restricted to the range (2.106).

At the level of vertex operators and for  $\omega > 0$ , the spectral flow operation introduced in (2.116) defines the so-called flowed primaries, whose OPEs with the currents take the form

$$j^{+}(z)V_{j,m}^{\omega}(w) = \frac{(m+1-j)V_{j,m+1}^{\omega}(w)}{(z-w)^{\omega+1}} + \sum_{n=1}^{\omega} \frac{(j_{n-1}^{+}V_{j,m}^{\omega})(w)}{(z-w)^{n}} + \dots, \quad (2.117a)$$

$$j^{3}(z)V_{j,m}^{\omega}(w) = \frac{\left(m + \frac{k}{2}\omega\right)V_{j,m}^{\omega}(w)}{(z-w)} + \dots,$$
 (2.117b)

$$j^{-}(z)V_{j,m}^{\omega}(w) = (z-w)^{\omega-1}(m-1+j)V_{j,m-1}^{\omega}(w) + \dots, \qquad (2.117c)$$

where the ellipses indicate higher-order terms. Similar expressions hold for  $\omega < 0$ , with the roles of  $j^+$  and  $j^-$  inverted. The operators  $V_{j,m}^{\omega}(z)$  are not affine primaries. They are, however, Virasoro primaries, with worldsheet conformal weight

$$\hat{\Delta} = -\frac{j(j-1)}{k-2} - m\omega - \frac{k}{4}\omega^2.$$
(2.118)

Note that for  $\omega > 0$  ( $\omega < 0$ ), independently of the characteristics of the original state, these correspond to lowest (highest) weight states, with SL(2, $\mathbb{R}$ ) spin

$$h = m + \frac{k}{2}\omega, \qquad (2.119)$$

 $(h = -m - k\omega/2, \text{ respectively})$ . The notation *h* anticipates that the SL(2,  $\mathbb{R}$ ) spin is identified with the holographic CFT conformal weight [110] (see also e.g. [99]), as we shall see in Eq. (2.121).

The flowed affine modules alluded above are built by acting with the currents on flowed primary states. In particular, the remaining states in the zero-mode algebra, which are obtained by acting with  $j_0^-$ , are *not* flowed primaries. Nevertheless, one can proceed as done for the unflowed states, and combine them into a local operator, defined initially for  $\omega > 0$  as

$$V_{j,h}^{w}(x,z) = \sum_{n \in \mathbb{N}_0} x^n \bar{x}^{\bar{n}} V_{j,h+n,h+\bar{n}}^{w}(z).$$
(2.120)

Moreover, by inverting  $x \to 1/x$  in the expansion, one also obtains the states in the highest-weight representation with the same spin and opposite  $\omega$  and m. This shows
that the resulting *x*-basis states are actually defined in terms of the absolute value of  $\omega$ , its sign being irrelevant. A direct *x*-basis definition for spectrally flowed vertex operators was recently derived in [39], extending the original proposal of [103] valid only for the singly flowed case.

The classical analog of the spectral flow operation (2.116) maps space-like geodesics of point-like strings into solutions in which a long string wound around the AdS<sub>3</sub> angular direction at large radius comes in to the centre of global AdS<sub>3</sub>, collapses to a point, and then re-expands to large radial distance [101]. The spectral flow parameter  $\omega$  is thus sometimes referred to in the literature as a "winding" number. Note that since the AdS<sub>3</sub> angular direction is contractible in the interior of global AdS<sub>3</sub>, the parameter  $\omega$  is not a conserved quantity. However, the *m*-basis two-point functions are diagonal in  $\omega$ : it was shown in [103] that the *m*-basis two-point function of flowed primaries is as in (2.114) with an extra factor of  $\delta_{\omega_1,-\omega_2}$  and the worldsheet conformal weight  $\Delta_1$  replaced by  $\hat{\Delta}_1$  given in Eq. (2.118). On the other hand, in the *x*-basis one finds

$$\langle V_{j_1,h_1}^{\omega_1}(x_1,z_1)V_{j_2,h_2}^{\omega_2}(x_2,z_2)\rangle = \frac{1}{|x_{12}|^{4h_1}} \frac{\langle V_{j_1,m_1}^{\omega_1}V_{j_2,m_2}^{\omega_2}\rangle}{V_{\text{conf}}}.$$
(2.121)

Thus, as mentioned above, the SL(2, $\mathbb{R}$ ) spin *h* is identified with the holographic CFT conformal weight [110], even though in the flowed sectors the spin is independent of the value of *j* of the corresponding unflowed operator. The factor  $V_{\text{conf}}$  stands for the divergent volume of the conformal group; it reflects the fact we are picking up the contribution from a pole, and it will cancel in the relevant computations that follow.

# 2.6.2 Bosonic WZW model for SU(2)

The bosonic SU(2) WZW model was studied in [111, 109]. We denote the generators of the current algebra by  $k^a$ , and for most quantities we use primes to distinguish them from their SL(2,  $\mathbb{R}$ ) counterparts. The currents satisfy the OPEs

$$k^{a}(z)k^{b}(w) \sim \frac{k'}{2} \frac{\delta^{ab}}{(z-w)^{2}} + \frac{f'^{ab}_{\ c}k^{c}(w)}{z-w}$$
, (2.122)

where k' is the level of the affine Lie algebra,  $\delta^{ab}$  is the Killing form, and  $f'^{abc}$  are the corresponding structure constants,

$$2\delta^{33} = \delta^{+-} = 2$$
,  $f'^{+-}_{3} = 2$ ,  $f'^{3+}_{+} = -f'^{3-}_{-} = 1$ . (2.123)

The energy-momentum tensor and central charge are

$$T_{\rm su}(z) = \frac{1}{k'+2} \left[ k^3(z)k^3(z) + \frac{1}{2}k^+(z)k^-(z) + \frac{1}{2}k^-(z)k^+(z) \right] , \qquad c_{\rm su} = \frac{3k'}{k'+2}.$$
(2.124)

We denote SU(2) vertex operators by  $V'_{j',m',\bar{m}'}(z,\bar{z})$ . Again, their zero-mode wavefunctions do not factorize into holomorphic and antiholomorphic parts, however we shall mostly work holomorphically and suppress antiholomorphic quantities ( $\bar{m}', \bar{z}$ ).

For SU(2), the unitary representations of the zero-mode algebra are labeled by

$$0 \le j' \le \frac{k'}{2}$$
,  $j' \in \mathbb{Z}/2$ , (2.125)

and their states are  $|j', m'\rangle$  with m' = -j', -j' + 1, ..., j' - 1, j'. Using conventions that mimic those used above for SL(2,  $\mathbb{R}$ ), we have

$$k_0^3 |j', m'\rangle = m' |j', m'\rangle,$$
 (2.126a)

$$k_{0}^{\pm}|j',m'\rangle = \begin{cases} (j'+1\pm m')|j',m'\pm 1\rangle & \text{if } m\neq \pm j \\ 0 & \text{if } m=\pm j \end{cases}$$
(2.126b)

$$k_n^a |j', m'\rangle = 0 \qquad \forall n > 0, \qquad (2.126c)$$

and

$$\Delta' = \frac{j'(j'+1)}{k'+2}.$$
(2.127)

Unlike  $SL(2, \mathbb{R})$ , in the SU(2) WZW model spectral flow is not necessary for constructing a consistent spectrum, due to the compactness of the group manifold. Indeed, the spectral flow automorphisms merely reshuffle primary and descendant fields, and they do not introduce new inequivalent representations. Nevertheless, for superstring theory applications it is of practical use to include it in the discussion [100, 52, 1]. We will discuss this in more detail shortly.

For SU(2), spectral flow is defined as

$$k^{\pm}(z) \to \tilde{k}^{\pm}(z) = z^{\mp w'} k^{\pm}(z) , \qquad k^{3}(z) \to \tilde{k}^{3}(z) = k^{3}(z) - \frac{k'\omega'}{2} \frac{1}{z} .$$
 (2.128)

In this case, however, it is possible to have  $\bar{\omega}' \neq \omega'$ . As before, spectrally flowed primaries  $V_{i',m'}^{\omega'}(z)$  are Virasoro primaries, with weight

$$\hat{\Delta}' = \frac{j'(j'+1)}{k'-2} + m'\omega' + \frac{k'}{4}\omega'^2, \qquad (2.129)$$

but they are not affine primaries, and for  $\omega' > 0$  they are defined in terms of the OPEs

$$k^{+}(z)V_{j',m'}^{\omega'}(w) = (z-w)^{\omega'-1}(j'+1+m')V_{j',m'+1}^{\omega'}(w) + \dots,$$
 (2.130a)

$$k^{3}(z)V_{j',m'}^{\omega'}(w) = \frac{\left(m' + \frac{k}{2}\omega'\right)V_{j',m'}^{\omega'}(w)}{(z-w)} + \dots,$$
(2.130b)

$$k^{-}(z)V_{j',m'}^{\omega'}(w) = \frac{(j'+1-m')V_{j',m'-1}^{\omega'}(w)}{(z-w)^{\omega'+1}} + \sum_{n=0}^{\omega'} \frac{(k_{n-1}^{-}V_{j',m'}^{\omega'})(w)}{(z-w)^{n}} + \dots$$
(2.130c)

The corresponding two-point functions are, again, the unflowed ones times  $\delta_{\omega_1,-\omega_2}$ , with the appropriate powers of  $z_{12}$ .

# **2.6.3** Superstrings in $AdS_3 \times S^3 \times T^4$

We now review supersymmetric generalizations of the bosonic WZW models discussed above. We introduce fermions  $\psi^a$  and  $\chi^a$  which are superpartners of the SL(2,  $\mathbb{R}$ ) and SU(2) currents  $J^a$  and  $K^a$  respectively. The appropriate  $\mathcal{N} = 1$  supersymmetric extensions of the affine sl(2, $\mathbb{R}$ )<sub>k</sub> and su(2)<sub>k'</sub> algebras are generated by the supercurrents  $\psi^a + \theta J^a$  and  $\chi^a + \theta K^a$ , where  $\theta$  is a Grassmann variable. The currents  $J^a$  and  $K^a$  satisfy the OPEs (2.101) and (2.122) respectively, with level  $n_5$  in both cases, and the OPEs involving the fermions  $\psi^a$  and  $\chi^a$  are

$$J^{a}(z)\psi^{b}(w) \sim \frac{f^{ab}_{\ c}\psi^{c}(w)}{(z-w)}, \qquad K^{a}(z)\chi^{b}(w) \sim \frac{f'^{ab}_{\ c}\chi^{c}(w)}{(z-w)}, \qquad (2.131a)$$

$$\psi^{a}(z)\psi^{b}(w) \sim \frac{n_{5}}{2} \frac{\eta^{ab}}{(z-w)}, \qquad \chi^{a}(z)\chi^{b}(w) \sim \frac{n_{5}}{2} \frac{\delta^{ab}}{(z-w)}.$$
(2.131b)

One can split the currents into two independent contributions via

$$J^{a} = j^{a} - \frac{1}{n_{5}} f^{a}_{\ bc} \psi^{b} \psi^{c} , \qquad K^{a} = k^{a} - \frac{1}{n_{5}} f'^{a}_{\ bc} \chi^{b} \chi^{c} . \qquad (2.132)$$

The "bosonic" currents  $j^a$  and  $k^a$  commute with the free fermions, and are currents of bosonic WZW models as described in Section 2.6, with levels  $k = n_5 + 2$  and  $k' = n_5 - 2$  respectively. In the fermionic sector, the spectral flow automorphisms act as

$$\tilde{\psi}^{\pm}(z) = z^{\mp\omega}\psi^{\pm}(z), \quad \tilde{\psi}^{3}(z) = \psi^{3}(z), \qquad \tilde{\chi}^{\pm}(z) = z^{\mp\omega}\chi^{\pm}(z), \quad \tilde{\chi}^{3}(z) = \chi^{3}(z).$$
(2.133)

The remaining flat compact directions are treated as usual. For the  $\mathbb{T}^4$ , we simply have four (canonically normalized) free bosons  $Y^i$  and their fermionic partners  $\lambda^i$  (i = 6, ..., 9), with OPEs

$$Y^{i}(z)Y^{j}(w) \sim -\delta^{ij}\log(z-w)$$
,  $\lambda^{i}(z)\lambda^{j}(w) \sim \frac{\delta^{ij}}{(z-w)}$ . (2.134)

~::

We can now write down the energy-momentum tensor *T* and the supercurrent *G* of the worldsheet theory for type II superstrings in  $AdS_3 \times S^3 \times T^4$ . The matter contributions read

$$T = \frac{1}{n_5} \left( j^a j_a - \psi^a \partial \psi_a + k^a k_a - \chi^a \partial \chi_a \right) + \frac{1}{2} \left( \partial Y^i \partial Y_i - \lambda^i \partial \lambda_i \right), \qquad (2.135)$$

$$G = \frac{2}{n_5} \left( \psi^a j_a - \frac{1}{3n_5} f_{abc} \psi^a \psi^b \psi^c + \chi^a k_a - \frac{1}{3n_5} f'_{abc} \chi^a \chi^b \chi^c \right) + i \,\lambda^j \partial Y_j \,, \quad (2.136)$$

and the resulting central charge is compensated by the usual *bc* and  $\beta\gamma$  ghost systems, leading to the BRST charge

$$Q = \oint dz \left( c \left( T + T_{\beta\gamma} \right) - \gamma G + c(\partial c)b - \frac{1}{4}b\gamma^2 \right).$$
(2.137)

Here  $T_{\beta\gamma}$  is the energy-momentum tensor of the  $\beta\gamma$  system, which is bosonized as

$$\beta = e^{-\varphi} \partial \xi$$
,  $\gamma = \eta e^{\varphi}$ , (2.138)

where  $\varphi(z)\varphi(w) \simeq -\ln(z-w)$  has background charge 2, and  $\xi(z)\eta(w) \sim (z-w)^{-1}$ . The (matter) central charge is

$$c = \frac{3(n_5+2)}{n_5} + \frac{3}{2} + \frac{3(n_5-2)}{n_5} + \frac{3}{2} + 6 = 15.$$
 (2.139)

For computational purposes it is useful to also bosonize the rest of the fermions [78, 99]. We thus define (canonically normalized) bosonic fields  $H_I$  with I = 1, ..., 5, and write

$$\hat{H}_I = H_I + \pi \sum_{J < I} N_J , \qquad N_J \equiv \oint i \partial H_J , \qquad (2.140)$$

where the number operators  $N_I$  are introduced in order to keep track of the cocycle factors, namely

$$e^{ia\hat{H}_{I}}e^{ib\hat{H}_{J}} = e^{ib\hat{H}_{J}}e^{ia\hat{H}_{I}}e^{i\pi ab}$$
, if  $I > J$ . (2.141)

We bosonize as

$$\psi^{\pm} = \sqrt{n_5} e^{\pm i\hat{H}_1}, \quad \chi^{\pm} = \sqrt{n_5} e^{\pm i\hat{H}_2}, \quad \lambda^6 \pm i\lambda^7 = e^{\pm i\hat{H}_4}, \quad \lambda^8 \pm i\lambda^9 = e^{\pm i\hat{H}_5}, \quad (2.142a)$$
$$\psi^3 = \frac{\sqrt{n_5}}{2} \left( e^{i\hat{H}_3} - e^{-i\hat{H}_3} \right), \qquad \chi^3 = \frac{\sqrt{n_5}}{2} \left( e^{i\hat{H}_3} + e^{-i\hat{H}_3} \right), \quad (2.142b)$$

where  $\hat{H}_{I}^{\dagger} = \hat{H}_{I}$  for  $I \neq 3$  and  $\hat{H}_{3}^{\dagger} = -\hat{H}_{3}$ . Then we have

$$i\partial \hat{H}_1 = \frac{1}{n_5}\psi^+\psi^-$$
,  $i\partial \hat{H}_2 = \frac{1}{n_5}\chi^+\chi^-$ ,  $i\partial \hat{H}_3 = \frac{2}{n_5}\psi^3\chi^3$ , (2.143a)

$$i\partial \hat{H}_4 = i\lambda^6\lambda^7$$
,  $i\partial \hat{H}_5 = i\lambda^8\lambda^9$ . (2.143b)

The phases in (2.141) ensure that bosonized fermions anticommute, and will be important when working with states in the Ramond sector. From now on we will simply omit the hats, and explicitly include the phase factors when they are needed.

The spacetime supercharges can be written as:

$$Q_{\varepsilon} = \oint dz \, e^{-\varphi/2} S_{\varepsilon} , \qquad S_{\varepsilon} = \exp\left(\frac{i}{2} \sum_{I=1}^{5} \varepsilon_{I} H_{I}\right), \qquad (2.144)$$

where  $S_{\varepsilon}$  are spin fields and  $\varepsilon_I = \pm 1$ . Imposing BRST invariance – where the relevant contributions come from the  $f_{abc}\psi^a\psi^b\psi^c$  and  $f'_{abc}\chi^a\chi^b\chi^c$  pieces of *G* in (2.136) – and mutual locality (chiral GSO) leads to the conditions

$$\prod_{I=1}^{3} \varepsilon_I = \prod_{I=1}^{5} \varepsilon_I = 1.$$
(2.145)

In the holomorphic sector this gives the expected four 'ordinary' supercharges and four 'superconformal' supercharges. The same applies in the antiholomorphic sector, giving the total 16 real supercharges of global  $AdS_3 \times S^3$  [78].

For later use, let us also recall that the R-symmetry of the boundary theory is generated on the worldsheet by the SU(2) currents. More precisely, the zero modes of the spacetime R-currents are given by the integrated worldsheet currents [78], i.e.

$$\mathbb{J}_0^a = \oint dz \, K^a(z) \,. \tag{2.146}$$

Consequently, the holomorphic R-charge in the holographic CFT is identified with m', which from now onwards denotes the eigenvalue of  $K_0^3$ . This is why we used the notation m' in Sections 2.5 and 2.5.1. Similarly, from now onwards m denotes the eigenvalue of  $J_0^3$ .

# 2.6.4 Vertex operators and two-point functions

We now discuss physical vertex operators and their two-point functions, both in NS and R sectors. This section is largely review, though we also give explicit expressions for some R sector operators that to our knowledge have not appeared before in the literature.

Our main interest is in worldsheet operators that correspond to chiral primaries of the holographic CFT. We thus focus on states belonging to the discrete representations of  $SL(2,\mathbb{R})$ , since it has been shown that the chiral primaries of the dual CFT are in one-to-one correspondence with worldsheet states in the discrete representation [100]. Also, as discussed in the previous sections, the continuous representation states are lifted at generic points of the moduli space, and, furthermore, are not part of the spectrum of the

dual CFT at the orbifold point. We also discuss the role of  $SL(2,\mathbb{R})$  and SU(2) spectral flows in the string theoretical construction. These vertex operators and their two-point functions will be used as building blocks for constructing the vertex operators and two-point functions of the null-gauged models.

# 2.6.4.1 NS sector

The unflowed NSNS sector was considered in [112], see also [99]. We continue to suppress antiholomorphic parts of the SL(2,  $\mathbb{R}$ ) and SU(2) vertex operators  $V_{j,m,\bar{m}}$  and  $V'_{j',m',\bar{m}'}$ . The complete NSNS vertex is obtained by including the antiholomorphic fermions and ghosts.

We work in the canonical "-1" ghost picture, and consider only states with vanishing momentum in the  $\mathbb{T}^4$  directions. Then the (holomorphic part of the) BRST invariant states with up to a single fermionic excitation include the tachyon (which is projected out by GSO),

$$\mathcal{T}_{j,m,j',m'} = e^{-\varphi} V_{j,m} V'_{j',m'}, \qquad (2.147)$$

and the following spacetime vectors. To write these, we denote the total spins corresponding to the supersymmetric currents  $J^a$  and  $K^a$  by  $J = j + \varepsilon$  and  $J' = j' + \varepsilon$ , with  $\varepsilon = \pm 1$  (recall that *m* and *m'* now denote the eigenvalues of  $J_0^3$  and  $K_0^3$  respectively, and that i = 6, ..., 9),

$$\mathcal{V}^{i}_{j,m,l',m'} = e^{-\varphi} \lambda^{i} V_{j,m} V'_{j',m'}, \qquad (2.148a)$$

$$\mathcal{W}_{J,m,J',m'}^{\varepsilon} = e^{-\varphi} \left( \psi V_j \right)_{j+\varepsilon,m} V'_{j',m'}, \qquad (2.148b)$$

$$\mathcal{X}_{J,m,J',m'}^{\varepsilon} = e^{-\varphi} V_{j,m}(\chi V_{j'}')_{j'+\varepsilon,m'},$$
 (2.148c)

where we have introduced the linear combinations

$$(\psi V_j)_{j+\varepsilon,m} = c_{\varepsilon}^r \psi^r V_{j,m-r} , \qquad (\chi V_{j'}')_{j'+\varepsilon,m'} = d_{\varepsilon}^r \chi^r V_{j',m'-r}' , \qquad (2.149)$$

in which a summation over r = +1, -1, 0 is implicit, "0" corresponding to the "3" direction of the respective algebras. These combine the products of bosonic primaries and free fermions into fields of total spins *J* and *J*' [112]. The Clebsh-Gordan coefficients are given in our conventions by

$$c_{-}^{r} = \left(\frac{1}{2}, \frac{1}{2}, -1\right), \quad d_{+}^{r} = \left(-\frac{1}{2}, \frac{1}{2}, 1\right),$$

$$c_{+}^{r} = \left(\frac{1}{2}(j+m)(j+m-1), \frac{1}{2}(j-m)(j-m-1), (j+m)(j-m)\right), \quad (2.150)$$

$$d_{-}^{r} = \left(\frac{1}{2}(j'-m')(j'-m'+1), -\frac{1}{2}(j'+m')(j'+m'+1), (j'-m')(j'+m')\right).$$

The Virasoro condition associated to all vertex operators in Eq. (2.148) reads

$$\frac{1}{2} + \frac{1}{2} - \frac{j(j-1)}{n_5} + \frac{j'(j'+1)}{n_5} = 1, \qquad (2.151)$$

and is solved by j = j' + 1 (or its reflection under  $j \to 1 - j$ ), thus implying that we are dealing with bosonic primaries in the discrete representations of SL(2, $\mathbb{R}$ )<sub>k</sub>.

Let us briefly discuss the worldsheet two-point functions involving these operators. The different bosonic sectors factorize and the fermions are free, so we can express the results directly in terms of the non-trivial contributions coming from the bosonic SL(2, $\mathbb{R}$ ) WZW model, namely Eq. (2.114). By construction, the only non-vanishing two-point functions are the diagonal ones. For the 6D scalars coming from the NSNS sector polarizations on the  $\mathbb{T}^4$ , (2.148a), using the shorthands  $V_i \equiv V_{j_i,m_i}$ ,  $V'_i \equiv V'_{j'_i,m'_i}$ , we have

$$\langle \mathcal{V}_1^{i\bar{i}} \mathcal{V}_2^{j\bar{j}} \rangle = \langle V_1 V_2 \rangle \langle V_1' V_2' \rangle \left[ \langle e^{-\varphi_1} e^{-\varphi_2} \rangle \langle \lambda_1^i \lambda_2^j \rangle \times \text{c.c.} \right] = \langle V_1 V_2 \rangle \langle V_1' V_2' \rangle \times \frac{\delta^{ij} \delta^{ij}}{|z_{12}|^4} .$$
(2.152)

Since we are dealing with discrete representations, the contact term in (2.114) vanishes, thus imposing  $j_1 = j_2 \equiv j$ . As discussed above Eq. (2.115), the conformal weight in the holographic CFT is to be identified with the SL(2, $\mathbb{R}$ ) spin, i.e. h = j. On the other hand, the R-charge is given by m', with  $|m'| \leq j' = j - 1$ . Thus  $h \neq |m'|$ , so  $\mathcal{V}^i$  cannot correspond to a chiral primary of the HCFT.

We now turn to the operators introduced in the second and third line of (2.148). When computing correlators of two W states, we must deal with expressions of the form

$$\langle \left(\psi_{1}V_{j_{1}}\right)_{j_{1}+\varepsilon_{1},m_{1}}\left(\psi_{2}V_{j_{2}}\right)_{j_{2}+\varepsilon_{2},m_{2}}\rangle = \sum_{r_{1,2}} c_{\varepsilon_{1}}^{r_{1}} c_{\varepsilon_{2}}^{r_{2}} \left\langle \psi^{r_{1}}\psi^{r_{2}} \right\rangle \left\langle V_{j_{1},m_{1}-r_{1}}V_{j_{2},m_{2}-r_{2}} \right\rangle.$$
(2.153)

We use the action of the bosonic currents (2.107) to express  $\langle V_{j_1,m_1-r_1}V_{j_2,m_2-r_2}\rangle$  in terms of  $\langle V_{j_1,m_1}V_{j_2,m_2}\rangle$ , insert the coefficients (2.150), and perform the sum. We obtain

$$\langle \mathcal{W}^{\varepsilon_{1}}\mathcal{W}^{\varepsilon_{2}}\rangle = \frac{n_{5}^{2}}{4|z_{12}|^{4}}\langle V_{1}V_{2}\rangle\langle V_{1}'V_{2}'\rangle \times \begin{cases} j_{1}(1-2j_{1})(j_{1}^{2}-m_{1}^{2})\times \text{c.c.} & \varepsilon_{1}=\varepsilon_{2}=1\\ \frac{(j_{1}-1)(1-2j_{1})}{(j_{1}-1)^{2}-m_{1}^{2}}\times \text{c.c.} & \varepsilon_{1}=\varepsilon_{2}=-1\\ 0 & \varepsilon_{1}=-\varepsilon_{2} . \end{cases}$$

$$(2.154)$$

From Eq. (2.114) we find that the coefficients are exactly those needed to produce the shift  $j \rightarrow j + \varepsilon$  in the two-point function. Hence, Eq. (2.113) shows that the weight of the corresponding holographic dual is  $h = j + \varepsilon$  [113]. In particular, the operator  $W^-$  with maximal SU(2) charge has h = j - 1 = m', and thus corresponds to a chiral primary operator of the holographic CFT.

The computation of the  $\langle \mathcal{X}\mathcal{X} \rangle$  correlators is analogous; we obtain

$$\langle \mathcal{X}^{\varepsilon_1} \mathcal{X}^{\varepsilon_2} \rangle = \frac{n_5^2}{4|z_{12}|^4} \langle V_1 V_2 \rangle \langle V_1' V_2' \rangle \times \begin{cases} \frac{(j_1'+1)(1+2j_1')}{(j_1'+1)^2 - m_1'^2} \times \text{c.c.} & \varepsilon_1 = \varepsilon_2 = 1\\ j_1'(1+2j_1')(j_1'^2 - m_1'^2) \times \text{c.c.} & \varepsilon_1 = \varepsilon_2 = -1\\ 0 & \varepsilon_1 = -\varepsilon_2 = \pm 1 \\ (2.155) \end{cases}$$

For  $\mathcal{X}^+$  at highest SU(2) weight we have h = j = j' + 1 = m', leading to a second family of spacetime chiral states. We will discuss the corresponding operators in the holographic CFT theory and fix their normalization below.

So far, we have constructed chiral operators whose boundary weights h = j - 1 and h = j are bounded from above by  $h < \frac{n_5+1}{2}$ , see Eq. (2.106). However, in the D1D5 CFT one can have chiral primaries in *n*-twisted sectors with *n* up to  $n_1n_5$ , and where *h* grows linearly with *n*, as discussed around Eq. (2.96). Thus, it seems that so far we are missing most of the heavier chiral operators. However, as discussed in [100], such states lie in the sectors of the worldsheet theory with non-trivial spectral flow charges, as we now review.

In the supersymmetric theory, spectrally flowed primary operators are built by combining the bosonic flowed primaries introduced in Eqs. (2.117) and (2.130) with fermionic excitations. The bosons  $H_I$  allow us to express the spectral flow operation in the fermionic sectors of SL(2, $\mathbb{R}$ ) and SU(2), Eq. (2.133), in the following form,

$$\psi_{\omega}^{\pm} = \psi^{\pm} e^{-i\omega H_1}$$
,  $\chi_{\omega'}^{\pm} = \chi^{\pm} e^{i\omega' H_2}$ , (2.156)

while the other fermions remain unchanged. Indeed, the OPEs between the operators in (2.156) and the fermionic currents are analogous to those in (2.117) and (2.130). Once factors of  $e^{-i\omega H_1}$  and  $e^{i\omega' H_2}$  are included, the corresponding weights take the form in (2.118) and (2.129), with  $k - 2 = n_5 = k' + 2$ .

In principle, one could simply ignore the possibility of including spectral flow in SU(2) since it does not give any new representations. However, for discrete series states it is useful to do so in order to solve the modified Virasoro condition, as discussed in [52, 1]. Bosonizing the currents  $j^3$  and  $k^3$  with canonically normalised scalars  $\phi$  and  $\phi'$  respectively, we use the spectral flow operator with equal amount of spectral flow in SL(2, $\mathbb{R}$ ) and SU(2),

$$\exp\left(-i\omega H_1 + \omega\sqrt{\frac{n_5+2}{2}}\phi + i\omega H_2 + i\omega\sqrt{\frac{n_5-2}{2}}\phi'\right),\qquad(2.157)$$

which is mutually local with the supercharges, thus producing flowed singly-excited

states which will also survive the GSO projection. The corresponding Virasoro condition for the flowed vertex operators is

$$\frac{1}{2} + \frac{1}{2} - \frac{j(j-1)}{n_5} - m\omega - \frac{n_5}{4}\omega^2 + \frac{j'(j'+1)}{n_5} + m'\omega + \frac{n_5}{4}\omega^2 = 1.$$
 (2.158)

We seek to solve this for general  $n_5$ . We thus impose j = j' + 1 as well as m = m'. The latter constraint is quite restrictive, since by definition we have

$\mathcal{V}$ -type operators:	$ m \geq j$ ,	$ m' \leq j'=j-1,$
<i>W</i> -type operators:	$ m \geq j-1$ ,	$ m'  \le j' = j - 1$ ,
$\mathcal X$ -type operators:	$ m \geq j$ ,	$ m'  \leq j' + 1 = j.$

Consequently, our only candidates are highest/lowest-weight  $W^-$ -type operators with m = m' = j' = j - 1 and  $\mathcal{X}^+$ -type operators with m = m' = j' + 1 = j. Their explicit expressions are given by

$$\mathcal{W}_{j}^{\omega} = e^{-\varphi} \psi^{-} e^{-i\omega H_{1}} e^{i\omega H_{2}} V_{j,j}^{\omega} V_{j-1,j-1}^{\prime \omega}, \qquad (2.159a)$$

$$\mathcal{X}_{j}^{\omega} = e^{-\varphi} e^{-i\omega H_{1}} \chi^{+} e^{i\omega H_{2}} V_{j,j}^{\omega} V_{j-1,j-1}^{\prime \omega}.$$
(2.159b)

These flowed states are also BRST-invariant [100] since the supercurrent G can be written in the flowed frame as

$$G(z) = \tilde{G}(z) + \frac{\omega}{z} \left(\chi^3 - \psi^3\right), \qquad (2.160)$$

such that the extra terms on the RHS of this equation act trivially on highest/lowest weight states.

The two-point functions of these spectrally flowed operators can be determined straightforwardly from the corresponding bosonic ones. This is because the latter impose  $\omega_2 = -\omega_1$ , such that the charge conservation rules for the  $H_I$  exponentials are automatically satisfied. Denoting the Hermitian conjugate operators of  $\mathcal{W}^{\omega}$  and  $\mathcal{X}^{\omega}$  by  $\hat{\mathcal{W}}^{\omega}$  and  $\hat{\mathcal{X}}^{\omega}$  respectively, we obtain

$$\langle \mathcal{W}_{1}^{\omega_{1}} \hat{\mathcal{W}}_{2}^{\omega_{2}} \rangle = \langle \mathcal{X}_{1}^{\omega_{1}} \hat{\mathcal{X}}_{2}^{\omega_{2}} \rangle = \langle V_{j_{1},j_{1}}^{\omega_{1}} V_{j_{1},-j_{1}}^{\omega_{2}} \rangle \langle V_{j_{1}-1,j_{1}-1}^{'\omega_{1}} V_{j_{1}-1,1-j_{1}}^{'\omega_{2}} \rangle \frac{n_{5}^{2}}{|z_{12}|^{4(1+\omega_{1}+\omega_{1}^{2})}}.$$
 (2.161)

Spectral flowed primaries are always annihilated by  $J_0^-$ , and are thus lowest-weight with respect to the SL(2,  $\mathbb{R}$ ) zero mode algebra. After the supersymmetric spectral flow (2.157), similarly to the bosonic transformations (2.119), (2.130b), the spectral flowed primaries have quantum numbers *h* and *m'* that have increased by  $\frac{n_5}{2}\omega$  from their values for the unflowed vertex operators. We thus conclude that these vertex operators

correspond exactly to the additional chiral operators we were looking for, with

$$\mathcal{W}_{j}^{\omega}$$
:  $h = m' = j - 1 + \frac{n_{5}}{2}\omega$ ,  $\mathcal{X}_{j}^{\omega}$ :  $h = m' = j + \frac{n_{5}}{2}\omega$ . (2.162)

These quantum numbers extend to large values, by raising  $\omega$ . In the holographic CFT there are states with conformal weight of order  $n_1n_5$ , however in our worldsheet models  $n_1$  is of order  $g_s^{-2}$  and we work in perturbation theory in  $g_s$ , so finite  $n_1$  physics is not accessible. Moreover, when considering holographic CFT operators with conformal weight of order  $n_1n_5$ , the dual bulk configuration is not a light probe on the original background, but rather a different background. The rest of the modes associated to such boundary operators are obtained by acting with the global current  $J_0^+$  as in Eq. (2.120), and do not have simple expressions in the *m*-basis since they are not flowed primaries.

# 2.6.4.2 Ramond sector

We now review the Ramond sector physical operators of the worldsheet theory, in the *m*-basis. To our knowledge, this construction has only been carried out explicitly in the literature for the case of highest/lowest-weight states [99, 100]; we shall present explicit expressions for more general Ramond sector operators.

We will make use of the spin fields introduced in (2.144), and distinguish the slightly more involved  $AdS_3 \times S^3$  sector, for which we write the relevant factors as

$$S_{\varepsilon_1\varepsilon_2\varepsilon_3} = e^{\frac{i}{2}(\varepsilon_1H_1 + \varepsilon_2H_2 + \varepsilon_3H_3)}.$$
(2.163)

We denote the  $AdS_3 \times S^3$  chirality by  $\varepsilon \equiv \varepsilon_1 \varepsilon_2 \varepsilon_3$ . We shall implement this by considering  $\varepsilon_3$  to be fixed to be  $\varepsilon_3 = \varepsilon \varepsilon_1 \varepsilon_2$ . We impose the chiral GSO projection via the mutual locality condition  $\prod_{I=1}^5 \varepsilon_I = 1$ , and we implement this by fixing  $\varepsilon_5 = \varepsilon \varepsilon_4$ . We introduce a generic linear combination of bosonic primaries and spin fields of  $AdS_3 \times S^3$  of fixed chirality,

$$\left(SVV'\right)_{J,m,J',m'}^{\varepsilon} = \sum_{\varepsilon_1,\varepsilon_2=\pm 1} f_{\varepsilon_1\varepsilon_2}^{\varepsilon} S_{\varepsilon_1\varepsilon_2\varepsilon_3} V_{j,m-\frac{\varepsilon_1}{2}} V'_{j',m'-\frac{\varepsilon_2}{2}}, \qquad \varepsilon_3 = \varepsilon \varepsilon_1 \varepsilon_2, \qquad (2.164)$$

where the total spins (J, J') will be related to (j, j') in various ways momentarily. Note that for highest/lowest weight states, there may be only one allowed choice of  $\varepsilon_1$  and/or  $\varepsilon_2$ , as we shall see in an example below. In the canonical " $-\frac{1}{2}$ " picture, the Ramond sector vertex operators then take the form

$$\mathcal{Y}_{J,m,J',m'}^{\varepsilon,\varepsilon_4} = e^{-\frac{\varphi}{2}} \left( SVV' \right)_{J,m,J',m'}^{\varepsilon} e^{\frac{i\varepsilon_4}{2}(H_4 + \varepsilon H_5)}.$$
(2.165)

The Clebsch–Gordan coefficients  $f_{\varepsilon_1 \varepsilon_2}^{\varepsilon}$  are computed by requiring that the  $\mathcal{Y}$  operators transform appropriately under the action of the currents  $J^{\pm}$ ,  $K^{\pm}$ . In our conventions, this gives four linear combinations. In the equations below, the first bracket specifies how (J, J') are related to j and j'; for instance, for case A, (J = j - 1/2, J' = j' + 1/2). For each case we write the coefficients as a list,  $f_{\varepsilon_1 \varepsilon_2}^{\varepsilon} = (f_{++}^{\varepsilon}, f_{+-}^{\varepsilon}, f_{-+}^{\varepsilon}, f_{--}^{\varepsilon})$ . We obtain

$$A: (j - \frac{1}{2}, j' + \frac{1}{2}), \quad f_{\varepsilon_1 \varepsilon_2}^{\varepsilon, A} = (1, i, \varepsilon, \varepsilon i), \qquad (2.166a)$$

B: 
$$(j + \frac{1}{2}, j' + \frac{1}{2}), \quad f_{\varepsilon_1 \varepsilon_2}^{\varepsilon, B} = (f^{B_1}, if^{B_1}, \varepsilon f^{B_2}, \varepsilon i f^{B_2}),$$
 (2.166b)

C: 
$$(j - \frac{1}{2}, j' - \frac{1}{2}), \quad f_{\varepsilon_1 \varepsilon_2}^{\varepsilon, C} = (f^{C_1}, -if^{C_2}, \varepsilon f^{C_1}, \varepsilon (-i)f^{C_2}),$$
 (2.166c)

D: 
$$(j + \frac{1}{2}, j' - \frac{1}{2}), \quad f_{\varepsilon_1 \varepsilon_2}^{\varepsilon, D} = \left( f^{B_1} f^{C_1}, (-i) f^{B_1} f^{C_2}, \varepsilon f^{B_2} f^{C_1}, \varepsilon (-i) f^{B_2} f^{C_2} \right), \quad (2.166d)$$

where

$$f^{B_1} = m + j - \frac{1}{2}, \quad f^{B_2} = m - j + \frac{1}{2}, \quad f^{C_1} = j' - m' + \frac{1}{2}, \quad f^{C_2} = j' + m' + \frac{1}{2}.$$
 (2.167)

We note that in all cases we have  $f_{\varepsilon_1\varepsilon_2}^+ = \varepsilon_1 f_{\varepsilon_1\varepsilon_2}^-$ . In addition, and using j = j' + 1, BRST-invariance gives four equations for each chirality, out of which only two are linearly independent, namely

$$f_{-+}^{\varepsilon} = \frac{1}{\left(j+m-\frac{1}{2}\right)} \left[ f_{++}^{\varepsilon} \left(\varepsilon \, m+m'\right) - i \, f_{+-}^{\varepsilon} \left(j'-m'+\frac{1}{2}\right) \right], \quad (2.168a)$$

$$f_{--}^{\varepsilon} = \frac{1}{(j+m-\frac{1}{2})} \left[ i f_{++}^{\varepsilon} \left( j'+m'+\frac{1}{2} \right) + f_{+-}^{\varepsilon} \left( \varepsilon \, m-m' \right) \right].$$
(2.168b)

These are satisfied by only half of the states in Eq. (2.166). The physical states in the " $-\frac{1}{2}$ " picture are given by the *A* and *D* states with  $\varepsilon = 1$  (and either choice of  $\varepsilon_4 = \pm 1$ ), plus the *B* and *C* states with  $\varepsilon = -1$  (and again either sign of  $\varepsilon_4$ ), making the correct eight physical polarizations.

The full list of expressions in (2.166) is useful in order to construct the representatives of such operators in the " $-\frac{3}{2}$ " ghost picture, necessary for computing two-point functions. To obtain the " $-\frac{3}{2}$ " picture operators, we make an educated guess for their expressions, and then apply the picture *raising* operator, i.e.

$$\Phi^{(-\frac{1}{2})}(w) = \lim_{z \to w} \left( e^{\varphi} G \right)(z) \, \Phi^{(-\frac{3}{2})}(w) \,. \tag{2.169}$$

In order to get a non-trivial propagator, and up to an overall constant, the appropriate guess is that they are given by the states with the same spins but opposite chirality. Explicitly, we have

$$\mathcal{Y}_{J,m,J',m'}^{\varepsilon,\varepsilon_4\,(-\frac{3}{2})} = \pm \frac{\sqrt{n_5}}{2j-1} e^{-\frac{3\varphi}{2}} \left(SVV'\right)_{J,m,J',m'}^{-\varepsilon} e^{\frac{i\varepsilon_4}{2}(H_4+\varepsilon H_5)}, \qquad (2.170)$$

where the negative (positive) sign holds for the cases *A* and *B* (*C* and *D*).

We can now compute the two-point functions in the unflowed Ramond sector. Only diagonal pairings are non-zero, by construction. Denoting the antiholomorphic sector contributions by "c.c.", we obtain

$$\langle \mathcal{Y}_{[A]}^{\varepsilon_{4,}(-\frac{1}{2})} \mathcal{Y}_{[A]}^{-\varepsilon_{4,}(-\frac{3}{2})} \rangle = \frac{n_5}{|z_{12}|^4} \langle V_1 V_2 \rangle \langle V_1' V_2' \rangle \left( \frac{(2j-1)}{(j-m-\frac{1}{2})(j'+m'+\frac{1}{2})} \times \text{c.c.} \right),$$
(2.171a)

$$\langle \mathcal{Y}_{[B]}^{\varepsilon_{4},(-\frac{1}{2})} \mathcal{Y}_{[B]}^{-\varepsilon_{4},(-\frac{3}{2})} \rangle = \frac{n_{5}}{|z_{12}|^{4}} \langle V_{1}V_{2} \rangle \langle V_{1}'V_{2}' \rangle \left( \frac{(2j-1)(j+m-\frac{1}{2})}{(j'+m'+\frac{1}{2})} \times \text{c.c.} \right), \quad (2.171b)$$

$$\langle \mathcal{Y}_{[C]}^{\varepsilon_{4,(-\frac{1}{2})}} \mathcal{Y}_{[C]}^{-\varepsilon_{4,(-\frac{3}{2})}} \rangle = \frac{n_5}{|z_{12}|^4} \langle V_1 V_2 \rangle \langle V_1' V_2' \rangle \left( \frac{(2j-1)(j'-m'+\frac{1}{2})}{(m-j+\frac{1}{2})} \times \text{c.c.} \right), \quad (2.171c)$$

$$\langle \mathcal{Y}_{[D]}^{\varepsilon_{4,}(-\frac{1}{2})} \mathcal{Y}_{[D]}^{-\varepsilon_{4,}(-\frac{3}{2})} \rangle = \frac{n_{5}}{|z_{12}|^{4}} \langle V_{1}V_{2} \rangle \langle V_{1}'V_{2}' \rangle \left( (2j-1)(m+j-\frac{1}{2})(j'-m'+\frac{1}{2}) \times \text{c.c.} \right),$$
(2.171d)

where here  $\langle V_1 V_2 \rangle = \langle V_{m_1-1/2} V_{-m_1+1/2} \rangle$  and  $\langle V'_1 V'_2 \rangle = \langle V'_{m'_1-1/2} V'_{-m'_1+1/2} \rangle$ . As expected, the coefficients resulting from the linear combinations effectively shift the spins  $j \to J$  and  $j' \to J'$  in the gamma functions coming from the bosonic correlators.

Among the states described above, the only chiral one corresponds to the SU(2) highestweight operator of type *A*. To simplify notation and for later convenience, we suppress the SU(2) labels and use the label *j* rather than *J* (here J = j - 1/2). This operator has quantum numbers

$$\mathcal{Y}_{j,m[A]}^{+,\epsilon_4}$$
:  $h = J = j - \frac{1}{2} = j' + \frac{1}{2} = J' = m'$ . (2.172)

The explicit form of this operator is simpler than the generic Ramond sector operator, and is given by

$$\mathcal{Y}_{j,m[A]}^{+,\varepsilon_4} = e^{-\frac{\varphi}{2}} \left( S_{+++} V_{j,m-\frac{1}{2}} + S_{-+-} V_{j,m+\frac{1}{2}} \right)_{j-\frac{1}{2},m} V_{j-1,j-1}' e^{\frac{i\varepsilon_4}{2}(H_4 + H_5)} \,. \tag{2.173}$$

As in the NS sector, the rest of the chiral operators belong to the spectrally flowed sectors. These are obtained by acting with the spectral flow operator (2.157). From the flowed Virasoro condition, similar to Eq. (2.158), the resulting operators must have m = m' (together with the relations in (2.172)), and so only the second term in (2.173) is non-vanishing, giving rise to the flowed Ramond operators (we now suppress also the label m = J = j - 1/2)

$$\mathcal{Y}_{j[A]}^{+,\varepsilon_{4},\omega} = e^{-\frac{\varphi}{2}} S_{-+-}^{\omega} V_{j,j}^{\omega} V_{j-1,j-1}^{'\omega} e^{\frac{i\varepsilon_{4}}{2}(H_{4}+H_{5})}, \qquad (2.174)$$

where

$$S^{\omega} \equiv e^{\frac{i}{2}[(1+2\omega)(-H_1+H_2)-H_3]}.$$
(2.175)

The corresponding two-point function is equivalent to the highest-weight case of (2.171a), up to the usual additional  $\delta_{\omega_1,-\omega_2}$  factor.

## 2.6.4.3 Holographic dictionary for light chiral primaries

We have reviewed three sets of *m*-basis vertex operators corresponding to chiral primaries of the holographic CFT. Two sets are in the NS sector:  $\mathcal{W}_{j,m}^-$  and  $\mathcal{X}_{j,m}^+$ , together with the corresponding spectral flowed operators  $\mathcal{W}_j^{\omega}$  and  $\mathcal{X}_j^{\omega}$ . The third set is in the Ramond sector,  $\mathcal{Y}_{j,m[A]}^{+,e_4}$  and its spectral flow,  $\mathcal{Y}_{j[A]}^{+,e_4,\omega}$ . From now on we shall omit the label *A* and the AdS<sub>3</sub> × S<sup>3</sup> chirality  $\epsilon = +$ , denoting this operator by  $\mathcal{Y}_j^{\omega,e_4}$ . Recall that in the spectral flowed sectors, the remaining states in the zero-mode algebra are obtained by acting with  $J_0^+$ , as discussed around Eqs. (2.120) and (2.162).

In order to reconstruct the corresponding local operators of the spacetime CFT, we need to combine such modes by going to the *x*-basis, as done in Eqs. (2.115) and (2.120) in the bosonic SL(2, $\mathbb{R}$ ) model.<sup>32</sup> For the operators at hand, the sum over *m* in the analog of Eqs. (2.115) and (2.120) factorizes between fermionic and bosonic contributions, leading to expressions of the following form:

$$\mathcal{W}_{j}^{\omega}(x) = e^{-\varphi}\psi^{\omega}(x)e^{i\omega H_{2}}V_{j}^{\omega}(x)V_{j-1,j-1}^{\prime\omega}, \qquad (2.176a)$$

$$\mathcal{X}_{j}^{\omega}(x) = e^{-\varphi}\psi^{\omega-1}(x)e^{i(\omega+1)H_{2}}V_{j}^{\omega}(x)V_{j-1,j-1}^{'\omega}, \qquad (2.176b)$$

$$\mathcal{Y}_{j}^{\omega,\varepsilon_{4}}(x) = e^{-\frac{\varphi}{2}} S^{\omega}(x) V_{j}^{\omega}(x) V_{j-1,j-1}^{'\omega} e^{\frac{i\varepsilon_{4}}{2}(H_{4}+H_{5})}.$$
(2.176c)

Here  $\psi^{\omega}(x)$  and  $S^{\omega}(x)$  are defined as follows. First, note that the fermions  $\psi^a$  introduced in (2.131), which generate an affine  $\widehat{\mathfrak{sl}(2,\mathbb{R})}_{-2}$  algebra with level  $k_{\psi} = -2$ , constitute affine primaries with spin  $J_{\psi} = -1$ , on which, however, the zero-mode currents act as in (2.107) but with<sup>33</sup>  $J \to 1 - J$ , i.e.  $J_0^{\pm}|J, m \pm 1\rangle = (m \pm J)|J, m \pm 1\rangle$ . As a consequence, and in contrast with what happens with bosonic primaries, the action  $J_0^+$ on  $\psi^-$ , the lowest-weight state, is truncated. Identifying  $\psi^{\omega=0}(0) = \psi^-$ , the resulting *x*-basis operator has only three terms:

$$\psi^{\omega=0}(x) \equiv \psi^{-}(x) = e^{xJ_{0}^{+}}\psi^{-}e^{-xJ_{0}^{+}} = \psi^{-} - 2x\psi^{3} + x^{2}\psi^{+}.$$
(2.177)

Of course, we already knew the action of the currents on  $\psi^a$  from (2.131), but the advantage of the above discussion is that it extends to the spectrally flowed sectors. Indeed,

 $<sup>^{32}</sup>$ In this thesis we are interested in operators of fixed R-charge. Hence, and in contrast to [99, 100], we do not introduce isospin variables in the SU(2) sector.

<sup>&</sup>lt;sup>33</sup>This convention is perhaps more natural from the SL(2, $\mathbb{R}$ ) point of view, but we have decided to employ the conventions used in the most relevant literature for us, i.e. [101, 103, 99, 100].

 $\psi^{\omega}(0) = \sqrt{n_5}e^{-i(1+\omega)H_1}$  is the lowest-weight component of a spin  $J^{\omega}_{\psi} = -1 - \omega$  field. The corresponding *x*-basis operator is of the form

$$\psi^{\omega}(x) \equiv \sqrt{n_5} e^{x J_0^+} e^{-i(1+\omega)H_1} e^{-x J_0^+}, \qquad (2.178)$$

and contains  $1 - 2J_{\psi}^{\omega} = 2\omega + 3$  terms.

Similarly, the spin field (2.175) is the lowest-weight component of a representation with SL(2,  $\mathbb{R}$ ) and SU(2) spins  $(J_S^{\omega}, J_S'^{\omega}) = (-\frac{1}{2} - \omega, \frac{1}{2} + \omega)$ , such that the *x*-basis operator is

$$S^{\omega}(x) \equiv e^{xJ_0^+} S^{\omega}_{-+-} e^{-xJ_0^+}, \qquad (2.179)$$

and contains  $2(1 + \omega)$  terms.

We thus have three types of vertices,  $W_j^{\omega}(x,z)$ ,  $\mathcal{X}_j^{\omega}(x,z)$  and  $\mathcal{Y}_j^{\omega,\pm}(x,z)$ , which correspond to local chiral primary operators of the boundary theory. As mentioned before, these should be completed with analogous antiholomorphic excitations, which have been omitted in the presentation. As discussed around Eqs. (2.162) and (2.172), their boundary weights are given by

$$h\left[\mathcal{W}_{j}^{\omega}\right] = j_{\omega} - 1, \qquad h\left[\mathcal{Y}_{j}^{\omega,\pm}\right] = j_{\omega} - \frac{1}{2}, \qquad h\left[\mathcal{X}_{j}^{\omega}\right] = j_{\omega}, \qquad (2.180)$$

where j = j' + 1 and so

$$j_{\omega} = j + \frac{n_5}{2}\omega, \qquad j = 1, \frac{3}{2}, \dots, \frac{n_5}{2}, \qquad \omega = 0, 1, \dots.$$
 (2.181)

Up to normalization, which will be fixed shortly, these operators are identified with the chiral primaries of the holographic CFT listed in Eq. (2.100). In the  $\mathcal{Y}_{j}^{\omega,\varepsilon_{4}}$  tower,  $\varepsilon_{4} = \pm$  is identified with the boundary quantum number  $\dot{A}$  in (2.99). Note also that the  $\mathcal{W}_{j}^{\omega}$  tower starts with the identity operator of the boundary theory.<sup>34</sup> The dictionary is summarized in Table 2.3.

Worldsheet	Weight h	Twist n	Dual Operator
$\mathcal{W}^{\omega}_{j}$	$j_\omega - 1$	$2j_{\omega}-1$	$O_n^-$
$\mathcal{Y}^{\omega, \varepsilon_4}_j$	$j_\omega - \frac{1}{2}$	$2j_{\omega}-1$	$O_n^{\dot{A}}$
$\mathcal{X}_{j}^{\omega}$	jω	$2j_{\omega}-1$	$O_n^+$

TABLE 2.3: Dictionary between worldsheet vertex operators and chiral primaries of the holographically dual CFT. Here  $j_{\omega} = j + \frac{n_5}{2}\omega$ ,  $j = 1, \frac{3}{2}, \dots, \frac{n_5}{2}$ , and  $\omega = 0, 1, \dots$ 

<sup>&</sup>lt;sup>34</sup>As discussed in [110], this is subtle, since the operator is actually a spectral-flow-sector dependent constant. This subtlety is related to the fact that spectral flow charge is not conserved in *n*-point functions with  $n \ge 3$ , and was resolved in [114] by performing a Legendre transform to the microcanonical ensemble, in which the total number of fundamental strings is fixed.

Although most of the chiral primaries of the holographic CFT are accounted for by considering the ranges given in (2.181), it is known that those belonging to the the *n*-twisted sectors with  $n = pn_5$  with  $p \in \mathbb{N}$  are still missing [99, 100]. These would correspond to operators sitting at the boundary of the allowed range of *j* in Eq. (2.106), at which the spectrum becomes degenerate and the continuous representations appear [104, 115]. The absence of these states in the worldsheet spectrum has been related to the fact that the NS5-F1 model sits at a singular point in the moduli space where all RR modes are turned off [50].

The twist *n* of the holographic CFT operators is identified as [99, 100]

$$n = 2j - 1 + n_5 \omega \,. \tag{2.182}$$

Let us make a side comment regarding the limit in which there is only a single NS5 brane sourcing the background,  $n_5 = 1$ . This model is special in that it corresponds to the tensionless limit of the theory. It has to be treated with care since the usual RNS formalism outlined above breaks down due to the fact that the bosonic SU(2) level would become negative. It was shown in [87, 116] that for  $n_5 = 1$  the worldsheet theory is exactly dual to the supersymmetric symmetric orbifold  $(\mathbb{T}^4)^{n_1}/S_{n_1}$ . In this model, the discrete series is absent, the spectrum truncates to j = 1/2, physical states have  $\omega > 0$ , and the spectral flow charge is identified with n, i.e.  $n = \omega$ . Eq. (2.182) is the known generalization of this relation for  $n_5 > 1$ .

In order to fix the normalization of the operators, we compute their two-point functions. Making use of

$$\langle \psi^{\omega_1}(x_1)\psi^{\omega_2}(x_2)\rangle \times \text{c.c.} = x_{12}^{2(\omega_1+1)} \langle e^{-i(1+\omega_1)H_1}e^{i(1-\omega_2)H_1}\rangle \times \text{c.c.} = \delta_{\omega_1,-\omega_2} \frac{|x_{12}|^{4(\omega_1+1)}}{|z_{12}|^{2(\omega_1+1)^2}}, \\ \langle S^{\omega_1,+}(x_1)S^{\omega_2,-}(x_2)\rangle \times \text{c.c.} = \frac{x_{12}^{2\omega_1+2}}{z_{12}} \langle S^{\omega_1}_{-+-}S^{\omega_2}_{+-+}\rangle \times \text{c.c.} = \delta_{\omega_1,-\omega_2} \frac{|x_{12}|^{4\omega_1+2}}{|z_{12}|^{4\omega_1+2}},$$

we obtain

$$\langle \mathcal{W}_{j_1}^{\omega_1}(x_1, z_1) \mathcal{W}_{j_2}^{\omega_2}(x_2, z_2) \rangle = \frac{n_5^2 B(j_1)}{16} \frac{\delta(j_1 - j_2) \delta_{\omega_1, -\omega_2}}{|x_{12}|^{4(j_1 - 1) + 2n_5 \omega_1} |z_{12}|^4},$$
(2.183)

$$\langle \mathcal{X}_{j_1}^{\omega_1}(x_1, z_1) \mathcal{X}_{j_2}^{\omega_2}(x_2, z_2) \rangle = \frac{n_5^2 B(j_1)}{16} \frac{\delta(j_1 - j_2) \delta_{\omega_1, -\omega_2}}{|x_{12}|^{4j_1 + 2n_5\omega_1} |z_{12}|^4},$$
(2.184)

$$\langle \mathcal{Y}_{j_1}^{\omega_1,\pm}(x_1,z_1)\mathcal{Y}_{j_2}^{\omega_2,\mp}(x_2,z_2)\rangle = \frac{n_5 B(j_1)}{(2j_1-1+n_5\omega_1)^2} \frac{\delta(j_1-j_2)\delta_{\omega_1,-\omega_2}}{|x_{12}|^{4j_1+2(n_5\omega_1-1)}|z_{12}|^4}$$
(2.185)

Here we have used that in the spectrally flowed R sectors the denominator in the extra factor of the corresponding vertex operator in the " $-\frac{3}{2}$ " picture is shifted as  $2j - 1 \rightarrow 2j - 1 + n_5\omega$  as compared to (2.170). Moreover, the  $V_{\text{conf}}$  factor in Eq. (2.121) is cancelled by the pole appearing in (2.114) upon setting  $m_1 = j_1$ .

The string two-point function is then obtained by including an extra factor  $g_s^{-2} \sim n_1/n_5$  as usual in string perturbation theory, fixing  $z_1 = 0$  and  $z_2 = 1$ , and dividing by a volume of the conformal group that leaves such worldsheet insertions fixed. As discussed in [110, 103], this cancels the divergence coming from  $\delta(j_1 - j_2)$ , leaving a constant *j*-dependent factor of the form  $(2j - 1 + n_5\omega)$ . As a consequence, the holographic dictionary reads

$$O_n^{--}(x,\bar{x}) \leftrightarrow A_{\rm NS}(j,\omega)\mathcal{W}_j^{\omega}(x,\bar{x}), \qquad O_n^{++}(x) \leftrightarrow A_{\rm NS}(j,\omega)\mathcal{X}_j^{\omega}(x,\bar{x}), \qquad (2.186)$$

$$O_n^{AB} \leftrightarrow A_{\mathbb{R}}(j,\omega)\mathcal{Y}_j^{\omega,AB}(x,\bar{x}),$$
(2.187)

with *n* related to the worldsheet quantum numbers as in (2.182) and where [113, 99]

$$A_{\rm NS}(j,\omega) = \frac{4g_s}{\sqrt{n_5^2 B(j)(2j-1+n_5\omega)}}, \qquad A_{\rm R}(j,\omega) = g_s \sqrt{\frac{(2j-1+n_5\omega)}{n_5 B(j)}}.$$
(2.188)

Of course, this identification is only expected to hold at small string coupling, i.e. for  $n_1 \gg n_5$ . Analysis and comparison of boundary and worldsheet three-point functions were carried out in [117, 99, 113].

# 2.7 Null-gauged WZW models

In this section we review some relevant aspects of the models that we shall study in this thesis. We will aim to be brief where possible; the interested reader can find the details in the works [51, 52, 53, 54].

We consider the (10+2)-dimensional upstairs target  $AdS_3 \times S^3 \times \mathbb{R}_t \times S_y^1 \times \mathbb{T}^4$ , where we have introduced coordinates *t* and *y* for the timelike  $\mathbb{R}$  and spacelike  $S^1$  factors respectively. Since the Cartan direction in  $SL(2, \mathbb{R})$  is timelike, and that of SU(2) is spacelike, and the levels are the same, one can form null linear combinations  $J_{sl}^3 \pm J_{su}^3$ in both holomorphic and antiholomorphic sectors of the worldsheet theory. Gauging such null currents leads [118] to the background sourced by a circular array of NS5 branes on their Coulomb branch [119, 120, 121].

The (10+2)-dimensional models have other null currents that are linear combinations of  $J_{sl}^3$ ,  $J_{su}^3$ ,  $\partial_t$  and  $\partial_y$ . It was recently found that particular linear combinations of these currents give rise to a family of backgrounds that include NS5-P and NS5-F1 BPS circular supertubes [122, 123], as well as NS5-F1-P BPS and non-BPS spectral flowed supertubes [124, 125, 37, 96, 94]. In the IR, the backgrounds become asymptotically AdS<sub>3</sub> × S<sup>3</sup>, and correspond to heavy states of the holographically dual CFT at the symmetric orbifold point [94].

# 2.7.1 Null-gauged sigma models

We now briefly review the null gauging formalism for general sigma models, before specialising to WZW models (see e.g. [126, 127, 128, 129]). In this passage we follow the presentation of [54]. We use units in which  $\alpha' = 1$ , and work at tree level in the string coupling  $g_s$ .

Consider the string worldsheet  $\mathcal{M}_2$  and an embedding map  $\varphi$  into a pseudo-Riemannian manifold  $\mathcal{N}$ , namely  $\varphi : \mathcal{M}_2 \to \varphi(\mathcal{M}_2) \subset \mathcal{N}$ . The target manifold  $\mathcal{N}$  is endowed with a metric with components  $G_{ij}$ . We wish to gauge a set of Killing vectors  $\xi_a$  generating isometries of  $\mathcal{N}$ , where *a* labels the different Killing vectors (in this thesis we will always have a = 1, 2). We introduce a set of independent worldsheet gauge fields  $\mathcal{A}^a$ , one corresponding to each Killing vector. Then the kinetic term in the string sigma model action is written in terms of the covariant derivative

$$\mathcal{D}\varphi^{i} = \partial\varphi^{i} - \mathcal{A}^{a}\xi^{i}_{a}, \qquad (2.189)$$

and takes the form

$$\mathcal{L}_{\mathrm{K}} = \mathcal{D}\varphi^{i} G_{ij} \overline{\mathcal{D}}\varphi^{j} = (\partial\varphi^{i} - \mathcal{A}^{a}\xi^{i}_{a}) G_{ij} (\bar{\partial}\varphi^{j} - \bar{\mathcal{A}}^{a}\xi^{j}_{a}) .$$
(2.190)

To write the gauged Wess-Zumino (WZ) term we introduce target-space one-forms  $\theta_a$  (we follow the notation of [129]), pulled back to the worldsheet. The WZ term can then be written as

$$\mathcal{L}_{WZ} = B_{ij}\partial\varphi^i\bar{\partial}\varphi^j + \mathcal{A}^a\theta_{a,i}\bar{\partial}\varphi^i - \bar{\mathcal{A}}^a\theta_{a,i}\partial\varphi^i + \xi^i_{[a}\theta_{b],i}\mathcal{A}^a\bar{\mathcal{A}}^b , \qquad (2.191)$$

where  $\theta_{a,i}$  denotes the *i*<sup>th</sup> component of the one-form  $\theta_a$ . For our null-gauged models, the target-space one-forms  $\theta_a$  are given by

$$\theta_a = (-1)^{a+1} \xi_a \cdot d\varphi \equiv (-1)^{a+1} \xi_a^i G_{ij} d\varphi^j, \qquad (a = 1, 2).$$
(2.192)

For a consistent gauging, the following conditions must hold:

$$\iota_a H = d\theta_a, \qquad \iota_a \theta_b = -\iota_b \theta_a, \qquad (2.193)$$

where H = dB. The expression (2.192) implies that half of the gauge field components decouple, so that they are naturally chiral. As a result, in our  $U(1) \times U(1)$  gauged models, the coefficient of the term quadratic in gauge fields is proportional to the quantity

$$\Sigma \equiv -\frac{1}{2}\xi_1^i G_{ij}\xi_2^j.$$
 (2.194)

All together, the terms in the action involving the gauge fields then reduce to

$$\mathcal{L}_{\mathcal{A}} = -2\mathcal{A}^2 \xi_2^i G_{ij} \bar{\partial} \varphi^j - 2\bar{\mathcal{A}}^1 \xi_1^i G_{ij} \partial \varphi^j - 4\mathcal{A}^2 \bar{\mathcal{A}}^1 \Sigma , \qquad (2.195)$$

and in the following, we shall denote  $\mathcal{A} \equiv \mathcal{A}^2$ ,  $\bar{\mathcal{A}} \equiv \bar{\mathcal{A}}^1$ .

We define the worldsheet currents  $\mathcal{J}$ ,  $\overline{\mathcal{J}}$  to be pull-backs of the target-space one-forms as follows<sup>35</sup>

$$\mathcal{J} \equiv -\theta_1 \cdot \partial \varphi \equiv -\theta_{1,i} \partial \varphi^i , \qquad \bar{\mathcal{J}} \equiv \theta_2 \cdot \bar{\partial} \varphi \equiv \theta_{2,i} \bar{\partial} \varphi^i . \tag{2.196}$$

By using (2.192), one can then write the gauge terms (2.195) as

$$\mathcal{L}_{\mathcal{A}} = 2\mathcal{A}\theta_{2,i}\bar{\partial}\varphi^{i} - 2\bar{\mathcal{A}}\theta_{1,i}\partial\varphi^{i} - 4\mathcal{A}\bar{\mathcal{A}}\Sigma \equiv 2\mathcal{A}\bar{\mathcal{J}} + 2\bar{\mathcal{A}}\mathcal{J} - 4\mathcal{A}\bar{\mathcal{A}}\Sigma.$$
(2.197)

Upon integrating out the gauge fields, the gauge terms in the action then become

$$\frac{\mathcal{J}\bar{\mathcal{J}}}{\Sigma} = -\frac{1}{\Sigma} (\theta_1 \cdot \partial \varphi) (\theta_2 \cdot \bar{\partial} \varphi) = \frac{1}{\Sigma} (\xi_1 \cdot \partial \varphi) (\xi_2 \cdot \bar{\partial} \varphi), \qquad (2.198)$$

where  $\xi_1 \cdot \partial \varphi \equiv \xi_1^i G_{ij} \partial \varphi^j$ . Thus the overall effect of the null gauging procedure is to add the term (2.198) to the ungauged sigma model lagrangian.

# 2.7.2 Null-gauged WZW models

We now specialise the discussion to the case where the upstairs theory is a WZW model whose target space is a Lie group  $\mathcal{G}$ , and thus we replace  $\varphi$  with a  $\mathcal{G}$ -valued function  $g : \mathcal{M}_2 \to \mathcal{G}$ . We will shortly consider  $\mathcal{G}$  to be a direct product of simple and abelian factors, but for the moment we focus on one of the simple factors. We follow in places the presentation in [53].

We wish to gauge the action of a subgroup  $\mathcal{H} \subset \mathcal{G}$ . Its action on  $\mathcal{G}$  is defined by the group homomorphism embeddings

$$\ell: \mathcal{H} \to \mathcal{G}_L$$
,  $r: \mathcal{H} \to \mathcal{G}_R$ , (2.199)

where  $G_L \times G_R$  is the standard left-right isometry group, and such that we will gauge the transformations

$$g \mapsto \ell(h) g r(h)^{-1}, \qquad h \in \mathcal{H}.$$
 (2.200)

The group embeddings  $\ell$  and r induce corresponding Lie algebra homomorphisms. Since the meaning will be clear from the context, it is convenient to abuse notation and

<sup>&</sup>lt;sup>35</sup>Note that  $\mathcal{J}, \overline{\mathcal{J}}$  are related to the CFT current operators  $J, \overline{J}$  by factors of i, e.g.  $J = i\mathcal{J}$ .

re-use the same symbols  $\ell$  and r for the induced Lie algebra homomorphisms,

$$\ell: \mathfrak{h} \to \mathfrak{g}, \qquad r: \mathfrak{h} \to \mathfrak{g}.$$
 (2.201)

To write the corresponding Killing vector field, let us denote the left(right)-invariant vector field corresponding to a generic  $X \in \mathfrak{g}$  by  $X^L$  ( $X^R$ ). Given a basis of  $\mathfrak{h}$ , for each element  $X_a$  there is a corresponding Killing vector field given by (see e.g. [129])

$$\xi_a \equiv -\ell (X_a)^R - r(X_a)^L \,. \tag{2.202}$$

Let us write the left and right Maurer-Cartan one-forms as

$$\theta_L = g^{-1} dg, \qquad \theta_R = -dg g^{-1}.$$
(2.203)

We denote by  $\langle \cdot, \cdot \rangle$  the standard inner product on g given by the Killing form. More explicitly, for matrix groups we use the normalisation  $\langle A, B \rangle = \text{Tr}(AB)$ . In terms of these, the one-forms  $\theta_a$  introduced in Eqs. (2.191)–(2.192) take the form

$$\theta_a = \langle \ell(X_a), \theta_R \rangle - \langle r(X_a), \theta_L \rangle .$$
(2.204)

### 2.7.3 The models we study

We work in Type II superstring theory, however we suppress worldsheet fermions in this section for ease of presentation. Worldsheet fermions will be discussed in detail in Section 3.2 below. We consider the cosets<sup>36</sup>

$$\mathcal{G}/\mathcal{H} \times \mathbb{T}^4 = \frac{SL(2,\mathbb{R}) \times SU(2) \times \mathbb{R}_t \times U(1)_y}{U(1)_L \times U(1)_R} \times \mathbb{T}^4.$$
(2.205)

To define the action of  $\mathcal{H} = U(1)_L \times U(1)_R$  we must specify the embedding into each of the four subgroups of the upstairs group  $\mathcal{G}$  that participate in the gauging. Parametrising  $SL(2,\mathbb{R})$  as SU(1,1), we introduce coordinates for the upstairs subgroup elements  $as^{37}$ 

$$g = (g_{\rm sl}, g_{\rm su}, g_t, g_y) = \left( e^{\frac{i}{2}(\tau - \sigma)\sigma_3} e^{\rho\sigma_1} e^{\frac{i}{2}(\tau + \sigma)\sigma_3}, e^{\frac{i}{2}(\psi - \phi)\sigma_3} e^{i(\frac{\pi}{2} - \theta)\sigma_1} e^{-\frac{i}{2}(\psi + \phi)\sigma_3}, e^t, e^{\frac{iy}{R_y}} \right), \qquad (2.206)$$

where  $\sigma_i$  denotes the *i*<sup>th</sup> Pauli matrix and  $y \in [0, 2\pi R_y)$ . At the level of the algebra, the chiral embeddings we consider are specified by eight arbitrary real parameters  $l_i$ ,  $r_i$ ,

<sup>&</sup>lt;sup>36</sup>More precisely, the upstairs model involves the universal cover of SL(2, R), and globally we gauge  $\mathbb{R} \times U(1)$ , as we discuss in more detail in Section 3.2.2; see also [52]. <sup>37</sup>We adopt the conventions in [2, 3].

i = 1, 2, 3, 4, as follows (the ordering of subgroups is as in Eq. (2.206)),

$$\ell(\alpha) = \left(i l_1 \alpha \sigma_3, -i l_2 \alpha \sigma_3, l_3 \alpha, -i \frac{l_4}{R_y} \alpha\right), \qquad r(\alpha) = 0,$$
  

$$r(\beta) = -\left(i r_1 \beta \sigma_3, -i r_2 \beta \sigma_3, r_3 \beta, -i \frac{r_4}{R_y} \beta\right), \qquad \ell(\beta) = 0,$$
(2.207)

where  $\alpha, \beta \in \mathbb{R}$  and the signs have been chosen for later convenience, in particular for Eq. (2.215) below. The group action (2.200) being gauged is then

$$g \mapsto \left(e^{il_1\alpha\sigma_3}g_{\rm sl}e^{ir_1\beta\sigma_3}, e^{-il_2\alpha\sigma_3}g_{\rm su}e^{-ir_2\beta\sigma_3}, e^{l_3\alpha}g_te^{r_3\beta}, e^{-i\frac{l_4}{R_y}\alpha}g_ye^{-i\frac{r_4}{R_y}\beta}\right).$$
(2.208)

The general gauge-invariant action for such asymmetric cosets can be found in [130]. We introduce two (independent)  $\mathfrak{h}$ -valued worldsheet gauge fields  $(\mathcal{A}_1, \mathcal{A}_1)$  and  $(\mathcal{A}_2, \mathcal{A}_2)$ . The gauged WZW action takes the form

$$S = \sum_{j} \operatorname{sgn}(\kappa_{j}) \frac{k_{j}}{\pi} \left( \int_{\mathcal{M}_{2}} \frac{1}{2} \operatorname{Tr} \left[ g^{-1} \partial g g^{-1} \overline{\partial} g \right]_{j} d^{2}z + i \int_{\mathcal{M}_{3}} \frac{1}{3!} \operatorname{Tr} \left[ g^{-1} dg \wedge g^{-1} dg \wedge g^{-1} dg \right]_{j} + \int_{\mathcal{M}_{2}} \operatorname{Tr} \left[ -\sum_{a=1}^{2} \left[ \ell(\bar{\mathcal{A}}_{a}) \partial g g^{-1} \right]_{j} + \sum_{a=1}^{2} \left[ r(\mathcal{A}_{a}) g^{-1} \overline{\partial} g \right]_{j} - \sum_{a,b=1}^{2} \left[ g^{-1} \ell(\bar{\mathcal{A}}_{a}) g r(\mathcal{A}_{b}) \right]_{j} d^{2}z \right),$$

$$(2.209)$$

where *j* runs over the Lie algebras,  $\mathcal{M}_3$  is a three-dimensional auxiliary space such that  $\mathcal{M}_2 = \partial \mathcal{M}_3$ ,  $k_j$  are the levels of the Kac-Moody algebras, and sgn( $\kappa_j$ ) are the signatures of the respective Killing forms, which in our conventions is positive for SL(2,  $\mathbb{R}$ ) and negative for (SU(2),  $\mathbb{R}_t$ ,  $\mathbb{S}_y^1$ ). Here the embeddings  $\ell$ , *r* should be understood as corresponding to each respective Lie subalgebra, i.e. the components of the right-hand sides of Eq. (2.207).

We now recall from the discussion of general gaugings of sigma models in Eqs. (2.189)–(2.198) that, since the gauge fields are null and chiral, one of their components simply drops out, such that we can set

$$A_1 = 0$$
,  $\bar{A}_2 = 0$ . (2.210)

The gauge field embeddings are then

$$\ell(\bar{A}_{1}) = \left(i l_{1} \bar{A}_{1} \sigma_{3}, -i l_{2} \bar{A}_{1} \sigma_{3}, l_{3} \bar{A}_{1}, -i \frac{l_{4}}{R_{y}} \bar{A}_{1}\right), \qquad \ell(A_{2}) = 0,$$

$$r(A_{2}) = -\left(i r_{1} A_{2} \sigma_{3}, -i r_{2} A_{2} \sigma_{3}, r_{3} A_{2}, -i \frac{r_{4}}{R_{y}} A_{2}\right), \qquad r(\bar{A}_{1}) = 0,$$
(2.211)

consistently with (2.207). As before, in order to lighten the notation we set  $A = A_2$ ,  $\bar{A} = \bar{A}_1$  from now on.

We introduce the currents (our conventions follow [54, App. A])

$$j_{\rm sl}^3 = k_{\rm sl} \,{\rm Tr}\left(-i\frac{\sigma_3}{2}\,\partial g_{\rm sl}\,g_{\rm sl}^{-1}\right) \,, \qquad \bar{j}_{\rm sl}^3 = k_{\rm sl} \,{\rm Tr}\left(-i\frac{\sigma_3}{2}\,g_{\rm sl}^{-1}\,\bar{\partial}g_{\rm sl}\right) \,, \tag{2.212}$$

and similarly for SU(2). Their explicit form in our coordinates is

$$j_{\rm sl}^3 = n_5 (\cosh^2 \rho \ \partial \tau + \sinh^2 \rho \ \partial \sigma) , \qquad \bar{j}_{\rm sl}^3 = n_5 (\cosh^2 \rho \ \bar{\partial} \tau - \sinh^2 \rho \ \bar{\partial} \sigma) ,$$

$$j_{\rm su}^3 = n_5 (\cos^2 \theta \ \partial \psi - \sin^2 \theta \ \partial \phi) , \qquad \bar{j}_{\rm su}^3 = -n_5 (\cos^2 \theta \ \bar{\partial} \psi + \sin^2 \theta \ \bar{\partial} \phi) .$$

$$(2.213)$$

We also define

$$\mathsf{P}_L^t = \partial t$$
,  $\mathsf{P}_R^t = \bar{\partial} t$ ,  $\mathsf{P}_L^y = \partial y$ ,  $\mathsf{P}_R^y = \bar{\partial} y$ . (2.214)

Note that, as usual, the bosonic subsector of the supersymmetric WZW model has  $k_{sl} = n_5 + 2$  and  $k_{su} = n_5 - 2$  while the full supersymmetric model has  $k_{sl} = k_{su} = n_5$ . As noted above, we are suppressing worldsheet fermions in the present section. The shift in the levels is important (see e.g. the discussion in [54]), and we will take care of this in detail when discussing results in the worldsheet CFT in later chapters. When discussing supergravity solutions we will work in the usual supergravity regime  $n_5 \gg 1$  (and in the fivebrane decoupling limit  $g_s \rightarrow 0$ ) and thus for our purposes in this section we can simply work with  $k_{sl} = k_{su} = n_5$ . To have canonical kinetic terms we set the (otherwise irrelevant)  $\widehat{u(1)}$  levels to be  $k_t = 2$ ,  $k_y = 2R_y^2$ .

The group action that we gauge, defined in Eq. (2.207), corresponds to gauging the currents

$$\mathcal{J} = l_1 j_{sl}^3 + l_2 j_{su}^3 + l_3 \mathsf{P}_{\mathsf{L}}^t + l_4 \mathsf{P}_{\mathsf{L}}^y, 
\bar{\mathcal{J}} = r_1 \bar{j}_{sl}^3 + r_2 \bar{j}_{su}^3 + r_3 \mathsf{P}_{\mathsf{R}}^t + r_4 \mathsf{P}_{\mathsf{R}}^y,$$
(2.215)

which we require to be null by imposing

$$n_5(l_1^2 - l_2^2) + l_3^2 - l_4^2 = 0$$
,  $n_5(r_1^2 - r_2^2) + r_3^2 - r_4^2 = 0$ . (2.216)

One can use these constraints to fix the overall normalization of the gauging parameters. We assume that  $l_1 = r_1 \neq 0$  and divide through by  $l_1^2$  and  $r_1^2$ , to work with the ratios

$$l_i = \frac{l_i}{l_1}, \qquad r_i = \frac{r_i}{r_1}, \qquad i = 2, 3, 4.$$
 (2.217)

In practice this has the same effect as setting  $l_1 = r_1 = 1$ , however we have introduced a separate notation for later convenience. Of course, one can modify this step accordingly to deal with models in which  $l_1 = r_1 = 0$ . For later use we record that the ratio parameters  $l_i$ ,  $r_i$ , i = 2, 3, 4 are subject to the constraints

$$n_5(1-l_2^2) + l_3^2 - l_4^2 = 0$$
,  $n_5(1-r_2^2) + r_3^2 - r_4^2 = 0$ . (2.218)

In the upstairs model, the line element and NSNS three-form flux are given by

$$ds^{2} = n_{5} \left( -\cosh^{2}\rho d\tau^{2} + d\rho^{2} + \sinh^{2}\rho d\sigma^{2} + d\theta^{2} + \cos^{2}\theta d\psi^{2} + \sin^{2}\theta d\phi^{2} \right) - dt^{2} + dy^{2},$$
  

$$H = n_{5} \left( \sinh 2\rho \ d\rho \wedge d\tau \wedge d\sigma + \sin 2\theta \ d\theta \wedge d\psi \wedge d\phi \right).$$
(2.219)

The Killing vectors associated to the group action (2.200) being gauged are

$$\begin{aligned} \xi_{\rm L} &= (\partial_{\tau} - \partial_{\sigma}) - l_2 (\partial_{\psi} - \partial_{\phi}) + l_3 \partial_t - l_4 \partial_y ,\\ \xi_{\rm R} &= (\partial_{\tau} + \partial_{\sigma}) + r_2 (\partial_{\psi} + \partial_{\phi}) + r_3 \partial_t - r_4 \partial_y , \end{aligned}$$
(2.220)

and so we obtain the one-forms  $\theta_a$ ,

$$\theta_{\rm L} = -n_5 \left[ \left( \cosh^2 \rho \, d\tau + \sinh^2 \rho \, d\sigma \right) + l_2 \left( \cos^2 \theta \, d\psi - \sin^2 \theta \, d\phi \right) \right] - \left( l_3 \, dt + l_4 \, dy \right),$$
  

$$\theta_{\rm R} = n_5 \left[ \left( \cosh^2 \rho \, d\tau - \sinh^2 \rho \, d\sigma \right) - r_2 \left( \cos^2 \theta \, d\psi + \sin^2 \theta \, d\phi \right) \right] + r_3 \, dt + r_4 \, dy \,.$$
(2.221)

The full null-gauged Wess-Zumino-Witten action is then

$$S = S_0^{\rm sl} + S_{\mathcal{A}}^{\rm sl} + S_0^{\rm su} + S_{\mathcal{A}}^{\rm su} + S_0^{t,y} + S_{\mathcal{A}}^{t,y}, \qquad (2.222)$$

with

$$S_{0}^{\rm sl} = \frac{n_{5}}{\pi} \int \left[ \partial \rho \bar{\partial} \rho + \mathrm{sh}^{2} \rho \, \partial \sigma \bar{\partial} \sigma - \mathrm{ch}^{2} \rho \, \partial \tau \bar{\partial} \tau - \mathrm{sh}^{2} \rho \left( \partial \sigma \bar{\partial} \tau - \partial \tau \bar{\partial} \sigma \right) \right] d^{2}z \,,$$

$$S_{\mathcal{A}}^{\rm sl} = \frac{2n_{5}}{\pi} \int \left[ \bar{\mathcal{A}} \left( \mathrm{sh}^{2} \rho \, \partial \sigma + \mathrm{ch}^{2} \rho \, \partial \tau \right) + \mathcal{A} \left( \mathrm{ch}^{2} \rho \, \bar{\partial} \tau - \mathrm{sh}^{2} \rho \, \bar{\partial} \sigma \right) - \mathcal{A} \bar{\mathcal{A}} \, \mathrm{ch}(2\rho) \right] d^{2}z \,,$$

$$S_{0}^{\rm su} = \frac{n_{5}}{\pi} \int \left[ \partial \theta \bar{\partial} \theta + c_{\theta}^{2} \partial \psi \bar{\partial} \psi + s_{\theta}^{2} \partial \phi \bar{\partial} \phi + c_{\theta}^{2} \left( \partial \phi \bar{\partial} \psi - \bar{\partial} \phi \partial \psi \right) \right] d^{2}z \,,$$

$$S_{\mathcal{A}}^{\rm su} = \frac{2n_{5}}{\pi} \int \left[ l_{2} \bar{\mathcal{A}} \left( c_{\theta}^{2} \partial \psi - s_{\theta}^{2} \partial \phi \right) - r_{2} \mathcal{A} \left( c_{\theta}^{2} \bar{\partial} \psi + s_{\theta}^{2} \bar{\partial} \phi \right) - l_{2} r_{2} \mathcal{A} \bar{\mathcal{A}} \cos(2\theta) \right] d^{2}z \,,$$

$$S_{0}^{t,y} = \frac{1}{\pi} \int \left[ -\partial t \bar{\partial} t + \partial y \bar{\partial} y \right] d^{2}z \,,$$

$$S_{\mathcal{A}}^{t,y} = \frac{2}{\pi} \int \left[ l_{3} \bar{\mathcal{A}} \partial t + r_{3} \mathcal{A} \bar{\partial} t + l_{4} \bar{\mathcal{A}} \partial y + r_{4} \mathcal{A} \bar{\partial} y - (l_{3} r_{3} - l_{4} r_{4}) \mathcal{A} \bar{\mathcal{A}} \right] d^{2}z \,,$$
(2.223)

where we have used the shorthands  $c_{\theta} = \cos \theta$  and  $s_{\theta} = \sin \theta$ .

We note that with the Killing vectors (2.220), the quantity  $\Sigma$  defined in Eq. (2.194) becomes

$$\Sigma = \frac{1}{2} \Big( n_5 \big[ \cosh(2\rho) + l_2 r_2 \cos(2\theta) \big] + l_3 r_3 - l_4 r_4 \Big).$$
 (2.224)

For convenience let us define the rescaled quantity

$$\Sigma_0 = \frac{1}{n_5}\Sigma = \sinh^2 \rho + l_2 r_2 \cos^2 \theta + \frac{1 - l_2 r_2}{2} + \frac{l_3 r_3 - l_4 r_4}{2n_5}.$$
 (2.225)

Upon integrating out the gauge fields, the gauging procedure effectively adds a term quadratic in the currents, resulting in an action of the schematic form

$$S_{\rm WZW} + \frac{2}{\pi} \int \frac{\mathcal{J}\bar{\mathcal{J}}}{\Sigma} d^2 z \,. \tag{2.226}$$

As we will discuss next, one can then read off the resulting line element and *B*-field of the gauged model. The change in the measure also generates a non-trivial dilaton, which can be obtained by solving for the vanishing of the appropriate worldsheet one-loop beta function. It is important to note that these solutions are exact in  $\alpha'$ , and thus may receive only non-perturbative corrections: this is a direct consequence of the exact nature of the gWZW models we consider.

# 2.7.4 Supergravity fields

By integrating out the gauge fields and choosing the gauge  $\sigma = \tau = 0$ , we obtain the following line element and *B*-field:

$$ds^{2} = -\frac{h_{t}}{\Sigma_{0}}dt^{2} + \frac{h_{y}}{\Sigma_{0}}dy^{2} + \frac{(l_{3}r_{4} + l_{4}r_{3})}{n_{5}\Sigma_{0}}dtdy + n_{5}(d\theta^{2} + d\rho^{2}) + n_{5}\frac{h_{\phi}}{\Sigma_{0}}\sin^{2}\theta d\phi^{2} + n_{5}\frac{h_{\psi}}{\Sigma_{0}}\cos^{2}\theta d\psi^{2} - \frac{1}{\Sigma_{0}}\left[(l_{2}r_{3} + l_{3}r_{2})dt + (l_{2}r_{4} + l_{4}r_{2})dy\right]\sin^{2}\theta d\phi + \frac{1}{\Sigma_{0}}\left[(l_{2}r_{3} - l_{3}r_{2})dt + (l_{2}r_{4} - l_{4}r_{2})dy\right]\cos^{2}\theta d\psi, \qquad (2.227)$$

$$B = \frac{(l_3r_4 - l_4r_3)}{2n_5\Sigma_0} dt \wedge dy + n_5 \frac{h_{\phi}}{\Sigma_0} \cos^2 \theta \ d\phi \wedge d\psi + \frac{1}{2\Sigma_0} \left[ (l_2r_3 - l_3r_2) dt + (l_2r_4 - l_4r_2) dy \right] \wedge \sin^2 \theta d\phi - \frac{1}{2\Sigma_0} \left[ (l_2r_3 + l_3r_2) dt + (l_2r_4 + l_4r_2) dy \right] \wedge \cos^2 \theta d\psi ,$$

where

$$h_{t} = \sinh^{2} \rho + l_{2} r_{2} \cos^{2} \theta + \frac{1 - l_{2} r_{2}}{2} - \frac{l_{3} r_{3} + l_{4} r_{4}}{2 n_{5}},$$

$$h_{y} = \sinh^{2} \rho + l_{2} r_{2} \cos^{2} \theta + \frac{1 - l_{2} r_{2}}{2} + \frac{l_{3} r_{3} + l_{4} r_{4}}{2 n_{5}},$$

$$h_{\phi} = \sinh^{2} \rho + \frac{1 + l_{2} r_{2}}{2} + \frac{l_{3} r_{3} - l_{4} r_{4}}{2 n_{5}},$$

$$h_{\psi} = \sinh^{2} \rho + \frac{1 - l_{2} r_{2}}{2} + \frac{l_{3} r_{3} - l_{4} r_{4}}{2 n_{5}}.$$
(2.228)

A non-trivial dilaton  $\Phi$  is generated as usual at one-loop level on the worldsheet. In the null-gauging formalism, this arises from a change in the measure in the path integral formulation. The most direct way to compute the dilaton is by considering the usual one-loop beta function (equivalently the supergravity equations of motion). This fixes  $e^{2\Phi}$  to be proportional to

$$e^{2\Phi} \sim \frac{1}{\Sigma_0} \,. \tag{2.229}$$

The overall normalization of the dilaton can be fixed by matching to the NS5-brane decoupling limit of known solutions; we shall discuss this in detail in Section 3.4. Nevertheless, let us make some preliminary comments on this in order to highlight the physical meaning of this constant. The simplest scenario corresponds to the solution sourced by a stack of  $n_5$  coincident fivebranes, which is described by using the harmonic function

$$H_5 = 1 + \frac{n_5}{r^2}, \qquad (2.230)$$

where *r* is a radial coordinate, and where the dilaton is given by  $e^{2\Phi} = g_s^2 H_5$ . The fivebrane decoupling limit corresponds to  $g_s \to 0$  with fixed  $r/g_s$  and fixed  $\alpha'$  [131] (recall that we have set  $\alpha' = 1$ ), which can be implemented via a scaling limit  $g_s \to \epsilon$ ,  $r \to \epsilon r$ , with  $\epsilon \to 0$ . This brings the dilaton to the form

$$e^{2\Phi} = \frac{n_5}{r^2}.$$
 (2.231)

Here we could have kept a fiducial rescaled  $\tilde{g}_s$  (i.e.  $g_s = \epsilon \tilde{g}_s$ ) as in [51, 52], but since the asymptotic value of  $e^{2\Phi}$  is zero this has no precise physical meaning. Next, for an array of  $n_5$  fivebranes in a circular,  $\mathbb{Z}_{n_5}$  symmetric configuration, the supergravity solution sees a smeared source and the relevant harmonic function is based on the function  $\tilde{\Sigma} = r^2 + a^2 \cos^2 \theta$ , where the scale *a* parametrises the radius of the circular array (see e.g. [51] and references within). In this case we take a double scaling limit given by  $g_s \to 0$  with fixed  $r/g_s$ , fixed  $a/g_s$  and fixed  $\alpha'$  [120, 121], which can be implemented via a scaling limit  $g_s \to \epsilon$ ,  $r \to \epsilon r$ ,  $a \to \epsilon a$  with  $\epsilon \to 0$ . Changing variables to  $r = a \sinh \rho$  in order to match the notation used above, we have  $\tilde{\Sigma} = a^2(\sinh^2 \rho + \cos^2 \theta) \equiv a^2 \tilde{\Sigma}_0$ . The harmonic function in (2.230) is replaced by [119, 118, 51]

$$H_5 = 1 + \frac{n_5}{a^2 \tilde{\Sigma}_0}, \qquad (2.232)$$

so in the decoupling limit the dilaton takes the form

$$e^{2\Phi} = \frac{n_5}{a^2 \tilde{\Sigma}_0}.$$
 (2.233)

The backgrounds we consider will turn out to be generalisations of the circular array of fivebranes, such that, as a general expectation, the normalisation constant for the exponentiated dilaton in Eq. (2.229) should be proportional to the number of NS5 branes

in the geometry. When F1 charge is also present this gets divided by  $n_1$ , giving a factor  $n_5/n_1$  (or  $\sim n_5/n_p$  in the NS5-P frame). Furthermore, there should also be a factor in the denominator given by the square of a length scale characterising the distribution of the sources. There will turn out be two lengthscales  $a_1$ ,  $a_2$  generalizing the scale a, and the decoupling limit involves scaling  $g_s \rightarrow \epsilon$ ,  $r \rightarrow \epsilon r$ ,  $a_1 \rightarrow \epsilon a_1$ ,  $a_2 \rightarrow \epsilon a_2$  with  $\epsilon \rightarrow 0$  [52]. At this point however, we are working generally, so we do not yet know the details of the underlying bound state of branes. We postpone the precise computation until Section 3.4.

Together with the constraints (2.216) on the  $l_i$ ,  $r_i$  parameters, the expressions for the supergravity fields (2.227), (2.228), (2.229) describe the most general backgrounds that can be obtained within the class of null-gauged models considered in this chapter, under our assumption  $l_1 \neq 0$ ,  $r_1 \neq 0$ . Models in which  $l_1 = 0$  or  $r_1 = 0$  can easily be treated as a special case and we shall not consider them further.

As mentioned above, it is known that these models include the JMaRT solutions and their limits [51, 52]. In the next chapter we shall prove that these are in fact all consistent solutions in this class of null-gauged models. Moreover, we will show that this conclusion can be reached either from consistency of the worldsheet CFT or from asking that the supergravity fields (2.227), (2.228), (2.229) are free of CTCs, horizonless and smooth up to physical sources of string theory (in our cases, orbifold singularities or NS5-brane singularities).

# **Chapter 3**

# Black hole microstates from the worldsheet

# 3.1 Overview of the JMaRT cosets

Bound states of D1 and D5-branes, or of NS5 branes and fundamental strings (F1), possibly also carrying momentum P in a compact direction, have been a very fruitful arena in which to study black hole microstates in string theory. Taking the D1-D5 (or NS5-F1) decoupling limit gives rise to configurations that are asymptotically  $AdS_3 \times S^3 \times M$ , where M is  $\mathbb{T}^4$  or K3. This is one of the original examples of holographic duality [16].

Configurations that have come to be known as circular supertubes [123, 122] were important early supergravity solutions describing specific microstates of the two-charge system, in particular in the D1-D5 or NS5-F1 duality frames. These solutions were generalized by Lunin and Mathur, and others, to the full class of two-charge microstates [65, 66, 67, 72, 70].

An equally important family of three-charge (D1-D5-P or NS5-F1-P) microstate solutions are known as spectral flowed circular supertubes, of which there are both BPS and non-BPS configurations [132, 124, 125, 37, 96]. In the AdS<sub>3</sub> decoupling limit, the general holographic description of these configurations is well understood [96, 94] and involves spectral flow in the  $\mathcal{N} = (4, 4)$  superconformal algebra. Moreover, two-charge circular supertubes have proven to be important seed solutions in the construction of much more general families of "superstratum" solutions (see e.g. [133, 134, 135, 92, 136, 137, 138, 139]).

The non-BPS spectral flowed circular supertubes in the family mentioned above are known as the JMaRT solutions, after the authors of [37]. These microstates emit ergoregion radiation, which has been interpreted (via holography) as an enhanced, unitary version of Hawking radiation [140, 93, 141]. The JMaRT solutions also contain the BPS two-charge circular supertubes and BPS three-charge spectral flowed circular supertubes as (non-trivial) limits. All of these will be included in our analysis.

Recently a worldsheet description of the JMaRT three-charge NS5-F1-P configurations, in the NS5-brane decoupling limit, was constructed [51]. This regime corresponds to little string theory, an example of stringy holography which remains poorly understood. The models of [51] make use of a well-known supersymmetric WZW theory, combining it with the null-gauging formalism. More precisely, they involve an auxiliary (10+2)-dimensional group manifold, which is reduced to the physical (9+1)-dimensional target space by gauging a pair of null chiral currents. The corresponding spectrum of perturbative strings and D-branes were studied respectively in [52, 53]. The null-gauging construction was further extended to encompass more general Lunin-Mathur solutions [54], which correspond to a larger family of gauged sigma models that are generically not cosets. Other coset models that describe wrapped and/or intersecting fivebranes have also recently been studied [142].

The underlying microscopic configurations involve bound states of NS5 branes (possibly with F1 and/or P charge). Generically, the low-energy supergravity description is not reliable near the fivebrane sources, however the worldsheet theory remains under control. Indeed, these coset theories are exact in  $\alpha'$ , thus extending the description of these families of black hole microstates beyond the supergravity limit.

The models considered in [51, 52, 53] have the following basic structure. The *upstairs* theory (i.e. before gauging) involves pure NSNS fluxes and is of the form  $AdS_3 \times S^3 \times \mathbb{R} \times S^1 \times \mathbb{T}^4$ , where the  $\mathbb{R}$  factor is timelike, and where the  $S^1$  is separated from the  $\mathbb{T}^4$  because it plays a preferred role. Indeed, the gauging does not involve the  $\mathbb{T}^4$  and we shall mostly work in six physical spacetime directions *downstairs* (i.e. after gauging). The non-trivial WZW model involved is that of the universal cover of  $SL(2, \mathbb{R})$  times SU(2), which constitutes a rich and well studied example of an exactly solvable model [78, 110, 101, 102, 103]. The currents to be gauged are specific null linear combinations of the Cartan currents of  $SL(2, \mathbb{R}) \times SU(2)$  and the chiral momenta on  $\mathbb{R} \times S^1$ , plus a similar (though generically not identical) linear combination of their anti-chiral counterparts.

Within this class of models, it is natural to ask what is the general family of wellbehaved backgrounds that can be obtained by considering the most general null linear combination of the currents just described. This question was not addressed in [51, 52, 53]. Developing a systematic method for classifying such backgrounds is important for three reasons. First, it offers the possibility of finding novel configurations. Second, it can sharpen our understanding of the general backgrounds and how their parameters are constrained by different consistency conditions, shedding further light on the interplay between worldsheet CFT and spacetime geometry. Third, such techniques may then be applied to other similar classes of gauged models.

In this chapter we provide the answer to the above question, by proving that the JMaRT solutions and their limits represent the complete set of supergravity configurations described by this family of coset models. We do so from two complementary but independent points of view, and, in doing so, we clarify several aspects of both the worldsheet models and the supergravity backgrounds.

First, we consider the most general family of worldsheet coset theories, and derive necessary and sufficient conditions that lead to a consistent physical spectrum. These consistency conditions are obtained from analysing the gauge orbits, relating different representatives of the same physical operators, combined with worldsheet spectral flow (not to be confused with the spacetime/holographic spectral flow discussed above). This includes not only the spectral flow operation that is an essential part of the SL(2,  $\mathbb{R}$ ) WZW model [101], but also that of SU(2). While SU(2) spectral flow does not generate new affine representations, it has proven to be quite useful for string theory applications [118, 100, 52, 54]. As it turns out, we obtain constraints that take the form of algebraic relations for the a priori continuous gauging parameters, which imply that they can be written in terms of four integers, k, m, n, p (of which only three are independent), plus  $R_y$ , the continuous modulus corresponding to the asymptotic proper radius of the S<sup>1</sup>. Furthermore, we derive restrictions on the parities of k, m, n, p.

We then show that the same conditions can equally be derived from the analysis of the set of supergravity backgrounds obtained from the general gauged models. More precisely, we show that imposing absence of horizons, absence of closed time-like curves (CTCs), and smoothness up to orbifold singularities in the corresponding classical geometries leads to an identical quantisation of the gauging parameters.

On the other hand, the JMaRT solutions are usually written in terms of their own set of seemingly continuous parameters, which however are known to be constrained by regularity and absence of CTCs to obey their own set of algebraic relations [37]. This parametrisation is quite awkward to work with, and obscures aspects of the physics.

We find that for the (NS5-decoupled) JMaRT solutions and their limits, one can completely bypass most of the seemingly continuous parameters. Let  $n_5$ ,  $n_1$ ,  $n_p$  denote respectively the quantised numbers of NS5 branes, fundamental strings, and units of momentum along S<sup>1</sup> present in the background. We show that, in the NS5-brane decoupling limit, the six-dimensional metric and the NSNS *B*-field can be expressed explicitly in terms of the same set of integers k, m, n, p introduced in the coset models, together with  $n_5$  and  $R_y$ . Although this result is strongly inspired by our worldsheet analysis, we have derived it independently and purely within supergravity, via a non-trivial manipulation of the above-mentioned algebraic constraints. For the dilaton, an extra parameter is necessary, which can be taken to be either  $n_1/V_4$  or  $n_p/V_4$ , where  $V_4$  is the volume of the  $\mathbb{T}^4$ . In the three-charge solutions, there is a constraint that relates  $n_1$  and  $n_p$ , meaning that only one can be chosen independently. We take  $V_4$  to be microscopic and fixed, and ignore it when counting parameters, so that we consider the independent ones to be  $(m, n, n_5, R_y)$  plus one of either k or p, plus one of either  $n_1$  or  $n_p$ . Without loss of generality, one can restrict the range of the integer parameters to be non-negative.

The resulting expressions for the supergravity fields are identical to those obtained from the general coset worldsheet actions, completing the proof that these are the unique backgrounds that arise in these coset theories. Note that for the latter,  $n_5$  defines the level of the SL(2,  $\mathbb{R}$ ) and SU(2) affine algebras.

The rewriting of these configurations in terms of the integer parametrisation significantly clarifies the properties of these solutions. In particular, it makes some of their symmetries and the action of T-duality manifest. It also sheds light on the somewhat delicate limits that lead to the two-charge configurations where either  $n_1$  or  $n_p$  is set to zero. As a result, this allows us to derive a novel and non-trivial two-charge non-BPS NS5-P limit of the general solutions in a straightforward way.

Finally, we also comment on a potential relation to recent investigations of the so-called single-trace  $T\bar{T}$  deformation of the (holographic) D1-D5 CFT [143, 144, 145]. In the worldsheet model, this can be described by using a null-gauging procedure similar to the formalism employed throughout this chapter, although in that context the SL(2,  $\mathbb{R}$ ) current involved in the gauging is not in the Cartan subalgebra.

The structure of this chapter is as follows. In Section 3.2 we review in more detail the  $SL(2, \mathbb{R})$  and SU(2) WZW models, and analyse the consistency of the CFT spectrum in terms of the gauging parameters. In Section 3.3 we analyse absence of horizons, absence of CTCs, and smoothness in the corresponding supergravity backgrounds. In Section 3.4 we firstly match the resulting worldsheet models to the general JMaRT solutions. We then discuss in detail their various limits, including two-charge (non-BPS), BPS, and AdS<sub>3</sub> limits. In Section 3.5 we further discuss our results.

# **3.2** Consistency of the worldsheet spectrum

Here we discuss how the BRST charges are modified and under which conditions the resulting background is supersymmetric. Then, we derive a series of constraints leading to a consistent gauge-invariant spectrum. Spectral flow considerations play a key role in the analysis below.

# 3.2.1 Superstring Theory in null-gauged models

We now proceed to analyse the class of gauged WZW models introduced in Section 2.7. In this section we work directly at the level of the coset CFT. We will see that a number of consistency conditions can be derived, which restrict the possible values of the parameters  $l_i$ ,  $r_i$  that define the embedding of the abelian subgroups being gauged.

Recall that the transformations we gauge are chiral and correspond to  $g \rightarrow h_L g h_R^{-1}$ , where  $g \in \mathcal{G} = SL(2, \mathbb{R}) \times SU(2) \times \mathbb{R}_t \times U(1)_y$  and  $h_{L(R)} \in \mathcal{H}_{L(R)} = U(1)_{L(R)}$ . Keeping the notation general, we have seen that introducing two independent gauge fields  $\mathcal{A}, \mathcal{A}$ transforming as

$$\mathcal{A} \to h_L \mathcal{A} h_L^{-1} + \partial h_L h_L^{-1} , \qquad \bar{\mathcal{A}} \to h_R \bar{\mathcal{A}} h_R^{-1} + \bar{\partial} h_R h_R^{-1} , \qquad (3.1)$$

leads to the gauge-invariant action [146]

$$S[g, \mathcal{A}, \bar{\mathcal{A}}] = S_{WZW}(g) + \frac{k}{\pi} \int d^2 z \operatorname{Tr} \left[ \mathcal{A}g^{-1}\bar{\partial}g - \bar{\mathcal{A}}\partial gg^{-1} - g^{-1}\bar{\mathcal{A}}g\mathcal{A} \right].$$
(3.2)

By parametrising the gauge fields as

$$\mathcal{A} = \partial H_L H_L^{-1} , \ \bar{\mathcal{A}} = \bar{\partial} H_R H_R^{-1} , \ (H_L \to h_L H_L , \ H_R \to h_R H_R) , \qquad (3.3)$$

and making use of the Polyakov-Wiegmann identity

$$S_{\text{WZW}}(ab) = S_{\text{WZW}}(a) + S_{\text{WZW}}(b) + \text{sgn}(\kappa)\frac{k}{\pi}\int d^2z \,\operatorname{Tr}\left[a^{-1}\bar{\partial}a\partial bb^{-1}\right],\qquad(3.4)$$

we can rewrite the gauged action as

$$S[g, \mathcal{A}, \bar{\mathcal{A}}] = S_{WZW} \left( H_L^{-1} g H_R \right) - S_{WZW} \left( H_L^{-1} \right) - S_{WZW} \left( H_R \right).$$
(3.5)

A crucial simplification occurs when the currents being gauged are null. In this case we have  $S_{WZW} (H_L^{-1}) = S_{WZW} (H_R) = 0$ . Moreover, the Jacobian associated to the change of variables in Eq. (3.3) can be seen to trivialise for the same reason, except for the appearance of the usual  $\tilde{b}\tilde{c}$  system [146]. Finally, one can change variables from gto the gauge-invariant  $\mathcal{G}$ -valued quantity  $\tilde{g} = H_L^{-1}gH_R$ . As a result, the path integral of the gauged theory is simply interpreted as that of the original upstairs WZW model on  $\mathcal{G}$  combined with the ghost contributions. The same holds for the supersymmetric case.

The consequences of the presence of the ghosts signalling the null gaugings can be understood intuitively as follows. Upon quantisation, we find that, on top of the usual string theory contributions to the BRST charges Q and  $\bar{Q}$ , it becomes necessary to include new chiral terms of the (schematic) form

$$\oint dz : \tilde{c} J : , \quad \oint dz : \bar{\bar{c}} \bar{J} : , \qquad (3.6)$$

together with their fermionic counterparts. This ensures that, under the gauging procedure outlined above, the spectrum of the coset model is built simply out of the vertex operators of the upstairs theory that are BRST-closed. In other words, physical operators must be gauge invariant.

We can now make this construction explicit for the models under consideration. These include black hole microstate solutions with up to three charges, and we view them in the NS5-F1-P duality frame, where the relevant fields are the metric, the *B*-field and the dilaton. The corresponding geometries were described in Section 2.7. We know from [51] that these models include both BPS and non-BPS black hole microstates. Given that all the necessary ingredients belong to the NSNS sector we expect to have a well-defined solvable worldsheet model describing strings propagating in these backgrounds. Indeed, as shown in [51, 52] and reviewed in Section 2.7 above, the world-sheet theory associated to the propagation of strings in this context corresponds to a coset CFT of the form<sup>38</sup>

$$\frac{SL(2,\mathbb{R})\times SU(2)\times \mathbb{R}_t \times U(1)_y}{U(1)_L \times U(1)_R} \times U(1)^4.$$
(3.7)

Let us first characterise the upstairs twelve-dimensional model. Here we simply add the extra time direction t and spatial circle y to the matter content employed in the previous section, together with the corresponding fermionic partners  $\lambda^t$  and  $\lambda^y$ . The latter are bosonized as  $i\partial H_6 = \lambda^t \lambda^y$ , with  $H_6^{\dagger} = -H_6$ . They give additional free field contributions to the matter T and G in (2.135) and (2.136). The null current operators being gauged are then

$$J = i\mathcal{J} = J^{3} + l_{2}K^{3} + l_{3}P_{t} + l_{4}P_{y,L},$$
  

$$\bar{J} = i\bar{\mathcal{J}} = \bar{J}^{3} + r_{2}\bar{K}^{3} + r_{3}P_{t} + r_{4}P_{y,R},$$
(3.8)

where  $P_t = i\partial t$ ,  $P_{y,L} = i\partial y$ , and  $P_{y,R} = i\bar{\partial}y$ . Together with the extra coordinates we also include the additional set of ghosts mentioned above, together with their fermionic partners. Note that it is necessary to take  $h[\tilde{c}] = 0$  and  $h[\tilde{\gamma}] = 1/2$ , such that the central charges  $c_{\tilde{b}\tilde{c}} = -2$  and  $c_{\tilde{\beta}\tilde{\gamma}} = -1$  cancel the additional matter contribution  $c_{ty} = 3$ . This also implies that the bosonization

$$\tilde{\beta} = e^{-\tilde{\varphi}} \partial \tilde{\xi} , \qquad \tilde{\gamma} = \tilde{\eta} \, e^{\tilde{\varphi}} \tag{3.9}$$

<sup>&</sup>lt;sup>38</sup>As mentioned around Eq. (2.205), the coset involves the universal cover of SL(2,  $\mathbb{R}$ ), and we are actually gauging the non-compact subgroup  $\mathbb{R} \times U(1)$ .

yields a canonically normalized scalar  $\tilde{\varphi}$  with no background charge. Consequently, we can work with  $\tilde{\varphi}$ -independent vertex operators in the NSNS sector. On the other hand, the definition of the spin fields and the would-be spacetime supercharges is modified to [54]

$$Q_{\varepsilon} = \oint dz \, e^{-(\varphi - \tilde{\varphi})/2} S_{\varepsilon} \,, \qquad S_{\varepsilon} = \exp\left(\frac{i}{2} \sum_{I=1}^{6} \varepsilon_{I} H_{I}\right), \tag{3.10}$$

where the contributions to the conformal dimension of the integrand in  $Q_{\varepsilon}$  from the  $\tilde{\varphi}$  and  $H_6$  exponentials cancel exactly. Note that the mutual locality condition now reads  $\prod_{I=1}^{6} \varepsilon_I = 1$ . An analogous formula defines the anti-holomorphic counterpart  $\bar{Q}_{\varepsilon}$ .

As stressed above, the present procedure can lead to both BPS and non-BPS backgrounds. This depends on whether the charges  $Q_{\varepsilon}$  turn out to be BRST invariant or not, according to the precise current we choose to gauge. The left-handed BRST charge takes the form

$$Q = \oint dz : \left[ c \left( T + T_{\beta \gamma \tilde{\beta} \tilde{\gamma}} \right) + \gamma G + \tilde{c} J + \tilde{\gamma} \lambda + \text{ghosts} \right] :, \qquad (3.11)$$

and similarly for the right-handed one. Here  $\lambda$  and  $\overline{\lambda}$  are the superpartners of the currents in Eq. (3.8), that is

$$\lambda = \psi^{3} + l_{2}\chi^{3} + l_{3}\lambda^{t} + l_{4}\lambda^{y} , \quad \bar{\lambda} = \bar{\psi}^{3} + r_{2}\bar{\chi}^{3} + r_{3}\bar{\lambda}^{t} + r_{4}\bar{\lambda}^{y}.$$
(3.12)

This ensures that only operators satisfying the usual Virasoro and  $\gamma G$ -invariance conditions that are also uncharged under the bosonic currents  $J, \bar{J}$  and annihilated by  $\tilde{\gamma}\lambda$ and  $\bar{\tilde{\gamma}}\lambda$  are physical. In particular, by using the  $\gamma G$ -invariance condition  $\varepsilon_1\varepsilon_2\varepsilon_3 = -1$ , the  $Q_{\varepsilon}$  supercharges survive the gauging if and only if [54]

$$\varepsilon_1 + \varepsilon_2 l_2 = 0$$
,  $l_4 + \varepsilon_6 l_3 = 0$ , (3.13)

where the former constraint comes from the bosonic current and the latter arises from the fermionic one. Actually, only one of these two restrictions is independent due to the null condition (2.216). Analogously, for the antiholomorphic supercharges one has

$$\varepsilon_1 + \varepsilon_2 r_2 = 0$$
,  $r_4 + \varepsilon_6 r_3 = 0$ , (3.14)

For instance, the cases with (4, 4) and (4, 0) spacetime supersymmetry were considered recently in [54, App. B]. In the present work we focus mainly on the more general non-supersymmetric case, in which there is no solution to this set of constraints since  $l_2 \neq \pm 1$  and  $r_2 \neq \pm 1$ .

We pause here to stress that we start from a (10+2)-dimensional model where all operators are taken to be mutually local. This includes the charges  $Q_{\varepsilon}$ ,  $\bar{Q}_{\varepsilon}$ . In particular, this amounts to imposing the analogue of the GSO projection in the upstairs theory. Consequently, there are no tachyons in the resulting spectrum, nor in that of the coset model [52, 54]. Indeed, even if no supersymmetry is preserved, the supergravity solutions we are dealing with are expected to be classically stable. Note that this is consistent with the fact that we only recover the fivebrane decoupling limit of JMaRT geometries, where there is no ergoregion [52]. The full solutions with flat asymptotics do exhibit the well-known ergoregion instability [147, 94], interpreted as an enhanced version of Hawking radiation [140, 93, 141].

# 3.2.2 Physical spectrum and consistency conditions

We now consider the physical states in the null-gauged theory. These are given in terms of vertex operators of the upstairs model that are BRST invariant as defined by the charge (3.11) and its anti-holomorphic counterpart.

The lightest physical states (with no winding) are given by unflowed operators with a single fermionic excitation. All such operators must satisfy the Virasoro condition

$$0 = -\frac{j(j-1)}{n_5} + \frac{j'(j'+1)}{n_5} - \frac{1}{4}E^2 + \frac{1}{4}P_y^2.$$
(3.15)

These are automatically invariant under the action of  $\lambda$  in (3.12), so that they are BRST-invariant iff the following null-gauging<sup>39</sup> constraints hold:

$$0 = m + l_2 m' + \frac{l_3}{2}E + \frac{l_4}{2}P_y, \qquad 0 = \bar{m} + r_2 \bar{m}' + \frac{r_3}{2}E + \frac{r_4}{2}P_y.$$
(3.16)

The simplest states of this sort are the 6D scalars, whose spectrum was shown to coincide with that of minimally coupled massless scalar perturbations on top of the JMaRT geometries in [52]. There are also the extremal-weight states, which for  $E = P_y = 0$  are half-BPS and were studied in [54] in the context of supertubes. Note that operators with  $\mathbb{T}^4$  polarisations can also give BPS states in specific models with non-trivial twisted sectors [112].

On the other hand, operators with more general projections need to be combined with polarisations on the extra directions t and y in order to achieve BRST-invariance. The corresponding coefficients are determined by invariance under  $\lambda$  together with transversality in the t, y directions. The anti-holomorphic variables must satisfy analogous conditions as well, together with the gauge invariance conditions (4.12). Consequently, these constraints restrict the possible polarisations to those expected in the (9+1)- dimensional setting.

Let us now consider spectrally flowed states. A particularly simple case corresponds to the circular array of fivebranes on the Coulomb branch, which is obtained by choosing

<sup>&</sup>lt;sup>39</sup>The factors of 2 in the free field terms of Eqs. (3.15) and (3.16) arise from the OPEs  $\partial t(z)\partial t(0) \sim -\frac{1}{2}\frac{1}{z^2}$ ,  $\partial y(z)\partial y(0) \sim \frac{1}{2}\frac{1}{z^2}$  (recall that we work with  $\alpha' = 1$ ).

 $l_1 = l_2 = r_1 = -r_2 = 1$  and  $l_3 = l_4 = r_3 = r_4 = 0$  [118, 51]. From the worldsheet point of view, this null-gauged model is analogous to the cigar construction used in [120, 121] provided we replace the extra circle by an S<sup>3</sup> and take  $J_y \rightarrow K^3$ . Thus, as in that example, introducing spectral flow with  $\omega = \omega' = \bar{\omega}'$  does not produce new physical states since it merely amounts to a large gauge transformation. In a more general context, however, this is not true anymore, and spectrally flowed sectors contribute non-trivially to the spectrum. At this point, we also allow for both momentum  $n_y$  and winding  $\omega_y$  on S<sup>1</sup><sub>u</sub>, and use the shorthands

$$P_{y,L/R} = \left(\frac{n_y}{R_y} \pm \omega_y R_y\right) , \ n_y, \omega_y \in \mathbb{Z}.$$
(3.17)

For generic states with spectral flow charges  $\omega$  on SL(2,  $\mathbb{R}$ ),  $(\omega', \bar{\omega}')$  on SU(2), and winding  $\omega_y$  on  $\mathbb{S}^1_y$ , the null-gauge constraints (4.12) read

$$0 = m + \frac{n_5}{2}\omega + l_2\left(m' + \frac{n_5}{2}\omega'\right) + \frac{l_3}{2}E + \frac{l_4}{2}P_{y,L}, \qquad (3.18a)$$

$$0 = \bar{m} + \frac{n_5}{2}\omega + r_2\left(\bar{m}' + \frac{n_5}{2}\bar{\omega}'\right) + \frac{r_3}{2}E + \frac{r_4}{2}P_{y,R}, \qquad (3.18b)$$

while the Virasoro constraints take the form

$$\frac{1}{2} = \frac{j'(j'+1) - j(j-1)}{n_5} - m\omega + m'\omega' + \frac{n_5}{4}\left(\omega'^2 - \omega^2\right) - \frac{1}{4}\left(E^2 - P_{y,L}^2\right) + N,$$
(3.19a)

$$\frac{1}{2} = \frac{j'(j'+1) - j(j-1)}{n_5} - \bar{m}\omega + \bar{m}'\bar{\omega}' + \frac{n_5}{4}\left(\bar{\omega}'^2 - \omega^2\right) - \frac{1}{4}\left(E^2 - P_{y,R}^2\right) + \bar{N}.$$
 (3.19b)

Here *N* and  $\overline{N}$  are the excitation numbers, and we have restricted to unflowed states with no fermion excitations for simplicity.

The discussion so far does not characterise the physical spectrum in a unique way: there is a residual discrete gauge orbit connecting equivalent representatives of the same physical state. This fact was noticed in [52], so let us first recall the observations made in that work before making a set of generalisations. First, spectral flow in the null direction corresponding to the gauge current is gauge-trivial. Second, the non-compactness of  $\mathbb{R}_t$  means that there cannot be independent left and right gauge spectral flow transformations, since these would shift the zero mode of *t* differently. Therefore, globally we work with the universal cover of SL(2,  $\mathbb{R}$ ), in which the left and right spectral flow parameters are constrained to be equal,  $\omega = \bar{\omega}$ . Moreover, globally we gauge  $\mathbb{R} \times U(1)$ , a (1+1)-dimensional cylinder composed of one compact spacelike direction and one non-compact timelike direction. The gauged model then has a single non-compact timelike direction.<sup>40</sup> Third, the non-compactness of the time coordinate *t* 

<sup>&</sup>lt;sup>40</sup>By contrast, in related models that do not include the  $\mathbb{R}_t$  factor in the upstairs model, the single cover of SL(2,  $\mathbb{R}$ ) has been considered [118].

moreover imposes

$$l_3 = r_3$$
, (3.20)

or, in terms of the original gauging parameters,  $l_3/l_1 = r_3/r_1$ . We will re-derive the condition  $l_3 = r_3$  from an independent point of view in the following section by imposing smoothness, absence of horizons, and absence of CTCs in the corresponding geometry.

We now make a more general analysis of this phenomenon in the general models defined in the previous section. Let us stress that the analysis of such gauge orbits is not simply about the counting of states. Indeed, being able to identify gauge-equivalent operators in terms of the quantum numbers of the WZW model is necessary for building a consistent theory, and we will show that it further constrains the allowed values for the gauging parameters  $l_i$ ,  $r_i$ .

Given a physical state, let us seek spectral flow transformations that result in the same operator. By subtracting the two equations in (3.19) we find

$$0 = \omega(\bar{m} - m) + m'\omega' - \bar{m}'\bar{\omega}' + \frac{n_5}{4}(\omega'^2 - \bar{\omega}'^2) + n_y\omega_y + N - \bar{N}, \qquad (3.21)$$

which plays the role of the level-matching condition in this context. In order to find solutions of Eq. (3.21),  $(\omega'^2 - \bar{\omega}'^2)$  must be a multiple of 4, so  $\omega' \pm \bar{\omega}'$  must be even. Note that this preserves the statistics of the SU(2) part of the state.

Let us consider a shift of the form<sup>41</sup>

$$\omega \to \omega + q, \qquad q \in \mathbb{Z}.$$
 (3.22)

We shall show that this can be compensated at the level of the null-gauge constraints (3.18) without altering the weights (3.19) by shifting the remaining quantum numbers appropriately. We begin with a general shift and show that only the shift in the null gauge direction achieves this. We allow for arbitrary multiples of *q* to shift  $\omega'$ ,  $\bar{\omega}'$ , *E*,  $n_y$  and  $\omega_y$  as well, namely

$$(\omega',\bar{\omega}',E,P_{y,L},P_{y,R}) \to (\omega'-a_2q,\bar{\omega}'-b_2q,E+a_3q,P_{y,L}-a_4q,P_{y,R}-b_4q).$$
(3.23)

<sup>&</sup>lt;sup>41</sup>We include both the bosonic and the free fermion spectral flows, so that the shifted charges and weights are written in terms of the supersymmetric level  $n_5$  as opposed to the bosonic levels  $k = n_5 + 2$  and  $k' = n_5 - 2$ .
For the weights (3.19) and gauge constraint (3.18) to remain unchanged for arbitrary q, we must have

$$0 = m + \frac{n_5}{2}\omega + a_2\left(m' + \frac{n_5}{2}\omega'\right) + \frac{a_3}{2}E + \frac{a_4}{2}P_{y,L},$$
  

$$0 = n_5(1 - a_2^2) + a_3^2 - a_4^2,$$
  

$$0 = n_5(1 - l_2a_2) + l_3a_3 - l_4a_4,$$
  
(3.24)

and the same with  $a_{2,4} \rightarrow b_{2,4}$ ,  $l_{2,4} \rightarrow r_{2,4}$ ,  $m' \rightarrow \bar{m}'$ ,  $\omega' \rightarrow \bar{\omega}'$  and  $P_{y,L} \rightarrow P_{y,R}$ . To satisfy the first of these three conditions for general states without over-restricting the spectrum, we must set  $a_i = l_i$  and  $b_i = r_i$ . Indeed, in this case the first condition becomes (3.18), while the last two conditions both reduce to (2.218). The compensating shifts then take the form

$$\omega' \to \omega' - l_2 q$$
,  $\bar{\omega}' \to \bar{\omega}' - r_2 q$ , (3.25)

for the left and right SU(2) spectral flow charges, respectively,

$$E \to E + l_3 q = E + r_3 q,$$
 (3.26)

for the energy, and

$$n_y \to n_y - \frac{R_y}{2} (l_4 + r_4) q$$
,  $\omega_y \to \omega_y - \frac{1}{2R_y} (l_4 - r_4) q$ , (3.27)

for the  $S_{y}^{1}$  quantum numbers.

For Eqs. (3.22) and (3.25)–(3.27) to make sense in terms of integer spectral flows and momentum/winding numbers, the gauging parameters must be quantised in a specific way. On the one hand, taking into account that, as argued above,  $\omega' \pm \bar{\omega}'$  must be even, for the SU(2) sector we find  $l_2 \pm r_2 \in 2\mathbb{Z}$ . We can thus write as a first pass (to be refined momentarily)

$$l_2 = m + n$$
,  $r_2 = -(m - n)$ ,  $m, n \in \mathbb{Z}$ , (3.28)

where the signs are chosen for later convenience. Furthermore, recall that in the SL(2,  $\mathbb{R}$ ) and SU(2) sectors the spectral flow operations do not act solely on the bosonic subalgebras. Indeed, they also shift the fermionic modes as in Eqs. (2.133). At the level of the vertex operators, this is accounted for by including the  $H_{1,2}$  exponentials introduced in (2.156), which were taken into account for computing the weights (3.19). However, if as in the computation above we start from an unflowed state with no fermionic excitations, and use the shifts (3.22) and (3.25) with, say, q = 1 (or any other odd value), the presence of these exponentials also indicates that the fermion numbers on the left- and right-handed components will *not* be preserved for arbitrary values of m  $\pm$  n. Thus, we see that it is necessary to make Eq. (3.28) more precise by restricting to

$$l_2 = m + n \in 2\mathbb{Z} + 1$$
,  $r_2 = -(m - n) \in 2\mathbb{Z} + 1$ ,  $m, n \in \mathbb{Z}$ . (3.29)

On the other hand, from (3.27) we also must have

$$\frac{1}{2R_y}(r_4 - l_4) = \mathsf{k} \in \mathbb{Z}, \qquad \frac{R_y}{2}(l_4 + r_4) = \mathsf{p} \in \mathbb{Z}, \qquad (3.30)$$

or equivalently

$$l_4 = -\left(kR_y - \frac{p}{R_y}\right), \qquad r_4 = kR_y + \frac{p}{R_y}.$$
 (3.31)

By plugging the expressions (3.29) and (3.31) into the null constraints (2.216), we now solve for  $l_3$ ,  $r_3$  and p. Firstly, we obtain

$$l_3 = r_3 = -\sqrt{k^2 R_y^2 + \frac{p^2}{R_y^2} + n_5 (m^2 + n^2 - 1)},$$
 (3.32)

where we have chosen the negative square root for  $l_3$ ,  $r_3$ , so that we gauge away the difference of the upstairs time directions. This fixes the energy shift (3.26) in terms of k, m, n, p. More interestingly, we also get

$$k p = n_5 m n, \qquad (3.33)$$

which shows that only three of the integers k, m, n, p are actually independent. Moreover, either k or p (or both) must be even.

We will show below that the integers m and n introduced above control the angular momenta of the classical configuration along the S<sup>3</sup> coordinates  $\phi$  and  $\psi$ , respectively. We will further argue in the following that the absolute value of the integer k is to be interpreted as an orbifold parameter. The meaning of the remaining integer p is slightly complicated to interpret in classical terms. This is due to its stringy nature, and it can be understood either holographically or in terms of T-duality, as follows.

For  $k \neq 0$ , from (3.33) we find that p is  $n_5$  times the momentum per strand mn/k in the holographic description of JMaRT states, as noted in [52]. This must be an integer since the holographic CFT is a symmetric product orbifold theory (see the discussions in [96, 94]). On the other hand, it is well known that the worldsheet theory is invariant under T-duality along a circular direction. In the language of gauged WZW models, T-dual models arise due to the equivalence of vector and axial gaugings. In the present context, T-duality along *y* amounts to  $l_4 \rightarrow -l_4$ . Together with the usual radius redefinition  $R_y \rightarrow 1/R_y$ , this exchanges the role of the integers k and p. Thus, depending on the choice of duality frame, either k or p are interpreted as an orbifold parameter, while the remaining integer is fixed in terms of  $n_5$ , m and n, and it controls the momentum charge.

While on the subject of the T-duality, let us also observe that (3.31) has special features at the self-dual radius, which in  $\alpha' = 1$  units is at  $R_y = 1$ . At the self-dual radius, in string theory one observes an enhancement of the abelian U(1) symmetry to a nonabelian SU(2) symmetry due to the presence of new massless fields. In our case, when  $R_y = 1$  the expressions of the gauging parameters  $l_4$ ,  $r_4$  corresponding to the  $U(1)_y$ component resemble those of their SU(2) counterparts  $l_2$ ,  $r_2$ . In addition, we see that it becomes possible to set either  $l_4$  or  $r_4$  to zero while keeping the other one non-trivial, which is not possible for generic values of  $R_y$ . Since the self-dual radius is associated with the appearance of new massless states (in the upstairs theory), this might lead to new solutions. We leave a more detailed exploration of such configurations for future work.

Note that the values of the gauging parameters (3.29)–(3.32) imply that

$$\Sigma_{0} \left( \rho = 0, \theta = 0 \right) = \frac{1 - l_{2}r_{2}}{2} + \frac{l_{3}r_{3} - l_{4}r_{4}}{n_{5}} = m^{2} + \frac{k^{2}R_{y}^{2}}{n_{5}},$$

$$\Sigma_{0} \left( \rho = 0, \theta = \frac{\pi}{2} \right) = \frac{1 + l_{2}r_{2}}{2} + \frac{l_{3}r_{3} - l_{4}r_{4}}{n_{5}} = n^{2} + \frac{k^{2}R_{y}^{2}}{n_{5}}.$$
(3.34)

The combinations in (3.34) correspond to the minimal values of the quantity  $\Sigma_0$  appearing in the denominator of various components of the supergravity fields (2.227). We have just shown that the consistency conditions of the spectrum imply that they are both non-negative quantities. This will be important when studying the corresponding geometry in the following section.

Let us summarise our results so far. Starting with a class of generic null gauged models, the gauged currents are defined in terms of eight parameters, namely  $l_i$ ,  $r_i$ , i = 1, 2, 3, 4. Since only the direction of the gauging matters, the overall scale becomes irrelevant. This means that we can work directly with the six ratios  $l_i$ ,  $r_i$ . These must satisfy the two null conditions (2.218) and also  $l_3 = r_3$  from the non-compactness of t. Finally, by focusing on the worldsheet CFT and relating the action of spectral flow in SL(2,  $\mathbb{R}$ ) to that of the gauge orbits, we have shown that the theory is consistent only if the remaining three parameters can be written in terms of three integers (in addition to  $R_y$ ), which we can take to be m, n and k (when  $k \neq 0$ ). Moreover, these must be chosen so that  $p = n_5 mn/k$  is also integer-valued. It is possible that the quantization conditions on k, m, n, p could alternatively be obtained by analyzing the global consistency of the gauging, see e.g. [148, 149, 150, 151, 152]. In the next section we will instead proceed to analyze the global geometry of the gauged target space.

The set of conditions (3.29)–(3.32) is one of the main results of this chapter. It will allow us to rewrite the general supergravity fields in Eqs. (2.227)–(2.229) in a simple way, making their main physical features and some of their symmetries manifest. Furthermore, this will lead to a complete characterisation of the full set of consistent solutions.

## 3.3 Analysis of the supergravity backgrounds

In the previous section we have shown that the gauging parameters  $l_i$ ,  $r_i$ , i = 1, 2, 3 can be defined in terms of k, m, n, p, all of which are integers that have a clear physics meaning. We now perform an independent supergravity analysis of the metrics introduced in Eq. (2.227), and show that imposing smoothness and absence of closed timelike curves provides an alternative derivation and complementary interpretation of the constraints (3.29)–(3.32).

#### 3.3.1 Eliminating potential closed timelike curves

To investigate potential closed timelike curves we complete the squares successively in the periodic variables  $\psi$ ,  $\phi$  and y, to rewrite the line element (2.227). We obtain

$$ds^{2} = -T(\rho)dt^{2} + Y(\rho) \left[ dy + A_{y}(\rho; dt) \right]^{2} + n_{5}(d\rho^{2} + d\theta^{2}) + n_{5}\sin^{2}\theta \frac{h_{\phi}}{\Sigma_{0}} \left[ d\phi + A_{\phi}(\rho; dt, dy) \right]^{2} + n_{5}\cos^{2}\theta \frac{h_{\psi}}{\Sigma_{0}} \left[ d\psi + A_{\psi}(\rho; dt, dy) \right]^{2}, \quad (3.35)$$

where the functions  $\Sigma_0$ ,  $h_{\phi}$  and  $h_{\psi}$  were defined in (2.224) and (2.228) respectively,  $A_y$ ,  $A_{\psi}$ , and  $A_{\phi}$  are one-forms depending only on the radial variable  $\rho$  and with legs in the appropriate arguments, while  $T(\rho)$  will turn out to be a non-negative function whose explicit expression we will not need. In order to ensure the absence of CTCs we must require the functions multiplying the squares in the periodic variables to be non-negative, i.e.

$$\frac{h_{\phi}}{\Sigma_0} \ge 0$$
,  $\frac{h_{\psi}}{\Sigma_0} \ge 0$ ,  $Y(\rho) \ge 0.$  (3.36)

We now show that asking for the inequalities (3.36) to hold everywhere in geometry is equivalent to imposing

$$l_3 = r_3.$$
 (3.37)

Let us first see that (3.36) implies (3.37). By combining the first two inequalities we find that the product  $h_{\phi}h_{\psi}$  is non-negative. On the other hand, the explicit expression of  $Y(\rho; l_i, r_i)$  reads

$$Y(\rho) = \frac{4\sinh^2\rho \left(n_5\cosh^2\rho + l_3r_3\right) + (n_5 + l_3r_3)^2 - (n_5l_2^2 + l_4^2)(n_5r_2^2 + r_4^2)}{4n_5^2h_{\phi}h_{\psi}}$$
$$= \frac{4n_5\sinh^2\rho \left(n_5\cosh^2\rho + l_3r_3\right) - (l_3 - r_3)^2}{4n_5h_{\phi}h_{\psi}}, \qquad (3.38)$$

where in the second line we have used (2.218). It follows from this last expression that the third inequality in (3.36) can only be satisfied at the origin  $\rho = 0$  if  $l_3 = r_3$ .

It remains to be seen that the implication holds in the other direction as well. For this, we note that the minimal value of  $\Sigma_0$  is given by

$$\Sigma_0^{\min} = \frac{1}{2n_5} \left[ n_5(1 - |l_2 r_2|) + l_3 r_3 - l_4 r_4 \right].$$
(3.39)

Using  $l_3 = r_3$  we can rewrite the null conditions (2.218) as

$$n_5 \left( l_2^2 - r_2^2 \right) + r_4^2 - l_4^2 = 0, \qquad l_3^2 + r_3^2 = 2l_3^2 = l_4^2 + r_4^2 + n_5 \left( l_2^2 + r_2^2 - 2 \right), \quad (3.40)$$

so that

$$2[n_5(1\pm l_2r_2)+l_3r_3-l_4r_4] = n_5(l_2\pm r_2)^2+(l_4-r_4)^2.$$
(3.41)

It follows that  $\Sigma_0 \ge 0$  everywhere. Given that  $h_{\phi} = \Sigma_0(\rho, \theta = 0)$  and  $h_{\psi} = \Sigma_0(\rho, \theta = \pi/2)$ , the same holds for these functions and the first two inequalities in (3.36) are thus satisfied. The third one also holds, as can be checked from (3.38).

This proves that in the asymptotically linear dilaton geometry the necessary and sufficient condition for avoiding CTCs is precisely  $l_3 = r_3$ , Eq. (3.37). This constraint was also obtained in the worldsheet analysis of Section 3.2 from the non-compactness of the *t* direction, Eq. (3.20). Moreover, once this is imposed we find that, as advertised above,  $T(\rho)$  is non-negative.

#### 3.3.2 Absence of horizons

Here and in the following subsection we perform an analysis which closely follows that of [37]. The determinant of the metric (2.227) reads

$$\det g = -\left(\frac{n_5^2 \sin(2\theta) \sinh(2\rho)}{4\Sigma_0}\right)^2,\tag{3.42}$$

where we have used the null-gauge constraints (2.218). Besides the usual zeros at the poles of the S<sup>3</sup>, this only vanishes at  $\rho = 0$ . Given that the determinant of the induced metric on surfaces of constant  $\rho$  is simply (3.42) divided by  $n_5$ , we see that  $\rho = 0$  corresponds to either a horizon or an origin of higher codimension. In order to distinguish between these two cases, we further compute the determinant of the induced metric on surfaces of constant  $\rho$  and t, and we evaluate it at  $\rho = 0$ , giving

$$\lim_{\rho \to 0} \det g \big|_{(y,\theta,\phi,\psi)} = -\left(\frac{n_5(l_3 - r_3)\sin(2\theta)}{4\Sigma_0(0,\theta)}\right)^2.$$
(3.43)

Hence, in order to obtain a horizonless and possibly smooth geometry we again need to impose  $l_3 = r_3$ , Eq. (3.37), such that (3.43) vanishes. Smoothness will then be achieved if some circle direction shrinks appropriately when  $\rho \rightarrow 0$ , as we discuss next.

#### 3.3.3 Smoothness and quantisation

When the background contains F1 charge, the supergravity solutions must be smooth up to possible orbifold singularities. In the absence of F1 charge, NS5-brane singularities will be present. We begin by treating the more general case in which F1 charge is present, and treat the latter as a special case.

We therefore focus on the periodic directions and consider a generic Killing vector of the form

$$\xi = \partial_y + \alpha \, \partial_\psi - \beta \, \partial_\phi \,, \qquad \alpha, \beta \in \mathbb{R}. \tag{3.44}$$

where the signs have been chosen for later convenience. To find smooth solutions we seek pairs of coefficients  $(\alpha, \beta)$  such that the norm of  $\xi$  vanishes  $\forall \theta \in [0, \frac{\pi}{2}]$  when we approach  $\rho = 0$ . From the metric (2.227) we find that this is indeed the case when

$$\alpha = \frac{l_2 r_4 - l_4 r_2}{2 n_5 \Sigma_0(0, \frac{\pi}{2})}, \qquad \beta = \frac{l_2 r_4 + l_4 r_2}{2 n_5 \Sigma_0(0, 0)}, \qquad (3.45)$$

since for these values, and upon using the null constraint (4.12), we obtain

$$\lim_{\rho \to 0} g_{ij} \, \xi^i \, \xi^j = -\frac{(l_3 - r_3)^2}{4n_5 \Sigma_0(0, 0) \, \Sigma_0(0, \frac{\pi}{2})} = 0 \,, \tag{3.46}$$

where  $i, j = y, \psi, \phi$ . In the last step we have used the no-CTC condition (3.37). Then, we define the following shifted coordinates

$$\hat{\psi} = \psi + \alpha y$$
,  $\hat{\phi} = \phi - \beta y$ , (3.47)

where the signs are chosen for later convenience. By examining the integral curves of  $\xi$ , we see that the direction that shrinks at  $\rho = 0$  is y at fixed  $\hat{\psi}$ ,  $\hat{\phi}$ . We find that near  $\rho = 0$  the line element at fixed  $(t, \theta, \hat{\psi}, \hat{\phi})$  is of the form

$$ds_{\rho\to 0}^2 \simeq n_5 \left[ d\rho^2 + \rho^2 d \left( \frac{y}{R} \right)^2 \right], \qquad R^2 = \left[ \frac{2n_5^2 \Sigma_0(0,0) \Sigma_0(0,\frac{\pi}{2})}{\left(n_5 + l_3^2\right) \left(l_4 - r_4\right)} \right]^2, \tag{3.48}$$

where we have used (2.216) and (3.37), and assumed  $l_4 \neq r_4$  (for now). Strictly speaking, a smooth geometry will be obtained only if the radius *R* coincides with  $R_y$ . Given that string theory is well-defined on orbifold backgrounds, as usual we allow for possible  $\mathbb{Z}_k$  orbifold singularities, k being the corresponding orbifold parameter. Thus, we relax this condition and impose  $R^2 = k^2 R_y^2$  for some positive integer k instead, i.e.

$$k^{2}R_{y}^{2} = \left[\frac{2n_{5}^{2}\Sigma_{0}(0,0)\Sigma_{0}(0,\frac{\pi}{2})}{\left(n_{5}+l_{3}^{2}\right)\left(l_{4}-r_{4}\right)}\right]^{2}.$$
(3.49)

Making use of the intuition developed in Section 3.2, we further rewrite the values of the parameters  $l_i$ ,  $r_i$  in terms of new quantities m, n, k and p as in Eqs. (3.28) and (3.30), namely

$$l_2 = m + n$$
,  $r_2 = -(m - n)$ ,  $\frac{1}{2}(l_4 - r_4) = -kR_y$ ,  $\frac{1}{2}(l_4 + r_4) = \frac{p}{R_y} = \frac{n_5}{R_y}\frac{mn}{k}$ . (3.50)

In (3.50) we could a priori have written k' instead of k. However, if we were to then substitute (3.50) into (3.49), we would find that  $k' = \pm k$ . In other words, within this parametrisation Eq. (3.49) is trivially satisfied.

Up to this point, the reparametrisation (3.50) does not assume that m and n are integers. However, the periodicities of the new angular variables  $\hat{\phi}$  and  $\hat{\psi}$  should be consistent with that of *y*. The corresponding quantisation conditions read

$$\alpha (\mathsf{k} R_{\mathsf{y}}) = \mathsf{m} \in \mathbb{Z} , \qquad \beta (\mathsf{k} R_{\mathsf{y}}) = \mathsf{n} \in \mathbb{Z} . \tag{3.51}$$

From the classical point of view, the values of the integers m and n seem otherwise unrestricted. However, we know from the discussion around Eq. (3.29) that one of m, n must be even and the other one must be odd. A geometric argument leading to this restriction was put forward in [37] in the JMaRT context. This is based on discussing the periodicity of the fermions along the  $S_y^1$  circle and the associated spin structure of the target space. Although it should be possible, we will not attempt to extend these arguments to the case with general gauging parameters. This is because in the next section we will directly match the JMaRT solutions to the supergravity fields of our coset models.

Moreover, out of the four parameters m, n, k and  $p = n_5 mn/k$ , a priori only the first three appear to be required to be integers from the above smoothness analysis, and nothing seems to prevent p from being a rational (not necessarily integer) number. The stringy nature of this parameter manifests itself in the fact that T-duality along  $S_y^1$ , namely  $l_4 \rightarrow -l_4$  and  $R_y \rightarrow 1/R_y$ , maps k  $\leftrightarrow$  p. Since in the T-dual geometries p is an orbifold parameter, it must also be quantised. Moreover, given that m and n may vanish or be non-vanishing integers with arbitrary signs, so can p, and consequently, the same applies to k. In due course we will restrict to non-negative values of k, m, n, p, without loss of generality.

Recall that in the previous passage we assumed  $l_4 \neq r_4$ . The case  $l_4 - r_4 = 0$ , which corresponds to k = 0 in the parametrisation (3.50), needs to be treated separately. When k = 0,  $\Sigma_0$  goes to zero at  $\rho = 0$  and either  $\theta = 0$  or  $\theta = \pi/2$ , see e.g. Eq. (3.34). The metric is singular as  $\Sigma_0 \rightarrow 0$ . We choose conventions in which the zero is at  $\theta = \pi/2$ . This corresponds to the location of the (smeared) NS5 brane source. As we shall see in the next section, this is because the F1 charge vanishes and the solutions are two-charge NS5-P (see Eqs. (3.69) and (3.76)–(3.77) below). Let us analyse the geometry away from

the source. The region of interest is the neighbourhood of  $\rho = 0$  for  $\theta \neq \pi/2$ . Note that, assuming  $l_3 = r_3$ , the null conditions (2.218) imply  $l_2 = \pm r_2$ , so that either m = 0 or n = 0. To have the source at  $\theta \neq \pi/2$ , we take  $l_2 = -r_2$ , i.e. n = 0. Then the norm of the Killing vector (3.44) is always non-vanishing. However, the  $\psi$  circle shrinks as  $\rho \rightarrow 0$ . Indeed, in this neighbourhood the line element at fixed  $(t, y, \theta, \phi)$  reads

$$ds_{\rho\to 0}^2 \simeq n_5 \left[ d\rho^2 + \rho^2 d \left(\frac{\psi}{\mathsf{m}}\right)^2 \right] , \qquad (3.52)$$

where we have used the parametrisation (3.50) without imposing the last equality, so that p is unconstrained. We conclude that for generic values of  $\theta$  and  $m = \pm 1$  the geometry is smooth. As before, we allow for orbifold singularities, so that m is again quantised: any other non-zero  $m \in \mathbb{Z}$  leads to a  $\mathbb{Z}_{|m|}$  orbifold structure. Had we chosen  $l_2 = r_2$  instead, the source would have been at  $\theta = 0$ , the  $\phi$  circle would have been the one shrinking at small  $\rho$ , and the parameter m would have been replaced by n.

#### 3.3.4 Killing Spinors

Finally, we discuss the relation that the embedding coefficient must satisfy in order to preserve a certain amount of supersymmetry. For simplicity, we work directly in the  $AdS_3 \times S^3$  limit, which will be shown to exist in all consistent cases. In order to achieve this, we first consider the Killing spinor in global AdS<sub>3</sub>, as given in [37]:

$$\epsilon_L^{\pm} = e^{\pm \frac{i}{2}\tilde{\phi}_L} e^{-\frac{i}{2}y} \epsilon_0, \qquad \epsilon_R^{\pm} = e^{\pm \frac{i}{2}\tilde{\phi}_R} e^{-\frac{i}{2}y} \epsilon_0, \qquad (3.53)$$

where  $\epsilon_0$  is a constant AdS<sub>3</sub> spinor. The dependence on the spacetime coordinates was derived in [153] and, in particular, the *y* dependence is such that the Killing spinors are regular near the origin (see also [123] and [154, App. D,E]). By a large gauge transformation one can induce the S<sup>3</sup> angular momenta starting from global AdS<sub>3</sub> × S<sup>3</sup> which, in terms of the  $\tilde{\phi}_{L,R}$  coordinates, translates into the following diffeomorphism

$$\tilde{\phi}_L = \phi + (m+n) y = \phi + l_2 y$$
,  $\tilde{\phi}_R = \phi + (m-n) y = \phi - r_2 y$ . (3.54)

This is known as spacetime spectral flow (see e.g. [96, 94]). Focusing on solutions that admit an asymptotically flat completion, the Killing spinor equations demand that the spinors be independent of y after performing the above large gauge transformation. This is obtained by imposing

$$|l_2| = 1$$
 or  $|r_2| = 1.$  (3.55)

By virtue of the null condition (2.216), the above constraint implies an analogous one for the remaining embedding coefficients, namely

$$|l_3| = |l_4|$$
 or  $|r_3| = |r_4|$ . (3.56)

We thus conclude that the above constraints must be satisfied in order to have nontrivial Killing spinors in spacetime. This is consistent with the discussion around Eqs. (3.13) and (3.14) above.

Summarising, we have imposed absence of CTCs, absence of horizons, and smoothness up to physical sources (corresponding to orbifold singularities or NS5 branes) of the general background (2.227). These conditions imply a set of constraints on the group-theoretic embedding coefficients  $l_i$ ,  $r_i$  parametrising the space of solutions in which the string propagates without pathologies. These consistency conditions take exactly the same form as those obtained from the worldsheet CFT analysis in Section 3.2, Eqs. (3.29)–(3.32). In passing, we note that recently a similar relation between consistency of the worldsheet theory and a well-behaved geometry was found in a related context in [155, 156].

#### 3.4 Matching to JMaRT and two-charge limits

In the previous sections we have shown that the class of null-gauged models defined in Section 2.7 in terms of the gauging parameters  $l_i$ ,  $r_i$  are consistent iff the latter can be written simply in terms of the four integers k, m, n and p. Out of these, only three are independent. Here we show that the resulting models correspond precisely to the full family of JMaRT solutions [37] and their various limits. The metric and *B*-field take a simple form in terms of these parameters, all of which have a clear physical meaning. Moreover, starting from the generic three-charge solution, we describe in detail the delicate limits that lead to the two-charge configurations and exhibit novel non-BPS NS5-P solutions.

#### 3.4.1 JMaRT metric and *B*-field

Let us recall the form of the NS5-decoupled limit of the NS5-F1-P JMaRT solutions, that is, the *S*-duals of the smooth and horizonless non-supersymmetric D1-D5-P back-grounds obtained in [37]. Note that we are working in units where  $\alpha' = 1$ , hence

 $Q_5 = n_5$ . In our conventions, these geometries take the form [52]

$$ds^{2} = \frac{f}{\tilde{H}_{1}} \left( -dt^{2} + dy^{2} \right) + \frac{M}{\tilde{H}_{1}} \left( c_{p}dt - s_{p}dy \right)^{2} + n_{5} \left( d\rho^{2} + d\theta^{2} \right) + \frac{n_{5}}{\tilde{H}_{1}} \left[ \left( r_{+}^{2} - r_{-}^{2} \right) \cosh^{2} \rho + r_{-}^{2} + a_{2}^{2} + Ms_{1}^{2} \right] \sin^{2} \theta \, d\phi^{2} + \frac{n_{5}}{\tilde{H}_{1}} \left[ \left( r_{+}^{2} - r_{-}^{2} \right) \sinh^{2} \rho + r_{+}^{2} + a_{1}^{2} + Ms_{1}^{2} \right] \cos^{2} \theta \, d\psi^{2} + \frac{2\sqrt{Mn_{5}}}{\tilde{H}_{1}} \left[ \left( a_{2}c_{1}c_{p} - a_{1}s_{1}s_{p} \right) dt + \left( a_{1}s_{1}c_{p} - a_{2}c_{1}s_{p} \right) dy \right] \sin^{2} \theta \, d\phi + \frac{2\sqrt{Mn_{5}}}{\tilde{H}_{1}} \left[ \left( a_{1}c_{1}c_{p} - a_{2}s_{1}s_{p} \right) dt + \left( a_{2}s_{1}c_{p} - a_{1}c_{1}s_{p} \right) dy \right] \cos^{2} \theta \, d\psi + ds_{\mathbb{T}^{4}}^{2},$$

$$(3.57)$$

$$B = -\frac{Ms_{1}c_{1}}{\tilde{H}_{1}} dt \wedge dy + \frac{n_{5}\cos^{2}\theta}{\tilde{H}_{1}} \Big[ (r_{+}^{2} - r_{-}^{2})\sinh^{2}\rho + r_{+}^{2} + a_{2}^{2} + Ms_{1}^{2} \Big] d\phi \wedge d\psi + \frac{\sqrt{Mn_{5}}}{\tilde{H}_{1}} \Big[ (a_{1}c_{1}c_{p} - a_{2}s_{1}s_{p})dt + (a_{2}s_{1}c_{p} - a_{1}c_{1}s_{p})dy \Big] \wedge \sin^{2}\theta \, d\phi$$
(3.58)  
$$+ \frac{\sqrt{Mn_{5}}}{\tilde{H}_{1}} \Big[ (a_{2}c_{1}c_{p} - a_{1}s_{1}s_{p})dt + (a_{1}s_{1}c_{p} - a_{2}c_{1}s_{p})dy \Big] \wedge \cos^{2}\theta \, d\psi,$$

together with the dilaton

$$e^{2\Phi} = g_s^2 \, \frac{n_5}{\tilde{H}_1}.\tag{3.59}$$

Here the charges are given in terms of the boost parameters  $\delta_{1,p}$ , that is

$$Q_i = M s_i c_i$$
,  $c_i = \cosh(\delta_i)$ ,  $s_i = \sinh(\delta_i)$ ,  $i = 1, p$ , (3.60)

and

$$f = \frac{1}{2} \left[ (r_{+}^{2} - r_{-}^{2}) \cosh(2\rho) + (a_{2}^{2} - a_{1}^{2}) \cos(2\theta) + r_{+}^{2} + r_{-}^{2} + a_{1}^{2} + a_{2}^{2} \right],$$
  

$$\tilde{H}_{1} = f + Ms_{1}^{2} = \frac{1}{2} \left[ (r_{+}^{2} - r_{-}^{2}) \cosh(2\rho) + (a_{2}^{2} - a_{1}^{2}) \cos(2\theta) + M^{2} \right] + Ms_{1}^{2}, \quad (3.61)$$

$$r_{\pm}^{2} = \frac{1}{2} \left[ (M - a_{1}^{2} - a_{2}^{2}) \pm \sqrt{(M - a_{1}^{2} - a_{2}^{2})^{2} - 4a_{1}^{2}a_{2}^{2}} \right] = -a_{1}a_{2} \left( \frac{s_{1}s_{p}}{c_{1}c_{p}} \right)^{\pm 1}.$$

In these formulas, parameters such as  $a_{1,2}$  are a priori thought of as continuous. As will be clear shortly, this is potentially misleading since they are constrained by the smoothness conditions and the absence of horizons. In this setup, these constraints read

$$a_{1}a_{2} = \frac{Q_{1}Q_{5}}{k^{2}R_{y}^{2}} \frac{s_{1}^{2}c_{1}^{2}s_{p}c_{p}}{(c_{1}^{2}c_{p}^{2} - s_{1}^{2}s_{p}^{2})},$$

$$M = a_{1}^{2} + a_{2}^{2} + r_{+}^{2} + r_{-}^{2} = a_{1}^{2} + a_{2}^{2} - a_{1}a_{2}\frac{c_{1}^{2}c_{p}^{2} + s_{1}^{2}s_{p}^{2}}{c_{1}c_{p}s_{1}s_{p}},$$
(3.62)

and

$$\mathsf{m} = \sqrt{\frac{M}{n_5} \frac{\mathsf{k}R_y s_p c_p}{(a_2 s_1 s_p - a_1 c_1 c_p)}} \in \mathbb{Z} , \quad \mathsf{n} = \sqrt{\frac{M}{n_5} \frac{\mathsf{k}R_y s_p c_p}{(a_2 c_1 c_p - a_1 s_1 s_p)}} \in \mathbb{Z}.$$
(3.63)

Here the integer numbers m and n again parametrise the angular momenta on  $S^3$ . Moreover, these constraints imply the important identity

$$\frac{Q_p}{Q_1} = \frac{n_5 \,\mathrm{mn}}{\mathrm{k}^2 R_y^2} = \frac{\mathrm{p}}{\mathrm{k}} \frac{1}{R_y^2} \tag{3.64}$$

with  $k p = n_5 m n$  as before, which relates non-trivially the three charges sourcing the configuration. We now show that the above set of conditions leads to a further set of relations which enable us to rewrite the JMaRT solutions in terms of the three integers k, m and n, together with a single dimensionful scale set by  $R_y$ . A related but different calculation was carried out in [157]. Defining

$$b^2 = r_+^2 - r_-^2 \Rightarrow f = b^2 f_0 \Rightarrow \tilde{H}_1 = b^2 \Sigma_0,$$
 (3.65)

the most useful relations are of the following form:

$$\begin{aligned} a_{2}^{2} - a_{1}^{2} &= b^{2}(\mathsf{m}^{2} - \mathsf{n}^{2}), \ f + M \, s_{p}^{2} = b^{2} \, h_{y}, \ f - M \, c_{p}^{2} = b^{2} \, h_{t}, \end{aligned} \tag{3.66} \\ M(c_{1}^{2} + s_{1}^{2}) &= b^{2} \left(\mathsf{m}^{2} + \mathsf{n}^{2} - 1 + \frac{2\mathsf{k}^{2}R_{y}^{2}}{n_{5}}\right), \ M(c_{p}^{2} + s_{p}^{2}) = b^{2} \left(\mathsf{m}^{2} + \mathsf{n}^{2} - 1 + \frac{2\mathsf{p}^{2}}{n_{5}R_{y}^{2}}\right) \\ a_{1}^{2} + r_{+}^{2} + M s_{1}^{2} = b^{2} \left(\mathsf{n}^{2} + \frac{\mathsf{k}^{2}R_{y}^{2}}{n_{5}}\right), \ a_{2}^{2} + r_{+}^{2} + M s_{1}^{2} = b^{2} \left(\mathsf{m}^{2} + \frac{\mathsf{k}^{2}R_{y}^{2}}{n_{5}}\right), \end{aligned}$$

where we have defined

$$\Sigma_0 = \sinh^2 \rho + (m^2 - n^2) \cos^2 \theta + n^2 + \frac{k^2 R_y^2}{n_5}, \qquad (3.67)$$

$$\Delta = \sqrt{n_5(\mathsf{m}^2 + \mathsf{n}^2 - 1) + \mathsf{k}^2 R_y^2 + \frac{\mathsf{p}^2}{R_y^2}},$$
(3.68)

such that  $\Sigma_0$  is the same quantity as in previous sections. Finally, we have

$$\frac{Q_1}{b^2} = \frac{\mathsf{k}R_y}{n_5}\,\Delta\,,\quad \frac{Q_p}{b^2} = \frac{\mathsf{p}}{n_5R_y}\,\Delta\,.\tag{3.69}$$

We note that for the metric and the *B*-field we do not need the individual charges  $Q_1$  and  $Q_p$ , but only the ratios (3.69). By using these formulas, the  $b^2$  factor cancels out

completely, and we finally obtain the six-dimensional fields

$$ds^{2} = n_{5}(d\theta^{2} + d\rho^{2}) + \frac{1}{\Sigma_{0}} \left[ -\left(\sinh^{2}\rho + (m^{2} - n^{2})\cos^{2}\theta + 1 - m^{2} - \frac{p^{2}}{n_{5}R_{y}^{2}}\right) dt^{2} \right]$$

$$(3.70)$$

$$+ \left(\sinh^{2}\rho + (m^{2} - n^{2})\cos^{2}\theta + n^{2} + \frac{p^{2}}{n_{5}R_{y}^{2}}\right) dy^{2} - 2\frac{p}{n_{5}R_{y}}\Delta dt dy$$

$$+ \left(n_{5}\sinh^{2}\rho + n_{5}m^{2} + k^{2}R_{y}^{2}\right)\sin^{2}\theta d\phi^{2} + \left(n_{5}\sinh^{2}\rho + n_{5}n^{2} + k^{2}R_{y}^{2}\right)\cos^{2}\theta d\psi^{2}$$

$$+ 2\left(m\Delta dt - \left(m\frac{p}{R_{y}} + nkR_{y}\right)dy\right)\sin^{2}\theta d\phi - 2\left(n\Delta dt - \left(n\frac{p}{R_{y}} + mkR_{y}\right)dy\right)\cos^{2}\theta d\psi\right],$$

$$B = \frac{1}{\Sigma_{0}} \left[ -\frac{kR_{y}}{n_{5}}\Delta dt \wedge dy + \left(n_{5}\sinh^{2}\rho + n_{5}m^{2} + k^{2}R_{y}^{2}\right)\cos^{2}\theta d\phi \wedge d\psi$$

$$+ \left(m\Delta dt - \left(m\frac{p}{R_{y}} + nkR_{y}\right)dy\right) \wedge \cos^{2}\theta d\psi - \left(n\Delta dt - \left(n\frac{p}{R_{y}} + mkR_{y}\right)dy\right) \wedge \sin^{2}\theta d\phi\right]$$

This is exactly the geometry we get from the null-gauge construction studied in the previous sections when inserting the parametrisation (3.50) for the  $l_i$ ,  $r_i$  gauging parameters in Eqs. (2.227).

It is worth discussing some interesting facts about the expressions we have presented in (3.70). First, we note the trivial symmetry associated to exchanging the two S<sup>3</sup> angular momenta. This corresponds to the re-labelling  $m \leftrightarrow n$  and  $\phi \leftrightarrow -\psi$ , which must be accompanied by the shift  $\theta \rightarrow \pi/2 - \theta$ . On the other hand, we note that while the usual JMaRT geometry is obtained by replacing  $p = n_5 mn/k$  in the expressions in (3.70), here we have chosen a slightly more general form by keeping p explicit. As a result, we easily find a symmetry that corresponds to exchanging  $k \leftrightarrow p$  and  $R_y \rightarrow 1/R_y$ , which we have identified above as T-duality. At the classical level, we can now see that this operation is equivalent the well-known Buscher rules [158], where  $g_{yy} \rightarrow 1/g_{yy}$ ,  $g_{ty} \rightarrow B_{ty}/g_{yy}$ , etc.

Importantly, by keeping p explicit in (3.70) we have presented expressions that are valid even for solutions where k = 0. As will be reviewed below, this includes the limit associated to the BPS and non-BPS two-charge NS5-P configurations.

#### 3.4.2 The dilaton

As described above, the JMaRT dilaton is of the form (3.59). The only coordinatedependent part of this expression corresponds to  $\Sigma_0$  as defined in (3.67), where we have used (3.65). This matches exactly with the expression obtained in Section 2.7, see Eq. (2.229), by considering the supergravity equations of motion, which provide the dilaton up to a multiplicative constant. The matching with the JMaRT backgrounds thus gives a criteria for choosing this constant appropriately: it is given by  $n_5/b^2$ , i.e.

$$e^{2\Phi} = \frac{n_5}{b^2 \Sigma_0}.$$
 (3.71)

In order to make the expression for the dilaton more transparent we proceed as follows. First, we introduce the canonical expressions for the charges

$$Q_1 = n_1 \frac{g_s^2}{V_4}, \quad Q_p = \frac{n_p}{R_y^2} \frac{g_s^2}{V_4},$$
 (3.72)

where  $V_4$  is the volume of the internal  $\mathbb{T}^4$ , while  $n_1$  is the number of fundamental string sources and  $n_p$  is the integer momentum charge. We observe that the key property (3.64) is equivalent to

$$\frac{\mathsf{k}R_y}{Q_1} = \frac{\mathsf{p}}{Q_p R_y}.\tag{3.73}$$

This ties in nicely with the fact that, as discussed above, T-duality interchanges  $Q_1 \leftrightarrow Q_p$  and  $kR_y \leftrightarrow p/R_y$ . It also justifies referring to p as being related to the momentum charge as we did in previous sections. We have seen that the parameters m and n are associated to the angular momenta of the geometry. Here we find in (3.73) that p and k relate to the momentum and F1 winding charges of the black hole microstate in question along the asymptotic *y*-circle.

Furthermore, while the  $b^2$  factor is irrelevant for writing down the metric and *B*-field, it does appear when computing the dilaton. This means that we need to work with the individual charges  $Q_1$  and  $Q_p$  as opposed to the ratios. By making use of Eqs. (3.69) we obtain two equivalent expressions for the dilaton, namely

$$e^{2\Phi} = \frac{\Delta}{\Sigma_0} \frac{\mathsf{k}R_y}{Q_1} = \frac{\Delta}{\Sigma_0} \frac{\mathsf{p}/R_y}{Q_p}, \qquad (3.74)$$

where  $\Delta$  was defined in (3.68). This shows that for the dilaton the Buscher rule  $\Phi \rightarrow \Phi - \frac{1}{2} \log g_{yy}$  is once again equivalent to the simultaneous replacements  $Q_1 \leftrightarrow Q_p$  and  $kR_y \leftrightarrow p/R_y$  since the constant prefactor in front of  $\Sigma_0$  is invariant by itself.

#### 3.4.3 JMaRT uniqueness

Having rewritten the NS5-decoupled JMaRT solutions in the form given in Eq. (3.70), we observed that these supergravity fields are exactly those we obtained from the null-gauge construction studied in the previous sections when inserting the parametrisation (3.29)–(3.32) for the  $l_i$ ,  $r_i$  gauging parameters in the general solutions in Eq. (2.227).

We have thus shown that we are able to reproduce the full family of supergravity backgrounds collectively denoted as JMaRT. We now argue that these solutions exhaust the full set of consistent null-gauged models considered in this chapter, by scrutinising the allowed ranges of the parameters and identifying physically equivalent solutions.

At first sight, by looking at the metric, B-field and dilaton given in Eqs. (3.70) and (3.74), one would expect that k, m, n and p could take any integer value. Additionally, in writing these expressions we have fixed an extra degree of freedom by choosing a positive sign for  $\Delta$ . However, this parameter space is constrained. In the general case, the parameter p is fixed in terms of the other three as in Eq. (3.33), though we will see below that in the limit in which the fundamental string charge  $Q_1$  vanishes, its value becomes arbitrary. On the other hand, choosing the opposite sign for  $\Delta$  simply results in an equivalent time-reversed configuration. This leaves us with arbitrary m, n and k (for negative k, the orbifold parameter is identified with its absolute value). Changing the sign of m, n or k can be compensated by choosing the orientation of the circle coordinate  $\psi$  or the S<sup>3</sup> angles  $\phi$  and  $\psi$ , respectively. Therefore we can take all of m, n, k to be nonnegative. In addition, due to the  $m \leftrightarrow n$  symmetry described below (3.70), we are free to restrict to  $m \ge n$ . Finally, based on spectral flow considerations we have argued in Section 3.2 that we must restrict to angular momenta such that  $m \pm n$  are odd, see the discussion below Eq. (3.28). This clarifies the discussion about spin structures in [37]. This excludes m = n and so we conclude that the set of inequivalent configurations is given by

$$k \ge 0$$
,  $m > n \ge 0$ ,  $m \pm n \in 2\mathbb{Z} + 1$ , (3.75)

which is precisely the principal range of values considered in [37]. As we shall discuss below, for k = 0 we take the limit such that one of the angular momenta vanishes and p is generically kept non-zero and finite.

#### 3.4.4 Two-charge limits and novel non-BPS NS5-P solutions

The expressions (3.70) and (3.74) we have obtained for the metric, *B*-field and dilaton generated in the classical limit of the null-gauged models match exactly the solutions given in [37], which have k > 0 (and where  $p = n_5 mn/k$ ). However, and as discussed above, they are presented in a form that is slightly more general and can be used to access somewhat delicate limits. In particular, we now examine two-charge limits.

There are two such limits that we can access. The first of these corresponds to NS5-F1 solutions, obtained by setting  $Q_p = 0$ , which were analysed in [37]. As shown by the identity (3.73), in order to keep  $Q_1$  finite and arbitrary we need to do this carefully. More precisely, we also need to take the limit  $p \rightarrow 0$  in such a way that the ratio  $p/(R_yQ_p) = kR_y/Q_1$  is finite. An analogous conclusion for taking  $n \rightarrow 0$  with  $p/n = n_5m/k$  fixed is obtained by considering (3.33). On the other hand, we can now similarly access a different limit leading to novel non-BPS NS5-P configurations. In this case, we take  $Q_1 = k = n = 0$  while keeping the ratios  $kR_y/Q_1 = p/(R_yQ_p)$  and  $k/n = n_5m/p$  fixed. This allows  $Q_p$  to take arbitrary values as needed. To the best of our knowledge, the metric and *B*-field for the non-BPS NS5-P solutions have not been presented in the literature. They take the following form:

$$ds^{2} = \frac{1}{\Sigma_{0}} \left[ -\left(\sinh^{2}\rho + m^{2}(\cos^{2}\theta - 1) + 1 - \frac{p^{2}}{n_{5}R_{y}^{2}}\right) dt^{2} - 2\frac{p}{n_{5}R_{y}}\Delta dtdy \qquad (3.76)$$

$$+ \left(\sinh^{2}\rho + m^{2}\cos^{2}\theta + \frac{p^{2}}{n_{5}R_{y}^{2}}\right) dy^{2} + 2m\left(\Delta dt - \frac{p}{R_{y}}dy\right)\sin^{2}\theta d\phi$$

$$+ n_{5}\left(\sinh^{2}\rho + m^{2}\right)\sin^{2}\theta d\phi^{2} + n_{5}\sinh^{2}\rho\cos^{2}\theta d\psi^{2} \right] + n_{5}(d\theta^{2} + d\rho^{2}),$$

$$B = \frac{n_{5}}{4\Sigma_{0}} \left[m^{2} - 1 + \cosh(2\rho)\right]\cos(2\theta) d\phi \wedge d\psi + \frac{m}{\Sigma_{0}}\left(\Delta dt - m\frac{p}{R_{y}}dy\right) \wedge \cos^{2}\theta d\psi,$$

with the dilaton given by the second expression in (3.74), and where we now have

$$\Sigma_0 = \sinh^2 \rho + m^2 \cos^2 \theta$$
,  $\Delta = \sqrt{n_5(m^2 - 1) + \frac{p^2}{R_y^2}}$ . (3.77)

Recall that, as discussed around Eq. (3.52), these solutions involve a fivebrane source at  $\rho = 0, \theta = \pi/2$  and a  $\mathbb{Z}_{|m|}$  orbifold singularity at  $\rho = 0, \theta \neq \pi/2$ .

Finally, we can restrict to the BPS cases by setting m = 1, as indicated by the supersymmetry conditions (3.13) and (3.14). Thus, in this limit we find  $\Delta = kR_y$  for the BPS NS5-F1 configuration, and  $\Delta = p/R_y$  for the NS5-P one. In the former case, the first expression for the dilaton in (3.74) then gives

$$e^{2\Phi}|_{\rm NS5-F1} = \frac{1}{Q_1} \frac{k^2 R_y^2}{\sinh^2 \rho + \cos^2 \theta + k^2 R_y^2 / n_5},$$
(3.78)

which coincides with that of [51], Eq. (4.13), while for the latter the alternative expression in (3.74) yields

$$e^{2\Phi}|_{\text{NS5-P}} = \frac{1}{Q_p} \frac{(p/R_y)^2}{\sinh^2 \rho + \cos^2 \theta},$$
 (3.79)

which coincides with that of [51], Eq. (4.2). One can check that in both cases the metric and *B*-field match as well.

#### 3.4.5 AdS<sub>3</sub> limit and holography

The AdS limit of the geometries under consideration is obtained by taking the large  $R_y$  limit, while keeping the charge  $Q_1$  fixed. This describes the region of small radial distances (as compared with  $Q_1$  and  $R_y$ ). The energy and momenta  $ER_y$  and  $P_yR_y$  also stay fixed, such that the coordinates

$$\tilde{t} = t/R_y, \ \tilde{y} = y/R_y \tag{3.80}$$

are better suited for this region. The six-dimensional metric (3.70) then takes the form of an orbifolded  $AdS_3 \times S^3$ , namely

$$ds^{2} = n_{5} \left[ -\frac{1}{k^{2}} \cosh^{2} \rho \, d\tilde{t}^{2} + \frac{1}{k^{2}} \sinh^{2} \rho \, d\tilde{y}^{2} + d\rho^{2} + d\theta^{2} + sin^{2} \theta \left( d\phi - \frac{n}{k} d\tilde{t} + \frac{m}{k} d\tilde{y} \right)^{2} + \cos^{2} \theta \left( d\psi + \frac{m}{k} d\tilde{t} - \frac{n}{k} d\tilde{y} \right)^{2} \right].$$

$$(3.81)$$

The orbifold singularity structure near  $\tilde{y} = 0$  depends on the common divisors between m, n, k and is described in [37, 94, 52]. By means of the large gauge transformation

$$\tilde{\psi} = \psi + \frac{m}{k}\tilde{t} - \frac{n}{k}\tilde{y}, \qquad \tilde{\phi} = \phi - \frac{n}{k}\tilde{t} + \frac{m}{k}\tilde{y}, \qquad (3.82)$$

one can formally re-absorb the contributions from the angular momenta, that is, the terms depending on m and n. This is related to the general holographic description of such configurations. They are interpreted as excited states in the holographic symmetric orbifold CFT which can be constructed by considering  $n_1n_5$ /k identical strands of length k in their NS vacuum state and performing left-right asymmetric fractional spectral flow [94]. The spectral flow charges are of the form

$$\alpha = \frac{m+n}{k}$$
,  $\bar{\alpha} = \frac{m-n}{k}$ , (3.83)

which matches the intuition derived from (3.82) and provides yet another interpretation for the gauging parameters  $l_2$  and  $r_2$ . Note that this is distinct from the *worldsheet* spectral flow that was used in Section 3.2. Interestingly, within this description the momentum per strand is given by mn/k and must be an integer number,

$$\frac{\mathsf{mn}}{\mathsf{k}} \in \mathbb{Z}, \tag{3.84}$$

which is a slightly more restrictive condition than the quantisation of p discussed above.

Furthermore, we also note that in this AdS limit there seems to be no particular issues with the solutions with even  $m \pm n$ , see the discussion around Eq. (3.29). In particular, the case m = n = 0 takes us back to the global AdS<sub>3</sub> vacuum. The present perspective shows that the cases that extend consistently to the full linear dilaton geometry, namely  $m \pm n \in 2\mathbb{Z} + 1$ , belong to the RR sector (in the covering space) of the holographic

CFT, while those that do not are characterised by having spectral flow charges (2.89) with even numerators, such that they correspond to NSNS states.

Moreover, in this AdS<sub>3</sub> limit the dilaton becomes constant, as can be seen from (3.74) since  $\Delta \rightarrow kR_y$  and  $\Sigma_0 \rightarrow k^2 R_y^2/n_5$ . In other words, the rescaled harmonic function  $\tilde{H}_1$  associated to the fundamental string charges approaches the constant value  $Q_1$ , see Eq. (3.69). In terms of the actual harmonic function  $H_1$ , this roughly corresponds to the usual *dropping the "1+"* term, as is usual in such decoupling limits (see e.g. [136]).

In terms of the null-gauged description, there is an intuitive way of understanding this  $AdS_3 \times S^3$  limit. Indeed, the upstairs model already contains an  $SL(2, \mathbb{R}) \times SU(2)$  factor, complemented by the novel  $\mathbb{R}_t \times S_y^1$  factor. For large  $R_y$ , the gauging parameters associated to the former are  $l_{1,2} \sim r_{1,2} \sim \mathcal{O}(1)$ , while those corresponding to the latter grow parametrically large as  $l_{3,4} \sim r_{3,4} \sim \mathcal{O}(R_y)$ . Thus, we are mostly gauging away the extra directions *t* and *y*.

#### 3.4.6 From AdS back to the linear dilaton background

At this point, it is interesting to go back to the intuition developed within the sigma model description presented at the beginning of Section 2.7.1. There, we argued that, after integrating out the gauge fields, the gauging procedure induces a deformation given by including an additional term of the form  $\mathcal{J}\bar{\mathcal{J}}/\Sigma$  to the action, see Eq. (2.198). We can simplify the discussion by working in the t = y = 0 gauge, such that the currents  $\mathcal{J}$  and  $\bar{\mathcal{J}}$  are nothing but linear combinations of the diagonal currents of the SL(2,  $\mathbb{R}$ ) and SU(2) WZW models. In the AdS limit, we have seen that the coefficient  $1/\Sigma$  approaches a constant value, such that the induced contribution becomes  $\mathcal{J}\bar{\mathcal{J}}$ , giving a simple marginal deformation of the worldsheet theory.

The situation is to be contrasted with that of [143, 144, 145]. There, the authors make use of the null-gauging formalism to introduce a  $J\bar{J}$  deformation for the AdS<sub>3</sub> worldsheet theory, the crucial difference being that the current under consideration corresponds to the  $J^-$  instead of  $J^3$ . For related work, see [159]. Based on [110], this procedure is interpreted as the dual of the so-called single-trace  $T\bar{T}$  irrelevant deformation of the holographic CFT. Such deformation triggers a controlled flow to the UV, which is realised in the dual geometry by effectively *reinserting the "1+" term* in the harmonic function associated to the fundamental string sources. This produces an asymptotically linear dilaton geometry, i.e. an NS5-decoupled background, such that the above UV flow is thought of as leading to a realisation of little string theory.

The parallel to our construction can thus be made more general. As described above, for large  $R_y$  we know that  $\Sigma$  becomes constant, and our  $\mathcal{J}\bar{\mathcal{J}}$  deformation on the world-sheet made up of  $J^3$  and  $K^3$  (together with their antiholomorphic counterparts) has a much less dramatic effect, as it produces a sort of large gauge transformation which,

however, does not further modify the AdS<sub>3</sub> asymptotics, where the dilaton stays constant. On the other hand, when moving away from the AdS limit by keeping  $R_y$  finite one recovers the full non-trivial coordinate dependence of the function  $\Sigma$ , which sits at the denominator of various terms in the supergravity fields. This modifies the effect of the  $\mathcal{J}\bar{\mathcal{J}}/\Sigma$ , which now *does* take us back to the full asymptotically linear dilaton background described by Eqs. (3.70) and (3.74). This effect has also recently been observed in a larger class of solutions in [54].

### 3.5 Discussion

In this chapter we have analysed all consistent backgrounds within a general class of null-gauged WZW models. We showed that the (NS5-decoupled) JMaRT family, and limits thereof, are the unique supergravity backgrounds that arise in these models. We also showed that the metric and *B*-field can be written explicitly in terms of the integers k, m, n,  $n_5$  and the modulus  $R_y$ , while for the dilaton one needs further include the ratio  $n_1/V_4$  (or  $n_p/V_4$ ).

Our analysis makes the connection between the worldsheet and geometric descriptions quite explicit. In the supergravity solutions, imposing absence of CTCs implies  $l_3 = r_3$ , which excludes horizons. The converse statements are also true: excluding horizons implies  $l_3 = r_3$ , which excludes CTCs. On the other hand, the condition  $l_3 = r_3$  is necessary but not sufficient for smoothness (up to orbifold singularities), which further requires the quantization of k, m, n. This is consistent with the interpretation of the allowed configurations as a family of black hole microstates.

At the level of the worldsheet CFT, we have shown that a consistent spectrum is obtained if and only if the gauging parameters are given in terms of three independent integers k, m, n together with  $n_5$  and  $R_y$ , as in (3.29)–(3.33). We explicitly rewrote the JMaRT metric and *B*-field in terms of these quantities, which enabled us to completely bypass the usual, somewhat cumbersome, supergravity parametrisation in Eqs. (3.57), (3.58).

Our parametrisation also provides a clear and important physical understanding of the quantity  $p = n_5 mn/k$ . In the AdS<sub>3</sub> decoupling limit, the quantity mn/k has previously been interpreted as being the momentum per strand in the holographically dual symmetric product orbifold CFT [94]. We have uncovered the direct role of p in the (asymptotically linear dilaton) supergravity solutions, as being the quantity that is T-dual to k, where the T-duality is performed along the *y* circle.

As we mentioned in the Introduction, one of the motivations of our systematic analysis was the possibility of finding new backgrounds. Although our uniqueness proof means that the set of models we analysed does not have more general backgrounds than the JMaRT family, we have exhibited a novel sub-family of two-charge non-BPS NS5-P backgrounds that arise from a non-trivial limit, see Eqs. (3.76) and (3.77). We observed that in the core of the solutions but away from the fivebrane source, the solutions involve a  $\mathbb{Z}_m$  orbifold singularity. To our knowledge, these solutions have not appeared before in the literature.

We expect that our results will be useful in analysing generalisations of the models studied here, either by changing the currents being gauged to include non-Cartan generators of the non-Abelian factors of the upstairs group, or by changing the upstairs group, or both. Our systematic approach should enable generalisations to be investigated in a similar way. For instance, there are multi-centre non-BPS generalizations of the JMaRT family [160, 161, 162].

Besides this, within the models considered in this work there remain several unanswered questions. For instance, since we have control over these theories exactly in  $\alpha'$ , there are many interesting correlation functions that can be computed. We intend to report an analysis of such correlators in the near future.

The results we have obtained, and the possibilities they open up for future work, offer the prospect of improving our understanding of little string theory and the corresponding non-AdS holography. Furthermore, it is tempting to wonder about extending some of these ideas beyond the fivebrane decoupling limit into the full asymptotically flat regime.

Having an exact worldsheet description of heavy pure states, far from the vacuum of the theory, is rare and valuable. Such models allow us to study aspects of black hole microstates that are smeared out in supergravity, and so cannot be studied with supergravity techniques. This offers the tantalising prospect of obtaining a quantitative understanding of the microscopic degrees of freedom of black holes.

# Chapter 4

# Worldsheet Correlators in Black Hole Microstates

String Theory provides a microscopic description of black holes as being bound states of strings and branes with an exponentially large number of internal microstates [19]. Amongst these microstates, there are coherent pure states, large families of which have been shown to be well-described by smooth and horizonless supergravity solutions, see e.g. [65, 66, 135, 92, 137, 138, 163]. Upon taking an appropriate AdS decoupling limit, these solutions are proposed to correspond to specific families of pure states in the holographically dual CFT (HCFT); precision holography has provided sharp evidence supporting this correspondence [164, 72, 165, 139, 166].

While supergravity constructions provide valuable insight into the structure of black hole microstates, it is natural to expect that string-theoretic physics beyond supergravity will be necessary to obtain a complete description of black hole microstructure. A fruitful arena in which to investigate such stringy physics is provided by bound states of NS5 branes carrying fundamental string (F1) and/or momentum charge (P). More specifically, we work in Type IIB compactified on  $\mathbb{S}^1 \times \mathbb{T}^4$ , with  $n_5$  NS5 branes wrapped on  $\mathbb{S}^1 \times \mathbb{T}^4$ ,  $n_1$  units of F1 winding on  $\mathbb{S}^1$ , and  $n_P$  units of momentum charge along  $\mathbb{S}^1$ .

Upon taking the fivebrane decoupling limit, one obtains asymptotically linear dilaton configurations, which are holographically dual to (doubly scaled) Little String Theory [120, 115]. In an appropriate region of the parameter space, there is an AdS<sub>3</sub> regime in the IR, and one can take a further AdS<sub>3</sub> decoupling limit [16]. Upon doing so, one obtains the well-studied NS5-F1 instance of AdS<sub>3</sub>/CFT<sub>2</sub> holography [78, 110].

The NSNS vacuum of the holographic CFT corresponds to the global  $AdS_3 \times S^3 \times \mathbb{T}^4$  background, whose worldsheet theory involves an  $SL(2,\mathbb{R})\times SU(2)$  Wess-Zumino-Witten (WZW) model [109, 106, 101, 102, 103]. In recent work, a family of gauged

WZW models involving the same Lie groups has been constructed and studied, providing an exact worldsheet description of a set of NS5-F1-P black hole microstates [51, 52, 53, 54, 1].

Processes in which light probes interact with a heavy background such as a black hole or a black hole microstate give rise to interesting dynamical observables, in particular mixed heavy-light (HL) correlators. Such correlators have been previously studied in holographic systems, see e.g. [167, 168, 169, 170, 171]. In the NS5-F1 system, there is a locus in moduli space at which the holographic CFT is conjectured to be an  $\mathcal{N} =$ (4,4) symmetric product orbifold CFT with target space  $(\mathbb{T}^4)^N / S_N$ , where  $N = n_1 n_5$ . There is now a substantial body of evidence for this conjecture, see e.g. [164, 72, 165, 139, 166, 172, 113, 99, 100]. For recent discussions of holography in related systems, see [90, 173, 174].

For instance, heavy-light-light-heavy (HLLH) four-point functions have been computed in the supergravity approximation and/or in the symmetric product orbifold CFT, for particular sets of heavy and light operators [167, 170, 171]. Having solvable worldsheet models associated to black hole microstates means we can go much further by taking into account  $\alpha'$  corrections [51]. Given a worldsheet model describing string dynamics on a heavy background, the relevant quantities correspond to integrated correlators of light operators in the worldsheet vacuum.

Worldsheet correlators in global AdS<sub>3</sub> were first studied in [103], building in part on [175, 106], and the role of the vertex operators associated with spectrally flowed representations was highlighted. Further studies include [176, 177, 178, 38, 179]. The spectrum of chiral primaries and their three-point functions in global AdS<sub>3</sub> × S<sup>3</sup> × T<sup>4</sup> were computed in [172, 113, 99, 100], and shown to match those of the symmetric product orb-ifold CFT, as studied in [180, 97].

The supergravity backgrounds we consider are known as NS5-F1 circular supertubes and spectral flows thereof [132, 124, 125, 37, 96, 94]. This includes non-BPS spectrally flowed supertubes, known as the JMaRT solutions, after the authors of [37]. The associated worldsheet models are null-gauged WZW models, where before gauging one considers a (10+2)-dimensional target space  $AdS_3 \times S^3 \times \mathbb{R}_t \times S_y^1 \times \mathbb{T}^4$ , and after gauging one obtains a (9+1)-dimensional spectrally flowed supertube solution. In the asymptotic linear dilaton region, the gauging is concentrated mostly in the time and angular directions of SL(2, $\mathbb{R}$ ), while in the IR AdS<sub>3</sub> region, the gauging is concentrated mostly along the *t* and *y* directions.

These coset models can also be thought of as marginal current-current deformations of the worldsheet theory for strings in  $AdS_3$ . These are instances of a larger class of deformations that undo the decoupling limit with respect to the F1 harmonic function, i.e. they "add back the 1+" in that function, leading to linear dilaton asymptotics; see

e.g. [54]. At the level of the dual field theory, a closely related procedure has been argued to correspond to the so-called single-trace  $T\bar{T}$  irrelevant deformation of the original holographic CFT [145, 143], flowing towards a non-local Little String Theory.

To study string correlators in these highly excited backgrounds, we first compute a large set of physical vertex operators, in both NSNS and RR sectors, building on [52, 54]. These describe linearized perturbations of the background configurations. We focus primarily on coset states in discrete series representations, including worldsheet spectral flow, that are dual to chiral primary operator excitations in the HCFT. When the background is BPS, a subset of these are BPS fluctuations.

The currents being gauged in these cosets are linear combinations of the Cartan generators of the symmetry algebra. Therefore the "*m*-basis" for vertex operators, in which the actions of these currents are diagonalized, is the natural framework to use. In the IR AdS<sub>3</sub> limit, we describe how these operators are related to their global  $AdS_3 \times S^3$ counterparts.

We then compute a large set of correlators in the  $AdS_3$  limit. It is well known that in worldsheet models of global  $AdS_3$ , one can define an "x" variable that corresponds to the local coordinate of the holographic CFT [110]. One of the main novelties of our approach is the identification of the analogous x variable in the coset models we study. This identification requires some care due to the gauging. Indeed, the construction of [110] breaks down, because the SL(2, $\mathbb{R}$ ) raising and lowering operators do not commute with the BRST charge. A considerable amount of interesting physics follows from this step. It leads, for instance, to the combination of seemingly simple *m*-basis two-point functions into spacetime-local *x*-basis correlators with highly non-trivial *x*-dependence.

Our first main result is of a family of HLLH correlators, for which we obtain fully explicit expressions. In doing so, we show that these correlators assume a remarkably simple structure when written in terms of a covering space related only to the heavy states. From this observation, we obtain our second main result: a closed-form expression for a set of HL worldsheet correlators with an arbitrary number *n* of massless insertions, in terms of a correlator consisting of *n* light insertions in global  $AdS_3 \times S^3$ . For *n* = 3 this result can be made completely explicit, and we present a particular example in full detail. This constitutes the first correlator in the literature involving three light worldsheet vertices on a black hole microstate background, dual to a heavy-light five-point function of the holographic CFT.

*A priori*, our worldsheet correlators give predictions for correlators of the dual holographic CFT at strong coupling. Generically, four-point (and higher-point) correlators are not protected across moduli space, however, a specific set of HLLH correlators have been shown to precisely agree between supergravity and the symmetric product orbifold CFT [167]. Similarly, the emission spectrum and rate for the unitary analog of Hawking radiation from the JMaRT solutions agrees between supergravity and symmetric product orbifold CFT [140, 93, 141, 94]. Thus it is natural to investigate more generally which HL correlators are protected (at large *N*) between worldsheet and symmetric product orbifold CFT, and which are not.

We carry out this comparison for three sub-families of our worldsheet correlators. Firstly, we compare various sets of HLLH correlators to the symmetric product orbifold CFT, finding exact agreement in all cases for which the orbifold CFT correlator is available in the literature. Importantly, this matching holds at leading order in large *N*, but exactly in  $\alpha'$ . This comparison includes a substantial generalization of the supergravity and holographic CFT correlators computed in [167]. Our comparisons notably include an example in which the light operators in the symmetric orbifold CFT are twist-two. In this case, and as shown recently in [181, 182], the Lunin-Mathur covering map used in the symmetric orbifold computation is different to the one appearing in the world-sheet computation, making the comparison highly non-trivial. Remarkably, both results agree exactly in the large *N* limit.

Secondly, we compute the five-point HLLLH symmetric orbifold CFT correlator corresponding to the three-point worldsheet correlator mentioned above, and also find exact agreement.

Finally, we compute the analogue of the Hawking radiation rate for the JMaRT solutions. Once again, we find perfect agreement with the dual symmetric product orbifold CFT, extending the supergravity and holographic CFT results of [140, 93, 141, 94].

A likely explanation for this remarkable agreement is that the heavy states we consider are quite special. Specifically, the heavy backgrounds are related to the global  $AdS_3 \times S^3$ vacuum via orbifolding and fractional spectral flow [96, 94]. This fact also underlies our general formula for the HL correlators with *n* light insertions. When *n* > 3, we do not expect these HL correlators to be generically protected across moduli space; we shall discuss this in detail in due course.

The structure of the chapter is as follows. In Section 4.1 we construct a large set of vertex operators of the worldsheet cosets we study, both in the NS and in the R sectors. We then examine their AdS<sub>3</sub> limit and relate these vertices to those constructed in Sec. 2.5.1. In Sections 4.2 and 4.3, we present our main results. We identify the "x" variable dual to the local coordinate of the holographic CFT, and obtain an extensive set of novel HLLH correlators, including massless insertions with arbitrary spacetime weights and charges. The final results are presented in Eqs. (4.81) and (4.93). We then compare a subset of these results to the symmetric product orbifold CFT, finding exact agreement for all correlators available in the literature. We present a closed formula for a large class of worldsheet correlators with an arbitrary number of massless insertions, Eq. (4.94). We compute a five-point correlator in the symmetric orbifold CFT and find agreement with our general worldsheet formula. Finally, we compute the amplitude

describing the unitary analogue of Hawking radiation for the JMaRT microstates. We discuss our results in Section 4.4.

# 4.1 Null-gauged description and worldsheet spectrum

We now proceed to describe massless excitations in the worldsheet theories associated with the heavy backgrounds we study. We describe in detail how to obtain the lowlying physical states via BRST quantization in these null-gauged models, in both the NSNS and RR sectors. In the subset of the backgrounds that preserve some supersymmetry, we discuss the BPS light excitations.

In the full null-gauged models, these massless vertex operators describe linearized fluctuations around the full asymptotically linear dilaton solutions describing the heavy states, and so can be thought of as worldsheet representatives of light states belonging to the Little String Theory living on the NS5 branes.

Our main interest in this work will be computing correlators in the IR AdS<sub>3</sub> limit, in which we have reviewed the fact that the backgrounds are related to orbifolded AdS<sub>3</sub> ×  $S^3 \times T^4$  via a spacetime spectral flow large coordinate transformation. We describe how the AdS<sub>3</sub> limit can be taken on the worldsheet vertex operators. This leads to states that can be understood holographically, in the spacetime (fractionally) spectrally flowed frame defining the heavy background, as discussed around Eq. (2.89).

#### 4.1.1 BRST quantization

We start by reviewing the quantization of the class of worldsheet coset models introduced in Section 2.7, which describe the propagation of superstrings in the JMaRT backgrounds and their (BPS and/or two-charge) limits [51, 52, 54, 1].

Before gauging, we have the WZW model associated to the (10+2)-dimensional group manifold  $SL(2,\mathbb{R}) \times SU(2) \times \mathbb{R}_t \times U(1)_y \times U(1)^4$  as introduced in (2.205). This is described simply by adding the extra time direction *t* and spatial circle *y* to the matter content employed in the previous section, together with the corresponding fermionic partners  $\lambda^t$  and  $\lambda^y$ . The latter are bosonized using a canonically normalised scalar  $H_6$ as

$$\lambda_t = \frac{1}{2} \left( e^{iH_6} - e^{-iH_6} \right) , \qquad \lambda_y = \frac{1}{2} \left( e^{iH_6} + e^{-iH_6} \right) , \qquad (4.1)$$

$$i\partial H_6 = 2\,\lambda^t \lambda^y$$
,  $H_6^\dagger = -H_6$ . (4.2)

The holomorphic parts of their OPEs are

$$-t(z)t(w) \sim y(z)y(w) \sim -\frac{1}{2}\log(z-w), \quad -\lambda^t(z)\lambda^t(w) \sim \lambda^y(z)\lambda^y(w) \sim \frac{1}{2}\frac{1}{(z-w)},$$
(4.3)

and they give additional free field contributions to the matter T and G in (2.135) and (2.136),

$$T_{(ty)} = \partial t \partial t - \partial y \partial y , \qquad G_{(ty)} = 2i \left( -\lambda^t \partial t + \lambda^y \partial y \right).$$
(4.4)

From now on, *T* and *G* will denote the stress tensor and supercurrent of the full model, including the above terms. Introducing the operators  $\hat{P}_t = i\partial t$ ,  $\hat{P}_t = i\bar{\partial} t$ ,  $\hat{P}_y = i\partial y$ , and  $\hat{P}_y = i\bar{\partial} y$ , we gauge the chiral null currents<sup>42</sup>

$$J = i\mathcal{J} = J^3 + l_2K^3 + l_3\hat{P}_t + l_4\hat{P}_y, \quad \bar{J} = i\bar{\mathcal{J}} = \bar{J}^3 + r_2\bar{K}^3 + r_3\hat{P}_t + r_4\hat{P}_y, \quad (4.5)$$

which are the quantum operator versions of the classical currents in Eq. (2.215). The supersymmetric partners of the currents *J* and  $\overline{J}$  are given in Eq. (3.12).

To perform the null gauging, one introduces additional fermionic and bosonic firstorder ghosts, denoted by  $(\tilde{b}, \tilde{c})$  and  $(\tilde{\beta}, \tilde{\gamma})$ , with conformal weights  $\Delta[\tilde{c}] = 0$  and  $\Delta[\tilde{\gamma}] =$ 1/2 [54]. The central charges  $c_{\tilde{b}\tilde{c}} = -2$  and  $c_{\tilde{\beta}\tilde{\gamma}} = -1$  cancel the additional matter contribution  $c_{ty} = 3$ . The  $(\tilde{\beta}, \tilde{\gamma})$  system has no background charge and is bosonized via

$$\tilde{\beta} = e^{-\tilde{\varphi}} \partial \tilde{\xi} , \qquad \tilde{\gamma} = \tilde{\eta} \, e^{\tilde{\varphi}} .$$
 (4.6)

We shall momentarily introduce a modified BRST charge that imposes invariance under the action of the null currents (3.8) and their supersymmetric partners (3.12). Physical operators in 9+1 dimensions will be given by states of the ungauged (10+2)-dimensional WZW model that survive the gauging procedure [54]. Of course, and as we shall see shortly, the Virasoro conditions and the expressions of the BRST-exact states will be modified accordingly. We consider a set of mutually local operators before the gaugings, i.e. we perform the analog of the GSO projection in the (10+2)-dimensional model. We thereby obtain a tachyon-free spectrum in the gauged models.

Underlying this procedure is the fact that for the case of chiral null gaugings, the Polyakov-Wiegmann identity allows one to rewrite the gauged action of the *downstairs* model into a form identical to that of the *upstairs* model in terms of a new gauge-invariant variables [146, 118], see also [1]. This is achieved at the level of the path integral by means of a field redefinition with a Jacobian that is almost trivial except for a factor which, when exponentiated, gives rise to the additional ghost fields described above.

<sup>&</sup>lt;sup>42</sup>The symbol *J* for the current operators should not be confused with the total SL(2, $\mathbb{R}$ ) spin that has appeared in Section 2.6.4. The meaning should be clear from the context.

Explicitly, physical operators in the coset model are defined by the cohomology classes of the BRST charge [54]

$$Q = \oint dz \left[ c \left( T + T_{\beta \gamma \bar{\beta} \bar{\gamma}} \right) + \gamma G + \tilde{c} J + \tilde{\gamma} \lambda + \text{ghosts} \right], \qquad (4.7)$$

where the last two terms implement the null-gauging procedure. Whether the resulting spectrum is supersymmetric or not depends on whether some linear combination(s) of the following supercharges are BRST-invariant [54],

$$Q_{\varepsilon} = \oint dz \, e^{-(\varphi - \tilde{\varphi})/2} S_{\varepsilon} \,, \qquad S_{\varepsilon} = \exp\left(\frac{i}{2} \sum_{I=1}^{6} \varepsilon_{I} H_{I}\right). \tag{4.8}$$

For these to be mutually local, we impose the analog of the GSO projection in (10+2) dimensions,

$$\prod_{I=1}^{6} \varepsilon_I = 1, \qquad (4.9)$$

where the sign choice is dictated by requiring compatibility with the (9+1)-d GSO projection (2.145) and our other conventions, as we shall see below. We shall discuss the conditions for spacetime supersymmetry after we have analyzed more general Ramond sector vertex operators, around Eq. (4.29). For now, we emphasize that only a subset of the backgrounds we consider preserve some spacetime supersymmetry.

#### 4.1.2 The unflowed NS sector

We now analyze physical NS sector states of the gauged models, focusing on states with no spectral flow charges in SL(2, $\mathbb{R}$ ) or SU(2), and no winding charge  $\omega_y$  around the *y*-circle. As usual, the lightest physical operators come with a single fermionic excitation on top of the tachyon state

$$\mathcal{T}_{j,m,j',m'} = e^{-\varphi} V_{j,m} V'_{j',m'} e^{i(-Et+P_y y)}.$$
(4.10)

Note that since *t* is a non-compact direction and  $\omega_y = 0$ , both *E* and  $P_y$  are identical on the left and on the right sectors. For massless states, the  $L_0$  and  $\bar{L}_0$  Virasoro constraints both read

$$0 = -\frac{j(j-1)}{n_5} + \frac{j'(j'+1)}{n_5} - \frac{1}{4}E^2 + \frac{1}{4}P_y^2.$$
(4.11)

Moreover, operators are uncharged with respect to the null-currents  $J, \bar{J}$  in (3.8) if and only if their quantum numbers are related by

$$0 = m + l_2 m' + \frac{l_3}{2}E + \frac{l_4}{2}P_y , \qquad 0 = \bar{m} + r_2 \bar{m}' + \frac{r_3}{2}E + \frac{r_4}{2}P_y . \qquad (4.12)$$

We will work in the canonical "-1" picture for the  $\varphi$  ghost. On the other hand, the fact that  $\tilde{\varphi}$  has background charge  $Q_{\tilde{\varphi}} = 0$  allows us to build NS states directly at  $\tilde{\varphi}$ -picture zero. BRST-closed operators must then have a vanishing second-order pole in their OPE with the supercurrent *G*, and vanishing first-order pole in their OPE with the fermionic current  $\lambda$  given in (3.12).

As can be expected from the fact that the  $\mathbb{T}^4$  is untouched by the gaugings, the simplest solutions are the 6D scalars

$$\mathcal{V}_{j,m,j',m'}^{i} = e^{-\varphi} \lambda^{i} V_{j,m} V_{j',m'}^{\prime} e^{i(-Et+P_{y}y)} , \qquad i = 6, \dots, 9.$$
(4.13)

These are direct analogs of the global  $AdS_3$  states defined in Eq. (2.148a). They were considered in detail in [52], and their energies were matched with those of the minimally coupled scalar perturbations on top of the JMaRT background as computed in supergravity.

The remaining massless vertex operators will constitute the beginning of the main new results of this work. They are slightly more involved to construct, due to the fact that their polarization lies in a direction in which the null currents act non-trivially. An important consequence is that now the raising/lowering operators  $J_0^{\pm}$  and  $K_0^{\pm}$  do not commute with the BRST charge Q. So, unlike in global  $AdS_3 \times S^3 \times \mathbb{T}^4$  as reviewed in Section 2.6, physical states need not have definite  $SL(2,\mathbb{R})$  and SU(2) spins. They will, however, have definite projections  $m, \bar{m}, m'$  and  $\bar{m}'$ , and also well-defined energy E and momentum  $P_{y}$ .

This situation is a consequence of the fact that the  $AdS_3 \times S^3$  isometries are absent in the asymptotically linear dilaton geometry. Nevertheless, these isometries are restored in the IR, by taking  $R_y$  large while keeping  $ER_y$  and  $P_yR_y$  fixed. In this regime, the vertex operators of the gauged models will reduce to the  $AdS_3 \times S^3$  expressions in Eqs. (2.148b) and (2.148c).

Let us consider a generic linear combination of NS sector vertex operators,

$$e^{-\varphi} \left[ \left( c^{r} \psi^{r} V_{j,m-r} V_{j',m'}' + d^{r} \psi^{r} V_{j,m} V_{j',m'-r}' \right) + \left( c^{t} \lambda^{t} + c^{y} \lambda^{y} \right) V_{j,m} V_{j',m'}' \right] e^{i(-Et+P_{y}y)}, \quad (4.14)$$

where the notation mirrors that of the  $AdS_3 \times S^3$  expressions in Eq. (2.149); in particular, summation over r = +1, -1, 0 is implicit, with "0" corresponding to the "3" direction of the respective algebras. Of these eight degrees of freedom, two are removed by the conditions arising from the *G* and  $\lambda$  terms in the BRST charge, which respectively read

$$0 = mc^{3} + (m-j)c^{+} + (m+j)c^{-} + m'd^{3} + (j'+m')d^{+} + (j'-m')d^{-} + c^{t}\frac{E}{2} + c^{y}\frac{P_{y}}{2},$$
(4.15)

and

$$0 = n_5 \left( -c^3 + l_2 d^3 \right) - l_3 c^t + l_4 c^y.$$
(4.16)

This leaves six states, of which two turn out to be BRST exact. The first exact state comes, as usual, from the action of *G* on the tachyon operator (4.10), while the second one has no global  $AdS_3$  counterpart and appears due to the action of  $\lambda$  on the same state. Their explicit expressions are

$$\Phi_{G} = e^{-\varphi} \left[ \frac{2}{n_{5}} \frac{1}{2} V_{j',m'}' \left( (m-j+1)\psi^{-}V_{j,m+1} + (m+j-1)\psi^{+}V_{j,m-1} - 2m\psi^{3}V_{j,m} \right) \right. \\ \left. + \frac{2}{n_{5}} \frac{1}{2} V_{j,m} \left( (j'+m'+1)\chi^{-}V_{j',m'+1}' + (j'-m'+1)\chi^{+}V_{j',m'-1}' + 2m'\chi^{3}V_{j',m'}' \right) \right. \\ \left. + \frac{1}{2} \left( -E\lambda^{t} + P_{y}\lambda^{y} \right) V_{j,m}V_{j',m'}' \right] e^{i(-Et+P_{y}y)},$$

$$(4.17)$$

and

$$\Phi_{\lambda} = e^{-\varphi} \left[ \psi^3 + l_2 \chi^3 + l_3 \lambda^t + l_4 \lambda^y \right] V_{j,m} V'_{j',m'} e^{i(-Et+P_y y)}, \tag{4.18}$$

respectively. Such states are trivially BRST invariant since *G* and  $\lambda$  square to the Virasoro constraint (4.11) and the null condition (2.216), and the relevant term in their product is  $G(z)\lambda(0) \sim \lambda(z)G(0) \sim J(0)/z$ , whose action vanishes by means of the condition (4.12). In the end, we are left with four physical vertex operators to add to the four from the  $\mathbb{T}^4$  directions to give the correct eight polarizations in the holomorphic sector in 9+1 dimensions.

We choose a basis for these four physical vertex operators such that, in the  $AdS_3$  limit, they reduce to the basis of global  $AdS_3$  vertex operators described around Eq. (2.149). We thus obtain

$$\mathcal{W}^{\varepsilon} = e^{-\varphi} \left[ (\psi V_j)_{j+\varepsilon,m} V'_{j',m'} + (c^t_{\varepsilon} \lambda^t + c^y_{\varepsilon} \lambda^y) V_{j,m} V'_{j',m'} \right] e^{i(-Et+P_y y)}, \quad (4.19a)$$

$$\mathcal{X}^{\varepsilon} = e^{-\varphi} \left[ V_{j,m}(\chi V'_{j'})_{j'+\varepsilon,m'} + \left( d^{t}_{\varepsilon} \lambda^{t} + d^{y}_{\varepsilon} \lambda^{y} \right) V_{j,m} V'_{j',m'} \right] e^{i\left(-Et+P_{y}y\right)}, \quad (4.19b)$$

where the SL(2, $\mathbb{R}$ ) and SU(2) coefficients are those given in (2.149)–(2.150), while the novel ones are<sup>43</sup>

$$c_{\varepsilon}^{t} = -c_{\varepsilon}^{3} \frac{n_{5} P_{y}}{l_{4} E + l_{3} P_{y}}, \qquad c_{\varepsilon}^{y} = c_{\varepsilon}^{3} \frac{n_{5} E}{l_{4} E + l_{3} P_{y}}, \qquad (4.20)$$

$$d_{\varepsilon}^{t} = d_{\varepsilon}^{3} \frac{n_{5} l_{2} P_{y}}{l_{4} E + l_{3} P_{y}}, \qquad \qquad d_{\varepsilon}^{y} = -d_{\varepsilon}^{3} \frac{n_{5} l_{2} E}{l_{4} E + l_{3} P_{y}}.$$
(4.21)

By construction, the resulting states are polarized transverse to the gauge directions. As anticipated, they are built out of a linear combination of terms of spin j and  $j + \varepsilon$  (j' and  $j' + \varepsilon$ ). Moreover, at leading order in the large  $R_y$  expansion, the coefficients in the t, y directions go to zero, since  $E, P_y \sim O(1/R_y), l_{3,4} \sim O(R_y)$ , and  $l_2 \sim O(1)$ .

<sup>&</sup>lt;sup>43</sup>The coefficients  $c^{t,y}$  and  $d^{t,y}$  were reported in the letter [2] with a slightly different notation, related by  $c^{t,y}_{\text{there}} = c^{t,y}_{\varepsilon,\text{here}}/c^3_{\varepsilon}$ , and likewise for  $d^{t,y}$ .

#### 4.1.3 The unflowed R sector

We now describe the physical states in the R sector of the null-gauged model. The computation turns out to be more involved than in the NS sector, since the spin fields necessarily involve all  $\varepsilon$ -chiralities. As a consequence, we will not find a situation akin to (4.19) in which a subset of coefficients are exactly those of the global AdS<sub>3</sub> × S<sup>3</sup> operators. However, we will again show that in the AdS<sub>3</sub> limit the vertex operators will reduce to their global AdS<sub>3</sub> × S<sup>3</sup> counterparts.

We introduce  $AdS_3 \times S^3$  and  $\mathbb{R}_t \times S^1_u \times \mathbb{T}^4$  spin fields,<sup>44</sup>

$$S_{\varepsilon_1\varepsilon_2\varepsilon_3} = e^{\frac{i}{2}(\varepsilon_1H_1 + \varepsilon_2H_2 + \varepsilon_3H_3)}, \qquad S_{\varepsilon_6\varepsilon_4\varepsilon_5} = e^{\frac{i}{2}(\varepsilon_6H_6 + \varepsilon_4H_4 + \varepsilon_5H_5)}.$$
(4.22)

Recalling the definition of the  $AdS_3 \times S^3$  chirality  $\varepsilon$  and the mutual locality / chiral GSO projections in 9+1 and 10+2 dimensions, (2.145), (4.9), we substitute away  $\varepsilon_3$  and  $\varepsilon_6$  via

$$\varepsilon_3 = \varepsilon \varepsilon_1 \varepsilon_2$$
,  $\varepsilon_6 = \varepsilon \varepsilon_4 \varepsilon_5$ . (4.23)

The  $H_{4,5}$  exponentials are spectators under the action of Q, so the parameters  $\varepsilon_4$ ,  $\varepsilon_5$  will label the vertex operators. For fixed  $\varepsilon_4$ ,  $\varepsilon_5$ , we consider  $\varepsilon_6$  to be controlled by  $\varepsilon$  through the second equation in (4.23), and we will form linear combinations of different values over  $\varepsilon_1$ ,  $\varepsilon_2$ ,  $\varepsilon$ .

We work with vertex operators in ghost pictures  $(q_{\varphi}, q_{\tilde{\varphi}}) = (-\frac{1}{2}, +\frac{1}{2})$ , for which the  $\lambda$ -constraint is non-trival, while there is no need to worry about BRST-exact states. We thus make an ansatz for R sector vertices of the following form:

$$\mathcal{Y}^{\varepsilon_{4},\varepsilon_{5}} = e^{-(\varphi-\tilde{\varphi})/2} \sum_{\varepsilon_{1},\varepsilon_{2},\varepsilon} F^{\varepsilon}_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{4}\varepsilon_{5}} S_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{3}} \mathcal{S}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} V_{j,m-\frac{\varepsilon_{1}}{2}} V'_{j',m'-\frac{\varepsilon_{2}}{2}} e^{i(-Et+P_{y}y)} .$$
(4.24)

Note that the coefficients  $F_{\varepsilon_1\varepsilon_2\varepsilon_4\varepsilon_5}^{\varepsilon}$  are not determined by the representation theory of SL(2,  $\mathbb{R}$ )×SU(2), since the states will not in general have definite spin.

The *cT* and  $\tilde{c}J$  terms of the BRST operator Q (4.7) act as in the NS sector. Hence the unflowed, non-winding states of the R sector also satisfy both the Virasoro condition (4.11) and the bosonic null-gauge constraint (4.12).

Next, the  $e^{\tilde{\varphi}}\lambda$  term in Q leaves  $\varepsilon_{1,2,4,5}$  unchanged, so for this term we can treat  $\varepsilon_{1,2,4,5}$  as fixed, and focus on the sum over  $\varepsilon = \pm$ . The resulting constraints on  $F_{\varepsilon_1\varepsilon_2\varepsilon_4\varepsilon_5}^{\pm}$  form a two-dimensional homogeneous linear system, which is degenerate due to the null

<sup>&</sup>lt;sup>44</sup>The order of the spin fields in  $S_{\epsilon_6\epsilon_4\epsilon_5}$  has been chosen for convenience in order to reduce clutter in computations involving cocycle factors.

condition on the gauge parameters, Eq. (2.216). For each choice of  $\varepsilon_{1,2,4,5}$ , we have

$$(l_3 + \varepsilon_4 \varepsilon_5 l_4) F^-_{\varepsilon_1 \varepsilon_2 \varepsilon_4 \varepsilon_5} - i\sqrt{n_5} (1 - \varepsilon_1 \varepsilon_2 l_2) F^+_{\varepsilon_1 \varepsilon_2 \varepsilon_4 \varepsilon_5} = 0,$$
  

$$i\sqrt{n_5} (1 + \varepsilon_1 \varepsilon_2 l_2) F^-_{\varepsilon_1 \varepsilon_2 \varepsilon_4 \varepsilon_5} - (l_3 - \varepsilon_4 \varepsilon_5 l_4) F^+_{\varepsilon_1 \varepsilon_2 \varepsilon_4 \varepsilon_5} = 0.$$
(4.25)

These constraints halve the degrees of freedom. When  $|l_2| = 1$  (and so  $|l_3| = |l_4|$ ), some of the  $F_{\varepsilon_1\varepsilon_2\varepsilon_4\varepsilon_5}^{\varepsilon}$  get set to zero. For a given  $\varepsilon_{1,2,4,5}$ , when neither of  $F_{\varepsilon_1\varepsilon_2\varepsilon_4\varepsilon_5}^{\pm}$  get set to zero, their ratio  $F_{\varepsilon_1\varepsilon_2\varepsilon_4\varepsilon_5}^{-}/F_{\varepsilon_1\varepsilon_2\varepsilon_4\varepsilon_5}^{+}$  becomes determined. So the 32 d.o.f. remaining after imposing GSO in (10+2) dimensions have now become 16, corresponding to  $\varepsilon_{1,2,4,5}$  in our parameterization.

Let us pause to discuss how Eq. (4.25) behaves in the large  $R_y$  limit. We have  $l_2 \sim O(1)$ and generically  $|l_2| \neq 1$ , while  $l_3 + l_4 \sim O(R_y)$  and  $l_3 - l_4 \sim O(1/R_y)$ , from (3.29)– (3.32). When  $\varepsilon_4\varepsilon_5 = +1$ , we obtain  $F^+ \sim O(1)$  and  $F^- \sim O(1/R_y)$ , so at leading order in large  $R_y$  we obtain an operator of purely positive AdS<sub>3</sub> × S<sup>3</sup> chirality  $\varepsilon$ . Similarly, when  $\varepsilon_4\varepsilon_5 = -1$ , at leading order in large  $R_y$  we obtain a purely negative chirality operator. So we obtain operators of definite AdS<sub>3</sub> × S<sup>3</sup> chirality  $\varepsilon$ , with  $\varepsilon_4\varepsilon_5 = \varepsilon$ , exactly as in Section 2.6, see Eq. (2.165). This explains the sign choice of the (10+2)-d GSO projection (4.9). As before, one of  $\varepsilon_4$  or  $\varepsilon_5$  remains unfixed, say  $\varepsilon_4$ .

We now examine the action of  $e^{\varphi}G$  on the R vertex operator ansatz (4.24). This will reduce the remaining 16 degrees of freedom to the correct 8 physical polarizations in the holomorphic sector. It leads to the following set of equations (we suppress the  $\varepsilon_4$ ,  $\varepsilon_5$ subscripts on the RHS for ease of notation):

$$\mathcal{B}^{\varepsilon}_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{4}\varepsilon_{5}} \equiv \left(m + \varepsilon_{1}j - \frac{\varepsilon_{1}}{2}\right)F^{\varepsilon}_{(-\varepsilon_{1})\varepsilon_{2}} + i\varepsilon_{1}\varepsilon_{2}\left(j' - \varepsilon_{2}m' + \frac{1}{2}\right)F^{\varepsilon}_{\varepsilon_{1}(-\varepsilon_{2})} - (\varepsilon m + \varepsilon_{1}\varepsilon_{2}m')F^{\varepsilon}_{\varepsilon_{1}\varepsilon_{2}} + \frac{i\sqrt{n_{5}}}{2}\left(\varepsilon_{4}\varepsilon_{5}P - \varepsilon E\right)F^{(-\varepsilon)}_{\varepsilon_{1}\varepsilon_{2}} = 0.$$
(4.26)

Comparing to the  $AdS_3 \times S^3$  BRST condition, the only new term is the fourth and final one, proportional to  $F_{\epsilon_1\epsilon_2}^{(-\epsilon)}$ , which has the effect of mixing the  $\epsilon$  chiralities. The first three terms are unchanged from the  $AdS_3 \times S^3$  BRST condition, so the  $AdS_3 \times S^3$  limit of this condition is simply to drop the fourth term. Note that Eq. (4.25) implies that half of these equations are redundant, and allows us to decouple the  $F^+$  from the  $F^-$  coefficients. Moreover, by using the Virasoro constraint (4.11), the bosonic null-gauge condition (4.12), and the null constraint on the gauge parameters (2.216), one can show that for fixed  $\epsilon_4$  and  $\epsilon_5$ , actually only two equations are linearly independent. For generic values of quantum numbers, such that all denominators appearing below are nonzero, the linearly independent equations can be taken to be

$$F_{-+}^{+} = -i\frac{j'-m'+\frac{1}{2}}{j+m-\frac{1}{2}}F_{+-}^{+} + \frac{-j(j-1)+j'(j'+1)+m^2-m'^2}{(j+m-\frac{1}{2})\left[m-m'+\frac{l_4-\varepsilon_4\varepsilon_5l_3}{2(l_2-1)}\left(\varepsilon_4\varepsilon_5E-P_y\right)\right]}F_{++}^{+},$$

$$F_{--}^{+} = \frac{m-m'+\frac{l_4-\varepsilon_4\varepsilon_5l_3}{2(l_2-1)}\left(\varepsilon_4\varepsilon_5E-P_y\right)}{j+m-\frac{1}{2}}F_{+-}^{+} + i\frac{j'+m'+\frac{1}{2}}{j+m-\frac{1}{2}}F_{++}^{+}.$$
(4.27)

Alternatively, the two linearly independent equations can generically be taken to be

$$F_{-+}^{-} = -i\frac{j'-m'+\frac{1}{2}}{j+m-\frac{1}{2}}F_{+-}^{-} + \frac{-j(j-1)+j'(j'+1)+m^2-m'^2}{(j+m-\frac{1}{2})\left[-m-m'+\frac{n_5(l_2-1)}{2(l_4-\epsilon_4\epsilon_5l_3)}\left(\epsilon_4\epsilon_5E+P_y\right)\right]}F_{+-}^{-},$$

$$F_{--}^{-} = \frac{-m-m'+\frac{n_5(l_2-1)}{2(l_4-\epsilon_4\epsilon_5l_3)}\left(\epsilon_4\epsilon_5E+P_y\right)}{j+m-\frac{1}{2}}F_{+-}^{-} + i\frac{j'+m'+\frac{1}{2}}{j+m-\frac{1}{2}}F_{++}^{-}.$$
(4.28)

Let us pause again to check consistency with the  $AdS_3 \times S^3$  limit. Setting  $\varepsilon_4 \varepsilon_5 = \varepsilon$  and taking the large  $R_y$  limit of Eqs. (4.27), we indeed find that a solution is given by setting (the  $AdS_3 \times S^3$  limit of)  $F_{\varepsilon_1 \varepsilon_2}^{\varepsilon}$  to be equal to the values  $f_{\varepsilon_1 \varepsilon_2}^{\varepsilon}$  specified in Eqs. (2.166)–(2.168).

Similarly to  $AdS_3 \times S^3$ , Eqs. (4.27) are two equations for four unknowns, so for each  $\varepsilon_4$ ,  $\varepsilon_5$  there is a two-parameter family of solutions, which we take to be parameterized by the values of  $F_{\pm\pm}^+$ . If working with Eqs. (4.28), we take the two-parameter family of solutions to be parameterized by the values of  $F_{\pm\pm}^-$ . Together with  $\varepsilon_4$ ,  $\varepsilon_5$ , this gives 8 physical polarizations.

In Section 2.6, for  $AdS_3 \times S^3$  these unfixed coefficients were chosen such that the vertex operators transform appropriately under the action of the currents  $J^{\pm}$  and  $K^{\pm}$ . However, in the null-gauged worldsheet theory associated to the full asymptotically linear dilaton geometry this need not necessarily be the case.

In the cosets we fix these coefficients by requiring a reasonable IR limit. We treat the different  $\varepsilon$  chiralities separately. For  $\varepsilon = 1$  we set the particular components  $F_{+\pm}^+$  equal to their values in the AdS limit,  $F_{+\pm}^+ = f_{+\pm}^+$ . The rest of the coefficients are then obtained using Eqs. (4.27) and (4.25). Alternatively, for  $\varepsilon = -1$  we set  $F_{+\pm}^- = f_{+\pm}^-$  and again solve for the remaining coefficients using Eqs. (4.28) and (4.25).

We now turn to the analysis of the spacetime supercharges preserved by the null gauging, following on from the initial discussion around Eq. (4.8) (see also [54]). The supercharge analysis corresponds to the limit of the Ramond sector analysis in which we take  $j = j' = E = P_y = 0$ ,  $m = \frac{\varepsilon_1}{2}$ ,  $m' = \frac{\varepsilon_2}{2}$ , as can be seen by comparing Eqs. (4.8) and (4.24). In this limit, the center-of-mass wavefunction trivializes, and we are left with integrated vertex operators involving only the spin fields. As before, we parameterize the  $\varepsilon_i$  according to (4.23);  $\varepsilon_4$  and  $\varepsilon_5$  are spectators that will label the supercharges; and we will sum over  $\varepsilon$ ,  $\varepsilon_1$ ,  $\varepsilon_2$  as in (4.24). The *J* constraint (4.12) reduces to

$$\varepsilon_1 + \varepsilon_2 \, l_2 = 0 \qquad \Rightarrow \qquad \varepsilon_1 \varepsilon_2 = -l_2 \,, \tag{4.29}$$

so supersymmetry is preserved in the holomorphic sector if and only if  $|l_2| = 1$ , and thus  $|l_3| = |l_4|$ . The  $\gamma G$  constraint (4.26) reduces directly to

$$\varepsilon = -1$$
, (4.30)

so all supercharges have negative  $AdS_3 \times S^3$  chirality. Then the  $\lambda$  constraint in the first line of (4.25), with  $F^+ = 0$ , reduces to

$$l_3 + \varepsilon_4 \varepsilon_5 \, l_4 = 0 \,. \tag{4.31}$$

So when  $|l_2| = 1$  and  $|l_3| = |l_4| \neq 0$ , there are four holomorphic supercharges, labelled by say  $\varepsilon_2$  and  $\varepsilon_4$ .

Combining this analysis with the corresponding one in the antiholomorphic sector, we observe consistency with the passage below Eq. (3.82) describing which subset of the backgrounds are supersymmetric. In terms of the spacetime spectral flow parameters s,  $\bar{s}$  introduced in Eq. (2.89), we have  $l_2 = 2s + 1$ ,  $r_2 = 2\bar{s} + 1$ . The circular supertube backgrounds of [122, 123] have  $s = \bar{s} = 0$  and preserve supersymmetry in both holomorphic and antiholomorphic sectors; the backgrounds of [132, 124, 125, 96] have  $\bar{s} = 0$ ,  $s \neq 0$  and so preserve supersymmetry only in the antiholomorphic sector; the general JMaRT backgrounds [37] have s and  $\bar{s}$  both nonzero, and preserve no supersymmetry. As we will see in the next chapter, the absence of supersymmetry will not obstruct us in the computation of correlation functions in the AdS<sub>3</sub> limit. However, given that the embedding coefficients  $l_i$ ,  $r_i$  enter the bosonic null gauge constraints Eq. (4.12), the number of admissible BRST-invariant spectrum of excitations on a given background will change.

#### Picture changing in the R sector

In order to compute two-point functions of operators  $\mathcal{Y}$  in the Ramond sector of the gauged model, we need to define their picture-changed versions. Propagators will be non-vanishing only if the total ghost charges add up to  $-Q_{\varphi} = -2$  and  $-Q_{\tilde{\varphi}} = 0$ . The picture-changing operators are given by  $P_{+1} \sim e^{\varphi}G$  and  $\tilde{P}_{+1} \sim e^{\tilde{\varphi}}\lambda$ . One possible natural choice would be to compute the two-point function (superscripts denote  $(q_{\varphi}, q_{\tilde{\varphi}})$  charges)

$$\langle \mathcal{Y}^{\left(-\frac{3}{2},-\frac{1}{2}\right)}(z) \mathcal{Y}^{\left(-\frac{1}{2},+\frac{1}{2}\right)}(w) \rangle$$
 (4.32)

However, it turns out that looking for an explicit expression for the state  $\mathcal{Y}^{\left(-\frac{3}{2},-\frac{1}{2}\right)}$  is not the simplest way to go. This is due to the fact that such a state is automatically BRST closed, so that it must be determined by the somewhat cumbersome procedure

of removing all the BRST-exact contributions. To avoid this issue, one can distribute the ghost charges in a different way inside the correlator, and consider instead the equivalent two-point function

$$\langle \mathcal{Y}^{\left(-\frac{3}{2},+\frac{1}{2}\right)}(z) \mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right)}(w) \rangle .$$
 (4.33)

Here  $\mathcal{Y}^{\left(-\frac{3}{2},+\frac{1}{2}\right)}$  is in the canonical  $\tilde{\varphi}$ -picture, while  $\mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right)}$  is in the canonical  $\varphi$ -picture. Thus, although this forces us to compute two additional R-sector operators instead of only one, these are constrained by the  $\tilde{\gamma}\lambda$  and  $\gamma G$  BRST constraints, respectively. The procedure is then similar to that employed above to construct  $\mathcal{Y}^{\left(-\frac{1}{2},+\frac{1}{2}\right)}$ . We thus make the Ansätze

$$\mathcal{Y}^{\left(-\frac{3}{2},+\frac{1}{2}\right),\varepsilon_{4}\varepsilon_{5}} = e^{-\left(\frac{3}{2}\varphi-\frac{1}{2}\tilde{\varphi}\right)} \sum_{\varepsilon,\varepsilon_{1},\varepsilon_{2}} \mathcal{L}^{\varepsilon}_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{4}\varepsilon_{5}} S_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{3}} \mathcal{S}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} V_{j,m-\frac{\varepsilon_{1}}{2}} V'_{j',m'-\frac{\varepsilon_{2}}{2}} e^{i\left(-Et+P_{y}y\right)} \mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right),\varepsilon_{4}\varepsilon_{5}} = e^{-\left(\frac{1}{2}\varphi+\frac{1}{2}\tilde{\varphi}\right)} \sum_{\varepsilon,\varepsilon_{1},\varepsilon_{2}} \mathcal{G}^{\varepsilon}_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{4}\varepsilon_{5}} S_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{3}} \mathcal{S}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} V_{j,m-\frac{\varepsilon_{1}}{2}} V'_{j',m'-\frac{\varepsilon_{2}}{2}} e^{i\left(-Et+P_{y}y\right)} \mathcal{I}^{\varepsilon}_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{4}\varepsilon_{5}} S_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{3}} \mathcal{S}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} V_{j,m-\frac{\varepsilon_{1}}{2}} V'_{j',m'-\frac{\varepsilon_{2}}{2}} e^{i\left(-Et+P_{y}y\right)} \mathcal{I}^{\varepsilon}_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{4}\varepsilon_{5}} S_{\varepsilon_{1}\varepsilon_{2}\varepsilon_{3}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} V_{j,m-\frac{\varepsilon_{1}}{2}} V'_{j',m'-\frac{\varepsilon_{2}}{2}} e^{i\left(-Et+P_{y}y\right)} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{4}\varepsilon_{5}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}\varepsilon_{6}\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon_{6}} \mathcal{I}^{\varepsilon}_{\varepsilon} \mathcal{I$$

where again  $\varepsilon_3$ ,  $\varepsilon_6$  are substituted away using (4.23). These must satisfy

$$\oint dz : e^{\tilde{\varphi}} \boldsymbol{\lambda} : (z) \mathcal{Y}^{\left(-\frac{3}{2},+\frac{1}{2}\right), \varepsilon_4 \varepsilon_5}(w) = 0,$$

$$\oint dz : e^{\varphi} G : (z) \mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right), \varepsilon_4 \varepsilon_5}(w) = 0,$$
(4.34)

and

$$: e^{\varphi}G:(z) \mathcal{Y}^{\left(-\frac{3}{2},+\frac{1}{2}\right),\varepsilon_{4}\varepsilon_{5}}(w) = \mathcal{Y}^{\left(-\frac{1}{2},+\frac{1}{2}\right),\varepsilon_{4}\varepsilon_{5}}(w),$$

$$: e^{\tilde{\varphi}}\lambda:(z) \mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right),\varepsilon_{4}\varepsilon_{5}}(w) = \mathcal{Y}^{\left(-\frac{1}{2},+\frac{1}{2}\right),\varepsilon_{4}\varepsilon_{5}}(w).$$
(4.35)

By solving the above constraints, all the coefficients  $\mathcal{L}^{\varepsilon}$  and  $\mathcal{G}^{\varepsilon}$  can be expressed explicitly in terms of the  $F^{\varepsilon}$  coefficients in Eq. (4.27).

In addition, one can explicitly check that in the AdS limit they correctly reproduce the expected behaviour. From the definition of the corresponding coset states  $\mathcal{Y}^{\left(-\frac{3}{2},+\frac{1}{2}\right)}$  and  $\mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right)}$ , one might reasonably expect that they would reduce to the states  $\mathcal{Y}^{\left(-\frac{3}{2}\right)}$  and  $\mathcal{Y}^{\left(-\frac{1}{2}\right)}$  of Section 2.6.4.2 respectively. However, care is needed when comparing both chiralities and normalisations in the UV and IR. To explain this, let us consider for instance  $\mathcal{Y}_{A}^{\left(-\frac{1}{2}\right)}$ , whose  $\varepsilon$ -chirality is  $\varepsilon = +1$ . (Analogous comments hold for other operators.) First of all, recall that already in the case of  $\mathrm{AdS}_3 \times \mathrm{S}^3 \times \mathrm{T}^4$ , the picture-changing operator induces a change in chirality of the state. Indeed, in case "A" of the analysis in (2.166), the physical states in the " $-\frac{3}{2}$ " picture have *negative*  $\varepsilon$ -chirality.

In the full coset models, an analogous pattern occurs with the two picture-changing operators  $P_{+1}$ ,  $\tilde{P}_{+1}$ . The coset state  $\mathcal{Y}^{\left(-\frac{3}{2},+\frac{1}{2}\right)}$  that correctly reduces to  $\mathcal{Y}^{\left(-\frac{3}{2}\right),\varepsilon=-1}_{A}$  in the AdS limit indeed has *positive*  $\varepsilon$ -chirality. This means that, in our case under study, the

coefficients  $\mathcal{L}_{\epsilon_1\epsilon_2}^+$  reduce to  $-\frac{\sqrt{n_5}}{j+j'}f_{\epsilon_1\epsilon_2}^-$ . Similarly, the coset state  $\mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right)}$  with negative  $\epsilon$ -chirality reduces to the  $\mathcal{Y}_A^{\left(-\frac{1}{2}\right)}$  state. However, in the latter case there is a normalisation factor  $(i \, kR_y)^{-1}$  to account for. This is removed by taking into account the normalisation of the picture-changing operator, which indeed contains a term which is dominant in the AdS limit,  $l_3\lambda_t + l_4\lambda_y \sim (kR_y)(\lambda_t + \lambda_y)$ .

We illustrate the example of the  $\mathcal{G}^-_{++}$  coefficient, appearing in the coset state  $\mathcal{Y}^{\left(-\frac{1}{2},-\frac{1}{2}\right)}$ . The argument holds analogously for all the other coefficients. The explicit expression of  $\mathcal{G}^-_{++}$  in terms of positive coefficients  $F^+_{\varepsilon_1\varepsilon_2}$  is obtained by solving the constraints Eq. (4.34) and Eq. (4.35), without using Eq. (4.11) and Eq. (4.12). For generic quantum numbers such that the denominators below are non-zero, one finds

$$\frac{(l_3+l_4)^2}{i(E-P_y)n_5(1+l_2)} \mathcal{G}_{++}^{-}$$

$$= \frac{(m+j-\frac{1}{2})F_{-+}^{+} + i(j'-m'+\frac{1}{2})F_{+-}^{+} - \left(\frac{2(j-j'+1)(j+j')(l_3+l_4)}{(E-P_y)n_5(1+l_2)} + \frac{(-1+l_2)}{(1+l_2)}(m-m')\right)F_{++}^{+}}{\left(-j(j-1)+j'(j'+1) + n_5\frac{(m+l_2m')(E-P_y)}{l_3+l_4} + n_5\frac{n_5(-1+l_2^2)(E-P_y)^2}{4(l_3+l_4)^2}\right)},$$
(4.36)

and thus for  $R_y \gg 1$  one has  $\mathcal{G}_{++}^- \simeq (i k R_y)^{-1} f_{++}^+ + \mathcal{O}(R_y^{-2})$ , as claimed above.

The expressions for the coefficients  $\mathcal{G}, \mathcal{L}$  are quite lengthy, and we leave the computation of correlators in the full coset model for future work. Nevertheless, it is easy to check that in all cases the coefficients reduce to the expected expressions when going into the IR regime. Consequently, we find that, to leading order in  $R_y$ , the coset two-point functions in the RR sector reproduce the *m*-basis expressions in Eq. (2.171), as they should. As will become clear in Section 4.2 below, this does *not* mean that the physics in the IR regime of the coset model is that of global  $AdS_3 \times S^3 \times \mathbb{T}^4$ ; the bosonic null gauge condition (4.12) will lead to substantially different correlators in the appropriately defined *x*-basis.

#### 4.1.4 Flowed/winding sectors

We now briefly discuss the states with non-trivial spectral flow we will be interested in, that is, those corresponding to the description of the higher-weight chiral primaries described in Section 2.6.

A generic state with excitation numbers  $(\frac{1}{2}, \frac{1}{2})$  in the null-gauged worldsheet theory must satisfy the the  $L_0$  and  $\bar{L}_0$  Virasoro constraints

$$0 = \frac{j'(j'+1) - j(j-1)}{n_5} - m\omega + m'\omega' + \frac{n_5}{4}\left(\omega'^2 - \omega^2\right) - \frac{1}{4}\left(E^2 - P_y^2\right), \quad (4.37a)$$

$$0 = \frac{j'(j'+1) - j(j-1)}{n_5} - \bar{m}\omega + \bar{m}'\bar{\omega}' + \frac{n_5}{4}\left(\bar{\omega}'^2 - \omega^2\right) - \frac{1}{4}\left(E^2 - \bar{P}_y^2\right), \quad (4.37b)$$

where

$$P_y = \frac{n_y}{R_y} - \omega_y R_y$$
,  $\bar{P}_y = \frac{n_y}{R_y} + \omega_y R_y$ , (4.38)

and we recall that  $\omega_y \in \mathbb{Z}$  is the winding on the *y*-circle. The level-matching  $L_0 - \bar{L}_0$  constraint thus reads

$$0 = \omega(\bar{m} - m) + m'\omega' - \bar{m}'\bar{\omega}' + \frac{n_5}{4}(\omega'^2 - \bar{\omega}'^2) - n_y\omega_y.$$
(4.39)

Let us try to follow the global AdS<sub>3</sub> procedure as close as possible, and consider states with  $\omega = \omega' = \bar{\omega}'$  and  $m = m' = \bar{m} = \bar{m}'$ . This will be general enough to describe all coset states corresponding to chiral primary operators of the spacetime theory. Then, (4.39) forces us to set  $\omega_y = 0$ . Moreover, for states constructed by spectrally flowing highest/lowest weight primaries, the discussion around Eq. (2.160) shows that the  $e^{\varphi}G$ part of the BRST charge acts as in the unflowed case. Moreover, the action of  $e^{\bar{\varphi}\lambda}$  is left unchanged. The derivation of the coefficients involved in the definition of the vertex operators constructed above then goes through without changes, and we only need to restrict to the highest (lowest) possible values of m (m') in each case.

Regarding the gauge constraints, recall that, both in SL(2, $\mathbb{R}$ ) and in SU(2), the different modes of the spectrally flowed operators are obtained by acting with the raising/lowering operators  $J_0^{\pm}$  and  $K_0^{\pm}$  on the flowed primary. Although this does not lead to operators that can be expressed in a simple way in the *m*-basis, the presence of these different modes is crucial in order to obtain the set of physical modes that satisfy the gauge constraints. Focusing on (lowest-weight) discrete states corresponding to operators of spacetime weight *h* in the chiral multiplets, this gives modes described by worldsheet operators with projections  $m_{\omega} = J + \frac{n_5}{2}\omega + n = h + n$  and  $m'_{\omega} = J' + \frac{n_5}{2}\omega - n'$ , with  $n, n' \in \mathbb{N}_0$ , and similarly in the antiholomorphic sector. The bosonic gauge constraints now read

$$0 = m_{\omega} + l_2 m'_{\omega} + \frac{l_3}{2}E + \frac{l_4}{2}P_y , \qquad 0 = \bar{m}_{\omega} + r_2 \bar{m}'_{\omega} + \frac{r_3}{2}E + \frac{r_4}{2}P_y . \qquad (4.40)$$

As discussed below Eq. (4.13), this implies that  $J_0^{\pm}$  and  $K_0^{\pm}$  do not commute with the BRST charge. Consequently and importantly, only the subset of modes satisfying Eqs. (4.40) will be physical. The implications will be discussed at length in the following section. Combining the gauge constraints with the conditions discussed below Eq. (4.39) is then sufficient to obtain the set of BRST invariant states we are interested in.

Before concluding this section, let us review the fact that there is a residual discrete gauge symmetry in these models, which implies that operators related by shifts of the following form describe the same physical state [52], see also [1]. Parameterizing ( $l_4$ ,  $r_4$ ) through p and k as in (3.31), the symmetry is

$$\delta(\omega, \omega', \bar{\omega}', E, n_y, \omega_y) = (1, -l_2, -r_2, l_3, -p, -k).$$
(4.41)
In particular, one can trade a unit of SL(2, $\mathbb{R}$ ) spectral flow (i.e. shift  $\omega$  by  $\delta \omega = -1$ ) for k units of *y*-winding  $\omega_y$ , together with the corresponding other shifts implied by Eq. (4.41). In particular, the energy acquires a term linear in  $R_y$ , namely  $\delta E = kR_y + \mathcal{O}(1/R_y)$ . The origin of the factor k relating  $\omega$  and  $\omega_y$  can be traced back to the  $\mathbb{Z}_k$  orbifold appearing in the IR, see Eq. (3.81). It reflects the fact that the CFT state associated with the background lives in the k-twisted sector of the D1D5 CFT.

The operators discussed in this section do not exhaust the spectrum of the worldsheet model; for instance we have not discussed operators that do not not satisfy  $\omega_y \equiv 0 \mod k$ , which were analyzed in [52]. However, the operators described above comprise a large set of light operators in spectral flowed sectors, in parallel to the analysis of global AdS<sub>3</sub> × S<sup>3</sup>, which will be general enough for our purposes in the present work.

# 4.2 Novel heavy-light correlators from the worldsheet

In this section we describe the computation of two-point correlators in the null gauged models, corresponding to HLLH correlators of the holographic CFT. To do so, we take a set of physical coset operators derived in the previous section and flow them to the IR, in which the geometry is locally an orbifold of  $AdS_3 \times S^3$ . We develop a proposal to define coset operators in an appropriate *x*-basis corresponding to local operators of the holographic CFT. We then use this definition to compute a large set of HLLH correlators. We observe precise agreement between a subset of these and known results computed in supergravity and holographic CFT, and significantly extend these results.

#### 4.2.1 Light states in the AdS<sub>3</sub> regime

We begin by describing in more detail the vertex operators of the null-gauged model in the AdS<sub>3</sub> limit. We send  $R_y \rightarrow \infty$ , keeping  $\tilde{t} = t/R_y$  and  $\tilde{y} = y/R_y$  fixed. After choosing the gauge  $\tau = \sigma = 0$ , this leads to a geometry described by the six-dimensional metric (3.81), which is related to  $\mathbb{Z}_k$ -orbifolded AdS<sub>3</sub> × S<sup>3</sup> × T<sup>4</sup> by the large coordinate transformation Eq. (3.82).

We focus initially on light states with no winding or worldsheet spectral flow. As we have argued in the previous section, the different polarizations and the associated coefficients simply reduce to those described in Section 2.6 in the  $AdS_3$  limit. Here we further describe what happens to their quantum numbers in the regime of interest. In general, the Virasoro condition (4.11) determines *j* via the solution

$$j = \frac{1}{2} + \sqrt{\left(j' + \frac{1}{2}\right)^2 + \frac{n_5}{4}\left(P_y^2 - E^2\right)},$$
(4.42)

where we have fixed the sign in order to have *j* in the range (2.106). As  $R_y \rightarrow \infty$ , we hold fixed the rescaled energy and momentum

$$\mathcal{E} = ER_y , \qquad n_y = P_y R_y . \tag{4.43}$$

Hence, the second term inside the square root in (4.42) is  $\mathcal{O}\left(1/R_y^2\right)$ , and at large  $R_y$  the solution becomes  $j = j' + 1 + \mathcal{O}\left(1/R_y^2\right)$ , which to leading order is the usual AdS<sub>3</sub> × S<sup>3</sup> relation. The  $\mathcal{O}\left(1/R_y^2\right)$  corrections to j are non-zero when  $|\mathcal{E}| \neq |n_y|$ , which is generically the case. To see this, note that at large  $R_y$  the gauging parameters associated to the t and y directions become

$$l_3 = r_3 = l_4 = -r_4 = -kR_y + \mathcal{O}(1/R_y).$$
(4.44)

On the other hand, those associated to the S<sup>3</sup> angular directions do not scale with  $R_y$ , and remain  $l_2 = 2s + 1$  and  $r_2 = 2\bar{s} + 1$ . Hence, at leading order at large  $R_y$ , Eqs. (4.12) take the form

$$0 = m + (2s+1) m' - \frac{k}{2} \left(\mathcal{E} + n_y\right) , \qquad 0 = \bar{m} + (2\bar{s}+1) \bar{m}' - \frac{k}{2} \left(\mathcal{E} - n_y\right) , \quad (4.45)$$

which fix  $\mathcal{E}$  and  $n_y$  in terms of m, m',  $\bar{m}$ ,  $\bar{m}'$ , such that indeed generically  $\mathcal{E} \neq \pm n_y$ .

Although for simplicity we restricted to light states with no winding or worldsheet spectral flow in Eqs. (4.42) and (4.45), the present discussion and the computations in the rest of this section are analogous for winding states after replacing the projections  $m \rightarrow m_{\omega}$ , etc, where  $m_{\omega}$  was defined above Eq. (4.40). Our results will be valid for the full set of chiral primaries that can be described within the usual  $AdS_3 \times S^3 \times \mathbb{T}^4$  worldsheet theory, as well as their descendants under the global part of the chiral algebra, that fill out the short multiplet. We shall comment further on states with non-trivial winding and/or worldsheet spectral flow in due course.

#### 4.2.2 Identifying the spacetime modes

Let us discuss the identification of the spacetime modes. We shall work in a gauge in which the upstairs SL(2, $\mathbb{R}$ ) time  $\tau$  and angular direction  $\sigma$  are fixed. Then, importantly, the asymptotic boundary of the physical downstairs AdS<sub>3</sub> is parameterized by  $t/R_y$  and  $y/R_y$ , at a fixed point on the S<sup>3</sup>. We therefore define

$$m_y = \frac{1}{2} \left( \mathcal{E} + n_y \right) , \qquad \bar{m}_y = \frac{1}{2} \left( \mathcal{E} - n_y \right) , \qquad (4.46)$$

and we interpret these as the asymptotic mode labels. We will see that this gives rise to a rich set of correlators that agree with and extend previous results.

Let us first consider k = 1, and continue to focus mainly on the holomorphic sector. Given a holographic CFT chiral primary with spacetime weight h and definite (left) R-charge m', we wish to construct the dual worldsheet operator by summing over the corresponding modes. As reviewed in Section (2.6), in global AdS<sub>3</sub> × S<sup>3</sup> this leads to x-basis operators of the form  $\mathcal{O}_{h,m'}(x) = \sum_m x^{m-h} \mathcal{O}_{h,m,m'}$ , where the modes  $\mathcal{O}_{h,m,m'}$  are identified as either  $\mathcal{W}_{j,m,j',m'}$ ,  $\mathcal{X}_{j,m,j',m'}$  or  $\mathcal{Y}_{j,m,j',m'}$ , where j and h are related by Eq. (2.180). For simplicity, we collect all of these modes under the notation  $\mathcal{V}_{j,m,\bar{m}}\mathcal{V}'_{j',m',\bar{m}'}$ . In the null-gauged models, we replace  $m \to m_y$  in the exponent of x in the sum, defining the worldsheet operator

$$\mathcal{O}_{h,m'}(x,\bar{x}) \equiv \frac{1}{\mathsf{k}^{h+\bar{h}}} \sum_{m_y,\bar{m}_y} x^{m_y-h} \bar{x}^{\bar{m}_y-\bar{h}} \mathcal{V}_{j,m,\bar{m}} \mathcal{V}'_{j',m',\bar{m}'} e^{-im_y(\bar{t}-\bar{y})} e^{-i\bar{m}_y(\bar{t}+\bar{y})} , \quad (4.47)$$

where we have temporarily included the antiholomorphic dependence to emphasize the coupling between the left- and right-moving sectors due to the null gauge constraints, Eq. (4.12). The normalization factors of k have been introduced for later convenience and will be discussed below Eq. (4.81). We emphasize that, from the point of view of the worldsheet theory, x is an auxiliary complex variable, while  $\tilde{t}$  and  $\tilde{y}$  are scalar fields.

Note that combining the bosonic null constraint (4.45) and the definition of the modes (4.46), we obtain  $m_y = m + (2s + 1)m'$ . This relation parallels the supergravity spectral flow large gauge transformation (3.82), and we will make use of this observation when discussing the relation to the holographic CFT in due course.

For k > 1 we follow the same logic, and make the same definition (4.47). This time however, combining Eqs. (4.45) and (4.46) we obtain

$$m_y = \frac{1}{k} \left( m + (2s+1)m' \right), \qquad \bar{m}_y = \frac{1}{k} \left( \bar{m} + (2\bar{s}+1)\bar{m}' \right).$$
 (4.48)

We shall shortly see that this gives rise to an important technical complication relating the holomorphic and antiholomorphic sectors.

Before computing our first example of a HLLH correlator, let us briefly return to operators with non-zero spectral flow and/or winding charge. In Section 4.1.4 we analyzed a set of coset vertex operators in sectors with non-zero worldsheet spectral flow, corresponding to chiral primaries of higher twists, similar to those in global  $AdS_3 \times S^3$ . In that section we worked with  $\omega_y = 0$ , and reviewed that the large gauge spectral flow transformation (4.41) relates these to operators with  $\omega_y \in k\mathbb{Z}$ .

When  $\omega_y \neq 0$ , in general one should be careful both when examining whether the states survive the AdS<sub>3</sub> limit, and also when defining the AdS<sub>3</sub> energy  $\mathcal{E}$  and angular momentum  $n_y$ , since fundamental string *y*-winding charge can be exchanged for background flux [52]. However, for operators which have  $\omega_y \in k\mathbb{Z}$ , the situation is

straightforward: we shall simply use gauge spectral flow to always work in a frame in which  $\omega_y = 0$ , and then use the definitions of  $\mathcal{E}$  and  $n_y$  in (4.43) and the modes  $m_y$ ,  $\bar{m}_y$  in (4.46).

The discussion becomes more complicated when considering global SL(2,  $\mathbb{R}$ )×SU(2) descendants of the spectrally flowed primaries. The isomorphism between the affine modules gives a simple identification for the highest/lowest weight states, but this structure becomes more complicated for rest of the multiplet. Indeed, global descendants of the spectrally flowed affine primary state are identified with affine/Virasoro descendants of the corresponding unflowed equivalent state with non-trivial *y*-winding, such that one needs to include string oscillator excitations. The situation is similar to what happens for the usual series identification  $V_{j,j}^{\omega=0} \sim V_{\frac{k}{2}-j,j-\frac{k}{2}}^{\omega=1}$  in bosonic SL(2,  $\mathbb{R}$ ), where, for instance, one has  $V_{j,m}^{\omega=0} \sim (j_0^+)^{m-j}V_{j,j}^{\omega=0} \sim (j_{-1}^+)^{m-j}V_{\frac{k}{2}-j,j-\frac{k}{2}}^{\omega=1}$ , see [54]. We leave a more detailed exploration of these features in the coset models for future work. In the remainder of this chapter we will work with operators that have  $\omega_y = 0$ .

#### 4.2.3 HLLH correlators: first example

We now compute a first explicit example of a worldsheet two-point function in the cosets corresponding to the heavy backgrounds under consideration. For this purpose, we focus on a particular light operator probing the backgrounds with  $\bar{s} = 0$  (hence  $r_2 = 1$ ), but general s. These describe supersymmetric spectral flowed supertubes [132, 124, 125, 37, 96]. We shall demonstrate that this worldsheet correlator agrees with both the supergravity and symmetric product orbifold CFT HLLH fourpoint functions computed in [167]. We will also significantly extend beyond the set of correlators computed in [167].

The light operator in this first example is a massless RR operator of  $\mathcal{Y}_{[A]}$  type with h = m' = 1/2 and hence j = 1, see Eq. (2.172). We shall denote this operator by  $\mathcal{Y}_{\frac{1}{2}}$ . Together with the analogous antiholomorphic part, this vertex operator is dual to a particular  $(h, \bar{h}) = (\frac{1}{2}, \frac{1}{2})$  chiral primary of the HCFT denoted by  $O^{++}$ , which we introduced in Eq. (2.92). In the six-dimensional supergravity arising from reduction on  $\mathbb{T}^4$ , this corresponds to a particular combination of fluctuations of a scalar and anti-self-dual two-form potential, in a tensor multiplet that is not turned on in the backgrounds we consider. In type IIB supergravity, in the present NS5-F1-P duality frame, these fields correspond to a particular combination of supergravity fluctuations of the RR axion and certain components of the RR two-form and four-form potentials. (In the D1-D5-P frame, the fluctuations are of the RR axion and certain components of the RR four-form and NSNS two-form potentials). The light operator corresponds to a particular scalar spherical harmonic of these fields [167, 139, 166].

In the large  $R_y$  limit, we have shown that the  $\mathcal{Y}$  operators simply reduce to their  $\operatorname{AdS}_3 \times \operatorname{S}^3$  cousins of Section 2.6, times the additional exponentials in t and y. These exponentials give trivial contributions to the two-point functions  $\langle \mathcal{Y}_{\frac{1}{2}} \mathcal{Y}_{\frac{1}{2}} \rangle$  once the charge conservation conditions  $m_{y,1} = -m_{y,2}$  and  $\bar{m}_{y,1} = -\bar{m}_{y,2}$  are imposed. So do the SU(2) parts of the vertex operators for  $m'_1 = -m'_2$ , upon using the appropriate normalization. Hence, at the level of the two-point function of m-basis operators, i.e. the mode correlators from the spacetime point of view, the only non-trivial contribution comes from the SL(2,  $\mathbb{R}$ ) part of the upstairs theory.

To illustrate this more explicitly and also more generally, we introduce the following notation:  $\mathcal{O}_{h,m'}$  will denote a generic massless worldsheet vertex operator in the AdS limit of the null-gauged model with spacetime weight *h* and spacetime R-charge *m'*. When m' = h, we shall suppress the label *m'* and write  $\mathcal{O}_h$ . The corresponding holographic CFT chiral primaries will be denoted by  $O_h$ . Let us then consider the following worldsheet correlator,

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h_1, m_1'}(x_1, z_1) \mathcal{O}_{h_2, m_2'}(x_2, z_2) | s, \bar{s}, \mathsf{k} \rangle$$

$$\equiv \frac{1}{\mathsf{k}^{2h_1 + 2h_2}} \sum_{m_{y,i}, \bar{m}_{y,i}} \prod_{i=1,2} x_i^{m_{y,i} - h_i} \bar{x}_i^{\bar{m}_{y,i} - h_i} \lim_{R_y \to \infty} \langle \mathcal{V}_1(z_1) \mathcal{V}_2(z_2) \rangle ,$$
(4.49)

where  $\mathcal{V}$  denotes a generic *m*-basis massless vertex operator of the full coset model. The spacetime correlator corresponds to the worldsheet-integrated version of (4.49).

As discussed below (2.185), the normalization of the vertex operators is chosen so that the overall factors coming from the worldsheet integration procedure cancel out. After setting  $h_1 = h_2 = h$ ,  $x_1 = 1$  and  $x_2 = x$ , computing the free-field correlators, and imposing the various charge conservations, the integrated correlator becomes

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h,m'}(1) \mathcal{O}_{h,m'}^{\dagger}(x) | s, \bar{s}, \mathsf{k} \rangle = \frac{1}{\mathsf{k}^{4h}} \sum_{m_y, \bar{m}_y} x^{m_y - h} \bar{x}^{\bar{m}_y - h} \lim_{R_y \to \infty} \langle \mathcal{V}_1 \mathcal{V}_2 \rangle, \qquad (4.50)$$

where  $\langle V_1 V_2 \rangle$  stands for the *m*-basis two-point function with the *z*-dependence stripped out, c.f. Eqs. (2.154), (2.171). Note that on the right-hand side, the sum over  $m_y$  involves the R-charge and other quantum numbers of the operator located at *x*.

In turn, the remaining correlator is particularly simple in our context. As argued around Eq. (2.171a), the *m*-basis two-point functions of worldsheet chiral primary operators reduce to the Gamma functions expression in the bulk term of (2.114) with the replacement  $j \mapsto J = h$ . These are simply the coefficients obtained by Mellin-transforming the usual propagator  $|1 - x|^{-4h}$ , which further all become equal to one when  $h = \frac{1}{2}$ . Thus, for  $\mathcal{Y}_{\frac{1}{2}}$  in the  $\bar{s} = 0$  backgrounds, we obtain

$$\langle s, \mathsf{k} | \mathcal{Y}_{\frac{1}{2}}(1) \mathcal{Y}_{\frac{1}{2}}^{\dagger}(x) | s, \mathsf{k} \rangle = \frac{1}{\mathsf{k}^2} \sum_{m_y, \bar{m}_y} x^{m_y - \frac{1}{2}} \bar{x}^{\bar{m}_y - \frac{1}{2}}.$$
(4.51)

To be fully explicit, we take the operator at  $x_2 = x$  to be a discrete series state  $\mathcal{D}_j^+$  corresponding to an anti-chiral primary which, having set  $J = h = \bar{h}$ , has m = h + n,  $\bar{m} = h + \bar{n}$  and  $m' = \bar{m}' = -h$ , where  $n, \bar{n}$  are non-negative integers. For the supersymmetric backgrounds with  $\bar{s} = 0$  and setting  $h = \frac{1}{2}$ , the relation (4.48) becomes

$$m_y = \frac{n-s}{\mathsf{k}}, \qquad \bar{m}_y = \frac{\bar{n}}{\mathsf{k}}. \tag{4.52}$$

Hence, the correlator takes the form

$$\langle s, \mathsf{k} | \mathcal{Y}_{\frac{1}{2}}(1) \mathcal{Y}_{\frac{1}{2}}^{\dagger}(x) | s, \mathsf{k} \rangle = \frac{1}{\mathsf{k}^2} \sum_{n, \bar{n}} x^{\frac{n-s}{\mathsf{k}} - \frac{1}{2}} \bar{x}^{\frac{\bar{n}}{\mathsf{k}} - \frac{1}{2}}, \qquad (4.53)$$

where we have denoted the sum with a prime because the range of summation over  $n, \bar{n}$  is constrained. We now determine this constraint. Subtracting the two equations in (4.52) and comparing with (4.46), we obtain

$$m_y - \bar{m}_y = \frac{1}{k}(n - \bar{n} - s) = n_y$$
  
$$\Rightarrow n - \bar{n} = kn_y + s.$$
(4.54)

Thus the allowed values of  $n - \bar{n}$  are constrained by  $n_y \in \mathbb{Z}$ . To see this in detail, let us first write  $s = kp + \hat{s}$  with  $0 \le \hat{s} < k$  and  $p \in \mathbb{N}$ . For convenience here and later, we define  $a \equiv k - \hat{s}$ , so that s = kp - a, and  $1 \le a \le k$ . Then the sum in (4.53) is restricted to be over non-negative integers  $n, \bar{n}$  satisfying

$$\bar{n} - n \equiv a \mod \mathsf{k}. \tag{4.55}$$

We now demonstrate that the correlator (4.53), (4.55) agrees precisely with the supergravity and orbifold CFT expressions derived in [167]. In the holographic CFT, we have a HLLH four-point function  $\langle O_H(x_3)O_L(x_1)O_L^{\dagger}(x_2)O_H^{\dagger}(x_4)\rangle$ . Using Möbius symmetry we set  $x_3 = 0$  and  $x_4 \rightarrow \infty$ , in which case the heavy operators are interpreted as in/out states which we similarly denote as  $|s, k\rangle$ , such that the four-point function becomes a two-point function in the heavy background. We further set  $x_1 = 1$ , while  $x_2 = x$ parametrizes the usual cross-ratio. Then the supergravity result of [167, Eq. (4.25)] is

$$\langle s, \mathsf{k} | O_{\frac{1}{2}}(1) O_{\frac{1}{2}}^{\dagger}(x) | s, \mathsf{k} \rangle = \frac{x^{(\hat{s}-s)/\mathsf{k}}}{|x||1-x|^2} \frac{1-|x|^{2(1-\hat{s}/\mathsf{k})} + \bar{x}\left(|x|^{-2\hat{s}/\mathsf{k}} - 1\right)}{1-|x|^{2/\mathsf{k}}} , \qquad (4.56)$$

where the overall normalization of the supergravity amplitude was not fixed in [167]. Eq (4.56) was further shown to coincide with the corresponding symmetric orbifold CFT calculation for the particular cases  $\hat{s} = 0$  and  $\hat{s} = k - 1$ .

To demonstrate agreement between (4.53) and (4.56), it is easier to work from (4.56) towards our expression (4.53). We start by rewriting (4.56) in terms of sums akin to those involved in the definition of the *x*-basis, i.e. the mode expansion of local operators

in spacetime as seen from the worldsheet theory. Recalling that  $\hat{s} = k - a$ , the correlator becomes

$$\langle s, \mathsf{k} | O_{\frac{1}{2}}(1) O_{\frac{1}{2}}^{\dagger}(x) | s, \mathsf{k} \rangle = \frac{x^{1-p}}{|x||1-x|^2} \frac{1-|x|^{2a/\mathsf{k}} + \bar{x}\left(|x|^{2a/\mathsf{k}-2} - 1\right)}{1-|x|^{2/\mathsf{k}}} \\ = \frac{x^{-p}}{|x|} \left[ \frac{1}{|1-x|^2} \frac{1-|x|^2}{1-|x|^{2/\mathsf{k}}} - \frac{1}{1-\bar{x}} \frac{1-|x|^{2a/\mathsf{k}}}{1-|x|^{2/\mathsf{k}}} \right].$$
(4.57)

Assuming |x| < 1, the RHS can then be expressed as

$$\sum_{\hat{n}=0}^{\infty} \left[ \sum_{\hat{n}=0}^{\infty} \sum_{\delta=0}^{k-1} -\delta_{\hat{n},0} \sum_{\delta=0}^{a-1} \right] x^{\hat{n}+\frac{\delta}{k}-p-\frac{1}{2}} \bar{x}^{\hat{n}+\frac{\delta}{k}-\frac{1}{2}} .$$
(4.58)

The second term in (4.58) is understood as subtracting the  $\hat{n} = 0$  and  $\delta = 0, ..., a - 1$  coming from the first term. As a consequence, we can further rewrite the sum over  $\hat{n}, \hat{n}$  and  $\delta$  in Eq. (4.58) as a restricted double sum,

$$\sum_{n,\bar{n}} x^{\frac{n-s}{k} - \frac{1}{2}} \bar{x}^{\frac{\bar{n}}{k} - \frac{1}{2}} , \qquad (4.59)$$

over pairs of non-negative integers  $(n, \bar{n})$  satisfying the following restrictions

$$\bar{n} - n \equiv a \mod k$$
,  $n, \bar{n} \in \mathbb{N}_0$ . (4.60)

Indeed, for a = k we simply parametrize  $n = k\hat{n} + \delta$  and  $\bar{n} = k\hat{n} + \delta$ , which enforces the mod k condition (4.60), so that the sum gives the first term in (4.58). On the other hand, for a < k we can take  $n = k\hat{n} + \delta - a$  and  $\bar{n} = k\hat{n} + \delta$ , so that the factors of a coming from s and n cancel each other out in the exponent. However, in this case we need to explicitly subtract the contribution of all pairs  $(\hat{n}, \delta)$  which lead to n < 0, thus again giving (4.58). Therefore we observe agreement between (4.53), (4.55) and (4.56), up to the overall normalization that was not fixed in the supergravity calculation of [167]. Here and throughout the chapter, we shall not keep track of overall normalization factors.

For  $\hat{s} = 0$  and  $\hat{s} = k - 1$ , since Eq. (4.56) matches the corresponding symmetric orbifold CFT correlator, we have demonstrated an explicit match between the worldsheet and symmetric orbifold CFT correlators. This is striking, as it is an agreement across moduli space for a correlator that a priori is not covered by an existing non-renormalization theorem.

This agreement is almost certainly due to the special nature of the heavy states we consider. Indeed, let us compare the worldsheet *x*-basis operator Eq. (4.47) with the discussion of holographic CFT spectral flow in [93, App. A]. Spectral flow in the holographic CFT is an automorphism of the small (4, 4) superconformal algebra, that is a useful tool to relate different states and operators. For instance, the heavy backgrounds

we consider are related by fractional spectral flow to the k-orbifolded NSNS vacuum, as discussed around Eq. (2.89).

Given a symmetric orbifold CFT correlator, one can perform spectral flow on both the operators and the background states. The value of the correlator is invariant under this operation. One can use this to map the correlator in one of our heavy backgrounds to a correlator in the k-orbifolded NSNS vacuum. Of course, for k = 1, after undoing the spectral flows one obtains a vacuum correlator.

Taking for simplicity k = 1 and  $\bar{s} = 0$ , the transformation of a chiral primary operator under this operation is [93, App. A]

$$\tilde{O}_h(x) = x^{(2s+1)m'} O_h(x).$$
(4.61)

The exponent of *x* directly parallels the *x* factors appearing in Eq. (4.47). This observation generalizes straightforwardly to  $\bar{s} \neq 0$  and to k > 1, whereupon operators have fractional modes taking values in  $\mathbb{Z}/k$ . We will comment further on the relation between worldsheet and symmetric product orbifold CFT correlators in due course.

## **4.2.4** Non-BPS HLLH correlators for $h = \frac{1}{2}$

The correlator presented in the previous subsection can be readily generalized to compute a set of novel HLLH correlators involving the same light operators, but probing the more general class of non-supersymmetric backgrounds given by the JMaRT solutions, in which both spacetime (fractional) spectral flow parameters s and  $\bar{s}$  are nontrivial. Note that the parameters  $s, \bar{s}, k$  defining the background must satisfy  $s(s + 1) - \bar{s}(\bar{s} + 1) \in k\mathbb{Z}$ , from combining Eqs. (3.84) and (2.89).

The same steps as described in the previous subsection lead directly to the following generalization of Eq. (4.53):

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{Y}_{\frac{1}{2}}(1) \mathcal{Y}_{\frac{1}{2}}^{\dagger}(x) | s, \bar{s}, \mathsf{k} \rangle = \frac{1}{\mathsf{k}^2} \sum_{n, \bar{n}} x^{\frac{n-s}{\mathsf{k}} - \frac{1}{2}} \bar{x}^{\frac{\bar{n}-\bar{s}}{\mathsf{k}} - \frac{1}{2}} .$$
(4.62)

To make precise the restricted summation, analogously to s = kp - a we write  $\bar{s} = k\bar{p} - \bar{a}$ , with  $1 \le \bar{a} \le k$ . We parametrize  $n = k\hat{n} + \delta - a$  and  $\bar{n} = k\hat{n} + \delta - \bar{a}$  in order to satisfy the condition coming from the subtracted gauge constraint generalizing Eq. (4.55), namely

$$\bar{n} - n \equiv (a - \bar{a}) \operatorname{mod} \mathsf{k}. \tag{4.63}$$

Then, by summing over all possible values of  $\hat{n}$ ,  $\hat{n}$  and  $\delta$  such that n and  $\bar{n}$  are non-negative and satisfy (4.63), defining  $b \equiv \min(a, \bar{a})$  we obtain

$$\langle s, \bar{s}, k | \mathcal{Y}_{\frac{1}{2}}(1) \mathcal{Y}_{\frac{1}{2}}^{\dagger}(x, \bar{x}) | s, \bar{s}, k \rangle = \frac{1}{k^2} \frac{x^{-p} \bar{x}^{-\bar{p}}}{|x|} \times$$

$$\left[ \frac{1}{|1-x|^2} \frac{1-|x|^2}{1-|x|^{2/k}} - \frac{1}{1-\bar{x}} \frac{1-|x|^{2a/k}}{1-|x|^{2/k}} - \frac{1}{1-x} \frac{1-|x|^{2\bar{a}/k}}{1-|x|^{2/k}} + \frac{1-|x|^{2b/k}}{1-|x|^{2/k}} \right].$$

$$(4.64)$$

As before, the second and third terms remove contributions for which either *n* or  $\bar{n}$  become negative, while the fourth one compensates for the over-counting of cases in which both *n* and  $\bar{n}$  are negative.

At first sight, Eq. (4.64) may seem to depend on the values of *a* and  $\bar{a}$  separately, in apparent contradiction with the fact that, as is implied by (4.63), only their difference matters. However, the RHS of Eq. (4.64) can be rewritten as

$$\frac{1}{k^{2}} \frac{x^{-s/k} \bar{x}^{-\bar{s}/k}}{|x||1-x|^{2}} \frac{1}{(1-|x|^{2/k})} \left(\frac{x}{\bar{x}}\right)^{(\bar{a}-a)/2k} \times (4.65) \\ \left[ (1-x)|x|^{(a-\bar{a})/k} + (1-\bar{x})|x|^{-(a-\bar{a})/k} - |1-x|^{2}|x|^{-|a-\bar{a}|/k} \right],$$

which explicitly depends only the orbifold parameter k, the spectral flow parameters *s* and  $\bar{s}$ , and the difference  $a - \bar{a} \equiv \bar{s} - s \mod k$ , as expected. Note that (4.65) is symmetric under the simultaneous replacements  $x \leftrightarrow \bar{x}$  and  $s \leftrightarrow \bar{s}$ .

The worldsheet correlators (4.64)–(4.65) are one of the main results of this thesis. Unlike the  $\bar{s} = 0$  supersymmetric example of the previous subsection, generically the corresponding supergravity or holographic CFT correlators have not been computed in the literature.

Since the backgrounds are non-supersymmetric when s and  $\bar{s}$  are both non-zero, again a priori there is no obvious reason to expect the correlators to be protected across moduli space.

However, better-than-expected agreement between supergravity and holographic CFT has already been observed for a closely related observable describing the analog of the Hawking radiation process [140, 141, 94]; we shall comment further on this in due course.

Moreover, for the particular cases in which *s* and  $\bar{s}$  are congruent to either 0 or k – 1 mod k, the holographic CFT correlator follows straightforwardly from the techniques in App. A of [167], providing another exact match even for the non-supersymmetric backgrounds. We shall generalize this further in the next section.

# 4.3 More general heavy-light correlators

In this section we present HLLH correlators for generic chiral primaries of conformal weights h > 1/2. We then progress to describe higher-point heavy-light correlators. As an application, we compute the analogue of the Hawking radiation process from the backgrounds under consideration. We also compute a five-point HLLLH correlator of the symmetric product orbifold CFT, and demonstrate precise agreement with the corresponding worldsheet correlator.

#### **4.3.1** HLLH correlators for general *h*

We now consider HLLH correlators where the light operators are chiral primaries with  $h \ge 1$ . For these correlators the *m*-basis SL(2,  $\mathbb{R}$ ) two-point functions do not trivialize, and the resulting sums become more complicated.

By following the method outlined in the previous sections, the worldsheet computation leads to restricted sums of the form

$$\frac{1}{k^{4h}} \sum_{n,\bar{n}} x^{\frac{n-2hs}{k} - h} \bar{x}^{\frac{\bar{n}-2hs}{k} - h} \frac{\Gamma(2h+n)\Gamma(2h+\bar{n})}{\Gamma(2h)^2 \Gamma(n+1)\Gamma(\bar{n}+1)} .$$
(4.66)

For k = 1, there are no restrictions on the allowed values of the mode numbers *n* and  $\bar{n}$ , and so the sum can be performed straightforwardly to obtain

$$\langle s, \bar{s}, \mathsf{k} = 1 | \mathcal{O}_h(1) \mathcal{O}_h^{\dagger}(x, \bar{x}) | s, \bar{s}, \mathsf{k} = 1 \rangle = \frac{x^{-2hs} \bar{x}^{-2h\bar{s}}}{|1-x|^{4h}}.$$
 (4.67)

This expression agrees with the corresponding HLLH correlator of the symmetric product orbifold CFT, as a direct consequence of the discussion around Eq. (4.61).

On the other hand, for k > 1 the sum becomes more difficult to carry out explicitly since *n* and  $\bar{n}$ , which appear in the arguments of the Gamma functions, must satisfy the constraint

$$\bar{n} - n \equiv 2h \, (\bar{s} - s) \mod \mathsf{k} \,, \tag{4.68}$$

which is the direct generalization of Eq. (4.63). We shall first briefly describe a method to construct these correlators iteratively, starting from the  $h = \frac{1}{2}$  case obtained above, then present an improved method.

Our iterative construction operates by expressing the additional coefficients in the sum in Eq. (4.66) in terms of differential operators acting on results for lower values of h. Let us illustrate how this works for the simplest non-trivial case h = 1. From the general

expression (4.66), we have

$$\begin{aligned} \langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h=1}(1) \mathcal{O}_{h=1}^{\dagger}(x, \bar{x}) | s, \bar{s}, \mathsf{k} \rangle &= \frac{1}{\mathsf{k}^4} \sum_{n, \bar{n}} x^{\frac{n-2s}{\mathsf{k}} - 1} \bar{x}^{\frac{\bar{n}-2\bar{s}}{\mathsf{k}} - 1} (n+1) (\bar{n}+1) \end{aligned} (4.69) \\ &= \frac{1}{\mathsf{k}^4} \frac{x^{-\frac{2s}{\mathsf{k}}} \bar{x}^{-\frac{2s}{\mathsf{k}}}}{|x|^2} (\mathsf{k} x \partial_x + 1) (\mathsf{k} \bar{x} \partial_{\bar{x}} + 1) \sum_{n, \bar{n}} x^{\frac{n}{\mathsf{k}}} \bar{x}^{\frac{\bar{n}}{\mathsf{k}}} .\end{aligned}$$

Hence, the differential operators act on a sum similar to the one analyzed in the previous section. For the general case, the procedure iterates. We redefine *a* and  $\bar{a}$  to be generalizations of the *a* and  $\bar{a}$  used in the h = 1/2 correlators (see above Eqs. (4.55) and (4.63)), where we replace  $s \mapsto 2hs$ ,  $\bar{s} \mapsto 2h\bar{s}$ , such that  $a - \bar{a} \equiv 2h(\bar{s} - s) \mod k$ . Then we obtain

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h}(1) \mathcal{O}_{h}^{\dagger}(x, \bar{x}) | s, \bar{s}, \mathsf{k} \rangle = \frac{1}{\mathsf{k}^{4h}} \frac{x^{-\frac{2h\bar{s}}{\mathsf{k}}} \bar{x}^{-\frac{2h\bar{s}}{\mathsf{k}}}}{|x|^{2h}} \frac{D_{h,\mathsf{k}} \bar{D}_{h,\mathsf{k}}}{\Gamma(2h)^{2}} \sum_{n,\bar{n}} x^{\frac{n}{\mathsf{k}}} \bar{x}^{\frac{\bar{n}}{\mathsf{k}}}, \qquad (4.70)$$

where we have introduced differential operators of order 2h - 1 defined as

$$D_{h,k} \equiv (kx\partial_x + 2h - 1) \cdots (kx\partial_x + 1), \qquad (4.71)$$

and where

$$\sum_{n,\bar{n}} x^{\frac{n}{k}} \bar{x}^{\frac{\bar{n}}{k}} = \frac{(1-x)|x|^{(a-\bar{a})/k} + (1-\bar{x})|x|^{-(a-\bar{a})/k} - |1-x|^2|x|^{-|a-\bar{a}|/k}}{|1-x|^2(1-|x|^{2/k})} \left(\frac{x}{\bar{x}}\right)^{(\bar{a}-a)/2k}.$$
(4.72)

Although it leads to correct results, the procedure outlined above quickly becomes cumbersome, and leads to seemingly complicated expressions for higher values of *h*. In addition, it does not appear to give any insight into whether the results are likely to match with computations in the symmetric product orbifold CFT. However, we can improve on both these points with a different method. We now describe this method by first rederiving the h = 1/2 correlators of the previous section, and then generalizing the improved method to arbitrary values of *h*, and also to higher-point heavy-light correlators.

Let us thus re-examine the general expression Eq. (4.66), and consider the case in which  $a - \bar{a} = 0$  for simplicity. We can take into account the restriction on the allowed values of *n* and  $\bar{n}$  by considering an *unrestricted* sum over arbitrary positive integers by making use of a "Kronecker comb". In other words, we impose that  $n - \bar{n} = 0 \mod k$  by including extra coefficients of the form

$$\sum_{q \in \mathbb{Z}} \delta_{n-\bar{n}, kq} = \frac{1}{k} \sum_{r=0}^{k-1} e^{2\pi i r \frac{n-\bar{n}}{k}}, \qquad (4.73)$$

where the final equality is obtained by Fourier transformation, and represents a simple form of the discrete Poisson summation formula. The RHS in Eq. (4.73) is interesting,

because the exponentials can be absorbed into terms involving powers of *x* and  $\bar{x}$ . Explicitly, we can rewrite the expression (4.66) with  $a - \bar{a} = 0$  as

$$\frac{1}{k^{4h+1}} \frac{x^{-\frac{2h\bar{s}}{k}} \bar{x}^{-\frac{2h\bar{s}}{k}}}{|x|^{2h}} \sum_{n,\bar{n}\geq 0} \sum_{r=0}^{k-1} u_r^n \bar{u}_r^{\bar{n}} \frac{\Gamma(2h+n)\Gamma(2h+\bar{n})}{\Gamma(2h)^2\Gamma(n+1)\Gamma(\bar{n}+1)} , \qquad (4.74)$$

where  $u_r$ ,  $\bar{u}_r$  are the k<sup>th</sup> roots of x and  $\bar{x}$ , respectively; writing  $x^{\frac{1}{k}} \equiv |x|^{\frac{1}{k}} e^{2\pi i \frac{\operatorname{Arg}(x)}{k}}$ ,

$$u_r \equiv x^{\frac{1}{k}} e^{2\pi i \frac{r}{k}} , \qquad \bar{u}_r \equiv \bar{x}^{\frac{1}{k}} e^{-2\pi i \frac{r}{k}} . \qquad (4.75)$$

Thus, inside the convergence region |x| < 1 we can exchange the order of the sums, such that the *unrestricted* sum over integers *n* and  $\bar{n}$  leads to the usual expression for the CFT two-point function. However, it is evaluated at the different values of  $u_r$ , instead of the insertion point itself. Thus, the expression (4.74) becomes

$$\frac{1}{k^{4h+1}} \frac{x^{-\frac{2hs}{k}} \bar{x}^{-\frac{2hs}{k}}}{|x|^{2h}} \sum_{r=0}^{k-1} \frac{1}{|1-u_r|^{4h}}.$$
(4.76)

In fact, we can rewrite this in a slightly more general form. Indeed, it is easy to "unfix" the first insertion point and write the full expression of the HLLH correlator in terms of  $x_1$  and  $x_2$ . To do so, we introduce k<sup>th</sup> roots of  $x_1$ ,  $x_2$ , and  $x_2/x_1$  via  $u_{1,r_1}^k = x_1$ ,  $u_{1,r_2}^k = x_2$ , and  $u_{21,r}^k = x_2/x_1$ , and then make use of the identity

$$\frac{1}{|x_1|^{\frac{4h}{k}}} \sum_{r=0}^{k-1} \frac{1}{|1-u_{21,r}|^{4h}} = \frac{1}{k} \sum_{r_1, r_2=0}^{k-1} \frac{1}{|u_{1,r_1}-u_{2,r_2}|^{4h}}.$$
(4.77)

This gives

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h}(x_{1}, \bar{x}_{1}) \mathcal{O}_{h}^{\dagger}(x_{2}, \bar{x}_{2}) | s, \bar{s}, \mathsf{k} \rangle \Big|_{2h(s-\bar{s}) = 0 \mod \mathsf{k}}$$

$$= \frac{1}{\mathsf{k}^{4h+2}} \left(\frac{x_{2}}{x_{1}}\right)^{-h\frac{(2s+1)}{\mathsf{k}}} \left(\frac{\bar{x}_{2}}{\bar{x}_{1}}\right)^{-h\frac{(2s+1)}{\mathsf{k}}} |x_{1}x_{2}|^{2h\left(\frac{1}{\mathsf{k}}-1\right)} \sum_{r_{1}, r_{2}=0}^{\mathsf{k}-1} \frac{1}{|u_{1,r_{1}} - u_{2,r_{2}}|^{4h}} .$$

$$(4.78)$$

The general case is computed entirely analogously. We must simply replace  $n - \bar{n} \mapsto n - \bar{n} + (a - \bar{a})$  in Eq. (4.73), which induces some extra phases. The appropriate generalization of Eq. (4.77) is given by

$$\frac{1}{|x_1|^{\frac{4h}{k}}} \sum_{r=0}^{k-1} \frac{e^{2\pi i r(a-\bar{a})/k}}{|1-u_{21,r}|^{4h}} = \frac{1}{k} \sum_{r_1, r_2=0}^{k-1} \frac{e^{2\pi i (r_2-r_1)(a-\bar{a})/k}}{|u_{1,r_1}-u_{2,r_2}|^{4h}}.$$
(4.79)

Then the HLLH correlator with generic values of the orbifold parameter k, the spectral flow parameters s and  $\bar{s}$ , and the weight of the light chiral primary operator h, takes the

form

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h}(x_{1}, \bar{x}_{1}) \mathcal{O}_{h}^{\dagger}(x_{2}, \bar{x}_{2}) | s, \bar{s}, \mathsf{k} \rangle$$

$$= \frac{1}{\mathsf{k}^{4h+2}} \left(\frac{x_{2}}{x_{1}}\right)^{-h\frac{(2s+1)}{\mathsf{k}}} \left(\frac{\bar{x}_{2}}{\bar{x}_{1}}\right)^{-h\frac{(2s+1)}{\mathsf{k}}} |x_{1}x_{2}|^{2h\left(\frac{1}{\mathsf{k}}-1\right)} \sum_{r_{1}, r_{2}=0}^{\mathsf{k}-1} \frac{e^{2\pi i (r_{2}-r_{1})(a-\bar{a})/\mathsf{k}}}{|u_{1,r_{1}}-u_{2,r_{2}}|^{4h}} ,$$

$$(4.80)$$

where the sum is over the k-th roots of the insertion points  $x_1$  and  $x_2$ , as defined above (4.77), and where  $2h(\bar{s} - s) \equiv a - \bar{a} \mod k$ .

Note that we can relax the chiral primary condition and consider operators in which  $m' \neq \pm h$ . We shall continue to focus on massless vertex operators, however this could be generalized further. In addition, by making use of the phases and the  $x_{1,2}$  powers on the RHS of (4.80), we can rewrite the result in a cleaner form,

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h,m'}(x_1, \bar{x}_1) \mathcal{O}_{h,m'}^{\dagger}(x_2, \bar{x}_2) | s, \bar{s}, \mathsf{k} \rangle$$

$$= \frac{1}{\mathsf{k}^2} \sum_{r_1, r_2=0}^{\mathsf{k}-1} \left( \frac{u_{2,r_2}}{u_{1,r_1}} \right)^{-m'(2s+1)} \left( \frac{\bar{u}_{2,r_2}}{\bar{u}_{1,r_1}} \right)^{-\bar{m}'(2\bar{s}+1)} \frac{|u_{1,r_1}u_{2,r_2}|^{2h(1-\mathsf{k})}}{\mathsf{k}^{4h}|u_{1,r_1}-u_{2,r_2}|^{4h}} .$$

$$(4.81)$$

#### 4.3.2 Matching between worldsheet and symmetric product orbifold

The appearance of the k<sup>th</sup> roots of the physical insertions  $x_{1,2}$  in Eq. (4.81) is related to the fact that the holographic description of the heavy backgrounds involves heavy states in k-twisted sectors of the boundary CFT. The same feature appears in certain computations performed using the Lunin-Mathur covering space technique [97], specifically when there are operators of twist k inserted at the origin and infinity of the CFT plane, and when there are other untwisted operators in the correlator. Then the coordinate transformation to the k-fold covering space is precisely  $x = u^k$ .

Thus, when the light worldsheet operators correspond to untwisted operators of the symmetric product orbifold CFT, it is natural to identify u with the coordinate on the k-fold covering space that trivializes the twist operators involved in the definition of the heavy states.

The sum over the different roots generates the usual phases included in the definition of fractional modes by summing over the different copies of the theory [97],

$$O_{\frac{m}{k}} = \oint \frac{dx}{2\pi i} \sum_{r=1}^{k} O_{(r)}(x) e^{\frac{2\pi i m}{k}(r-1)} x^{h+\frac{m}{k}-1}.$$
(4.82)

Moreover, the *fractional* spectral flow defining the background, when mapped to a k-fold covering space, becomes *integer* spectral flow with parameters 2s + 1 and  $2\bar{s} + 1$  [96, 94]. Hence, one can generalize the discussion around Eq. (4.61) and simply consider the appropriate powers of  $u_{i,r_i}$  to arise from performing spacetime spectral flow

on the operators, in the k-fold covering space. Finally, the last factor on the RHS of (4.81) corresponds to the usual two-point function evaluated at the roots, including the necessary Jacobian factors arising from mapping to the k-fold covering space,  $|\partial u/\partial x|^{2h}$ . Obtaining precisely this Jacobian is the justification for the factors of k introduced in the definition of the *x*-basis operators in Eq. (4.47). Thus, we see that symmetric orbifold CFT HLLH correlators for which the covering space is  $x = u^k$  agree in both structure and value with the worldsheet correlator (4.81).

By contrast, for twisted operators, the interpretation of our worldsheet result (4.81) is more involved: the Lunin-Mathur covering map for such correlators is *not*  $x = u^k$ . To understand the precise relation, we focus on light operators of twist two, and show that Eq. (4.81) nevertheless matches with the symmetric orbifold CFT also in this case. The relevant four-point function was studied recently in [183, 181, 182] in the Sym<sup>N</sup>( $\mathbb{T}^4$ ) CFT. At leading order in large *N*, the correlator is dominated by a contribution from a covering space with genus zero, where the copy indices of the light twist-two operator act on different k-strands corresponding to the heavy state. One of the light insertions effectively joins together two k-strands into a 2k-strand, and the other light insertion effectively cuts the 2k-strand back to two strands of length k. For this process, and setting for simplicity  $s = \bar{s} = 0$  as done in [181], the relation between the physicalspace cross-ratio *x* and the covering-space cross ratio *v* is<sup>45</sup>

$$x(v) = \left(\frac{v+1}{v-1}\right)^{2k}$$
 (4.83)

The correlator of interest is written in terms v(x) as defined through Eq. (4.83), and involves the sum over the 2k pre-images of x and also an N- and k-dependent overall factor. However, due to the  $v \rightarrow 1/v$  symmetry of the map (4.83), there are actually only k distinct contributions [182], corresponding to distinct ramified coverings of the base space. In more detail, up to overall normalization the correlator takes the form [182, Eq. (4.17)]

$$\langle \mathsf{k} | \mathcal{O}_{h,m'}(x,\bar{x}) \mathcal{O}_{h,m'}^{\dagger}(1) | \mathsf{k} \rangle \sim \sum_{r=0}^{\mathsf{k}-1} \left[ v_r^{K_0} (v_r-1)^{K_-} (v_r+1)^{K_+} \times \text{c.c.} \right],$$
 (4.84)

where  $|k\rangle = |s = \bar{s} = 0, k\rangle$ . The exponents  $K_0$  and  $K_{\pm}$  are defined in [182, Eq. (4.17b)]; for this particular correlator, they evaluate to  $K_0 = -2h$  and  $K_{\pm} = 2h(1 \mp k) \pm 2m'$ . Upon inserting the explicit solutions

$$v_r(x) = \frac{x^{\frac{1}{2k}}e^{\frac{i\pi r}{k}} + 1}{x^{\frac{1}{2k}}e^{\frac{i\pi r}{k}} - 1}, \qquad r = 0, \dots, k - 1, \qquad (4.85)$$

<sup>&</sup>lt;sup>45</sup>Note that in [181, 182], the base (physical) space coordinates are denoted by z or u rather than our x, while the covering space coordinates are denoted by t or x rather than our v.

where, as before,  $x^{\frac{1}{2k}}$  stands for a particular  $(2k)^{\text{th}}$  root of x, the final expression remarkably coincides with Eq. (4.81). The analysis for the JMaRT states and for more general light insertions can be carried out analogously.

Recall that, as reviewed in Section 2.5.1, at a generic spacetime dimension *h* there is a degeneracy in the twist *n* of light states in the symmetric product orbifold CFT. An interesting feature of the worldsheet correlator (4.81) is that it is independent of this twist *n*. Recall also that, for untwisted light operators, the worldsheet correlator has the same structure as the covering space method of the symmetric product orbifold CFT. The fact that the agreement of HLLH correlators extends to (at least some) twisted light operators is thus remarkable from the point of view of the holographic CFT. Despite the more complicated covering map, the above discussion demonstrates how, for these correlators, the end result agrees with an expression whose structure is that of the simple map  $x = u^k$ .

## 4.3.3 Higher-point heavy-light correlators

Our general expression for HLLH correlators, Eq. (4.81), together with the matching to the symmetric product orbifold CFT that we have observed so far, motivate a deeper exploration. Thus, we now describe how local *x*-basis operators are seen from the spectrally flowed frame as indicated by our null-gauged worldsheet models. This will allow us to extract consequences for worldsheet three-point and higher-point functions, corresponding to holographic CFT correlators with two heavy states and three or more light operators.

The AdS<sub>3</sub> limit of the holomorphic gauge condition, (4.45), upon using the definition of  $m_y$  in Eq. (4.46), reads

$$0 = m + (2s+1)m' - km_y.$$
(4.86)

We wish to re-interpret this constraint in the local coordinate basis of the holographic CFT. A priori, it is perhaps not obvious that this is a useful thing to do, since the usual *x*-basis operators are constructed by resumming the action of  $J_0^{\pm}$ , which does not commute with the BRST charge in the coset theory. However we shall see that it will be very useful.

Let us observe that there are two notions of *x*-type local coordinates in the worldsheet model. The one used so far in Section 4.2 and the present section is the physical *x* coordinate of the gauged models. However, before gauging, there is an analogous coordinate for the *upstairs* SL(2, $\mathbb{R}$ ) algebra. We will denote the associated coordinate by the complex variable *u*; we will see momentarily that  $u^k = x$ , so that there will be no clash with the *u* used above.

The differential operator  $x\partial_x + h$  corresponds to the quantity  $m_y$ , as can be seen by comparing Eqs. (2.110), (2.115), (4.47). On the other hand, the upstairs SL(2,  $\mathbb{R}$ ) projection mcorresponds to an analogous operator in the u variable: we write this as  $u\partial_u + h - \beta$ , where we have allowed for a shift  $\beta$ , whose precise form will become clear shortly, as will the reason for its existence. Then Eq. (4.86) can be expressed in terms of these differential operators as

$$kx\partial_x = u\partial_u + (2s+1)m' + h(1-k) - \beta.$$
(4.87)

In order that this condition is solved by  $u^k = x$ , we choose

$$\beta = h(1 - \mathsf{k}) + (2s + 1)m'. \tag{4.88}$$

Thus the role of  $\beta$  is two-fold. On the one hand, the first term in (4.88) effectively replaces the weight by  $h \rightarrow h_u \equiv kh$ , which further supports the discussion above about *u* corresponding to a covering space coordinate in the holographic CFT. It also generates the Jacobian factor obtained in Eq. (4.81). On the other hand, the second term in (4.88) takes into account the shift arising from spacetime spectral flow.

We now use this to obtain an improved construction of gauge-invariant operators directly in the *x*-basis, built upon *u*-basis operators of the upstairs SL(2, $\mathbb{R}$ ), i.e. without relying on their spacetime Virasoro mode expansion as in (4.47). Although such a construction gives equivalent results at the level of worldsheet two-point functions (4.81), its importance for higher-point functions was highlighted recently in [38]. The construction proceeds as follows:

1. We consider an operator whose upstairs SL(2, $\mathbb{R}$ ) part is expressed in the usual local SL(2, $\mathbb{R}$ ) basis,  $\mathcal{V}_h(u, \bar{u})$ , where for simplicity we set  $h = \bar{h}$ . We multiply this by an SU(2) vertex operator  $\mathcal{V}'_{j',m',\bar{m}'}$ . We suppress the exponentials of t and y, since they have weight zero in the AdS<sub>3</sub> limit, and their only effect is taken into account through (4.86) and its antiholomorphic counterpart. We introduce the notation

$$\hat{\mathcal{O}}_{h,m',\bar{m}'}(u,\bar{u}) \equiv \mathcal{V}_h(u,\bar{u})\mathcal{V}'_{j',m',\bar{m}'}.$$
(4.89)

- 2. We introduce the above  $\beta$ -shift by multiplying by an extra factor  $u^{\beta}\bar{u}^{\bar{\beta}}$ .
- 3. We sum the resulting operator over all insertion points *u* such that  $u^k = x$ .

Explicitly, we define

$$\mathcal{O}_{h,m',\bar{m}'}(x,\bar{x}) \equiv \frac{1}{\mathsf{k}^{2h+1}} \sum_{u^{\mathsf{k}}=x} u^{\beta} \bar{u}^{\bar{\beta}} \hat{\mathcal{O}}_{h,m',\bar{m}'}(u,\bar{u}) \,. \tag{4.90}$$

Comparing with Eq. (4.47), using the Kronecker comb (4.73) to impose the constraints as above, we indeed have

$$\mathcal{O}_{h,m',\bar{m}'}(x,\bar{x}) \equiv \frac{1}{\mathsf{k}^{h+\bar{h}}} \sum_{m_y,\bar{m}_y} x^{m_y-h} \bar{x}^{\bar{m}_y-h} \mathcal{V}_{j,m,\bar{m}} \mathcal{V}'_{j',m',\bar{m}'}$$

$$= \frac{1}{\mathsf{k}^{2h}} \sum_{m,\bar{m}} 'x^{\frac{1}{\mathsf{k}}[m+(2s+1)m']-h} \bar{x}^{\frac{1}{\mathsf{k}}[\bar{m}+(2\bar{s}+1)\bar{m}']-h} \mathcal{V}_{j,m,\bar{m}} \mathcal{V}'_{j',m',\bar{m}'}$$

$$= \frac{1}{\mathsf{k}^{2h+1}} \sum_{u^{\mathsf{k}}=x} \sum_{m,\bar{m}} u^{m-h+(2s+1)m'+h(1-\mathsf{k})} \bar{u}^{\bar{m}-h+(2\bar{s}+1)\bar{m}'+h(1-\mathsf{k})} \mathcal{V}_{j,m,\bar{m}} \mathcal{V}'_{j',m',\bar{m}'}$$

$$= \frac{1}{\mathsf{k}^{2h+1}} \sum_{u^{\mathsf{k}}=x} u^{\beta} \bar{u}^{\bar{\beta}} \hat{\mathcal{O}}_{h,m',\bar{m}'}(u,\bar{u}).$$
(4.91)

We note that in the symmetric product orbifold CFT, when mapping the  $S_k$ -invariant untwisted operators  $O(x) = \sum_{r=1}^{k} O_{(r)}(x)$  to the k-fold covering space, using the inverse relation to (4.82),

$$O_{(r)}(x) = \frac{1}{k} \sum_{m} O_{\frac{m}{k}} x^{-\frac{m}{k} - h} e^{-\frac{2\pi i m}{k} (r-1)}, \qquad (4.92)$$

one obtains an expression closely analogous to Eq. (4.90).

We now exploit the expression (4.91) to study higher-point functions. We first rewrite the HLLH correlator (4.81) in the simple form

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h_1, m_1'}(x_1, \bar{x}_1) \mathcal{O}_{h_2, m_2'}^{\dagger}(x_2, \bar{x}_2) | s, \bar{s}, \mathsf{k} \rangle = \frac{1}{\mathsf{k}^{4h+2}} \sum_{u_i^{\mathsf{k}} = x_i} \frac{u_1^{\beta_1} \bar{u}_1^{\bar{\beta}_1} u_2^{\beta_2} \bar{u}_2^{\bar{\beta}_2}}{|u_1 - u_2|^{4h}}, \qquad (4.93)$$

with  $\beta_i = h_i(1 - k) + (2s + 1)m'_i$ ,  $\bar{\beta}_i = h_i(1 - k) + (2\bar{s} + 1)\bar{m}'_i$ , and  $h_1 = h_2 = h$ , and where the charge conservation  $m'_1 + m'_2 = 0$  is understood. We then deduce that the worldsheet correlator with *n* light insertions with weights  $h_i$  and charges  $m'_i$ ,  $\bar{m}'_i$  is given by the following straightforward generalization of (4.93) (in which we partially suppress antiholomorphic quantities):

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h_1, m_1'}(x_1) \dots \mathcal{O}_{h_n, m_n'}(x_n) | s, \bar{s}, \mathsf{k} \rangle$$

$$= \frac{1}{\mathsf{k}^{H+\bar{H}+n}} \sum_{u_i^{\mathsf{k}}=x_i} \left( \prod_{\ell=1}^n u_\ell^{\beta_\ell} \bar{u}_\ell^{\bar{\beta}_\ell} \right) \langle \hat{\mathcal{O}}_{h_1, m_1'}(u_1) \dots \hat{\mathcal{O}}_{h_n, m_n'}(u_n) \rangle, \quad (4.94)$$

where  $H = h_1 + \cdots + h_n$ , and  $\langle \hat{\mathcal{O}}_{h_1,m'_1}(u_1) \dots \hat{\mathcal{O}}_{h_n,m'_n}(u_n) \rangle$  stands for the global AdS<sub>3</sub> × S<sup>3</sup> vacuum *n*-point function evaluated at the roots of the original insertion points.

The expression (4.94), which holds for generic values of  $s, \bar{s}$ , k and generic light weights and charges  $h_i, m'_i, \bar{m}_i$ , constitutes one of the main results of this thesis. In Eq. (4.94), the n = 3 case can be made quite explicit, as we shall do so in the next subsection.

The above result can straightforwardly seen to include spectrally-flowed vertex operators, as follows. Setting  $\omega' = \bar{\omega}' = \omega$  for simplicity, the bosonic null-gauge condition

Eq. (4.40) in the AdS limit becomes

$$0 = m_{\omega} + (2s+1)m'_{\omega} - \mathsf{k}m_y , \qquad 0 = \bar{m}_{\omega} + (2\bar{s}+1)\bar{m}'_{\omega} - \mathsf{k}\bar{m}_y , \qquad (4.95)$$

where, for discrete states in the lowest weight representation,  $m_{\omega} = h_{\omega} + n$ ,  $h_{\omega} = J + n_5 \omega/2$  and  $m'_{\omega} = h'_{\omega} + n_5 \omega/2 - n'$ . As a consequence, the exponent  $\beta$  of the covering space coordinate u gets replaced by  $\beta \mapsto \beta_{\omega} = h_{\omega}(1 - k) + (2s + 1)m'_{\omega}$ , and the power of k in the normalisation factor is modified accordingly. Thus, for the vertex operators in the coset models, the net effects of the spectral flow procedure are the replacements  $h \mapsto h_{\omega}, m \mapsto m'_{\omega}$ . This is understood by the fact that when a boundary light operator has the spacetime dimension h = J that renders the SL(2, \mathbb{R}) spin above the unitary bound Eq. (2.106), it corresponds holographically to a spectrally-flowed worldsheet vertex operator [101]. This implies that the structure of the correlator in Eq. (4.94) is not drastically modified when  $\omega \neq 0$ .

It is important to note, however, that the entire computational complication due to worldsheet spectral flow remains present in the resulting *vacuum* correlator. Indeed, we see that the *n*-point function on the heavy state is now written in terms of a vacuum *n*-point function of spectrally-flowed states. It is thus natural to expect that the  $AdS_3$  selection rules carry over to *n*-point functions in the JMaRT microstates. We conclude that the generalisation of Eq. (4.94) to the case of worldsheet spectrally-flowed states reads

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}_{h_1, m'_1}^{\omega_1}(x_1) \dots \mathcal{O}_{h_n, m'_n}^{\omega_n}(x_n) | s, \bar{s}, \mathsf{k} \rangle =$$

$$\frac{1}{\mathsf{k}^{H_\omega + \bar{H}_\omega + n}} \sum_{u_i^k = x_i} \left( \prod_{\ell=1}^n u_\ell^{\beta_{\omega, \ell}} \bar{u}_\ell^{\bar{\beta}_{\omega, \ell}} \right) \langle \hat{\mathcal{O}}_{h_1, m'_1}^{\omega_1}(u_1) \dots \hat{\mathcal{O}}_{h_n, m'_n}^{\omega_n}(u_n) \rangle ,$$

$$(4.96)$$

where  $H_{\omega} = \sum_{i} h_{\omega,i}$  and the light operators  $\mathcal{O}_{h_i,m'_i}^{\omega_i}$  are *x*-basis spectrally-flowed world-sheet vertex operators.

We emphasize that the construction we have outlined in this section only holds in the IR  $AdS_3 \times S^3$  limit. In the full asymptotically linear dilaton geometry, the identification of the modes  $m_y$  and  $\bar{m}_y$  as defined in (4.46) breaks down, and the *t* and *y* exponentials can no longer be ignored. This is consistent with the fact that in the UV the dual holographic theory is not a CFT, but is instead a little string theory. Since little string theories are non-local, it is correct that the above definition of local operators does not apply. Note, however, that the *mode* correlators computed in the *m*-basis still make perfect sense, and carry information about string perturbation theory in the full geometry.

Let us speculate on which subset of the above correlators can be expected to agree with those of the symmetric product orbifold theory. Since our expressions for the general correlators (4.94), (4.96) involve vacuum correlators, it is natural to conjecture that for these particular heavy backgrounds, the heavy-light correlator is protected whenever

the global  $AdS_3 \times S^3$  vacuum correlator appearing in (4.94), (4.96) is protected. Recall that, in the global  $AdS_3 \times S^3$  vacuum, two-point and three-point correlation functions of chiral primaries are protected [184], while four-point and higher-point functions are generically renormalized. So heavy-light correlators with two or three light insertions on these backgrounds may be protected between worldsheet and symmetric product orbifold CFT. It may even be possible to prove a non-renormalization theorem generalizing [184]; work in this direction is in progress. For now however, we next compute a heavy-light correlator with three light insertions in both worldsheet and holographic CFT.

#### 4.3.4 An HLLLH correlator in worldsheet and holographic CFT

We now investigate the general expression for our worldsheet correlator (4.94), in a particular example with three light insertions, and compare it to the symmetric product orbifold CFT. We shall observe another highly non-trivial agreement.

We consider three light insertions living in the untwisted sector of the holographic CFT, with weights  $(h_1, h_2, h_3) = (\frac{1}{2}, \frac{1}{2}, 1)$ . In the dual CFT notation, we are then interested in computing the correlator  $\langle O_{\frac{1}{2}}(x_1)O_{\frac{1}{2}}(x_2)O_1^{\dagger}(x_3)\rangle_H$ . We further focus on heavy backgrounds with s = kp with  $p \in \mathbb{Z}$  and  $\bar{s} = 0$ .

We start by evaluating the general expression (4.94) for this particular worldsheet correlator. In the worldsheet theory associated to the global  $AdS_3 \times S^3$ , the  $O_{\frac{1}{2}}$  correspond to two RR states, while  $O_1$  is an NSNS state polarized on the  $S^3$  directions. The (integrated) vacuum three-point functions for these chiral primaries were studied in [99]. In our notation, they take the form

$$\langle \mathcal{O}_{h_1}^{\text{RR}}(x_1)\mathcal{O}_{h_2}^{\text{RR}}(x_2)\mathcal{O}_{h_3}^{+\text{NSNS}}(x_3)\rangle = \frac{1}{N^{1/2}} \frac{\sqrt{(2j_1-1)(2j_2-1)(2j_3-1)^{-1}}}{|x_{12}|^{2(h_1+h_2-h_3)}|x_{13}|^{2(h_1+h_3-h_2)}|x_{23}|^{2(h_2+h_3-h_1)}},$$
(4.97)

where  $j_1 = h_1 + \frac{1}{2}$ ,  $j_2 = h_2 + \frac{1}{2}$ , and  $j_3 = h_3$ . The relevant values for us are simply  $j_i = 1$ , and upon a global SL(2, $\mathbb{C}$ ) transformation to set  $x_3 = 1$ , we have

$$\langle \mathcal{O}_{\frac{1}{2}}(x_1)\mathcal{O}_{\frac{1}{2}}(x_2)\mathcal{O}_{1}^{\dagger}(1)\rangle = \frac{1}{N^{1/2}}\frac{1}{|1-x_1|^2|1-x_2|^2}.$$
 (4.98)

In order to compute the HLLLH correlator in the worldsheet coset models corresponding to the JMaRT backgrounds, we must sum (4.98) evaluated at all k<sup>th</sup>-roots of the insertion points. An explicit expression can be obtained following the arguments of Sec. 4.2.3, using the following generalisation of Eq. (4.77) and Eq. (4.79) for the case of three insertions,

$$\frac{\mathsf{k}}{|x_3|^{\frac{2(\alpha+\beta)}{\mathsf{k}}}} \sum_{r_{1,2}=0}^{\mathsf{k}-1} \frac{1}{|1-u_{13,r}|^{2\alpha}|1-u_{23,r}|^{2\beta}} = \sum_{r_{1,2,3}=0}^{\mathsf{k}-1} \frac{1}{|u_{3,r_3}-u_{1,r_1}|^{2\alpha}|u_{3,r_3}-u_{2,r_2}|^{2\beta}}, \quad (4.99)$$

where  $u_{i,r_i} = x_i^{1/k} e^{2\pi i r_i/k}$  and  $u_{j\ell,r} = (x_j/x_\ell)^{1/k} e^{2\pi i r/k}$  with  $\alpha = \beta = 1$ , one obtains

$$\langle \mathcal{O}_{\frac{1}{2}}(x_1)\mathcal{O}_{\frac{1}{2}}(x_2)\mathcal{O}_{1}^{\dagger}(1)\rangle_H = \frac{1}{\mathsf{k}^7} \frac{(x_1 x_2)^p |x_1 x_2|^{\frac{2}{\mathsf{k}}-1}}{|1-x_1|^2 |1-x_2|^2} \frac{1-|x_1|^2}{1-|x_1|^{\frac{2}{\mathsf{k}}}} \frac{1-|x_2|^2}{1-|x_2|^{\frac{2}{\mathsf{k}}}}.$$
(4.100)

The result (4.100) constitutes the first computation of a heavy-light worldsheet correlator with three light insertions probing a black hole microstate.

We now show that the same result can be obtained from the HCFT at the symmetric orbifold point. We follow the method used in [167, App. A] for the HLLH correlator reviewed in Section 4.2.3. The heavy states we use indicate that we should work in the k-twisted sector of the theory. The operators can be written in terms of the fermions introduced in Eq. (2.97). For the  $h = \frac{1}{2}$  chiral primaries, and for each strand of k copies of the theory, this simply reads

$$O_{\frac{1}{2}} = \sum_{r=1}^{k} O_{\frac{1}{2},(r)} = -\frac{i}{\sqrt{2}} \sum_{r=1}^{k} \psi_{(r)}^{+\dot{A}} \tilde{\psi}_{(r)}^{+\dot{B}} \epsilon_{\dot{A}\dot{B}} = -\frac{i}{\sqrt{2}} \sum_{\rho=0}^{k-1} \psi_{\rho}^{+\dot{A}} \tilde{\psi}_{\rho}^{+\dot{B}} \epsilon_{\dot{A}\dot{B}} , \qquad (4.101)$$

while for the h = 1 operator we find

$$O_{1}^{\dagger} = \sum_{r=1}^{k} O_{1,(r)} = \frac{1}{4} \sum_{r=1}^{k} \psi_{(r)}^{-\dot{A}} \psi_{(r)}^{-\dot{B}} \tilde{\psi}_{(r)}^{-\dot{C}} \tilde{\psi}_{(r)}^{-\dot{D}} \epsilon_{\dot{A}\dot{B}} \epsilon_{\dot{C}\dot{D}}$$
$$= \frac{1}{4k} \sum_{\rho_{i}=0}^{k-1} \delta_{\rho_{1}+\rho_{2},\rho_{3}+\rho_{4}} \psi_{\rho_{1}}^{-\dot{A}} \psi_{\rho_{2}}^{-\dot{B}} \tilde{\psi}_{\rho_{3}}^{-\dot{C}} \tilde{\psi}_{\rho_{4}}^{-\dot{D}} \epsilon_{\dot{A}\dot{B}} \epsilon_{\dot{C}\dot{D}}.$$
(4.102)

We will work in the bosonized language, in which

$$\psi_{\rho}^{+\dot{1}} = ie^{iH_{\rho}}, \qquad \psi_{\rho}^{-\dot{2}} = ie^{-iH_{\rho}}, \qquad \psi_{\rho}^{+\dot{2}} = e^{iK_{\rho}}, \qquad \psi_{\rho}^{-\dot{1}} = e^{-iK_{\rho}}, \qquad (4.103)$$

Here  $H_{\rho}$  and  $K_{\rho}$  are canonically normalized bosonic fields, in terms of which the (unit normalized) heavy states take the form [167]

$$|H\rangle = |s = \mathsf{k}p, \mathsf{k}\rangle = \left[\Sigma_{\mathsf{k}}\tilde{\Sigma}_{\mathsf{k}}\prod_{\rho=0}^{\mathsf{k}-1} e^{i\left(p+\frac{1}{2}-\frac{\rho}{\mathsf{k}}\right)(H_{\rho}+K_{\rho})} e^{i\left(\frac{1}{2}-\frac{\rho}{\mathsf{k}}\right)(\tilde{H}_{\rho}+\tilde{K}_{\rho})}\right]^{\frac{N}{\mathsf{k}}}|0\rangle, \qquad (4.104)$$

where  $\Sigma_k$  and  $\tilde{\Sigma}_k$  are the twist operators. Note that the contribution of  $\Sigma_k$  and  $\tilde{\Sigma}_k$  to the correlators will simply factorize, since the  $\psi_\rho$  fermions diagonalize the twisted boundary conditions. Choosing the labelling of the insertion points for later convenience, the

correlator to be computed is then

$$\langle O_{\frac{1}{2}}(x_1)O_{\frac{1}{2}}(x_2)O_{1}^{\dagger}(x_3)O_{H}(x_4)O_{H}^{\dagger}(x_5)\rangle = \frac{1}{\mathsf{k}}\prod_{\rho,\rho'=0}^{\mathsf{k}-1}\sum_{\rho_i=0}^{\mathsf{k}-1}\delta_{\rho_3+\rho_4,\rho_5+\rho_6}$$
(4.105)

$$\left\langle \left[ \psi_{\rho_{1}}^{+\dot{A}_{1}} \tilde{\psi}_{\rho_{1}}^{+\dot{B}_{1}} \epsilon_{\dot{A}_{1}\dot{B}_{1}} \right] (x_{1}) \left[ \psi_{\rho_{2}}^{+\dot{A}_{2}} \tilde{\psi}_{\rho_{2}}^{+\dot{B}_{2}} \epsilon_{\dot{A}_{2}\dot{B}_{2}} \right] (x_{2}) \left[ \psi_{\rho_{3}}^{-\dot{A}_{3}} \psi_{\rho_{4}}^{-\dot{B}_{3}} \tilde{\psi}_{\rho_{5}}^{-\dot{C}_{3}} \tilde{\psi}_{\rho_{6}}^{-\dot{D}_{3}} \epsilon_{\dot{A}_{3}\dot{B}_{3}} \epsilon_{\dot{C}_{3}\dot{D}_{3}} \right] (x_{3}) \\ e^{i \left( p + \frac{1}{2} - \frac{\rho}{k} \right) (H_{\rho} + K_{\rho})} (x_{4}) e^{-i \left( p + \frac{1}{2} - \frac{\rho'}{k} \right) (H_{\rho'} + K_{\rho'})} (x_{5}) e^{i \left( \frac{1}{2} - \frac{\rho}{k} \right) (\ddot{H}_{\rho} + \tilde{K}_{\rho})} (x_{4}) e^{-i \left( \frac{1}{2} - \frac{\rho'}{k} \right) (\ddot{H}_{\rho'} + \tilde{K}_{\rho'})} (x_{5}) \right\rangle.$$

Clearly, charge conservation implies  $\rho = \rho'$ . For the same reason, the correlator vanishes unless  $\rho_1 = \rho_3$  and  $\rho_2 = \rho_4$ , or  $\rho_1 = \rho_4$  and  $\rho_2 = \rho_3$ , or both. An analogous statement holds with  $\rho_3$ ,  $\rho_4$  replaced by  $\rho_5$ ,  $\rho_6$ , hence all contributions trivially satisfy the  $\rho_3 + \rho_4 = \rho_5 + \rho_6$  constraint. Consequently, we only really need to sum over all possible values of, say,  $\rho_1$  and  $\rho_2$ , and also compute the product over  $\rho$ . In this way, up to an irrelevant numerical factor, the holomorphic free field contractions give

$$\sum_{\rho_{1},\rho_{2}=0}^{k-1} \frac{1}{|x_{45}^{2h_{H}}x_{13}x_{23}|^{2}} \left(\frac{x_{41}x_{35}}{x_{51}x_{34}}\right)^{p+\frac{1}{2}-\frac{\rho_{1}}{k}} \left(\frac{x_{42}x_{35}}{x_{52}x_{34}}\right)^{p+\frac{1}{2}-\frac{\rho_{2}}{k}} \left(\frac{\bar{x}_{41}\bar{x}_{35}}{\bar{x}_{51}\bar{x}_{34}}\right)^{\frac{1}{2}-\frac{\rho_{1}}{k}} \left(\frac{\bar{x}_{42}\bar{x}_{35}}{\bar{x}_{52}\bar{x}_{34}}\right)^{\frac{1}{2}-\frac{\rho_{2}}{k}}$$
(4.106)

where  $h_H$  is the weight of the heavy state.

We can now take  $x_3 \rightarrow 1$ ,  $x_4 \rightarrow 0$  and  $x_5 \rightarrow \infty$ , and perform the sums over  $\rho_1$  and  $\rho_2$  explicitly. Upon doing so, we find that the structure of this orbifold CFT correlator Eq. (4.106) precisely matches the worldsheet correlator (4.100).

#### 4.3.5 Hawking radiation from the worldsheet

As a final application of our results, we now use the HLLH correlator (4.81) to compute the amplitude that describes the analogue of the Hawking radiation process for the JMaRT backgrounds [140, 141, 93, 94]. In the bulk, this process is ergoregion radiation, which is a feature of the full asymptotically flat JMaRT solutions [147]. The ergoregion does not survive the fivebrane decoupling limit [52] or AdS<sub>3</sub> decoupling limit, however aspects of the process can still be studied quantitatively in those limits. This process has been interpreted as an enhanced analogue of Hawking radiation, since both are described by the same microscopic process in the holographic CFT [140]. Indeed, acting on a thermal state, this vertex operator gives precisely the spectrum and rate of Hawking radiation of the corresponding black hole, while acting on the states dual to the JMaRT solutions yields their characteristic spectrum and rate of emission [140, 141, 93, 94].

The emission spectrum and rate for general k,  $s, \bar{s}$  was computed in supergravity and symmetric product orbifold CFT in [94], building on the results of [140, 141, 93]. We will reproduce these results from the worldsheet CFT.

We start with a specific HLLH correlator in which the light operators are given by minimally coupled scalars in six dimensions, after reducing on the  $\mathbb{T}^4$ . The corresponding vertex operators were defined in Eq. (4.13). These are not chiral primaries of the boundary theory, but are their superdescendants within the short multiplet, so the holographic correlator arising in the AdS<sub>3</sub> limit is easily computed by using the techniques outlined in the previous sections. The amplitude of interest involves an initial state consisting of a probe excitation on top of the JMaRT background, a vertex operator  $\mathcal{V}$  associated to a light insertion, and a final state given by the black hole microstate. Schematically we have

$$\langle s, \bar{s}, \mathsf{k} | \mathcal{O}(x) | s, \bar{s}, \mathsf{k} + \mathsf{probe} \rangle = \langle s, \bar{s}, \mathsf{k} | \mathcal{O}(x) \mathcal{O}^{\dagger}(0) | s, \bar{s}, \mathsf{k} \rangle .$$
(4.107)

To begin with, we work with k = 1. Up to an overall sign, and considering the lowest energy state, the holomorphic part of the amplitude for the Hawking emission of a single quanta of dimension  $h = \frac{l}{2} + 1$  and whose corresponding vertex operator has charge  $m' = k - \frac{l}{2}$  reads [93]

$$\mathcal{A}_{L}(x) = \frac{1}{x^{(1+\alpha)\frac{l}{2} - \alpha k + 1}} = \frac{1}{x^{\frac{l}{2} + 1 - \alpha(k - \frac{l}{2})}}.$$
(4.108)

Here  $\frac{1}{2}$  denotes the total angular momentum of the probe on the S<sup>3</sup> part of the geometry, while *k* is the number of  $\mathbb{J}_0^+$  operators acting on the state with the lowest projection, appearing in the definition of the vertex operator. To compare their computation with our worldsheet result, one uses the following (notation) map:

$$k - \frac{l}{2} \mapsto m'$$
,  $\frac{l}{2} + 1 \mapsto h$ ,  $\alpha \mapsto l_2 = m + n = 2s + 1$ . (4.109)

Taking care of the cylinder-to-plane conversion factor  $x^{-\frac{l}{2}-1}$ , one obtains

$$\mathcal{A}_L(x) = \frac{1}{x^{2h - m'(2s+1)}}.$$
(4.110)

We now perform the analogous computation in the worldsheet cosets. From Eq. (4.107), in the worldsheet formalism all we need to do is to insert the second operator at the boundary origin, i.e. to take the  $x_2 \rightarrow 0$  limit in Eq. (4.81) (with k = 1 for now). This gives

$$\lim_{x_2 \to 0} x_2^{m'(2s+1)} \bar{x}_2^{\bar{m}'(2\bar{s}+1)} \langle s, \bar{s}, 1 | \mathcal{O}_h^{(m',\bar{m}')}(x_1) \mathcal{O}_h^{(m',\bar{m}')^{\dagger}}(x_2) | s, \bar{s}, 1 \rangle$$
$$= \frac{1}{x^{2h-m'(2s+1)}} \frac{1}{\bar{x}^{2h-\bar{m}'(2\bar{s}+1)}}, \qquad (4.111)$$

in agreement with (4.110) upon including the antiholomorphic contribution.

The procedure is analogous for general k, s,  $\bar{s}$ . We again evaluate the amplitude for  $x_2 \rightarrow 0$  by including the appropriate Jacobian factor for the light state, and obtain

$$\begin{split} \lim_{x_{2}\to0} \mathsf{k}^{2h} x_{2}^{m_{2}'\frac{(2s+1)}{\mathsf{k}}+h(1-\frac{1}{\mathsf{k}})} \bar{x}_{2}^{\bar{m}_{2}'\frac{(2s+1)}{\mathsf{k}}+h(1-\frac{1}{\mathsf{k}})} \langle s,\bar{s},\mathsf{k}| \mathcal{O}_{h}^{(m',\bar{m}')}(x_{1}) \mathcal{O}_{h}^{(m',\bar{m}')\dagger}(x_{2})|s,\bar{s},\mathsf{k}\rangle \\ &= \frac{1}{\mathsf{k}^{2h+2}} \sum_{r_{1}=0}^{\mathsf{k}-1} \frac{e^{2\pi i \frac{r_{1}}{\mathsf{k}}(m'(2s+1)-\bar{m}'(2\bar{s}+1))}}{x_{1}^{h(1+\frac{1}{\mathsf{k}})-m'\frac{(2s+1)}{\mathsf{k}}} \bar{x}_{1}^{h(1+\frac{1}{\mathsf{k}})-\bar{m}'\frac{(2\bar{s}+1)}{\mathsf{k}}}} \sum_{r_{2}=0}^{\mathsf{k}-1} e^{2\pi i \frac{r_{2}}{\mathsf{k}}(-m'(2s+1)+\bar{m}'(2\bar{s}+1))} \\ &= \frac{1}{\mathsf{k}^{2h}} \frac{1}{x^{h(1+\frac{1}{\mathsf{k}})-m'\frac{(2s+1)}{\mathsf{k}}} \bar{x}^{h(1+\frac{1}{\mathsf{k}})-\bar{m}'\frac{(2\bar{s}+1)}{\mathsf{k}}}} \sum_{\ell\in\mathbb{Z}} \delta_{m'(2s+1)-\bar{m}'(2\bar{s}+1),\mathsf{k}\ell}, \end{split}$$
(4.112)

where in the first equality we have exchanged the finite sum with the limit, and  $x = x_1$ ,  $m = m_1 = -m_2$ . When k = 1, this reduces to Eq. (4.111). The Kronecker comb enforces the constraint  $(2s + 1)m' - (2\bar{s} + 1)\bar{m}' \in k\mathbb{Z}$ , which is a direct consequence of the difference between left and right null-gauge constraints (4.45) in the regime of interest. Moreover, by first multiplying the correlator in Eq. (4.81) by  $x^n \bar{x}^{\bar{n}}$  we can also consider descendant insertions. This condition is in agreement with the results present in [141, 94] (see also [52]), where our  $n_y$  has to be identified with their  $\lambda$  from the supergravity analysis.

When considering the case of multi-particle emission, the above amplitude must be multiplied by a combinatorial factor, as explained in [140, 93, 141]. To obtain the emission rate, one needs to consider the unit amplitude evaluated at  $(x, \bar{x}) = (1, 1)$ , implying that the spatial dependence trivialises. Nevertheless, the crucial feature related to the presence of the prefactor  $k^{-2h}$ , which enters the final expression of the emission rate<sup>46</sup>, is reproduced by (4.112).

Even though the spatial dependence of the two-point function Eq. (4.112) plays a trivial role in the emission rate, the power of x has a precise meaning in terms of the energy spectrum of the nearly unstable Hawking quanta [93, 94]. Indeed, consider the holomorphic part of the energy of these modes. In the conventions of [94], the corresponding spectrum reads

$$\omega \, \mathsf{k} R_y \,=\, \frac{1}{2} \alpha \mathsf{k} (m_\phi - m_\psi) - \frac{1}{2} \bar{\alpha} \mathsf{k} (m_\phi + m_\psi) - 2 \left(\frac{l}{2} + 1\right) \,. \tag{4.113}$$

In our notation,  $(m_{\phi} - m_{\psi}) = 2m'$ ,  $(m_{\phi} + m_{\psi}) = -2\bar{m}'$ , and  $\frac{l}{2} + 1 = h$ , so this becomes

$$-\omega R_y = \frac{2h}{k} - \alpha \, m' - \bar{\alpha} \, \bar{m}' \,, \qquad (4.114)$$

<sup>&</sup>lt;sup>46</sup>To compare to the final results of [141, 94] one must include the additional factor  $\sqrt{k\nu}$ , where  $\nu$  is related to the Bose enhancement, which is not visible for a single-particle process.

where  $\alpha$ ,  $\bar{\alpha}$  are the same as in Eq. (2.89). Finally, taking care of the cylinder-to-plane conversion factor for a field of spacetime conformal dimension *h*, we obtain

$$-\omega R_y = 2h\left(1+\frac{1}{k}\right) - \alpha \, m' - \bar{\alpha} \, \bar{m}' \,. \tag{4.115}$$

The RHS is exactly the sum of the exponents of *x* and  $\bar{x}$  in Eq. (4.112). Furthermore, we note that this relation is precisely the sum of the left and right bosonic null gauge constraints Eq. (4.45) for discrete states with  $n = \bar{n} = 0$ .

The emission takes place when the energy is positive,  $\omega > 0$ , and corresponds to quanta leaving the AdS region; in a near-decoupling limit, these quanta escape into an asymptotically flat region. Indeed, the exponent of *x* in Eq. (4.112) becomes positive and the amplitude diverges at large *x*, such that the energy indeed turns from negative to positive. This is consistent with the description of the ergoregion radiation process as pair creation [185].

# 4.4 Discussion

In this chapter we have computed a large set of worldsheet correlators describing the dynamics of light modes probing a class of highly excited supergravity backgrounds, the JMaRT solutions, in the fivebrane decoupling limit. The results are exact in  $\alpha'$  and were obtained by exploiting the solvability of the null-gauged WZW models corresponding to these backgrounds.

These coset models provide a powerful method to calculate HLLH correlators, since the heavy states are already taken into account in the worldsheet CFT itself. Thus spacetime HLLH correlators are two-point functions on the worldsheet, which can be computed once the vertex operators have been constructed.

We constructed physical vertex operators in both NS and R sectors, and then computed several families of correlators in the full coset models. We primarily focused on short strings belonging to discrete representations of the affine  $SL(2,\mathbb{R})$  algebra, as well as a tower of modes generated by worldsheet spectral flow. Our main techniques can also be employed in more general sectors of the theory.

In the IR AdS<sub>3</sub> limit, due to the non-trivial gauging, the identification of the *x* variable dual to the local coordinate of the holographic CFT requires some care. Once we made this identification, we computed several non-trivial HLLH correlators explicitly, and analyzed them in the context of  $AdS_3/CFT_2$ .

Vertex operators that are local on the AdS<sub>3</sub> boundary are constructed by summing over all allowed values of the spacetime modes. An important step in our analysis consists of

identifying these modes. We chose a gauge in which the IR AdS<sub>3</sub> boundary coordinates are (t, y) of the timelike  $\mathbb{R}$  and spacelike S<sup>1</sup> directions of the (10+2)-dimensional model before gauging. We therefore identified the spacetime mode indices with the quantum numbers  $m_y$  and  $\bar{m}_y$  defined in (4.46). Then the gauge constraints (4.12) satisfied by the physical states imply that the  $m_y$  mode numbers take values in  $\mathbb{Z}/k$ . This is how the worldsheet coset models capture the fact that when k > 1, the heavy background states of the symmetric product orbifold CFT are in the k-twisted sector [96, 94].

We observed that, at large N, several correlators agree exactly between worldsheet and symmetric product orbifold CFT. The fact that our correlators are exact in  $\alpha'$  significantly strengthens previous results that compared HLLH correlators between the separate supergravity and symmetric product orbifold CFT regimes.

To demonstrate our method, we presented a detailed example with an  $(h, \bar{h}) = (\frac{1}{2}, \frac{1}{2})$  chiral primary. The worldsheet correlator involves a non-trivial structure in terms of the boundary coordinate x, Eq. (4.56). When the background is BPS, the correlator agrees precisely with the supergravity and symmetric product orbifold CFT correlators computed in [167]. The non-BPS JMaRT backgrounds were not considered in [167], however we demonstrated that the agreement extends also to those backgrounds.

Similarly to correlators on the background of the global AdS<sub>3</sub> vacuum, the holomorphic and antiholomorphic sectors are related through the constraint  $m_y - \bar{m}_y = n_y$ , where  $n_y$  is the quantized momentum on the *y* circle. Thus, while the spacetime modes  $m_y$ ,  $\bar{m}_y$  are fractional, their difference must be an integer. This mirrors the  $m - \bar{m} \in \mathbb{Z}$ condition in the SL(2, $\mathbb{R}$ )/U(1) cigar coset and in global AdS<sub>3</sub>, which ensures that the wavefunctions are single-valued. In our models, the difference of the left and right gauge constraints leads to the mod k condition in Eq. (4.68), constraining which of the SL(2, $\mathbb{R}$ ) modes can contribute. The HLLH correlator is then obtained by summing over a specific linear combination of *m*-basis worldsheet two-point functions.

Our analysis of these correlators involving the h = 1/2 light operator indicated a way to obtain similar expressions for more general correlators. We considered general massless vertex operators, which correspond to symmetric product orbifold CFT operators in short multiplets whose top component is a chiral primary of arbitrary weight h, including those that live in twisted sectors. We computed all HLLH correlators where the light operators are massless, and where the heavy states correspond to any of the general family of orbifolded JMaRT configurations, including their BPS limits. The result assumes a remarkably simple form, presented in Eq. (4.81). It is built from three distinct factors: (1) the global  $AdS_3 \times S^3$  *vacuum* two-point function of the light operators inserted at the k-th roots of the original insertion points  $x_i$ , (2) the Jacobian factor associated with the corresponding change of coordinates, and (3) an additional factor coming from the way in which operators of definite R-charge transform under spectral flow. The product of these factors is then summed over all such roots. This structure reflects that one can formulate the computation in a k-fold covering space of the target space.

We then obtained a similar expression for all higher-point functions of the schematic form  $\langle H|O_1(x_1, \bar{x}_1) \dots O_n(x_n, \bar{x}_n)|H\rangle$ , with heavy JMaRT states, and *n* massless insertions. This is presented in Eq. (4.94). We expect this to be valid for an arbitrary number of massless insertions of weights  $h_i$  and charges  $m'_i$  and  $\bar{m}'_i$ , and also arbitrary parameters  $(k, s, \bar{s})$  for which a consistent background exists. In this way, we have provided a recipe for computing such (n + 2)-point heavy-light correlators in terms of *n*-point global AdS<sub>3</sub> × S<sup>3</sup> vacuum correlation functions of the corresponding light insertions.

It is known that vacuum two- and three-point functions of chiral primary operators are protected [184]. We therefore conjectured that heavy-light correlators in JMaRT heavy states are protected whenever the corresponding vacuum correlator in our general formula (4.94) is protected. We investigated a particular HLLLH five-point function—the first of its kind in the literature—and found that worldsheet and symmetric product results agree. We leave a more general investigation of this proposal to future work.

As an application, we have shown that our results describe the analog of the Hawking radiation process for the general family of non-BPS JMaRT black hole microstates, generalizing the analysis in [93, 46, 94].

In addition to these main results, our work has clarified some important technical details. For instance, since the full asymptotically linear dilaton JMaRT backgrounds do not have  $AdS_3 \times S^3$  isometries, the  $SL(2,\mathbb{R})$  and SU(2) raising and lowering operators  $J^{\pm}$ ,  $K^{\pm}$  of the (10+2)-dimensional ungauged worldsheet model do not commute with the gauging. Thus, the NS sector vertex operators of the cosets do not have well-defined  $SL(2,\mathbb{R})$  or SU(2) spins, see for instance Eq. (4.19). The same holds for the chirality quantum number  $\varepsilon$  in the R sector, as discussed around Eq. (4.24).

The absence of the SL(2, $\mathbb{R}$ ) spin has important implications also from the holographic point of view. It underlines the fact that the construction of *x*-basis operators is only appropriate in the AdS<sub>3</sub> limit, and breaks down otherwise. The breakdown of the *x* coordinate is a signal of the non-locality of the non-gravitational little string theory that lives on the worldvolume of the NS5 branes, dual to the full asymptotically lineardilaton models. Thus the states we have constructed contain valuable information about the dual LST and, more generally, about non-AdS holography [145, 143, 144]. We have nevertheless demonstrated how, in the AdS<sub>3</sub> limit, our vertex operators acquire definite spins and reduce to the appropriate expressions.

Our results suggest several directions for future investigations. First, it would be interesting to compute more general worldsheet correlators, both in the AdS<sub>3</sub> limit and in the full models. Our correlators are likely to generalize to a larger set of worldsheet vertex operators that correspond to operators in the symmetric product orbifold CFT that transform nicely under spacetime spectral flow [186]. In global AdS<sub>3</sub>, correlators are known to involve a highly non-trivial structure related to the non-conservation of the spectral flow number [103]. More generally, one would like to describe the physics of long/winding strings and their correlators in these backgrounds. A number of interesting techniques recently developed in [38, 179] (for the bosonic case) are likely to have interesting implications for computations in the coset theories, for which SL(2, $\mathbb{R}$ ) constitutes a crucial building block.

It would also be interesting to study such correlators by using conformal perturbation theory on top of a putative dual CFT explicitly associated to the NSNS singular point [50] of the moduli space, defined along the lines of [174, 90]. Doing so would require an understanding of how to define the JMaRT heavy states in such a theory. Separately, it would be interesting to investigate the case  $n_5 = 1$ , which would require going beyond the RNS formalism, as done in related recent developments [187, 87, 188]. Here one should go though the coset construction starting with the supergroup PSU(1,1—2).

In the full asymptotically linear dilaton models, more general correlators can be computed by using the vertex operators constructed in Section 4.1. However, a shift in perspective will be needed, since the *x* coordinate seems unlikely to be of any use in this regime. Although *a priori* in our case it is more natural to work in the *m*-basis, it seems plausible to relate our results to the momentum-space correlators studied in [143, 145], see also [189]. In those papers, the authors work with a related null-gauged model, and further interpret their holographic LST correlators in terms of an irrelevant (single-trace)  $T\bar{T}$ -deformation of the IR CFT<sub>2</sub>.

Separately, it will be interesting to investigate our proposal for the subset of heavy-light correlators that we expect to be protected by considering the dual computations in the symmetric product orbifold CFT.

Last, but not least, one would like to explore further how these correlation functions encode more detailed information about the physics of the microstate backgrounds we are working with. For instance, two-point functions are expected to probe the multipole ratios of the geometry [190, 191], while certain worldsheet three-point functions should be related to the Penrose process in the JMaRT backgrounds [192].

Although the JMaRT backgrounds are atypical microstates, the HLLH correlators we have computed approach black-hole-like behaviour at large k, reflecting the properties of the backgrounds in this limit. We expect that the techniques developed in this work will help further the study of more typical black hole microstates in string theory.

# Chapter 5

# A proof for string three-point functions in AdS<sub>3</sub>

In most known cases of the AdS/CFT duality, the bulk theory is approximated by supergravity, due to the notorious difficulty in performing stringy computations. The case of string theory in AdS<sub>3</sub> with pure NS fluxes is a notable exception. The dynamics of closed strings propagating on this background can be described at the worldsheet level in terms of the Wess-Zumino-Witten (WZW) model built upon the universal cover of SL(2,  $\mathbb{R}$ ). This model is believed to be exactly solvable, hence providing a concrete scenario in which the AdS/CFT duality might be proven at finite 't Hooft coupling.

The worldsheet CFT, being both Lorentzian and non-compact, enjoys a number of unusual features. Perhaps the most important one is the non-trivial action of the so-called spectral flow outer-automorphisms of the affine  $sl(2,\mathbb{R})_k$  algebra. Spectral flow plays a central role in the determination of the string spectrum and partition function [101, 102]. In a semi-classical description of long string states, the spectral flow charge  $\omega$  can be thought of as a winding number around the asymptotic boundary of AdS<sub>3</sub>. However, it is not a conserved quantity since the associated circle becomes contractible in the AdS<sub>3</sub> interior.

Spectral flow also introduces important complications, especially concerning the computation of worldsheet correlation functions [103]. Indeed, while being Virasoro primaries, spectrally flowed vertex operators are *not* affine primaries. Their operator product expansions (OPEs) with the conserved currents become increasingly complicated with growing  $\omega$ , and contain many unknown terms. Hence, some conventional techniques of two-dimensional CFT can not be applied directly.

The interest in correlation functions of the SL(2, $\mathbb{R}$ )-WZW model is explained by their important holographic applications, especially in the context of superstring theory in AdS<sub>3</sub> × S<sup>3</sup> × T<sup>4</sup> (or K3) [78, 110, 100, 113], and for the dynamics of light probes in black

hole microstates [51, 52, 53, 54, 193, 1, 2, 3]. Worldsheet vertex operators are functions of the worldsheet coordinate z, and of an additional continuous parameter x, which plays the role of the holomorphic coordinate on the conformal boundary. Hence, z-integrated correlators are identified with n-point functions of local operators in the holographic CFT. Although the precise definition of the latter remains elusive, a concrete albeit per-turbative proposal was put forward recently in [90, 174].

The generators of the spacetime Virasoro algebra are in one-to-one correspondence with those of the  $sl(2, \mathbb{R})_k$  algebra, and the worldsheet quantum numbers determine the conformal weight *h* of dual CFT operators [78]. Suppressing the anti-holomorphic quantities, we denote the spectrally-flowed worldsheet vertex operators in the so-called *x*-basis as  $V_{ih}^{\omega}(x, z)$ , where *j* is the SL(2,  $\mathbb{R}$ ) spin, while *h* is the spacetime weight.

In this chapter, we consider genus-zero worldsheet correlators of spectrally flowed vertex operators. Recently, progress in the computation of three- and four-point functions has been made in [88, 38, 179], thanks to the systematic use of the constraints deriving from global and local symmetries. In particular, the latter imply a set of complicated linear recursion relations among correlators involving different assignments of spectral flow charges. These are difficult to derive in general. However, in [38] it has been shown that they can be recast in the form of partial differential equations thanks to the introduction of the so-called '*y*-variable'. In this new basis, a vertex operator is now defined as a coherent superposition of states with different spacetime weights, denoted as  $V_j^{\omega}(x, y, z)$ . The authors of [38] were able to infer a closed-form expression for all three-point functions

$$\langle V_{j_1}^{\omega_1}(x_1; y_1; z_1) V_{j_2}^{\omega_2}(x_2; y_2; z_2) V_{j_3}^{\omega_3}(x_3; y_3; z_3) \rangle .$$
(5.1)

This proposal can be understood intuitively from the existence, in some cases, of a holomorphic mapping of the worldsheet to the AdS<sub>3</sub> boundary, which has branching points of order  $\omega_i$  at each of the corresponding insertions. The same authors provided a closed-form expression for four-point functions in a follow-up work [179]. These proposals satisfy a number of highly non-trivial consistency checks. However, in both cases, these expressions were derived on a case-by-case basis, and for a finite set of sufficiently low spectral flow charges.

In this work, we provide a proof for the conjecture of [38] concerning the three-point functions (5.1). Our methods rely heavily on the so-called 'series identifications' of SL(2, $\mathbb{R}$ ), originally formulated in [175, 101], which constitute a set of isomorphisms among affine modules of the sl(2,  $\mathbb{R}$ )<sub>k</sub> algebra with adjacent spectral flow charges. The corresponding identities at the level of *y*-basis operators were obtained recently in [39]. This allows us to derive the differential equations satisfied by (and the resulting *y*-dependence of) all spectrally flowed three-point functions, including those that do not admit a holomorphic cover.

The structure of the chapter is as follows. In section 5.1, we establish our conventions and review the derivation of the recursion relations satisfied by the SL(2, $\mathbb{R}$ ) correlation functions of [88]. We also introduce the *y*-basis operators of [38] together with their conjecture for three-point functions. Our main results are presented in section 5.2, where derive the partial differential equations satisfied by all *y*-basis three-point functions, and show that all the corresponding solutions are compatible with the proposal of [38]. In particular, generic odd parity correlators are considered in section 5.2.1, while even parity correlators are obtained in section 5.2.2. Edge cases and correlators with unflowed insertions are treated in sections 5.2.3 and 5.2.4, respectively. Finally, in section 5.2.5 we fix the *y*-independent normalisation factors, following the arguments presented in [39]. We conclude with some discussions and outlook for future work in section 5.3.

# 5.1 Conventions and brief review of the conjecture

Let us consider the bosonic SL(2, $\mathbb{R}$ )-WZW model at level k > 3. In this section, we introduce the spectrum and the associated vertex operators. We also briefly review the analysis of [88] leading to the recursion relations satisfied by the correlation functions of the model, and present the conjecture put forward in [38] for spectrally flowed three-point functions.

#### 5.1.1 Vertex operators

We focus mainly on the holomorphic sector. Compared to previous chapters, the present focuses on the purely bosonic sector of the theory. For this reason, in this chapter the bosonic currents will be denoted by *J*.

The conserved currents satisfy

$$J^{a}(z)J^{b}(w) \sim \frac{\eta^{ab}k/2}{(z-w)^{2}} + \frac{f^{ab}_{\ c}J^{c}(w)}{z-w},$$
(5.2)

with a = +, -, 3,  $\eta^{+-} = -2\eta^{33} = 2$ ,  $f^{+-}_{3} = -2$  and  $f^{3+}_{+} = -f^{3-}_{-} = 1$ . In the so-called *x*-basis, vertex operators are denoted as  $V_{jh}^{\omega}(x, z)$ . They depend both on the worldsheet coordinate *z* and on *x*, the coordinate on the boundary of AdS<sub>3</sub> associated with the holographic CFT. Moreover, *h* and *j* denote the spacetime weight and the SL(2,  $\mathbb{R}$ ) spin, respectively. The worldsheet conformal weight is

$$\Delta = -\frac{j(j-1)}{k-2} - h\omega + \frac{k}{4}\omega^2, \qquad (5.3)$$

while  $\omega \ge 0$  is the spectral flow charge.

Two types of states must be taken into account [101]. For operators defined upon unflowed states in the continuous representations  $C_i^{\alpha}$ , the relevant quantum numbers are

$$j \in \frac{1}{2} + i\mathbb{R}$$
 and  $m = \alpha \pm n$ , with  $\alpha \in [0,1)$  and  $n \in \mathbb{N}_0$ , (5.4)

where *m* stands for the spin projection of the corresponding unflowed state. On the other hand, for the unflowed discrete highest/lowest-weight representations  $D_j^{\pm}$  we have

$$\frac{1}{2} < j < \frac{k-1}{2} \quad \text{and} \quad m = \pm (j+n), \quad \text{with} \quad j \in \mathbb{R} \quad \text{and} \quad n \in \mathbb{N}_0.$$
 (5.5)

In both cases we have  $h = m + \frac{k}{2}\omega$  when  $\omega > 0$ , while h = j for  $\omega = 0$ .

The vertex operators are defined by means of their OPEs with the currents:

$$J^{+}(w)V_{jh}^{\omega}(x,z) = \sum_{n=1}^{\omega+1} \frac{\left(J_{n-1}^{+}V_{jh}^{\omega}\right)(x,z)}{(w-z)^{n}} + \cdots,$$
(5.6a)

$$J^{3}(w)V_{jh}^{\omega}(x,z) = x\sum_{n=2}^{\omega+1} \frac{\left(J_{n-1}^{+}V_{jh}^{\omega}\right)(x,z)}{(w-z)^{n}} + \frac{\left(J_{0}^{3}V_{jh}^{\omega}\right)(x,z)}{(w-z)} + \cdots,$$
(5.6b)

$$J^{-}(w)V_{jh}^{\omega}(x,z) = x^{2}\sum_{n=2}^{\omega+1} \frac{\left(J_{n-1}^{+}V_{jh}^{\omega}\right)(x,z)}{(w-z)^{n}} + \frac{\left(J_{0}^{-}V_{jh}^{\omega}\right)(x,z)}{(w-z)} + \cdots, \quad (5.6c)$$

where the ellipsis indicates higher order terms in (w - z). Unflowed vertex operators will be denoted by  $V_i(x, z)$ . The zero modes act as differential operators in x,

$$\left(J_0^+ V_{jh}^\omega\right)(x,z) = \partial_x V_{jh}^\omega(x,z), \qquad (5.7a)$$

$$\left(J_0^3 V_{jh}^{\omega}\right)(x,z) = (x\partial_x + h)V_{jh}^{\omega}(x,z), \qquad (5.7b)$$

$$\left(J_0^- V_{jh}^\omega\right)(x,z) = (x^2 \partial_x + 2hx) V_{jh}^\omega(x,z), \qquad (5.7c)$$

while

$$\left(J_{\pm\omega}^{\pm}V_{jh}^{\omega}\right)(x,z) = \left[h - \frac{k}{2}\omega \pm (1-j)\right]V_{j,h\pm1}^{\omega}(x,z).$$
(5.8)

Importantly, in terms of the currents

$$J^{+}(x,z) = J^{+}(z), \quad J^{3}(x,z) = J^{3}(z) - xJ^{+}(z), \quad J^{-}(x,z) = J^{-}(z) - 2xJ^{3}(z) + x^{2}J^{+}(z)$$
(5.9)

we get

$$I^{3}(x,w)V_{jh}^{\omega}(x,z) = \frac{h}{(w-z)}V_{jh}^{\omega}(x,z) + \cdots,$$
 (5.10a)

$$J^{-}(x,w)V_{jh}^{\omega}(x,z) = (w-z)^{\omega-1} \left(J_{-w}^{-}V_{jh}^{\omega}\right)(x,z) + \cdots .$$
 (5.10b)

An alternative (equivalent) definition was given recently in [39], based on [103]. This is to be understood as a point-splitting procedure between the corresponding unflowed vertex and the so-called generalized spectral flow operator  $V_{\frac{k}{2},\frac{k}{2}\omega}^{\omega-1}(x,z)$ , and reads

$$V_{jh}^{\omega}(x,z) = \lim_{\varepsilon,\bar{\varepsilon}\to 0} \varepsilon^{m\omega} \bar{\varepsilon}^{\bar{m}\omega} \int d^2y \, y^{j-m-1} \bar{y}^{j-\bar{m}-1} V_j(x+y,z+\varepsilon) V_{\frac{k}{2},\frac{k}{2}\omega}^{\omega-1}(x,z) \,.$$
(5.11)

We will come back to this shortly in section 5.1.3.

#### 5.1.2 **Recursion relations among correlators**

We now discuss the correlation functions of the model. It was shown in [88] that they must satisfy a set of recursion relations. Let us briefly review how this works for three-point functions of the form

$$F = \left\langle \prod_{j=1}^{3} V_{j_j h_j}^{\omega_j}(x_j, z_j) \right\rangle.$$
(5.12)

We define

$$F_n^i = \left\langle \left( J_n^+ V_{j_i h_i}^{\omega_i} \right) (x_i, z_i) \prod_{j \neq i} V_{j_j h_j}^{\omega_j} (x_j, z_j) \right\rangle,$$
(5.13)

so that, in particular,

$$F_0^i = \partial_{x_i} F \,, \tag{5.14}$$

and

$$F_{\omega_i}^i = \left(h_i - \frac{k}{2}\omega_i + 1 - j_i\right) \left\langle V_{j_i,h_i+1}^{\omega_i}(x_i, z_i) \prod_{j \neq i} V_{j_jh_j}^{\omega_j}(x_j, z_j) \right\rangle,$$
(5.15)

as implied by Eqs. (5.7a) and (5.8) respectively. Note that all  $F_n^i$  with  $n = 1, ..., \omega_i - 1$  are, in principle, unknown.

By using the OPEs in Eq. (5.6), one finds that correlators involving a current insertion can be expanded as

$$\left\langle J^{+}(z)\prod_{j=1}^{3}V_{j_{j}h_{j}}^{\omega_{j}}(x_{j},z_{j})\right\rangle = \sum_{i=1}^{3}\left[\frac{\partial_{x_{i}}F}{z-z_{i}} + \sum_{n=1}^{\omega_{i}}\frac{F_{n}^{i}}{(z-z_{i})^{n+1}}\right] + \cdots,$$
(5.16a)

$$\left\langle J^{3}(z)\prod_{j=1}^{3}V_{j_{j}h_{j}}^{\omega_{j}}(x_{j},z_{j})\right\rangle = \sum_{i=1}^{3}\left[\frac{(h_{i}+x_{i}\partial_{x_{i}})F}{z-z_{i}} + \sum_{n=1}^{\omega_{i}}\frac{x_{i}F_{n}^{i}}{(z-z_{i})^{n+1}}\right] + \cdots, \quad (5.16b)$$

$$\left\langle J^{-}(z)\prod_{j=1}^{3}V_{j_{j}h_{j}}^{\omega_{j}}(x_{j},z_{j})\right\rangle = \sum_{i=1}^{3}\left[\frac{(2h_{i}x_{i}+x_{i}^{2}\partial_{x_{i}})F}{z-z_{i}} + \sum_{n=1}^{\omega_{i}}\frac{x_{i}^{2}F_{n}^{i}}{(z-z_{i})^{n+1}}\right] + \cdots (5.16c)$$

Combining these expressions, we get

$$G_{j}(z) \equiv \left\langle J^{-}(x_{j},z) \prod_{l=1}^{3} V_{j_{l}h_{l}}^{\omega_{l}}(x_{l},z_{l}) \right\rangle = \sum_{i \neq j} \left[ \frac{(2h_{i}x_{ij} + x_{ij}^{2}\partial_{x_{i}})F}{z - z_{i}} + \sum_{n=1}^{\omega_{i}} \frac{x_{ij}^{2}F_{n}^{i}}{(z - z_{i})^{n+1}} \right] + \cdots$$
(5.17)

where  $x_{ij} = x_i - x_j$ .

On the other hand, Eq. (5.10b) imposes stringent restrictions on the behavior of  $G_j(z)$  when z is close to  $z_j$ . More precisely, in this regime, we must have

$$(z-z_j)^{1-\omega_j}G_j(z) = \left(h_j - \frac{k}{2}\omega_j - 1 + j_j\right) \left\langle V_{j_j,h_j-1}^{\omega_j}(x_j,z_j) \prod_{i \neq j} V_{j_ih_i}^{\omega_i}(x_i,z_i) \right\rangle + \cdots,$$
(5.18)

where we have used (5.8). As it turns out, the regularity of  $(z - z_j)^{1-\omega_j}G_j(z)$  at  $z = z_j$  as implied by (5.18) for all j = 1, 2, 3 provides enough information to solve for all the unknown  $F_n^i$  in terms of F, its  $x_i$ -derivatives, and  $F_{\omega_i}^i$ . Upon inserting the resulting expressions into Eq. (5.18), one obtains a set of complicated linear relations between correlators involving spectrally flowed vertex operators with consecutive spacetime weights and their  $x_i$  derivatives. Note that the latter are under control since the global Ward identities uniquely fix the  $x_i$ -dependence of all such correlators.

#### 5.1.3 The conjectured solution

The non-trivial OPEs in Eqs. (5.6) render the computation of correlation functions involving spectrally flowed insertions quite complicated. It will be useful to work with somewhat unusual linear combinations of the operators  $V_{jh}^{\omega}(x, z)$  with the same *j* and  $\omega$  but different values of *h*. These are the *y*-basis operators  $V_j^{\omega}(x, y, z)$  introduced in [38]. In [39], these were shown to be precisely the integrands on the RHS of Eq. (5.11), namely

$$V_{j}^{\omega}(x,y,z) \equiv \lim_{\varepsilon,\bar{\varepsilon}\to 0} |\varepsilon|^{2j\omega} V_{j}(x+y\varepsilon^{\omega},z+\varepsilon) V_{\frac{k}{2},\frac{k}{2}\omega}^{\omega-1}(x,z).$$
(5.19)

This can be understood directly from Eq. (5.11) above. Indeed, upon changing variables  $y \rightarrow y \varepsilon^{\omega}$  this can be re-written as

$$V_{jh}^{\omega}(x,z) = \int d^2y \, y^{j-m-1} \bar{y}^{j-\bar{m}-1} V_j^{\omega}(x,y,z) \,, \tag{5.20}$$

which coincides with the so-called *y*-transform of [38]. More details can be found in [38, 39].

The OPEs of *y*-basis operators with the conserved currents are

$$J^{+}(w)V_{j}^{\omega}(x,y,z) = \sum_{n=1}^{\omega+1} \frac{\left(J_{n-1}^{+}V_{jh}^{\omega}\right)(x,y,z)}{(w-z)^{n}} + \cdots,$$
(5.21a)

$$J^{3}(x,w)V_{j}^{\omega}(x,y,z) = \frac{y\partial_{y} + j + \frac{k}{2}\omega}{(w-z)}V_{j}^{\omega}(x,y,z) + \cdots,$$
(5.21b)

$$J^{-}(x,w)V_{j}^{\omega}(x,y,z) = (w-z)^{\omega-1} \left(J_{-w}^{-}V_{j}^{\omega}\right)(x,y,z) + \cdots .$$
 (5.21c)

While the zero modes still act as in (5.7), the main motivation for using the *y*-variable is that  $J_{\pm\omega}^{\pm}$  act as differential operators in *y*. More precisely, we have

$$\left(J_{\omega}^{+}V_{j}^{\omega}\right)(x,y,z) = \partial_{y}V_{j}^{\omega}(x,y,z), \qquad (5.22a)$$

$$\left(J^{-}_{-\omega}V^{\omega}_{j}\right)(x,y,z) = (y^{2}\partial_{y}+2jy)V^{\omega}_{j}(x,y,z).$$
(5.22b)

This allows one to re-write the recursion relations derived formally in the previous section as differential equations for correlators of the form

$$F_{y} \equiv \langle V_{j_{1}}^{\omega_{1}}(x_{1}, y_{1}, z_{1}) V_{j_{2}}^{\omega_{2}}(x_{2}, y_{2}, z_{2}) V_{j_{3}}^{\omega_{3}}(x_{3}, y_{3}, z_{3}) \rangle.$$
(5.23)

The *x*-basis correlators in Eq. (5.12) can be obtained from these by means of  $(5.20)^{47}$ . This integration procedure can be complicated, and was only carried out explicitly in [38] for a subset of cases.

In order to discuss the structure of these differential equations and their solutions, it will be useful to make use of the conformal invariance on the worldsheet and boundary CFT to fix  $x_1 = z_1 = 0$ ,  $x_2 = z_2 = 1$  and  $x_3 = z_3 = \infty$ , and consider

$$\hat{F}_{y} = \langle V_{j_{1}}^{\omega_{1}}(y_{1})V_{j_{2}}^{\omega_{2}}(y_{2})V_{j_{3}}^{\omega_{3}}(y_{3})\rangle \equiv \langle V_{j_{1}}^{\omega_{1}}(0,y_{1},0)V_{j_{2}}^{\omega_{2}}(1,y_{2},1)V_{j_{3}}^{\omega_{3}}(\infty,y_{3},\infty)\rangle.$$
(5.24)

<sup>&</sup>lt;sup>47</sup>The procedure is slightly different for flowed discrete and continuous states. The former arise as residues from poles of the integrand in (5.20) around y = 0 or  $y = \infty$ , depending on whether the corresponding unflowed vertex operator belongs to a lowest- or highest-weight representation. For the flowed continuous states one must integrate over the full complex plane.

The latter is related to the original correlator in Eq. (5.23) by

$$\left\langle V_{j_{1}}^{\omega_{1}}(x_{1},y_{1},z_{1})V_{j_{2}}^{\omega_{2}}(x_{2},y_{2},z_{2})V_{j_{3}}^{\omega_{3}}(x_{3},y_{3},z_{3})\right\rangle = \frac{x_{21}^{h_{3}^{0}-h_{1}^{0}-h_{2}^{0}}x_{31}^{h_{2}^{0}-h_{1}^{0}-h_{3}^{0}}x_{32}^{h_{1}^{0}-h_{3}^{0}-h_{2}^{0}-h_{3}^{0}}}{z_{21}^{\Delta_{1}^{0}+\Delta_{2}^{0}-\Delta_{3}^{0}}z_{31}^{\Delta_{1}^{0}+\Delta_{3}^{0}-\Delta_{2}^{0}}z_{32}^{\Delta_{2}^{0}+\Delta_{3}^{0}-\Delta_{1}^{0}}} \times \left\langle V_{j_{1}}^{\omega_{1}}\left(0,y_{1}\frac{x_{32}z_{21}^{\omega_{1}}z_{31}^{\omega_{1}}}{x_{21}x_{31}z_{32}^{\omega_{1}}},0\right)V_{j_{2}}^{\omega_{2}}\left(1,y_{2}\frac{x_{31}z_{21}^{\omega_{2}}z_{32}^{\omega_{2}}}{x_{21}x_{32}z_{31}^{\omega_{2}}},1\right)V_{j_{3}}^{\omega_{3}}\left(\infty,y_{3}\frac{x_{21}z_{31}^{\omega_{3}}z_{32}^{\omega_{3}}}{x_{31}x_{32}z_{21}^{\omega_{3}}},\infty\right)\right\rangle,$$

$$(5.25)$$

where

$$h_i^0 = j_i + \frac{k}{2}\omega_i$$
,  $\Delta_i^0 = -\frac{j_i(j_i - 1)}{k - 2} - j_i\omega_i - \frac{k}{4}\omega_i^2$ . (5.26)

Except for a certain subfamily of correlators which will be discussed below, spectrally flowed *y*-basis three-point functions and their associated differential equations were studied in [38] on a case-by-case basis. This was done for sufficiently low values of the spectral flow charges  $\omega_i$ , thus leading the authors to conjecture a general solution for the *y*-dependence of these correlators. The proposed expressions (not including the right-movers, and up to an overall normalization constant to be discussed below) read as follows. For *odd parity* correlators, namely when  $\omega_1 + \omega_2 + \omega_3 \in 2\mathbb{Z} + 1$ , one has

$$\left\langle V_{j_1}^{\omega_1}(0, y_1, 0) \, V_{j_2}^{\omega_2}(1, y_2, 1) \, V_{j_3}^{\omega_3}(\infty, y_3, \infty) \right\rangle = X_{123}^{\frac{k}{2} - j_1 - j_2 - j_3} \prod_{i=1}^3 X_i^{-\frac{k}{2} + j_1 + j_2 + j_3 - 2j_i} , \quad (5.27)$$

while for the *even parity* case, i.e. when  $\omega_1 + \omega_2 + \omega_3 \in 2\mathbb{Z}$ ,

$$\left\langle V_{j_1}^{\omega_1}(0, y_1, 0) \, V_{j_2}^{\omega_2}(1, y_2, 1) \, V_{j_3}^{\omega_3}(\infty, y_3, \infty) \right\rangle = X_{\emptyset}^{j_1 + j_2 + j_3 - k} \prod_{i < \ell} X_{i\ell}^{j_1 + j_2 + j_3 - 2j_i - 2j_\ell} \,. \tag{5.28}$$

Here, for any subset  $I \subset \{1, 2, 3\}$ ,

$$X_I(y_1, y_2, y_3) \equiv \sum_{i \in I: \ \varepsilon_i = \pm 1} P_{\omega + \sum_{i \in I} \varepsilon_i e_i} \prod_{i \in I} y_i^{\frac{1 - \varepsilon_i}{2}}.$$
(5.29)

The spectral flow parameters are chosen as  $\boldsymbol{\omega} = (\omega_1, \omega_2, \omega_3)$ , while  $e_1 = (1, 0, 0)$ ,  $e_2 = (0, 1, 0)$  and  $e_3 = (0, 0, 1)$ . The numbers  $P_{\boldsymbol{\omega}}$  are defined as

$$P_{\omega} = 0$$
 for  $\sum_{j} \omega_{j} < 2 \max_{i=1,2,3} \omega_{i}$  or  $\sum_{i} \omega_{i} \in 2\mathbb{Z} + 1$  (5.30)

and

$$P_{\omega} = S_{\omega} \frac{G\left(\frac{-\omega_1 + \omega_2 + \omega_3}{2} + 1\right) G\left(\frac{\omega_1 - \omega_2 + \omega_3}{2} + 1\right) G\left(\frac{\omega_1 + \omega_2 - \omega_3}{2} + 1\right) G\left(\frac{\omega_1 + \omega_2 + \omega_3}{2} + 1\right)}{G(\omega_1 + 1)G(\omega_2 + 1)G(\omega_3 + 1)},$$
(5.31)

otherwise, where G(n) is the Barnes G function

$$G(n) = \prod_{i=1}^{n-1} \Gamma(i)$$
 (5.32)
for positive integer values, while  $S_{\omega}$  is a phase depending on  $\omega$  mod 2. For more details, see [38].

Regarding the overall constants, their precise form was also conjectured in [38] and later proven in [39]. These structure constants are

$$C_{\omega}(j_{1}, j_{2}, j_{3}) = \begin{cases} C(j_{1}, j_{2}, j_{3}), & \text{if } \omega_{1} + \omega_{2} + \omega_{3} \in 2\mathbb{Z}, \\ \mathcal{N}(j_{1})C\left(\frac{k}{2} - j_{1}, j_{2}, j_{3}\right), & \text{if } \omega_{1} + \omega_{2} + \omega_{3} \in 2\mathbb{Z} + 1, \end{cases}$$
(5.33)

where  $C(j_1, j_2, j_3)$  are the structure constants of the unflowed three-point functions defined in terms of Barnes double Gamma functions in [106, 103]. Finally,  $\mathcal{N}(j_1)$  is defined in terms of the reflection coefficient appearing the unflowed two-point functions, namely

$$\mathcal{N}(j) = \sqrt{\frac{B(j)}{B\left(\frac{k}{2} - j\right)}},$$
(5.34)

with<sup>48</sup>

$$B(j) = \frac{2j-1}{\pi} \frac{\Gamma[1-b^2(2j-1)]}{\Gamma[1+b^2(2j-1)]} \nu^{1-2j}, \quad \nu = \frac{\Gamma[1-b^2]}{\Gamma[1+b^2]}, \quad b^2 = (k-2)^{-1}.$$
(5.35)

As shown in [38], the proposal given in Eqs. (5.27)-(5.33) passes a number of non-trivial consistency checks, including bosonic exchange symmetry and reflection symmetry for continuous states. However, no general expression for the *y*-basis differential equations is known. Hence, so far, this conjecture remains to be proven.

In the remainder of the chapter, we prove that this solution is indeed correct. In doing so, we highlight the role of holomorphic covering maps. Although these maps appear to be essential for the study of four-point functions [179], we show that they also play a key role in the present context. Furthermore, when treating cases where there is no well-defined covering map available, we will make use of the so-called series identifications for spectrally flowed vertex operators constructed upon states belonging to the discrete representations of  $SL(2,\mathbb{R})$  [175, 39].

### 5.2 The proof for *y*-basis three-point functions

In this section we prove the conjecture put forward in [38]. It was shown in [103] that all non-vanishing spectrally flowed three-point functions in the  $SL(2,\mathbb{R})$  model must

<sup>&</sup>lt;sup>48</sup>To be precise,  $\nu$  is actually a free parameter of the model, which essentially plays the role of the string coupling. Here we simply reproduce the value originally advocated in [106]. For a related discussion, see [99, 90]. We thank A. Dei and L. Eberhardt for pointing this out.

satisfy the following condition:

$$\omega_i + \omega_j \ge \omega_k - 1 \qquad \forall \ i \ne j \ne k. \tag{5.36}$$

We first consider the subfamily of odd parity correlators for which an associated holomorphic covering map exists [88, 38]. We then show how to compute all remaining non-vanishing correlators with three non-trivial spectral flow charges. This includes even parity correlators, and also those we denote as edge correlators. The latter correspond to correlators for which either the inequality in (5.36) for odd parity assignments or the analogous inequality for even parity cases saturate. These need to be treated with special care. Finally, we also discuss correlators with unflowed insertions and the overall normalizations.

#### 5.2.1 Holomorphic covering maps and odd parity correlators

For concreteness, and with no loss of generality, we take  $\omega_3$  to be the largest spectral flow charge, i.e.  $\omega_3 \ge \omega_{1,2}$ . Let us consider correlators satisfying

$$\omega_1 + \omega_2 + \omega_3 \in 2\mathbb{Z} + 1$$
,  $\omega_1 + \omega_2 > \omega_3 - 1$ ,  $\omega_i \ge 1$ ,  $\forall i$ . (5.37)

It was shown in [180] that there exists a unique holomorphic covering map  $\Gamma[\omega_1, \omega_2, \omega_3](z) \equiv \Gamma(z)$  from the worldsheet to the AdS<sub>3</sub> boundary such that

$$\Gamma(z) \sim x_i + a_i(z - z_i)^{\omega_i} + \cdots$$
 when  $z \sim z_i$ , (5.38)

where the ellipsis indicates higher order terms in  $(z - z_i)$ . This is a rational function which approaches a constant  $\Gamma_{\infty}$  as  $z \to \infty$ . One can show that it develops *N* single poles, counted by the Riemann-Hurwitz formula

$$N = \frac{1}{2} \left( \omega_1 + \omega_2 + \omega_3 - 1 \right).$$
 (5.39)

The coefficients  $a_i$  appearing in Eq. (5.38) take the form

$$a_{i} = \begin{pmatrix} \frac{\omega_{i} + \omega_{i+1} + \omega_{i+2} - 1}{2} \\ \frac{-\omega_{i} + \omega_{i+1} + \omega_{i+2} - 1}{2} \end{pmatrix} \begin{pmatrix} \frac{-\omega_{i} + \omega_{i+1} - \omega_{i+2} - 1}{2} \\ \frac{\omega_{i} + \omega_{i+1} - \omega_{i+2} - 1}{2} \end{pmatrix}^{-1} \frac{x_{i,i+1} x_{i+2,i}}{x_{i+1,i+2}} \left( \frac{z_{i+1,i+2}}{z_{i,i+1} z_{i+2,i}} \right)^{\omega_{i}}, (5.40)$$

where the subscripts are understood mod 3. Note that the last two factors in (5.40) simplify to 1 upon setting  $x_1 = z_1 = 0$ ,  $x_2 = z_2 = 1$  and  $x_3 = z_3 = \infty$ . We will use the same notation  $a_i$  for the resulting purely numerical coefficients.

We now derive the differential equations satisfied by *y*-basis three-point functions satisfying (5.37). Although the presentation here is slightly different, this was already done

in [88]. Consider the operator  $J^{-}(\Gamma(z), z)$ , where we use the notation of Eq. (5.10b), namely

$$J^{-}(\Gamma(z), z) = J^{-}(z) - 2\Gamma(z)J^{3}(z) + \Gamma^{2}(z)J^{+}(z).$$
(5.41)

In order to obtain the recursion relations for the correlators under consideration, we compute following the contour integral:

$$\oint_{z_i} \frac{dz}{(z-z_i)^{\omega_i}} \left\langle J^-(\Gamma(z),z) \, V_{j_1}^{\omega_1}(x_1,y_1,z_1) V_{j_2}^{\omega_2}(x_2,y_2,z_2) V_{j_3}^{\omega_3}(x_3,y_3,z_3) \right\rangle \,. \tag{5.42}$$

Similarly to what was done in Sec. 5.1.2 above, we do this in two different ways. First, we note that, near  $z_i$ , we have

$$J^{-}(\Gamma(z),z) = J^{-}(x_{i},z) - 2a_{i}(z-z_{i})^{\omega_{i}}J^{3}(x_{i},z) + a_{i}^{2}(z-z_{i})^{2\omega_{i}}J^{+}(z) + \cdots$$
 (5.43)

Hence, by using (5.21) we find that

$$(5.42) = \left[ \left( 2j_i y_i + y_i^2 \partial_{y_i} \right) - 2a_i \left( j_i + \frac{k}{2} \omega_i + y_i \partial_{y_i} \right) + a_i^2 \partial_{y_i} \right] F_y , \qquad (5.44)$$

where  $F_y$  was defined in Eq. (5.23). On the other hand, using (5.41) together with the OPEs in Eq. (5.6), one finds that

$$\left\langle J^{-}(\Gamma(z),z) V_{j_{1}}^{\omega_{1}}(x_{1},y_{1},z_{1}) V_{j_{2}}^{\omega_{2}}(x_{2},y_{2},z_{2}) V_{j_{3}}^{\omega_{3}}(x_{3},y_{3},z_{3}) \right\rangle = \\ = \sum_{j=1}^{3} \left\{ -\frac{2[\Gamma(z)-x_{j}](y_{j}\partial_{y_{j}}+j_{j}+\frac{k}{2}\omega_{j})}{z-z_{j}} F_{y} + \sum_{n=1}^{\omega_{j}} \frac{[\Gamma(z)-x_{j}]^{2}}{(z-z_{j})^{n+1}} F_{y,n}^{i} \right\},$$
(5.45)

where  $F_{y,n}^i$  stands for the *y*-basis analogues of the  $F_n^i$  defined in Eq. (5.13). As discussed in [88], the RHS of (5.45) is a rational function of *z* which, as implied by the constraint equations, has zeros of order  $\omega_i - 1$  at all  $z_i$ . It also has double poles at the *N* simple poles of  $\Gamma(z)$ , and further goes to zero as  $z^{-2}$  for  $z \to \infty$  due to the global Ward identities. This implies that it must be proportional to the derivative of the covering map, namely

$$\left\langle J^{-}(\Gamma(z),z) \, V_{j_1}^{\omega_1}(x_1,y_1,z_1) V_{j_2}^{\omega_2}(x_2,y_2,z_2) V_{j_3}^{\omega_3}(x_3,y_3,z_3) \right\rangle = \alpha \, \partial \Gamma(z) \,, \tag{5.46}$$

where  $\alpha$  must be independent of *z*. This coefficient was also computed in [88]. When working in the *y*-basis it takes the following form:

$$\alpha = -\frac{1}{N} \sum_{j=1}^{3} \left( (y_j - a_j) \partial_{y_j} + j_j + \frac{k}{2} \omega_j \right) F_y.$$
 (5.47)

This allows us to provide an alternative expression for the contour integral (5.42). Indeed, the behavior of the covering map near the insertion points showcased in (5.38) implies that

$$(5.42) = -\frac{a_i \omega_i}{N} \sum_{j=1}^3 \left( (y_j - a_j) \partial_{y_j} + j_j + \frac{k}{2} \omega_j \right) F_y.$$
(5.48)

By combining the results in Eqs. (5.44) and (5.48), and further fixing the insertion points as in (5.24), one obtains the following differential equations:

$$\left\{ (y_i - a_i)^2 \partial_{y_i} + 2j_i (y_i - a_i) + \frac{a_i \omega_i}{N} \left[ \sum_{j=1}^3 \left( (y_j - a_j) \partial_{y_j} + j_j \right) + \frac{k}{2} \right] \right\} \hat{F}_y = 0, \quad (5.49)$$

for i = 1, 2, 3. This was derived originally in this form in [38]<sup>49</sup>. In this way, the use of the covering map and its derivative allows one to avoid dealing with the cumbersome unknowns discussed in Sec. 5.1.2.

The system of equations encoded in (5.49) uniquely fixes the dependence of the corresponding correlators on  $y_1$ ,  $y_2$  and  $y_3$ . Up to some overall normalization, which will be discussed in section 5.2.5 below, the solution of (5.49) is

$$\langle V_{j_1}^{\omega_1}(y_1)V_{j_2}^{\omega_2}(y_2)V_{j_3}^{\omega_3}(y_3)\rangle = (y_1 - a_1)^{j_2 + j_3 - j_1 - \frac{k}{2}}(y_2 - a_2)^{j_3 + j_1 - j_2 - \frac{k}{2}}(y_3 - a_3)^{j_1 + j_2 - j_3 - \frac{k}{2}} \\ \times \left[\sum_{\varepsilon_{1,2,3} = \pm 1} 2\frac{N_1^{\varepsilon_1, \varepsilon_2, \varepsilon_3}}{N}\prod_{i=1}^3 a_i^{\frac{\varepsilon_i + 1}{2}}y_i^{\frac{\varepsilon_i - 1}{2}}\right]^{\frac{k}{2} - j_1 - j_2 - j_3}.$$
(5.50)

An equivalent, perhaps simpler expression for the same solution is given by

$$\langle V_{j_1}^{\omega_1}(y_1)V_{j_2}^{\omega_2}(y_2)V_{j_3}^{\omega_3}(y_3)\rangle = (y_1 - a_1)^{-2j_1}(y_2 - a_2)^{-2j_2}(y_3 - a_3)^{-2j_3}$$

$$\times \left(\omega_1 \frac{y_1 + a_1}{y_1 - a_1} + \omega_2 \frac{y_2 + a_2}{y_2 - a_2} + \omega_3 \frac{y_3 + a_3}{y_3 - a_3} - 1\right)^{\frac{k}{2} - j_1 - j_2 - j_3}.$$

$$(5.51)$$

Note that this formulation makes the bosonic exchange symmetry manifest.

At first sight, this expression might seem very different from that in Eq. (5.27). The connection comes from the fact that while the  $P_{\omega}$  defined in (5.31) are somewhat complicated, their ratios are actually much simpler. For instance, consider the  $X_i$  term appearing in (5.27). From Eqs. (5.29) and (5.31), up to an overall sign, we find that

$$X_{i} = P_{\omega - e_{i}} y_{i} + P_{\omega + e_{i}} = P_{\omega - e_{i}} \left( y_{i} + \frac{P_{\omega + e_{i}}}{P_{\omega - e_{i}}} \right) = P_{\omega - e_{i}} \left( y_{i} - a_{i} \right) .$$
(5.52)

A similar result holds for  $X_{123}$ , showing that the *y*-dependence of the expression in (5.51) is consistent with that of Eq. (5.27).

As shown in Eq. (5.51), *y*-basis three-point functions diverge whenever a variable  $y_i$  approaches the corresponding coefficient  $a_i$ . Thus, the  $a_i$  are very special points in the

<sup>&</sup>lt;sup>49</sup>Note that we have corrected a couple of typos in their presentation.

*y*-plane, which signal the existence of an appropriate holomorphic covering map. An even more extreme situation takes place in the tensionless limit, which corresponds to k = 3 in the bosonic language [194, 187, 87, 88]. There, spectrally flowed correlators are non-vanishing *only* when  $y_i = a_i$  for all *i*. The corresponding recursion relations (5.49) are then satisfied provided<sup>50</sup>  $j_1 + j_2 + j_3 = \frac{3}{2}$ . It would be interesting to fully understand the relation between *y*-variables and covering map coefficients in the general case.

We end this section by noting that the same discussion can not be applied directly to odd parity correlators saturating the inequality in Eq. (5.36), i.e. those with  $\omega_3 = \omega_1 + \omega_2 + 1$ . These will be discussed in detail in section 5.2.3 below. Correlators with unflowed insertions are further considered in section 5.2.4.

#### 5.2.2 Series identification and even parity correlators

We now prove that the conjecture of [38] also holds for correlators satisfying

$$\omega_1 + \omega_2 + \omega_3 \in 2\mathbb{Z} , \qquad \omega_1 + \omega_2 > \omega_3 , \qquad \omega_i \ge 1 , \ \forall i. \tag{5.53}$$

These include all non-vanishing spectrally flowed three-point functions for which the total spectral flow charge is even, except for the edge cases where  $\omega_3 = \omega_1 + \omega_2$  and/or some of the vertex operators are unflowed, which will be treated separately.

When the  $\omega_i$  are as in (5.53), it is not possible to construct a covering map such as the one used in the previous section. Thus, one might wonder if differential equations similar to those in (5.49) can be deduced in this context. Indeed, the corresponding recursion relations have only been obtained on a case-by-case basis and for sufficiently low values of the spectral flow charges [38].

Nevertheless, we observe that the procedure outlined in Sec. 5.1.2 guarantees that, provided the system is compatible, and once the  $F_n^i$  have been solved for, the resulting *y*-basis recursions must take the form

$$\left[y_i(y_i\partial_{y_i}+2j_i)+\sum_{j=1}^3(A_{ij}y_j-B_{ij})\partial_{y_j}+C_i\right]\langle V_{j_1}^{\omega_1}(y_1)V_{j_2}^{\omega_2}(y_2)V_{j_3}^{\omega_3}(y_3)\rangle=0,$$
 (5.54)

where  $A_{ij}$ ,  $B_{ij}$  and  $C_i$  are some numerical constants to be determined, which depend on the spins  $j_i$  and the charges  $\omega_i$ . The rationale behind the structure of Eq. (5.54), which is the *y*-basis version of Eq. (5.18), goes as follows. First, note that, upon using Eq. (5.22b), the operator  $y_i(y_i\partial_{y_i} + 2j_i)$  is identified with the RHS of (5.18). Second, recall that the recursion relations were derived by expressing the OPEs of the vertex operators with the conserved currents in terms of the unknowns coming from the action of the modes

<sup>&</sup>lt;sup>50</sup>In the supersymmetric case, the RNS formalism breaks down for the tensionless theory. It was shown in [87], using the so-called hybrid formalism, that in this case only vertex operators with  $j = \frac{1}{2}$  are allowed.

of  $J^+(z)$ , see Eq. (5.17). This implies that the term involving the action of  $J^-(x, z)$  does not mix with the rest. By using the Möbius-fixed expression for the three-point function in Eq. (5.25), one can see that the terms in the recursion relations involving unknowns and *x*-derivatives of the correlator are mapped to operators of the schematic form  $y\partial_y$ and  $\partial_y$ , as well as *y*-independent multiplicative factors. In Eq. (5.54) we have allowed for generic coefficients  $A_{ij}$ ,  $B_{ij}$  and  $C_i$  in front of the corresponding contributions.

Moreover, we note that the way in which these equations are derived only depends on the values of the spectral flow charges involved in a given correlator. In other words, for a given set of  $\omega_i$ , the recursion relations are independent of whether the vertex operators involved belong to spectrally flowed discrete or continuous representations. These two observations will allow us to obtain all *y*-basis differential equations associated with even parity correlators in closed form.

As it turns out, even and odd parity cases are not completely disconnected. For the discrete representations, affine modules in spectral flow sectors with one unit of difference in the spectral flow charge are identifiable. For *y*-basis operators the corresponding series identifications read [39]

$$V_{j}^{\omega}(x, y = 0, z) = \mathcal{N}(j) \lim_{y \to \infty} y^{k-2j} V_{\frac{k}{2}-j}^{\omega+1}(x, y, z),$$
(5.55)

and

$$\lim_{y \to \infty} y^{2j} V_j^{\omega}(x, y, z) = \mathcal{N}(j) V_{\frac{k}{2} - j}^{\omega - 1}(x, y = 0, z),$$
(5.56)

where  $\mathcal{N}(j)$  was defined in Eq. (5.34). It was shown recently in [39] that, assuming the *y*-dependence proposed in [38] for all three-point functions, these relations fix the *y*-basis structure constants in terms of the unflowed ones, which can be found in [103]. Here we show that Eqs. (5.55) and (5.56) are actually much more powerful: they allow us to fix the *y*-dependence as well. More explicitly, we use them to derive all unknown coefficients  $A_{ij}$ ,  $B_{ij}$ , and  $C_i$  appearing in (5.54).

By means of Eqs. (5.55) and (5.56), we find that all even parity correlators can be related to (at least) three *different* situations where a covering map satisfying (5.38) and (5.40) exists. Explicitly, given  $(\omega_1, \omega_2, \omega_3)$  satisfying (5.53), it follows that the adjacent assignments  $(\omega_1 + 1, \omega_2, \omega_3), (\omega_1, \omega_2 + 1, \omega_3)$  and  $(\omega_1, \omega_2, \omega_3 - 1)$  satisfy all conditions in (5.37). Let us denote the corresponding covering maps as follows:

$$\Gamma_1^+ \equiv \Gamma[\omega_1 + 1, \omega_2, \omega_3], \qquad \Gamma_2^+ \equiv \Gamma[\omega_1, \omega_2 + 1, \omega_3], \qquad \Gamma_3^- \equiv \Gamma[\omega_1, \omega_2, \omega_3 - 1].$$
(5.57)

Then, the relations (5.55) and (5.56) provide the following set of identities:

$$\langle V_{j_{1}}^{\omega_{1}}(0)V_{j_{2}}^{\omega_{2}}(y_{2})V_{j_{3}}^{\omega_{3}}(y_{3})\rangle = \lim_{y_{1}\to\infty} y_{1}^{k-2j_{1}}\mathcal{N}(j_{1})\langle V_{\frac{k}{2}-j_{1}}^{\omega_{1}+1}(y_{1})V_{j_{2}}^{\omega_{2}}(y_{2})V_{j_{3}}^{\omega_{3}}(y_{3})\rangle$$

$$\langle V_{j_{1}}^{\omega_{1}}(y_{1})V_{j_{2}}^{\omega_{2}}(0)V_{j_{3}}^{\omega_{3}}(y_{3})\rangle = \lim_{y_{2}\to\infty} y_{2}^{k-2j_{2}}\mathcal{N}(j_{2})\langle V_{j_{1}}^{\omega_{1}}(y_{1})V_{\frac{k}{2}-j_{2}}^{\omega_{2}+1}(y_{2})V_{j_{3}}^{\omega_{3}}(y_{3})\rangle$$

$$\lim_{y_{3}\to\infty} y_{3}^{2j_{3}}\langle V_{j_{1}}^{\omega_{1}}(y_{1})V_{j_{2}}^{\omega_{2}}(y_{2})V_{j_{3}}^{\omega_{3}}(y_{3})\rangle = \mathcal{N}(j_{3})\langle V_{j_{1}}^{\omega_{1}}(y_{1})V_{j_{2}}^{\omega_{2}}(y_{2})V_{\frac{k}{2}-j_{3}}^{\omega_{3}+1}(0)\rangle.$$
(5.58c)

Having excluded the even edge cases, the same holds for the adjacent assignments  $(\omega_1 - 1, \omega_2, \omega_3), (\omega_1, \omega_2 - 1, \omega_3)$  and  $(\omega_1, \omega_2, \omega_3 + 1)$ , the corresponding maps being

$$\Gamma_1^- \equiv \Gamma[\omega_1 - 1, \omega_2, \omega_3], \qquad \Gamma_2^- \equiv \Gamma[\omega_1, \omega_2 - 1, \omega_3], \qquad \Gamma_3^+ \equiv \Gamma[\omega_1, \omega_2, \omega_3 + 1].$$
(5.59)

These lead to

$$\lim_{y_{1}\to\infty} y_{1}^{2j_{1}} \langle V_{j_{1}}^{\omega_{1}}(y_{1}) V_{j_{2}}^{\omega_{2}}(y_{2}) V_{j_{3}}^{\omega_{3}}(y_{3}) \rangle = \mathcal{N}(j_{1}) \langle V_{\frac{k}{2}-j_{1}}^{\omega_{1}-1}(0) V_{j_{2}}^{\omega_{2}}(y_{2}) V_{j_{3}}^{\omega_{3}}(y_{3}) \rangle$$
(5.60a)  
$$\lim_{y_{2}\to\infty} y_{2}^{2j_{2}} \langle V_{j_{1}}^{\omega_{1}}(y_{1}) V_{j_{2}}^{\omega_{2}}(y_{2}) V_{j_{3}}^{\omega_{3}}(y_{3}) \rangle = \mathcal{N}(j_{2}) \langle V_{j_{1}}^{\omega_{1}}(y_{1}) V_{\frac{k}{2}-j_{2}}^{\omega_{2}-1}(0) V_{j_{3}}^{\omega_{3}}(y_{3}) \rangle,$$
(5.60b)  
$$\langle V_{j_{1}}^{\omega_{1}}(y_{1}) V_{j_{2}}^{\omega_{2}}(y_{2}) V_{j_{3}}^{\omega_{3}}(0) \rangle = \lim_{y_{3}\to\infty} y_{3}^{k-2j_{3}} \mathcal{N}(j_{3}) \langle V_{j_{1}}^{\omega_{1}}(y_{1}) V_{j_{2}}^{\omega_{2}}(y_{2}) V_{\frac{k}{2}-j_{3}}^{\omega_{3}+1}(y_{3}) \rangle$$
(5.60c)

All expressions on the RHS of Eqs. (5.58) and (5.60) are limits of correlators discussed in the previous section. Hence, they must satisfy the appropriate limits of the differential equations given in (5.49). For instance,  $\langle V_{\frac{k}{2}-j_1}^{\omega_1-1}(0)V_{j_2}^{\omega_2}(y_2)V_{j_3}^{\omega_3}(y_3)\rangle$  is annihilated by the differential operators

$$y_{2}(y_{2}\partial_{y_{2}}+2j_{2}) + (\omega_{1}-\omega_{2}-\omega_{3})^{-1} \{ (\omega_{1}+\omega_{2}-\omega_{3})a_{2}[\Gamma_{1}^{-}]^{2}\partial_{y_{2}} \\ -2a_{2}[\Gamma_{1}^{-}] \left[ (\omega_{1}-\omega_{3})(j_{2}+y_{2}\partial_{y_{2}}) + \omega_{2}(j_{3}+y_{3}\partial_{y_{3}}-a_{3}[\Gamma_{1}^{-}]\partial_{y_{3}}+j_{1}+j_{3}) \right] \}$$
(5.61)

and

$$y_{3}(y_{3}\partial_{y_{3}}+2j_{3}) + (\omega_{1}-\omega_{2}-\omega_{3})^{-1} \{ (\omega_{1}-\omega_{2}+\omega_{3})a_{3}[\Gamma_{1}^{-}]^{2}\partial_{y_{3}} - 2a_{3}[\Gamma_{1}^{-}] [(\omega_{1}-\omega_{2})(j_{3}+y_{3}\partial_{y_{3}}) + \omega_{3}(j_{2}+y_{2}\partial_{y_{2}}-a_{2}[\Gamma_{1}^{-}]\partial_{y_{2}}+j_{1}+j_{2})] \},$$
(5.62)

where  $a_i[\Gamma_1^-]$  denotes the coefficient  $a_i$  associated with the map  $\Gamma_1^-$ . One can obtain analogous equations from the other odd parity correlators involved in (5.58). This leads to twelve differential operators, which must coincide with the appropriate limits of those provided in Eqs. (5.54). As an example, in the situation considered above we should match (5.61) and (5.62) with

$$y_{2}(y_{2}\partial_{y_{2}}+2j_{2}) + A_{22}y_{2}\partial_{y_{2}} + A_{23}y_{3}\partial_{y_{3}} - B_{22}\partial_{y_{2}} - B_{23}\partial_{y_{3}} - 2A_{21}j_{1} + C_{2},$$
  

$$y_{3}(y_{3}\partial_{y_{3}}+2j_{3}) + A_{33}y_{3}\partial_{y_{3}} + A_{32}y_{2}\partial_{y_{2}} - B_{33}\partial_{y_{3}} - B_{32}\partial_{y_{2}} - 2A_{31}j_{1} + C_{3}.$$
(5.63)

After identifying all linearly independent terms in these 12 equations we find a total of 60 conditions. In the end, 21 of these 60 conditions can be used to solve explicitly for all the coefficients  $A_{ij}$ ,  $B_{ij}$  and  $C_i$  in (5.54). Consistency of the system demands that the remaining 39 conditions must hold. We emphasize that the fact that these are identically satisfied is a highly non-trivial check of the logic behind our proof, and also a remarkable consequence of the identities relating the  $a_i$  coefficients of the different covering maps involved.

There are many equivalent ways to write the resulting coefficients. We find that the simplest one is as follows:

$$A = \begin{pmatrix} \frac{2(\omega_{3}-\omega_{2})}{\omega_{1}+\omega_{2}-\omega_{3}}a_{1}[\Gamma_{3}^{-}] & \frac{2\omega_{1}}{\omega_{1}+\omega_{2}-\omega_{3}}a_{1}[\Gamma_{3}^{-}] & \frac{2\omega_{1}}{\omega_{1}-\omega_{2}+\omega_{3}}a_{1}[\Gamma_{2}^{-}] \\ \frac{2\omega_{2}}{\omega_{1}+\omega_{2}-\omega_{3}}a_{2}[\Gamma_{3}^{-}] & \frac{2(\omega_{1}-\omega_{3})}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{2}[\Gamma_{1}^{-}] & \frac{2\omega_{2}}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{2}[\Gamma_{1}^{-}] \\ \frac{2\omega_{3}}{\omega_{1}-\omega_{2}+\omega_{3}}a_{3}[\Gamma_{2}^{-}] & \frac{2\omega_{3}}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{3}[\Gamma_{1}^{-}] & \frac{2(\omega_{1}-\omega_{2})}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{3}[\Gamma_{1}^{-}] \end{pmatrix}, \quad (5.64)$$

$$B = \begin{pmatrix} \frac{(\omega_{1}-\omega_{2}+\omega_{2})}{\omega_{1}+\omega_{2}-\omega_{3}}a_{1}[\Gamma_{3}^{-}]^{2} & \frac{2\omega_{1}}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{3}[\Gamma_{1}^{-}] & \frac{2\omega_{1}}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{3}[\Gamma_{1}^{-}] \\ \frac{2\omega_{2}}{\omega_{1}+\omega_{2}-\omega_{3}}a_{1}[\Gamma_{3}^{-}]a_{2}[\Gamma_{3}^{-}] & \frac{(\omega_{1}+\omega_{2}-\omega_{2})}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{2}[\Gamma_{1}^{-}]^{2} & \frac{2\omega_{2}}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{2}[\Gamma_{1}^{-}]a_{3}[\Gamma_{1}^{-}] \\ \frac{2\omega_{3}}{\omega_{1}-\omega_{2}+\omega_{3}}a_{1}[\Gamma_{2}^{-}]a_{3}[\Gamma_{2}^{-}] & \frac{2\omega_{3}}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{2}[\Gamma_{1}^{-}]a_{3}[\Gamma_{1}^{-}] & \frac{(\omega_{1}-\omega_{2}+\omega_{2})}{-\omega_{1}+\omega_{2}+\omega_{3}}a_{3}[\Gamma_{1}^{-}]^{2} \end{pmatrix}, \quad (5.65)$$

and

$$C = \begin{pmatrix} \frac{4\omega_{1j_3}}{\omega_1 - \omega_2 + \omega_3} a_1[\Gamma_2^-] + \frac{2\omega_1(j_2 + j_3) + 2j_1(\omega_3 - \omega_2)}{\omega_1 + \omega_2 - \omega_3} a_1[\Gamma_3^-] \\ \frac{4\omega_{2j_1}}{\omega_1 + \omega_2 - \omega_3} a_2[\Gamma_3^-] - \frac{2\omega_2(j_1 + j_3) + 2j_2(\omega_1 - \omega_3)}{\omega_1 - \omega_2 - \omega_3} a_2[\Gamma_1^-] \\ \frac{4\omega_{3j_1}}{\omega_1 - \omega_2 + \omega_3} a_3[\Gamma_2^-] - \frac{2\omega_3(j_1 + j_2) + 2j_3(\omega_1 - \omega_2)}{\omega_1 + \omega_2 - \omega_3} a_3[\Gamma_1^-] \end{pmatrix}.$$
(5.66)

Consequently, we can write the differential equations for the even parity correlators as

$$\left\{ \left( y_1 - a_1[\Gamma_3^-] \right)^2 \partial_{y_1} + 2j_1 \left( y_1 - a_1[\Gamma_3^-] \right) + \frac{2a_1[\Gamma_3^-]\omega_1}{\omega_1 + \omega_2 - \omega_3} \left[ \left( y_1 - a_1[\Gamma_3^-] \right) \partial_{y_1} + j_1 + \left( y_2 - a_2[\Gamma_3^-] \right) \partial_{y_2} + j_2 - \left( y_3 - a_3[\Gamma_2^-] \right) \partial_{y_3} - j_3 \right] \right\} \left\langle V_{j_1}^{\omega_1}(y_1) V_{j_2}^{\omega_2}(y_2) V_{j_3}^{\omega_3}(y_3) \right\rangle = 0,$$

$$(5.67)$$

$$\left\{ \left( y_2 - a_2[\Gamma_3^-] \right)^2 \partial_{y_2} + 2j_2 \left( y_2 - a_2[\Gamma_3^-] \right) + \frac{2a_2[\Gamma_3^-]\omega_2}{\omega_1 + \omega_2 - \omega_3} \left[ \left( y_1 - a_1[\Gamma_3^-] \right) \partial_{y_1} + j_1 + \left( y_2 - a_2[\Gamma_3^-] \right) \partial_{y_2} + j_2 - \left( y_3 - a_3[\Gamma_1^-] \right) \partial_{y_3} - j_3 \right] \right\} \left\langle V_{j_1}^{\omega_1}(y_1) V_{j_2}^{\omega_2}(y_2) V_{j_3}^{\omega_3}(y_3) \right\rangle = 0,$$

$$(5.68)$$

 $\left\{ \left( y_3 - a_3[\Gamma_1^-] \right)^2 \partial_{y_3} + 2j_3 \left( y_3 - a_3[\Gamma_1^-] \right) + \frac{2a_3[\Gamma_1^-]\omega_3}{\omega_1 - \omega_2 - \omega_3} \left[ (y_1 - a_1[\Gamma_2^-])\partial_{y_1} + j_1 - (y_2 - a_2[\Gamma_1^-])\partial_{y_2} - j_2 - (y_3 - a_3[\Gamma_1^-])\partial_{y_3} - j_3 \right] \right\} \langle V_{j_1}^{\omega_1}(y_1) V_{j_2}^{\omega_2}(y_2) V_{j_3}^{\omega_3}(y_3) \rangle = 0.$ 

The structural similarity of these equations with those of the odd case in Eq. (5.49) is striking. This suggest that there must be a way to derive Eqs. (5.67), (5.68) and (5.69) directly from the *adjacent* covering maps along the lines of Sec. 5.2.1. We will not attempt to do this here.

By solving Eqs. (5.67), (5.68) and (5.69), we find that, up to an overall constant, even parity correlators satisfying (5.53) take the form

$$\langle V_{j_{1}}^{\omega_{1}}(y_{1})V_{j_{2}}^{\omega_{2}}(y_{2})V_{j_{3}}^{\omega_{3}}(y_{3})\rangle = \left(1 - \frac{y_{2}}{a_{2}[\Gamma_{3}^{+}]} - \frac{y_{3}}{a_{3}[\Gamma_{2}^{+}]} + \frac{y_{2}y_{3}}{a_{2}[\Gamma_{3}^{-}]a_{3}[\Gamma_{2}^{+}]}\right)^{j_{1}-j_{2}-j_{3}} \\ \times \left(1 - \frac{y_{1}}{a_{1}[\Gamma_{3}^{+}]} - \frac{y_{3}}{a_{3}[\Gamma_{1}^{+}]} + \frac{y_{1}y_{3}}{a_{1}[\Gamma_{3}^{-}]a_{3}[\Gamma_{1}^{+}]}\right)^{j_{2}-j_{3}-j_{1}} (5.70) \\ \times \left(1 - \frac{y_{1}}{a_{1}[\Gamma_{2}^{+}]} - \frac{y_{2}}{a_{2}[\Gamma_{1}^{+}]} + \frac{y_{1}y_{2}}{a_{1}[\Gamma_{2}^{+}]a_{2}[\Gamma_{1}^{-}]}\right)^{j_{3}-j_{1}-j_{2}}.$$

As in the odd case, one can check that this is consistent with the conjectured expressions in Eq. (5.28) by using the relations between covering map coefficients and ratios of the numbers  $P_{\omega}$  defined in (5.31). It is also straightforward to see that (5.70) matches all relevant limits of the corresponding odd cases in Eq. (5.50), as implied by the different series identifications in Eqs. (5.58) and (5.60). We also note that, upon using some identities among the  $a_i$  coefficients such as  $a_1[\Gamma_2^+]a_2[\Gamma_1^-] = a_1[\Gamma_2^-]a_2[\Gamma_1^+]$ , the solution in Eq. (5.70) manifestly enjoys bosonic exchange symmetry.

#### 5.2.3 Edge cases

In this subsection, we consider spectral flow assignments lying at the "edge" of the inequalities displayed in Eqs. (5.37) and (5.53). More explicitly, we consider three-point functions with spectral flow charges satisfying either

$$\omega_3 = \omega_1 + \omega_2$$
 or  $\omega_3 = \omega_1 + \omega_2 + 1$ ,  $\omega_i \ge 1$ ,  $\forall i$ . (5.71)

Following the nomenclature of the previous sections, we refer to these as the even and odd edge cases, respectively. Note that, according to the fusion rules in Eq. (5.36), these include all possibly non-vanishing correlators which were not included in sections 5.2.1 and 5.2.2 above, except for correlators with unflowed insertions, which will be discussed later on.

(5.69)

and

The treatment of the edge cases is slightly different from what we have discussed so far. Indeed, the general method based on series identifications employed in Sec. 5.2.2 breaks down when considering the even edge cases. We find that many of the coefficients in Eqs. (5.64), (5.65) and (5.66) become either divergent or indeterminate when  $\omega_1 + \omega_2 = \omega_3$ , related to the fact that three of the six adjacent maps cease to exist, namely those in (5.59). Although, in principle, the existence of the three maps in (5.57) could give enough constraints, in practice one runs into similar problems with divergent or indeterminate coefficients. As for the odd edge cases, it turns out that the associated covering map does not exist. Moreover, the former are only related to even edge cases by the SL(2, R) series identifications.

We now show that alternative techniques involving current insertions can be used to derive the relevant differential equations satisfied by the *y*-basis edge correlators. Some of these equations are easier to derive in the limit where the first two vertex operators collide. As discussed in [38], one can take the vertex operators to be inserted at  $(x_1, x_2, x_3) = (0, x, \infty)$ , and then consider the limit  $x \to 0$ . More explicitly, we have [38]

$$\langle V_{j_1}^{\omega_1}(0, y_1, 0) V_{j_2}^{\omega_2}(x, y_2, 1) V_{j_3}^{\omega_3}(\infty, y_3, \infty) \rangle$$
  
=  $x^{-j_1 - j_2 + j_3 + \frac{k}{2}(-\omega_1 - \omega_2 + \omega_3)} \langle V_{j_1}^{\omega_1}\left(0, \frac{y_1}{x}, 0\right) V_{j_2}^{\omega_2}\left(1, \frac{y_2}{x}, 1\right) V_{j_3}^{\omega_3}(\infty, y_3 x, \infty) \rangle .$  (5.72)

The three-point functions that remain well-defined in this limit are those for which

$$|\omega_1 + \omega_2 - \omega_3| \le 1, \tag{5.73}$$

including both edge cases in (5.71). In these instances, it is possible to obtain the correlator with vertex operators inserted at generic points  $(x_1, x_2, x_3)$  from the one in the colliding limit with  $(x, x, x_3)$  by means of the global Ward identities. We will derive some of the relevant differential equations satisfied by the edge correlators in the above collision limit, and only recover the full correlators at the end. Note that taking  $x \rightarrow 0$  and  $x_3 \rightarrow \infty$  precisely corresponds to the limit in which these correlators can be interpreted as *m*-basis correlators of flowed primaries as in [103, 178], where they were denoted as spectral flow *conserving* and spectral flow *violating* three-point functions, depending on the overall parity of the spectral flow charges.

To illustrate how this works, let us derive a constraint that will be satisfied by both edge cases. We set  $x_1 = x_2 = x$  and consider the integral

$$\oint_{\mathcal{C}} \left\langle J^3(x,z) V_{j_1}^{\omega_1}(x,y_1,z_1) V_{j_2}^{\omega_2}(x,y_2,z_2) V_{j_3}^{\omega_3}(x_3,y_3,z_3) \right\rangle dz \,, \tag{5.74}$$

where C denotes a contour encircling all three insertion points. This trivially vanishes as there is no residue at infinity. On the other hand, the action of the current on the vertex operators at *x* can be read off directly from Eq. (5.21), while for the one inserted

at  $x_3$  we use

$$J^{3}(x,z) = J^{3}(x_{3},z) + (x_{3}-x)J^{+}(x_{3},z), \qquad (5.75)$$

which follows from (5.9). We obtain the following differential equation:

$$\left[\sum_{i=1}^{3} \left(y_i \partial_{y_i} + j_i + \frac{k}{2}\omega_i\right) + (x_3 - x)\partial_{x_3}\right] \langle V_{j_1}^{\omega_1}(x, y_1, z_1) V_{j_2}^{\omega_2}(x, y_2, z_2) V_{j_3}^{\omega_3}(x_3, y_3, z_3) \rangle = 0.$$
(5.76)

We now Möbius-fix the worldsheet insertions to  $(0, 1, \infty)$ , and further consider the limit  $(x, x_3) \rightarrow (0, \infty)$ . The  $y_i$  coordinates then get rescaled according to Eq. (5.25). This leads to

$$\begin{bmatrix} y_1 \partial_{y_1} + y_2 \partial_{y_2} - y_3 \partial_{y_3} + j_1 + j_2 - j_3 \\ + \frac{k}{2} (\omega_1 + \omega_2 - \omega_3) \end{bmatrix} \langle V_{j_1}^{\omega_1}(0, y_1, 0) V_{j_2}^{\omega_2}(0, y_2, 1) V_{j_3}^{\omega_3}(\infty, y_3, \infty) \rangle = 0.$$

$$(5.77)$$

We find that Eq. (5.77) is the *y*-basis version of the usual charge-conservation equation for *m*-basis three-point functions of spectrally flowed primaries. This holds for all correlators satisfying (5.73), including both edge cases.

#### 5.2.3.1 Even edge cases

In this subsection we focus on the even edge cases, where  $\omega_3 = \omega_1 + \omega_2$ . We will derive the remaining two differential equations by considering correlators with an extra insertion of the  $J^-(x,z)$  current multiplied by the appropriate ratio of worldsheet coordinates. This is similar to what was used in [107] when discussing the proof of the *m*-basis spectral flow *violation* rules, and also more recently in [195] in the context of the supersymmetric version of this model.

The integral

$$\oint_{\mathcal{C}} \left\langle J^{-}(x_{3},z) V_{j_{1}}^{\omega_{1}}(x_{1},y_{1},z_{1}) V_{j_{2}}^{\omega_{2}}(x_{2},y_{2},z_{2}) V_{j_{3}}^{\omega_{3}}(x_{3},y_{3},z_{3}) \right\rangle \frac{(z-z_{1})^{\omega_{1}}(z-z_{2})^{\omega_{2}}}{(z-z_{3})^{\omega_{3}}} dz \quad (5.78)$$

vanishes since there is no pole at infinity. Using the OPEs of  $J^{-}(x_3, z)$  with the vertex operators, this yields

$$\begin{bmatrix} x_{31}^2 \frac{z_{12}^{\omega_2}}{z_{13}^{\omega_3}} \partial_{y_1} + x_{32}^2 \frac{z_{21}^{\omega_1}}{z_{23}^{\omega_3}} \partial_{y_2} + z_{31}^{\omega_1} z_{32}^{\omega_2} (y_3^2 \partial_{y_3} + 2j_3 y_3) \end{bmatrix} \\ \langle V_{j_1}^{\omega_1}(x_1, y_1, z_1) V_{j_2}^{\omega_2}(x_2, y_2, z_2) V_{j_3}^{\omega_3}(x_3, y_3, z_3) \rangle = 0.$$
 (5.79)

Proceeding similarly with

$$\oint_{\mathcal{C}} \left\langle J^{-}(x,z) V_{j_{1}}^{\omega_{1}}(x,y_{1},z_{1}) V_{j_{2}}^{\omega_{2}}(x,y_{2},z_{2}) V_{j_{3}}^{\omega_{3}}(x_{3},y_{3},z_{3}) \right\rangle \frac{(z-z_{3})^{\omega_{3}}}{(z-z_{1})^{\omega_{1}}(z-z_{2})^{\omega_{2}}} dz \,, \quad (5.80)$$

where we have imposed the collision limit mentioned above, we find

$$\begin{bmatrix} \frac{z_{13}^{\omega_3}}{z_{12}^{\omega_2}} (y_1^2 \partial_{y_1} + 2j_1 y_1) + \frac{z_{23}^{\omega_3}}{z_{21}^{\omega_2}} (y_2^2 \partial_{y_2} + 2j_2 y_2) + \frac{(x - x_3)^2}{z_{31}^{\omega_1} z_{32}^{\omega_2}} \partial_{y_3} \end{bmatrix}$$
  
$$\langle V_{j_1}^{\omega_1} (x, y_1, z_1) V_{j_2}^{\omega_2} (x, y_2, z_2) V_{j_3}^{\omega_3} (x_3, y_3, z_3) \rangle = 0.$$
(5.81)

We now fix the worldsheet coordinates to  $(0, 1, \infty)$  while sending  $x \to 0$  and  $x_3 \to \infty$ , and use Eq. (5.25) for the corresponding rescaling of the *y* variables. Including the charge conservation condition (5.77), the system of differential equations satisfied by the even edge correlator in the collision limit is then

$$0 = \left[ j_1 + j_2 - j_3 + y_1 \partial_{y_1} + y_2 \partial_{y_2} - y_3 \partial_{y_3} \right] \langle \dots \rangle ,$$
  

$$0 = \left[ (-1)^{\omega_1} \partial_{y_1} + (-1)^{\omega_3} \partial_{y_2} + (y_3^2 \partial_{y_3} + 2j_3 y_3) \right] \langle \dots \rangle ,$$
  

$$0 = \left[ (-1)^{\omega_1} (y_1^2 \partial_{y_1} + 2j_1 y_1) + (-1)^{\omega_3} (y_2^2 \partial_{y_2} + 2j_2 y_2) + \partial_{y_3} \right] \langle \dots \rangle .$$
  
(5.82)

where  $\langle ... \rangle$  stands for  $\langle V_{j_1}^{\omega_1}(0, y_1, 0) V_{j_2}^{\omega_2}(0, y_2, 1) V_{j_3}^{\omega_3}(\infty, y_3, \infty) \rangle$ . Note that only the second equation in (5.82) remains valid away from the collision limit.

Up to an overall *y*-independent constant, the general solution of the system (5.82) can be written as follows:

This matches the result of [38], see their Eq. (5.37b).

As mentioned above, for more general values of the insertion points the corresponding three-point functions follow from the global Ward identities. As it turns out, we can infer the result in a heuristic way by looking at the general expression given in Eq. (5.70). Indeed, one can verify that, upon setting  $\omega_3 = \omega_1 + \omega_2$ , the coefficients  $a_1(\Gamma_3^+), a_2(\Gamma_3^+)$  and  $a_2(\Gamma_1^-)$  diverge, which is a manifestation of the fact that the associated covering maps do not exist. Since all other coefficients remain finite, we obtain

$$\langle V_{j_{1}}^{\omega_{1}}(y_{1})V_{j_{2}}^{\omega_{2}}(y_{2})V_{j_{3}}^{\omega_{3}}(y_{3})\rangle = \left(1 + (-1)^{\omega_{1}}\frac{(\omega_{1} + \omega_{2} - 1)!}{(\omega_{1} - 1)!\omega_{2}!}y_{3} + (-1)^{\omega_{3}}y_{2}y_{3}\right)^{j_{1} - j_{2} - j_{3}} \\ \times \left(1 + (-1)^{\omega_{1} + 1}\frac{(\omega_{1} + \omega_{2} - 1)!}{\omega_{1}!(\omega_{2} - 1)!}y_{3} + (-1)^{\omega_{1}}y_{1}y_{3}\right)^{j_{2} - j_{3} - j_{1}} \\ \times \left(1 + (-1)^{\omega_{1} + 1}\frac{\omega_{1}!\omega_{2}!}{(\omega_{1} + \omega_{2})!}((-1)^{\omega_{1}}y_{1} - (-1)^{\omega_{3}}y_{2})\right)^{j_{3} - j_{1} - j_{2}}.$$

$$(5.84)$$

Up to the normalisation, to be discussed below, this precisely matches the conjecture (5.28).

#### 5.2.3.2 Odd edge cases

We now turn to the odd edge cases, where  $\omega_3 = \omega_1 + \omega_2 + 1$ . Since the procedure is analogous to what we just described we will skip some of the intermediate steps.

In addition to (5.77), we find two differential equations by considering contour integrals very similar to that in Eq. (5.78). We first take

$$\oint_{\mathcal{C}} \left\langle J^{-}(x_{3},z) V_{j_{1}}^{\omega_{1}}(x_{1},y_{1},z_{1}) V_{j_{2}}^{\omega_{2}}(x_{2},y_{2},z_{2}) V_{j_{3}}^{\omega_{3}}(x_{3},y_{3},z_{3}) \right\rangle \frac{(z-z_{1})^{\omega_{1}+1}(z-z_{2})^{\omega_{2}}}{(z-z_{3})^{\omega_{3}}} dz,$$
(5.85)

which again vanishes due to the absence of a residue at infinity. The same holds for

$$\oint_{\mathcal{C}} \left\langle J^{-}(x_{3},z) V_{j_{1}}^{\omega_{1}}(x_{1},y_{1},z_{1}) V_{j_{2}}^{\omega_{2}}(x_{2},y_{2},z_{2}) V_{j_{3}}^{\omega_{3}}(x_{3},y_{3},z_{3}) \right\rangle \frac{(z-z_{1})^{\omega_{1}}(z-z_{2})^{\omega_{2}+1}}{(z-z_{3})^{\omega_{3}}} dz.$$
(5.86)

Hence, we find the following system of differential equations:

$$0 = \left[ j_1 + j_2 - j_3 - \frac{k}{2} + y_1 \partial_{y_1} + y_2 \partial_{y_2} - y_3 \partial_{y_3} \right] \langle \dots \rangle ,$$
  

$$0 = \left[ (-1)^{\omega_3} \partial_{y_2} + (y_3^2 \partial_{y_3} + 2j_3 y_3) \right] \langle \dots \rangle ,$$
  

$$0 = \left[ (-1)^{\omega_1} \partial_{y_1} + (y_3^2 \partial_{y_3} + 2j_3 y_3) \right] \langle \dots \rangle .$$
  
(5.87)

where  $\langle ... \rangle$  again denotes  $\langle V_{j_1}^{\omega_1}(0, y_1, 0) V_{j_2}^{\omega_2}(0, y_2, 1) V_{j_3}^{\omega_3}(\infty, y_3, \infty) \rangle$ . In this case, only the first of these equations gets eventually modified away from the collision limit.

We find that, up to an overall normalization, the odd edge three-point functions read

$$\langle V_{j_1}^{\omega_1}(0, y_1, 0) V_{j_2}^{\omega_2}(0, y_2, 1) V_{j_3}^{\omega_3}(\infty, y_3, \infty) \rangle$$

$$= y_3^{j_1 + j_2 - j_3 - \frac{k}{2}} (1 + (-1)^{\omega_1} y_1 y_3 + (-1)^{\omega_3} y_2 y_3)^{\frac{k}{2} - j_1 - j_2 - j_3},$$
(5.88)

in the collision limit, thus matching the result in [38], see their Eq. (5.37c). Moreover, as in the even edge case, we can infer the solution for generic insertion points from the general expression in Eq. (5.51). For this, we set  $\omega_3 = \omega_1 + \omega_2 + 1$  and carefully take the limit  $a_3 \rightarrow 0$ ,  $a_{1,2} \rightarrow \infty$  with the products  $a_1a_3$  and  $a_2a_3$  fixed. This yields

$$\langle V_{j_1}^{\omega_1}(0, y_1, 0) V_{j_2}^{\omega_2}(1, y_2, 1) V_{j_3}^{\omega_3}(\infty, y_3, \infty) \rangle = y_3^{j_1 + j_2 - j_3 - \frac{k}{2}}$$

$$\times \left( 1 + (-1)^{\omega_1 + 1} \frac{(\omega_1 + \omega_2)!}{\omega_1! \omega_2!} y_3 + (-1)^{\omega_1} y_1 y_3 + (-1)^{\omega_3} y_2 y_3 \right)^{\frac{k}{2} - j_1 - j_2 - j_3} .$$

$$(5.89)$$

One can check that this matches the *y*-dependence given in the conjecture of Eq. (5.27) for correlators with appropriate spectral flow assignments. Moreover, upon using Eq. (5.55) and (5.56) we also see that, as expected, the expressions in Eq. (5.70) and Eq. (5.89) are related via series identifications.

#### 5.2.4 Three-point functions with unflowed insertions

So far, we have considered three-point functions where all vertex operators had nonzero spectral flow charges. However, it is natural to expect that the above results include the special cases where some of the insertions are unflowed. We now show how the latter are obtained. Note that we still assume  $\omega_3 \ge \omega_{1,2}$ , as in the previous sections.

Let us start by discussing the case of a single unflowed insertion, namely  $\omega_1 = 0$ . The fusion rules in Eq. (5.36) can then be satisfied iff  $\omega_3 = \omega_2$  or  $\omega_3 = \omega_2 + 1$ . These are exactly the two cases that were computed in [178] in full generality from *m*-basis techniques. In this sense, the results presented in this section are not new, but we include them for completeness. Hence, the relevant correlators correspond to the spectral flow assignments  $(0, \omega, \omega + 1)$  and  $(0, \omega, \omega)$ . By means of

$$V_{j}(x,z) = \mathcal{N}(j) V_{\frac{k}{2}-j,h=j}^{1}(x,z) = \mathcal{N}(j) \lim_{y \to \infty} y^{k-2j} V_{\frac{k}{2}-j}^{1}(x,y,z),$$
(5.90)

which is a particular case of (5.55), these can be obtained from the three-point functions with charges  $(1, \omega, \omega + 1)$  and  $(1, \omega, \omega)$ , respectively. More precisely, we have

$$\langle V_{j_1}V_{j_2}^{\omega}(y_2)V_{j_3}^{\omega+1}(y_3)\rangle = \mathcal{N}(j_1)\lim_{y_1\to\infty}y_1^{k-2j_1}\langle V_{\frac{k}{2}-j_1}^1(y_1)V_{j_2}^{\omega}(y_2)V_{j_3}^{\omega+1}(y_3)\rangle, \quad (5.91)$$

and

$$\langle V_{j_1}V_{j_2}^{\omega}(y_2)V_{j_3}^{\omega}(y_3)\rangle = \mathcal{N}(j_1)\lim_{y_1\to\infty} y_1^{k-2j_1}\langle V_{\frac{k}{2}-j_1}^1(y_1)V_{j_2}^{\omega}(y_2)V_{j_3}^{\omega}(y_3)\rangle,$$
(5.92)

where we have abbreviated  $V_{j_1}(0,0) \equiv V_{j_1}$ .

Focusing on (5.91), the RHS involves an even edge correlator. We thus need to consider the appropriate limit of Eq. (5.84), which gives

$$\langle V_{j_1}V_{j_2}^{\omega}(y_2)V_{j_3}^{\omega+1}(y_3)\rangle = y_3^{j_1+j_2-j_3-\frac{k}{2}} \left(1-y_3-(-1)^{\omega}y_2y_3\right)^{\frac{k}{2}-j_1-j_2-j_3}.$$
(5.93)

up to an overall constant. On the other hand, the RHS of (5.92) is obtained as the appropriate limit of the solution in Eq. (5.50), namely

$$\langle V_{j_1}^1(y_1) V_{j_2}^{\omega}(y_2) V_{j_3}^{\omega}(y_3) \rangle = (y_1 - \omega)^{j_2 + j_3 - j_1 - \frac{k}{2}} (y_2 + (-1)^{\omega})^{j_1 + j_3 - j_2 - \frac{k}{2}} (y_3 - 1)^{j_1 + j_2 - j_3 - \frac{k}{2}} \\ \times \left( (-1)^{\omega + 1} (\omega + 1) + (-1)^{\omega} y_1 - y_2 + (-1)^{\omega} y_3 - (\omega - 1) y_2 y_3 + y_1 y_2 y_3 \right)^{\frac{k}{2} - j_1 - j_2 - j_3}.$$

$$(5.94)$$

Hence, we get

$$\langle V_{j_1}V_{j_2}^{\omega}(y_2)V_{j_3}^{\omega}(y_3)\rangle = (y_2 + (-1)^{\omega})^{j_3 - j_1 - j_2} (y_3 - 1)^{j_2 - j_1 - j_3} ((-1)^{\omega} + y_2 y_3)^{j_1 - j_2 - j_3} .$$
(5.95)

up to the overall constant. Finally, we consider correlators with exactly two unflowed insertions,  $\omega_1 = \omega_2 = 0$ . According to (5.36) this can only be non-trivial for  $\omega_3 = 1$ . By using again Eq. (5.90) we get

$$\langle V_{j_1} V_{j_2} V_{j_3}^1(y_3) \rangle = \mathcal{N}(j_2) \lim_{y_2 \to \infty} y_2^{k-2j_2} \langle V_{j_1} V_{\frac{k}{2}-j_2}^1(y_2) V_{j_3}^1(y_3) \rangle$$
  
=  $y_3^{j_1+j_2-j_3-\frac{k}{2}} (y_3-1)^{\frac{k}{2}-j_1-j_2-j_3}$ , (5.96)

where in the last line we have ignored an overall normalization factor. In this way, we match all the corresponding results of [38], where the authors showed that this further reproduces the original computations of [103, 178].

#### 5.2.5 Normalization

So far we have focused on the dependence of the *y*-basis spectrally flowed correlators on the variables  $y_1$ ,  $y_2$  and  $y_3$ , and shown that it matches precisely the predictions of [38]. We now describe how the overall normalizations in Eqs. (5.27)-(5.33) are obtained<sup>51</sup>.

Once again, the argument relies on the SL(2, $\mathbb{R}$ ) series identifications. Indeed, identities such as those in Eqs. (5.58) and (5.60) must hold exactly, including the normalization factors. Having fixed the *y*-dependence, we can thus determine the normalizations recursively, starting from the unflowed three-point functions of [106, 103]. For instance, we consider the following identity:

$$\lim_{y_3 \to \infty} y_3^{2j_3} \left\langle V_{j_1}^{\omega_1}(y_1) \, V_{j_2}^{\omega_2}(y_2) \, V_{j_3}^{\omega_3}(y_3) \right\rangle = \mathcal{N}(j_3) \left\langle V_{j_1}^{\omega_1}(y_1) \, V_{j_2}^{\omega_2}(y_2) \, V_{\frac{k}{2}-j_3}^{\omega_3-1}(0) \right\rangle \,, \quad (5.97)$$

which will give us a recursion relation for  $C_{\omega}(j_1, j_2, j_3)$ . Since the latter is independent of the  $y_i$ , we can set  $y_1 = y_2 = 0$ . Using the *y*-dependence derived above, written as in

 $<sup>^{51}</sup>$ This was already discussed in [39], assuming the *y*-dependence of the correlators was as in [38]. We reproduce the argument here for completeness

Eqs. (5.27) and (5.28), one finds that the product of  $X_I$  factors on the left- an right-hand sides of (5.97), both reduce to either

$$P_{\omega}^{j_1+j_2+j_3-k} P_{\omega+e_1+e_2}^{j_3-j_1-j_2} P_{\omega+e_2-e_3}^{j_1-j_2-j_3} P_{\omega+e_1-e_3}^{j_2-j_3-j_1},$$
(5.98)

or

$$P_{\omega+e_{1}+e_{2}-e_{3}}^{\frac{k}{2}-j_{1}-j_{2}-j_{3}}P_{\omega+e_{1}}^{-j_{1}+j_{2}+j_{3}-\frac{k}{2}}P_{\omega+e_{2}}^{j_{1}-j_{2}+j_{3}-\frac{k}{2}}P_{\omega-e_{3}}^{j_{1}+j_{2}-j_{3}-\frac{k}{2}},$$
(5.99)

depending on the overall parity of the spectral flow charges. Consequently, in both cases we find that Eq. (5.97) holds iff

$$C_{\omega}(j_1, j_2, j_3) = \mathcal{N}(j_3) C_{\omega - e_3}\left(j_1, j_2, \frac{k}{2} - j_3\right).$$
(5.100)

Analogous statements can be derived by shifting the spectral flow charges  $\omega_1$  and  $\omega_2$  instead. Moreover, one has the identity

$$\mathcal{N}(j_1)C\left(\frac{k}{2} - j_1, j_2, j_3\right) = \mathcal{N}(j_2)C\left(j_1, \frac{k}{2} - j_2, j_3\right) = \mathcal{N}(j_3)C\left(j_1, j_2, \frac{k}{2} - j_3\right).$$
 (5.101)

Since  $\mathcal{N}(j)\mathcal{N}(\frac{k}{2}-j) = 1$ , it follows that, as stated in (5.33),  $C_{\omega}(j_1, j_2, j_3)$  can only be  $C(j_1, j_2, j_3)$ , i.e. the unflowed three-point function, or  $\mathcal{N}(j_1)C(\frac{k}{2}-j_1, j_2, j_3)$ , depending on the parity of  $\omega_1 + \omega_2 + \omega_3$ . To be precise, this argument is valid for discrete representations, although we expect that it holds also for the continuous series by analytic continuation in j [103, 196]. This concludes our computation of three-point functions with arbitrary spectral flow charges.

### 5.3 Discussion

In this chapter, we have computed the *y*-basis string three-point function in AdS<sub>3</sub> involving vertex operators with arbitrary spectral flow charges. This provides a proof for the conjecture put forward recently in [38], thus establishing integral expressions for all (primary) three-point functions of the SL(2, $\mathbb{R}$ )-WZW model at level *k*, for all *k* > 3.

The subfamily of (odd parity) correlators for which a holomorphic covering map from the worldsheet to the  $AdS_3$  boundary exists had been obtained in [88, 38]. Here we have relied on the general structure of local Ward identities (in their *y*-basis formulation) and made extensive use of the  $SL(2,\mathbb{R})$  series identifications, whose importance was recently highlighted in [39]. This allowed us to extend the methods based on covering maps to all other non-vanishing correlators, as defined by the fusion rules computed in [103].

Our strategy can be summarised as follows. We first argued that the differential equations satisfied by all *y*-basis three-point functions must take the form given in Eq. (5.54). Obtaining the general expressions for these equations for all even parity correlators then reduces to computing all unknown coefficients in (5.54). We have provided the relations among correlators with adjacent spectral flow assignments that follow from  $SL(2,\mathbb{R})$  series identifications in Eqs. (5.58) and (5.60). These provide a considerable number of identities between even and odd parity correlators in the limit where one of the *y* variables is taken to either zero or infinity. This allowed us to evaluate all relevant coefficients  $A_{ij}$ ,  $B_{ij}$  and  $C_i$  in closed form, as given in Eqs. (5.64)-(5.66). The derivation of these 21 coefficients involves solving a total of 60 conditions, of which 39 can be taken as consistency checks. The latter turn out to be satisfied in a highly non-trivial manner, related to the existence of a set of identities relating the behaviour of different covering maps in the vicinity of the insertion points.

The resulting differential equations satisfied by even parity correlators are provided in Eqs. (5.67)-(5.69). These show a striking similarity with the cases of odd total spectral flow, a hallmark of the existence of a more direct derivation by means of *adjacent* covering maps. We leave this for future work. Here we have shown that the general solution to these equations, namely Eq. (5.70), is compatible with the proposal of [38].

We have also discussed the so-called edge cases, whose spectral flow assignments saturate the fusion rules in Eqs. (5.37) and (5.53). Some subtleties arise when trying to apply the general method described above in this context. For these cases we have provided an alternative approach, based on an improved version of the *m*-basis methods [103, 178]. We have then described how to obtain correlators involving unflowed insertions. Finally, we fixed the overall normalization of all *y*-basis three-point functions, following the arguments of [39].

At this point, it is natural to ask if an analogous story holds for four-point functions, which encode crucial dynamical information about the theory. A closed formula for four-point functions in the *y*-basis with arbitrary spectral flow assignments in terms of the corresponding unflowed correlator was conjectured in [179]. Here the situation is more subtle: on top of the four  $y_i$  variables, four-point functions also depend nontrivially on the worldsheet and spacetime cross-ratios, and must satisfy the corresponding Knizhnik-Zamolodchikov equations. It has been known for some time [197, 198] that the latter intertwine non-trivially with the recursion relations of the type described in section 5.1.2. If the structure put forward in [179] is correct, its proof is likely to work in two steps. First, one should use arguments similar to those we have considered in this chapter to show that solutions to the y-basis differential equations associated with the four-point functions consist of various prefactors given by powers of the generalized differences  $X_I$ , defined in Eq. (5.29), multiplied by an arbitrary function of the so-called generalized cross-ratio. Second, one should prove that this arbitrary function must satisfy the same Knizhnik-Zamolodchikov equation as the corresponding unflowed four-point function. The extension of the proof for the case of correlation functions involving four spectrally-flowed vertex operators is a work in progress.

## Chapter 6

## Conclusions

In this final chapter we summarise the findings presented above, and we will discuss some possible future projects and generalisations.

In Chapter 3 we have shown, among other things, how the consistency of the worldsheet CFT is in one to one correspondence with absence of CTCs, singularities and horizon. We have proven that the quantisation conditions derived from the worldsheet BRST analysis are exactly the same as those obtained in the gravity description when demanding absence of singularities, horizons and CTCs. Using these results, we showed that the metric, *B*-field and dilaton obtained from the coset model can be rewritten in terms of the integers k, m, n,  $n_5$  and the modulus  $R_y$ . These background fields precisely coincide with the NS5-decoupled JMaRT background [37], thus proving that the latter is the most general consistent supergravity background described by the coset Eq. (2.205). Despite this 'no-go theorem', we were able to derive a novel subfamily of two-charge non-BPS NS5-P backgrounds not appeared before in the literature, see Eqs. (3.76) and (3.77).

The computation of stringy observables involving light probes in the NS5-decoupled JMaRT background was presented in Chapter 4. Here we first constructed the physical vertex operators (in both NS and R sectors) in the AdS limit of the coset model, and we carefully identified the worldsheet *x* variable dual to the local coordinate of the holographic CFT. Our findings heavily generalise the results available in the literature [167, 183, 181]. We first extended the computation of HLLH correlators to the full non-supersymmetric AdS limit of the JMaRT solution. The light massless probes involved in the correlation function possess any conformal dimension and R-charge. Secondly, we further generalised the results to an arbitrary number of light insertions, providing an elegant formula that express the coset correlator as a function of the vacuum  $AdS_3 \times S^3$  correlator, see Eq. (4.94). This is valid for an arbitrary number of massless insertions of weights  $h_i$  and charges  $m'_i$  and  $\bar{m}'_i$ , and also arbitrary parameters (k, s,  $\bar{s}$ ) for

which a consistent background exists. We also observed that, at large *N*, certain worldsheet correlators exactly match the that of the dual orbifold CFT. The non-BPS JMaRT backgrounds were not considered in [167], however we demonstrated that the agreement extends also to those backgrounds. It is known that vacuum two- and three-point functions of chiral primary operators are protected [184]. We thus conjectured that heavy-light correlators in JMaRT heavy states are protected whenever the corresponding vacuum correlator in our general formula (4.94) is protected. We investigated a particular HLLLH five-point function—the first of its kind in the literature—and found that worldsheet and symmetric product results agree.

Given its importance for coset correlators and more broadly for the  $AdS_3/CFT_2$  correspondence, in Chapter 5 we presented the computation of the *y*-basis string three-point function in AdS<sub>3</sub> involving vertex operators with arbitrary spectral flow charges. This proves the conjectured integral expressions for all primary three-point functions of the SL(2,  $\mathbb{R}$ )-WZW model at level *k*, for all *k* > 3, first appeared in [38]. In order to achieve this goal we exploited the general *y*-basis structure of local Ward identities and, crucially, made extensive use of the *y*-basis SL(2,  $\mathbb{R}$ ) series identifications [39].

A number of possible ideas for future research projects can originate from this thesis. Let us discuss some possible developments.

First of all, it would be tempting to generalise the results of Chapter 3 to different backgrounds. The fascinating relation between the worldsheet consistency and absence of singularities, horizons and CTCs is very suggestive and perfectly in line with the Fuzzball proposal mentioned in the Introduction. However, great care is due. At the point of writing it is far from obvious how to generalise these results to all the backgrounds that admit a worldsheet description of some sort. Hence, an obvious task would be that of finding more general (and more typical) microstate backgrounds and verify the absence of horizons, singularities and CTCs. Finding new worldsheet models is notoriously difficult. However, a natural extension of our work is immediate when involving the  $\mathbb{T}^4$  in the gauging. In the latter case, we strongly believe that the majority of the results will be unchanged: the background will be absent of singularities, horizons and CTCs. Most of the computations presented in Chapter 3 will follow through.

Nevertheless, we expect that our results will be useful in analysing generalisations of the models studied here, either by changing the currents being gauged to include non-Cartan generators of the non-Abelian factors of the upstairs group, or by changing the upstairs group, or both. Our systematic approach should enable generalisations to be investigated in a similar way. For instance, there are multi-centre non-BPS generalizations of the JMaRT family [160, 161, 162] or the newly discovered NS-NS superstrata [163].

Once new worldsheet descriptions for microstate backgrounds are found, it is natural to try to generalise the results we have presented in Chapter 4. The computation of heavy-light correlators will provide valuable information about possible non-trivial properties of the background and be of great interest in the light of holography in backgrounds different from the global vacuum. We expect that our results may provide a guideline for these possible developments.

The reader should note that the computations of Chapter 4 were restricted to the AdS limit of the coset model. However, a natural question to ask is how to extend the results to the UV theory. In the full asymptotically linear dilaton models, correlators can be computed by using the vertex operators of Section 4.1. However, the x coordinate seems unlikely to be of any use in this regime given the non-local nature of the dual theory.

The results in Chapter 5 concern the proof of the conjecture for three point functions. It is very natural to ask if an analogous story holds for four-point functions, which encode crucial dynamical information about the theory. A closed formula for four-point functions in the *y*-basis with arbitrary spectral flow assignments in terms of the corresponding unflowed correlator was conjectured in [179]. Here the situation is more subtle: on top of the four  $y_i$  variables, four-point functions also depend non-trivially on the worldsheet and spacetime cross-ratios, and must satisfy the corresponding Knizhnik-Zamolodchikov equations. It has been known for some time [197, 198] that the latter intertwine non-trivially with the recursion relations of the type described in Section 5.1.2.

In this thesis we have shown that worldsheet techniques, when available, are tremendously effective and powerful. By contrast, trying to use a field theory description for black hole interior and horizon physics has proven so far to be largely unsuccessful. Being able to compute observables to all orders in  $\alpha'$  allows us to depart from (super-)gravity results, and hence be able to directly address problems associated to black hole physics with the power of string theory. Indeed, on top of the infinite degrees of freedom provided by stringy excitations, it is the extended nature of strings and branes that allows us to tame issues related to curvature singularities and black hole paradoxes.

Despite its success, string theory is nowadays still in large development. Many aspects of it are still not clear, especially in strong coupling and less supersymmetric settings, two regimes that may be crucial for an appropriate description of black holes and their interior. However, there are indications that string theory will shed a definite light on problems involving short distance physics, strong gravity, singularities, and various other puzzles and paradoxes. A deeper understanding of the whole theoretical framework will necessarily help in the resolution of the information paradox and the entropy puzzle: these issues afflict black holes since the very first computations that followed from the tremendous insight and intuition of Stephen Hawking.

# References

- [1] D. Bufalini, S. Iguri, N. Kovensky and D. Turton, *Black hole microstates from the worldsheet*, *JHEP* 08 (2021) 011 [2105.02255].
- [2] D. Bufalini, S. Iguri, N. Kovensky and D. Turton, Worldsheet Correlators in Black Hole Microstates, Phys. Rev. Lett. 129 (2022) 121603 [2203.13828].
- [3] D. Bufalini, S. Iguri, N. Kovensky and D. Turton, *Worldsheet computation of heavy-light correlators*, *JHEP* **03** (2023) 066 [2210.15313].
- [4] D. Bufalini, S. Iguri and N. Kovensky, A proof for string three-point functions in AdS<sub>3</sub>, JHEP 02 (2023) 246 [2212.05877].
- [5] LIGO SCIENTIFIC COLLABORATION AND VIRGO COLLABORATION collaboration, *Observation of gravitational waves from a binary black hole merger*, *Phys. Rev. Lett.* **116** (2016) 061102.
- [6] EVENT HORIZON TELESCOPE collaboration, First M87 Event Horizon Telescope Results. I. The Shadow of the Supermassive Black Hole, Astrophys. J. Lett. 875 (2019) L1 [1906.11238].
- [7] J. D. Bekenstein, Black hole hair: twenty-five years after, arXiv preprint gr-qc/9605059 (1996).
- [8] R. Ruffini and J. Wheeler, Introducing the black hole, Physics Today 24 (2009).
- [9] J. D. Bekenstein, Black holes and entropy, Phys. Rev. D7 (1973) 2333.
- [10] S. W. Hawking, Black hole explosions, Nature 248 (1974) 30.
- [11] S. W. Hawking, Particle Creation by Black Holes, Commun. Math. Phys. 43 (1975) 199.
- [12] S. W. Hawking, Breakdown of Predictability in Gravitational Collapse, Phys. Rev. D14 (1976) 2460.
- [13] R. M. Wald, On Particle Creation by Black Holes, Commun. Math. Phys. 45 (1975) 9.

- [14] G. 't Hooft, Dimensional reduction in quantum gravity, Conf. Proc. C 930308 (1993) 284 [gr-qc/9310026].
- [15] L. Susskind, The world as a hologram, Journal of Mathematical Physics 36 (1995) 6377–6396.
- [16] J. M. Maldacena, The Large N limit of superconformal field theories and supergravity, Int. J. Theor. Phys. 38 (1999) 1113 [hep-th/9711200].
- [17] E. Witten, Anti-de Sitter space and holography, Adv. Theor. Math. Phys. 2 (1998) 253 [hep-th/9802150].
- [18] S. S. Gubser, I. R. Klebanov and A. M. Polyakov, *Gauge theory correlators from non-critical string theory*, *Phys. Lett.* B428 (1998) 105 [hep-th/9802109].
- [19] A. Strominger and C. Vafa, Microscopic Origin of the Bekenstein-Hawking Entropy, Phys. Lett. B379 (1996) 99 [hep-th/9601029].
- [20] S. De Haro, J. van Dongen, M. Visser and J. Butterfield, Conceptual analysis of black hole entropy in string theory, Studies in History and Philosophy of Science Part B: Studies in History and Philosophy of Modern Physics 69 (2020) 82–111.
- [21] S. D. Mathur, What Exactly is the Information Paradox?, Lect. Notes Phys. **769** (2009) 3 [0803.2030].
- [22] S. D. Mathur, The fuzzball proposal for black holes: An elementary review, Fortsch. Phys. 53 (2005) 793 [hep-th/0502050].
- [23] Y. Aharonov, A. Casher and S. Nussinov, *The Unitarity Puzzle and Planck Mass Stable Particles*, *Phys. Lett. B* **191** (1987) 51.
- [24] J. D. Bekenstein, Universal upper bound on the entropy-to-energy ratio for bounded systems, Phys. Rev. D 23 (1981) 287.
- [25] J. D. Bekenstein, Entropy bounds and black hole remnants, Phys. Rev. D 49 (1994) 1912 [gr-qc/9307035].
- [26] L. Susskind, Trouble for remnants, hep-th/9501106.
- [27] S. B. Giddings, The Black hole information paradox, in PASCOS / HOPKINS 1995 (Joint Meeting of the International Symposium on Particles, Strings and Cosmology and the 19th Johns Hopkins Workshop on Current Problems in Particle Theory), pp. 415–428, 8, 1995, hep-th/9508151.
- [28] A. Almheiri, N. Engelhardt, D. Marolf and H. Maxfield, *The entropy of bulk quantum fields and the entanglement wedge of an evaporating black hole*, *JHEP* 12 (2019) 063 [1905.08762].

- [29] G. Penington, Entanglement Wedge Reconstruction and the Information Paradox, *JHEP* 09 (2020) 002 [1905.08255].
- [30] G. Penington, S. H. Shenker, D. Stanford and Z. Yang, *Replica wormholes and the black hole interior*, *JHEP* **03** (2022) 205 [1911.11977].
- [31] A. Almheiri, R. Mahajan, J. Maldacena and Y. Zhao, *The Page curve of Hawking radiation from semiclassical geometry*, *JHEP* **03** (2020) 149 [1908.10996].
- [32] A. Almheiri, T. Hartman, J. Maldacena, E. Shaghoulian and A. Tajdini, *Replica wormholes and the entropy of hawking radiation*, *Journal of High Energy Physics* 2020 (2020).
- [33] N. Engelhardt and A. C. Wall, *Quantum extremal surfaces: holographic entanglement entropy beyond the classical regime, Journal of High Energy Physics* **2015** (2015) .
- [34] B. Guo, M. R. R. Hughes, S. D. Mathur and M. Mehta, Contrasting the fuzzball and wormhole paradigms for black holes, Turk. J. Phys. 45 (2021) 281 [2111.05295].
- [35] K. Skenderis and M. Taylor, The fuzzball proposal for black holes, Phys. Rept. 467 (2008) 117 [0804.0552].
- [36] S. D. Mathur, The information paradox: A pedagogical introduction, Class. Quant. Grav. 26 (2009) 224001 [0909.1038].
- [37] V. Jejjala, O. Madden, S. F. Ross and G. Titchener, Non-supersymmetric smooth geometries and D1-D5-P bound states, Phys. Rev. D71 (2005) 124030 [hep-th/0504181].
- [38] A. Dei and L. Eberhardt, String correlators on AdS<sub>3</sub>: three-point functions, JHEP 08 (2021) 025 [2105.12130].
- [39] S. Iguri and N. Kovensky, On spectrally flowed local vertex operators in AdS<sub>3</sub>, 2208.00978.
- [40] M. R. Douglas, Branes within branes, NATO Sci. Ser. C 520 (1999) 267 [hep-th/9512077].
- [41] A. Marrani, P. Truini and M. Rios, *The Magic of Being Exceptional*, J. Phys. Conf. Ser. **1194** (2019) 012075 [1811.11208].
- [42] R. Gilmore, Lie Groups, Lie Algebras & Some of Their Applications. Dover Books, 1974.
- [43] C. M. Hull and P. K. Townsend, Unity of superstring dualities, Nucl. Phys. B 438 (1995) 109 [hep-th/9410167].
- [44] F. Larsen and E. J. Martinec, U(1) charges and moduli in the D1 D5 system, JHEP
   06 (1999) 019 [hep-th/9905064].

- [45] J. Polchinski, String theory. Vol. 2: Superstring Theory and Beyond. Cambridge Univ. Pr., 1998.
- [46] S. G. Avery, Using the D1D5 CFT to Understand Black Holes, 1012.0072.
- [47] L. J. Romans, Selfduality for Interacting Fields: Covariant Field Equations for Six-dimensional Chiral Supergravities, Nucl. Phys. B 276 (1986) 71.
- [48] S. Deger, A. Kaya, E. Sezgin and P. Sundell, Spectrum of D = 6, N=4b supergravity on AdS in three-dimensions x S\*\*3, Nucl. Phys. B 536 (1998) 110 [hep-th/9804166].
- [49] R. Dijkgraaf, Instanton strings and hyperKahler geometry, Nucl. Phys. B543 (1999) 545 [hep-th/9810210].
- [50] N. Seiberg and E. Witten, The D1 / D5 system and singular CFT, JHEP 04 (1999) 017 [hep-th/9903224].
- [51] E. J. Martinec and S. Massai, *String Theory of Supertubes*, *JHEP* 07 (2018) 163
   [1705.10844].
- [52] E. J. Martinec, S. Massai and D. Turton, *String dynamics in NS5-F1-P geometries*, *JHEP* 09 (2018) 031 [1803.08505].
- [53] E. J. Martinec, S. Massai and D. Turton, *Little Strings, Long Strings, and Fuzzballs*, *JHEP* 11 (2019) 019 [1906.11473].
- [54] E. J. Martinec, S. Massai and D. Turton, Stringy Structure at the BPS Bound, JHEP 12 (2020) 135 [2005.12344].
- [55] S. Ferrara and R. Kallosh, Supersymmetry and attractors, Phys. Rev. D 54 (1996) 1514 [hep-th/9602136].
- [56] E. Witten, On the conformal field theory of the Higgs branch, JHEP 07 (1997) 003 [hep-th/9707093].
- [57] J. M. Maldacena, J. Michelson and A. Strominger, Anti-de Sitter fragmentation, JHEP 02 (1999) 011 [hep-th/9812073].
- [58] J. R. David, G. Mandal and S. R. Wadia, Microscopic formulation of black holes in string theory, Phys. Rept. 369 (2002) 549 [hep-th/0203048].
- [59] M. Cvetic and D. Youm, General Rotating Five Dimensional Black Holes of Toroidally Compactified Heterotic String, Nucl. Phys. B476 (1996) 118 [hep-th/9603100].
- [60] M. Cvetic and F. Larsen, Near horizon geometry of rotating black holes in five dimensions, Nucl. Phys. B531 (1998) 239 [hep-th/9805097].
- [61] T. Azeyanagi, N. Ogawa and S. Terashima, *The Kerr/CFT Correspondence and String Theory*, *Phys. Rev. D* 79 (2009) 106009 [0812.4883].

- [62] J. Breckenridge, D. Lowe, R. C. Myers, A. Peet, A. Strominger et al., Macroscopic and microscopic entropy of near extremal spinning black holes, Phys.Lett. B381 (1996) 423 [hep-th/9603078].
- [63] C. G. Callan and J. M. Maldacena, D-brane Approach to Black Hole Quantum Mechanics, Nucl. Phys. B472 (1996) 591 [hep-th/9602043].
- [64] O. Lunin and S. D. Mathur, Metric of the multiply wound rotating string, Nucl. Phys. B610 (2001) 49 [hep-th/0105136].
- [65] O. Lunin and S. D. Mathur, AdS / CFT duality and the black hole information paradox, Nucl. Phys. B 623 (2002) 342 [hep-th/0109154].
- [66] O. Lunin, J. M. Maldacena and L. Maoz, Gravity solutions for the D1-D5 system with angular momentum, hep-th/0212210.
- [67] M. Taylor, General 2 charge geometries, JHEP 03 (2006) 009 [hep-th/0507223].
- [68] V. S. Rychkov, D1-D5 black hole microstate counting from supergravity, JHEP 01 (2006) 063 [hep-th/0512053].
- [69] A. Sen, Two Charge System Revisited: Small Black Holes or Horizonless Solutions?, JHEP 05 (2010) 097 [0908.3402].
- [70] S. D. Mathur and D. Turton, The fuzzball nature of two-charge black hole microstates, Nuclear Physics B 945 (2019) 114684.
- [71] C. G. Callan, Jr., J. A. Harvey and A. Strominger, Supersymmetric string solitons, hep-th/9112030.
- [72] I. Kanitscheider, K. Skenderis and M. Taylor, *Fuzzballs with internal excitations*, *JHEP* 06 (2007) 056 [0704.0690].
- [73] O. Lunin and S. D. Mathur, Statistical interpretation of Bekenstein entropy for systems with a stretched horizon, Phys. Rev. Lett. 88 (2002) 211303
   [hep-th/0202072].
- [74] G. T. Horowitz, J. M. Maldacena and A. Strominger, Nonextremal black hole microstates and U duality, Phys. Lett. B383 (1996) 151 [hep-th/9603109].
- [75] J. Breckenridge, R. C. Myers, A. Peet and C. Vafa, *D-branes and spinning black holes*, *Phys.Lett.* B391 (1997) 93 [hep-th/9602065].
- [76] J. M. Maldacena, A. Strominger and E. Witten, Black hole entropy in M-theory, JHEP 12 (1997) 002 [hep-th/9711053].
- [77] C. Vafa, Instantons on D-branes, Nucl. Phys. B463 (1996) 435 [hep-th/9512078].

- [78] A. Giveon, D. Kutasov and N. Seiberg, Comments on string theory on AdS(3), Adv. Theor. Math. Phys. 2 (1998) 733 [hep-th/9806194].
- [79] J. M. Maldacena, Black holes in string theory, hep-th/9607235.
- [80] R. Blumenhagen, D. Lüst and S. Theisen, *Basic concepts of string theory*, Theoretical and Mathematical Physics. Springer, Heidelberg, Germany, 2013, 10.1007/978-3-642-29497-6.
- [81] E. Witten, Some comments on string dynamics, in STRINGS 95: Future Perspectives in String Theory, pp. 501–523, 7, 1995, hep-th/9507121.
- [82] N. Seiberg, Observations on the Moduli Space of Superconformal Field Theories, Nucl. Phys. B 303 (1988) 286.
- [83] S. Cecotti, N=2 Landau-Ginzburg versus Calabi-Yau sigma models: Nonperturbative aspects, Int. J. Mod. Phys. A 6 (1991) 1749.
- [84] J. de Boer, J. Manschot, K. Papadodimas and E. Verlinde, *The chiral ring of AdS3/CFT2 and the attractor mechanism*, *JHEP* **03** (2009) 030 [0809.0507].
- [85] L. Eberhardt and K. Ferreira, Long strings and chiral primaries in the hybrid formalism, JHEP 02 (2019) 098 [1810.08621].
- [86] M. R. Gaberdiel, R. Gopakumar and C. Hull, *Stringy AdS<sub>3</sub> from the worldsheet*, *JHEP* 07 (2017) 090 [1704.08665].
- [87] L. Eberhardt, M. R. Gaberdiel and R. Gopakumar, *The Worldsheet Dual of the Symmetric Product CFT*, JHEP 04 (2019) 103 [1812.01007].
- [88] L. Eberhardt, M. R. Gaberdiel and R. Gopakumar, *Deriving the AdS<sub>3</sub>/CFT<sub>2</sub> correspondence*, *JHEP* 02 (2020) 136 [1911.00378].
- [89] A. Pakman, L. Rastelli and S. S. Razamat, Extremal Correlators and Hurwitz Numbers in Symmetric Product Orbifolds, Phys. Rev. D80 (2009) 086009 [0905.3451].
- [90] L. Eberhardt, A perturbative CFT dual for pure NS–NS AdS<sub>3</sub> strings, J. Phys. A 55 (2022) 064001 [2110.07535].
- [91] A. Dei and L. Eberhardt, *String correlators on AdS*<sub>3</sub>: *analytic structure and dual CFT*, 2203.13264.
- [92] I. Bena, S. Giusto, E. J. Martinec, R. Russo, M. Shigemori, D. Turton et al., Asymptotically-flat supergravity solutions deep inside the black-hole regime, JHEP 02 (2018) 014 [1711.10474].
- [93] S. G. Avery, B. D. Chowdhury and S. D. Mathur, *Emission from the D1D5 CFT*, *JHEP* **10** (2009) 065 [0906.2015].

- [94] B. Chakrabarty, D. Turton and A. Virmani, *Holographic description of non-supersymmetric orbifolded D1-D5-P solutions*, *JHEP* 11 (2015) 063
   [1508.01231].
- [95] E. J. Martinec and W. McElgin, String theory on AdS orbifolds, JHEP 04 (2002) 029 [hep-th/0106171].
- [96] S. Giusto, O. Lunin, S. D. Mathur and D. Turton, *D1-D5-P microstates at the cap*, *JHEP* **1302** (2013) 050 [1211.0306].
- [97] O. Lunin and S. D. Mathur, Three point functions for M(N) / S(N) orbifolds with N=4 supersymmetry, Commun. Math. Phys. 227 (2002) 385 [hep-th/0103169].
- [98] E. Moscato, *Black hole microstates and holography in the D1D5 CFT*, Ph.D. thesis, Queen Mary, U. of London, 9, 2017.
- [99] A. Dabholkar and A. Pakman, Exact chiral ring of AdS(3) / CFT(2), Adv. Theor. Math. Phys. 13 (2009) 409 [hep-th/0703022].
- [100] G. Giribet, A. Pakman and L. Rastelli, Spectral Flow in AdS(3)/CFT(2), JHEP 06 (2008) 013 [0712.3046].
- [101] J. M. Maldacena and H. Ooguri, Strings in AdS(3) and SL(2,R) WZW model 1.: The Spectrum, J. Math. Phys. 42 (2001) 2929 [hep-th/0001053].
- [102] J. M. Maldacena, H. Ooguri and J. Son, Strings in AdS(3) and the SL(2,R) WZW model. Part 2. Euclidean black hole, J. Math. Phys. 42 (2001) 2961 [hep-th/0005183].
- [103] J. M. Maldacena and H. Ooguri, Strings in AdS(3) and the SL(2,R) WZW model. Part 3. Correlation functions, Phys. Rev. D65 (2002) 106006 [hep-th/0111180].
- [104] J. Teschner, On structure constants and fusion rules in the SL(2,C) / SU(2) WZNW model, Nucl. Phys. B546 (1999) 390 [hep-th/9712256].
- [105] J. Teschner, The Minisuperspace limit of the sl(2,C) / SU(2) WZNW model, Nucl. Phys. B 546 (1999) 369 [hep-th/9712258].
- [106] J. Teschner, Operator product expansion and factorization in the H+(3) WZNW model, Nucl. Phys. B571 (2000) 555 [hep-th/9906215].
- [107] W. McElgin, Notes on the SL(2,R) CFT, 1511.07256.
- [108] D. Harlow, J. Maltz and E. Witten, Analytic Continuation of Liouville Theory, JHEP 12 (2011) 071 [1108.4417].
- [109] A. Zamolodchikov and V. Fateev, Operator Algebra and Correlation Functions in the Two-Dimensional Wess-Zumino SU(2) x SU(2) Chiral Model, Sov. J. Nucl. Phys. 43 (1986) 657.

- [110] D. Kutasov and N. Seiberg, More comments on string theory on AdS(3), JHEP 04 (1999) 008 [hep-th/9903219].
- [111] V. Fateev and A. Zamolodchikov, Parafermionic Currents in the Two-Dimensional Conformal Quantum Field Theory and Selfdual Critical Points in Z(n) Invariant Statistical Systems, Sov. Phys. JETP 62 (1985) 215.
- [112] D. Kutasov, F. Larsen and R. G. Leigh, String theory in magnetic monopole backgrounds, Nucl. Phys. B 550 (1999) 183 [hep-th/9812027].
- [113] M. R. Gaberdiel and I. Kirsch, Worldsheet correlators in AdS(3)/CFT(2), JHEP 04 (2007) 050 [hep-th/0703001].
- [114] J. Kim and M. Porrati, *On the central charge of spacetime current algebras and correlators in string theory on AdS*<sub>3</sub>, *JHEP* **05** (2015) 076 [1503.07186].
- [115] A. Giveon and D. Kutasov, Notes on AdS(3), Nucl. Phys. B621 (2002) 303 [hep-th/0106004].
- [116] L. Eberhardt, Partition functions of the tensionless string, JHEP 03 (2021) 176 [2008.07533].
- [117] A. Pakman and A. Sever, Exact N=4 correlators of AdS(3)/CFT(2), Phys. Lett. B 652 (2007) 60 [0704.3040].
- [118] D. Israel, C. Kounnas, A. Pakman and J. Troost, The Partition function of the supersymmetric two-dimensional black hole and little string theory, JHEP 06 (2004) 033 [hep-th/0403237].
- [119] K. Sfetsos, Branes for Higgs phases and exact conformal field theories, JHEP 01 (1999) 015 [hep-th/9811167].
- [120] A. Giveon and D. Kutasov, Little string theory in a double scaling limit, JHEP 10 (1999) 034 [hep-th/9909110].
- [121] A. Giveon and D. Kutasov, Comments on double scaled little string theory, JHEP 01 (2000) 023 [hep-th/9911039].
- [122] J. M. Maldacena and L. Maoz, Desingularization by rotation, JHEP 12 (2002) 055 [hep-th/0012025].
- [123] V. Balasubramanian, J. de Boer, E. Keski-Vakkuri and S. F. Ross, Supersymmetric conical defects: Towards a string theoretic description of black hole formation, Phys. Rev. D 64 (2001) 064011 [hep-th/0011217].
- [124] S. Giusto, S. D. Mathur and A. Saxena, Dual geometries for a set of 3-charge microstates, Nucl. Phys. B701 (2004) 357 [hep-th/0405017].

- [125] S. Giusto, S. D. Mathur and A. Saxena, 3-charge geometries and their CFT duals, Nucl. Phys. B710 (2005) 425 [hep-th/0406103].
- [126] A. A. Tseytlin, Conformal sigma models corresponding to gauged Wess-Zumino-Witten theories, Nucl. Phys. B 411 (1994) 509 [hep-th/9302083].
- [127] C. Klimcik and A. A. Tseytlin, Exact four-dimensional string solutions and Toda like sigma models from 'null gauged' WZNW theories, Nucl. Phys. B 424 (1994) 71
   [hep-th/9402120].
- [128] C. Hull and B. J. Spence, *The Gauged Nonlinear*  $\sigma$  *Model With Wess-Zumino Term*, *Phys. Lett. B* **232** (1989) 204.
- [129] J. M. Figueroa-O'Farrill and N. Mohammedi, Gauging the Wess-Zumino term of a sigma model with boundary, [HEP 08 (2005) 086 [hep-th/0506049].
- [130] T. Quella and V. Schomerus, Asymmetric cosets, JHEP 02 (2003) 030 [hep-th/0212119].
- [131] D. Kutasov, Introduction to little string theory, ICTP Lect. Notes Ser. 7 (2002) 165.
- [132] O. Lunin, Adding momentum to D1-D5 system, JHEP 04 (2004) 054 [hep-th/0404006].
- [133] I. Bena, S. Giusto, R. Russo, M. Shigemori and N. P. Warner, *Habemus Superstratum! A constructive proof of the existence of superstrata, JHEP* 05 (2015) 110
   [1503.01463].
- [134] I. Bena, E. Martinec, D. Turton and N. P. Warner, *Momentum Fractionation on Superstrata*, JHEP 05 (2016) 064 [1601.05805].
- [135] I. Bena, S. Giusto, E. J. Martinec, R. Russo, M. Shigemori, D. Turton et al., Smooth horizonless geometries deep inside the black-hole regime, Phys. Rev. Lett. 117 (2016) 201601 [1607.03908].
- [136] I. Bena, P. Heidmann and D. Turton, AdS<sub>2</sub> holography: mind the cap, JHEP 12 (2018) 028 [1806.02834].
- [137] N. Ceplak, R. Russo and M. Shigemori, Supercharging Superstrata, JHEP 03 (2019) 095 [1812.08761].
- [138] P. Heidmann and N. P. Warner, Superstratum Symbiosis, JHEP 09 (2019) 059 [1903.07631].
- [139] S. Giusto, S. Rawash and D. Turton, *Ads*<sub>3</sub> *holography at dimension two*, *JHEP* 07 (2019) 171 [1904.12880].
- [140] B. D. Chowdhury and S. D. Mathur, Radiation from the non-extremal fuzzball, Class. Quant. Grav. 25 (2008) 135005 [0711.4817].

- [141] S. G. Avery and B. D. Chowdhury, Emission from the D1D5 CFT: Higher Twists, JHEP 01 (2010) 087 [0907.1663].
- [142] T. D. Brennan and E. J. Martinec, Wrapped fivebranes redux, 2020.
- [143] A. Giveon, N. Itzhaki and D. Kutasov, TT and LST, JHEP 07 (2017) 122 [1701.05576].
- [144] M. Asrat, A. Giveon, N. Itzhaki and D. Kutasov, *Holography Beyond AdS*, *Nucl. Phys. B* 932 (2018) 241 [1711.02690].
- [145] A. Giveon, N. Itzhaki and D. Kutasov, A solvable irrelevant deformation of AdS<sub>3</sub>/CFT<sub>2</sub>, JHEP 12 (2017) 155 [1707.05800].
- [146] S.-w. Chung and S. H. H. Tye, Chiral gauged WZW theories and coset models in conformal field theory, Phys. Rev. D 47 (1993) 4546 [hep-th/9202002].
- [147] V. Cardoso, O. J. C. Dias, J. L. Hovdebo and R. C. Myers, Instability of non-supersymmetric smooth geometries, Phys. Rev. D 73 (2006) 064031 [hep-th/0512277].
- [148] K. Gawedzki, Noncompact WZW conformal field theories, in NATO Advanced Study Institute: New Symmetry Principles in Quantum Field Theory, 10, 1991, hep-th/9110076.
- [149] C. M. Hull, Global aspects of T-duality, gauged sigma models and T-folds, JHEP 10 (2007) 057 [hep-th/0604178].
- [150] C. M. Hull, Doubled Geometry and T-Folds, JHEP 07 (2007) 080 [hep-th/0605149].
- [151] K. Gawedzki, R. R. Suszek and K. Waldorf, Global Gauge Anomalies in Two-Dimensional Bosonic Sigma Models, Commun. Math. Phys. 302 (2011) 513 [1003.4154].
- [152] P. de Fromont, K. Gawedzki and C. Tauber, Global gauge anomalies in coset models of conformal field theory, Commun. Math. Phys. 328 (2014) 1371 [1301.2517].
- [153] J. M. Izquierdo and P. K. Townsend, *Supersymmetric spacetimes in* (2+1) *ads-supergravity models, Classical and Quantum Gravity* **12** (1995) 895–924.
- [154] G. Mandal, S. Raju and M. Smedbäck, Supersymmetric giant graviton solutions inads3, Physical Review D 77 (2008).
- [155] S. Chakraborty, A. Giveon and D. Kutasov,  $J\overline{T}$  deformed  $CFT_2$  and string theory, *JHEP* **10** (2018) 057 [1806.09667].
- [156] S. Chakraborty, A. Giveon and D. Kutasov, *TT̄*, *JT̄*, *TJ̄* and String Theory, J. Phys. A 52 (2019) 384003 [1905.00051].

- [157] E. G. Gimon, T. S. Levi and S. F. Ross, *Geometry of non-supersymmetric three-charge bound states*, *JHEP* **08** (2007) 055 [0705.1238].
- [158] T. H. Buscher, A Symmetry of the String Background Field Equations, Phys. Lett. B 194 (1987) 59.
- [159] S. Chakraborty,  $\frac{SL(2,\mathbb{R})\times U(1)}{U(1)}$  CFT, NS5+F1 system and single trace  $T\overline{T}$ , JHEP 03 (2021) 113 [2012.03995].
- [160] I. Bena, G. Bossard, S. Katmadas and D. Turton, Non-BPS multi-bubble microstate geometries, JHEP 02 (2016) 073 [1511.03669].
- [161] I. Bena, G. Bossard, S. Katmadas and D. Turton, *Bolting Multicenter Solutions*, *JHEP* 01 (2017) 127 [1611.03500].
- [162] G. Bossard, S. Katmadas and D. Turton, *Two Kissing Bolts*, *JHEP* 02 (2018) 008 [1711.04784].
- [163] B. Ganchev, A. Houppe and N. P. Warner, *Elliptical and purely NS superstrata*, *JHEP* 09 (2022) 067 [2207.04060].
- [164] I. Kanitscheider, K. Skenderis and M. Taylor, *Holographic anatomy of fuzzballs*, *JHEP* 04 (2007) 023 [hep-th/0611171].
- [165] S. Giusto, E. Moscato and R. Russo, AdS<sub>3</sub> holography for 1/4 and 1/8 BPS geometries, JHEP 11 (2015) 004 [1507.00945].
- [166] S. Rawash and D. Turton, *Supercharged AdS*<sub>3</sub> *Holography*, *JHEP* 07 (2021) 178
   [2105.13046].
- [167] A. Galliani, S. Giusto, E. Moscato and R. Russo, Correlators at large c without information loss, JHEP 09 (2016) 065 [1606.01119].
- [168] A. L. Fitzpatrick and J. Kaplan, *On the Late-Time Behavior of Virasoro Blocks and a Classification of Semiclassical Saddles, JHEP* **04** (2017) 072 [1609.07153].
- [169] V. Balasubramanian, A. Bernamonti, B. Craps, T. De Jonckheere and F. Galli, *Heavy-Heavy-Light-Light correlators in Liouville theory*, *JHEP* 08 (2017) 045 [1705.08004].
- [170] A. Galliani, S. Giusto and R. Russo, *Holographic 4-point correlators with heavy states*, *JHEP* **10** (2017) 040 [1705.09250].
- [171] A. Bombini and A. Galliani,  $AdS_3$  four-point functions from  $\frac{1}{8}$  -BPS states, JHEP **06** (2019) 044 [1904.02656].
- [172] R. Argurio, A. Giveon and A. Shomer, Superstrings on AdS(3) and symmetric products, JHEP 12 (2000) 003 [hep-th/0009242].

- [173] L. Eberhardt and M. R. Gaberdiel, *String theory on AdS*<sub>3</sub> *and the symmetric orbifold of Liouville theory*, *Nucl. Phys.* **B948** (2019) 114774 [1903.00421].
- [174] B. Balthazar, A. Giveon, D. Kutasov and E. J. Martinec, *Asymptotically free AdS<sub>3</sub>/CFT<sub>2</sub>*, *JHEP* 01 (2022) 008 [2109.00065].
- [175] V. Fateev, A. Zamolodchikov and A. Zamolodchikov, Unpublished notes, .
- [176] G. Giribet and C. A. Nunez, Correlators in AdS(3) string theory, JHEP 06 (2001) 010 [hep-th/0105200].
- [177] C. A. Cardona and I. Kirsch, Worldsheet four-point functions in AdS<sub>3</sub>/CFT<sub>2</sub>, JHEP 01 (2011) 015 [1007.2720].
- [178] Y. Cagnacci and S. M. Iguri, *More AdS*<sub>3</sub> correlators, *Phys. Rev. D* 89 (2014) 066006
   [1312.3353].
- [179] A. Dei and L. Eberhardt, String correlators on AdS<sub>3</sub>: four-point functions, JHEP 09 (2021) 209 [2107.01481].
- [180] O. Lunin and S. D. Mathur, Correlation functions for M\*\*N / S(N) orbifolds, Commun. Math. Phys. 219 (2001) 399 [hep-th/0006196].
- [181] A. A. Lima, G. M. Sotkov and M. Stanishkov, On the dynamics of protected ramond ground states in the D1-D5 CFT, JHEP 07 (2021) 120 [2103.04459].
- [182] A. Alves Lima, G. M. Sotkov and M. Stanishkov, Four-point functions with multi-cycle fields in symmetric orbifolds and the D1-D5 CFT, JHEP 05 (2022) 106 [2202.12424].
- [183] A. A. Lima, G. M. Sotkov and M. Stanishkov, Correlation functions of composite Ramond fields in deformed D1-D5 orbifold SCFT<sub>2</sub>, Phys. Rev. D 102 (2020) 106004 [2006.16303].
- [184] M. Baggio, J. de Boer and K. Papadodimas, A non-renormalization theorem for chiral primary 3-point functions, JHEP 07 (2012) 137 [1203.1036].
- [185] B. D. Chowdhury and S. D. Mathur, *Pair creation in non-extremal fuzzball geometries*, *Class. Quant. Grav.* **25** (2008) 225021 [0806.2309].
- [186] B. Guo, M. R. R. Hughes, S. D. Mathur and M. Mehta, Universal lifting in the D1-D5 CFT, JHEP 10 (2022) 148 [2208.07409].
- [187] M. R. Gaberdiel and R. Gopakumar, *Tensionless string spectra on AdS*<sub>3</sub>, *JHEP* 05 (2018) 085 [1803.04423].
- [188] M. R. Gaberdiel and B. Nairz, BPS correlators for AdS<sub>3</sub>/CFT<sub>2</sub>, JHEP 09 (2022) 244 [2207.03956].

- [189] G. Giribet, TT-deformations, AdS/CFT and correlation functions, JHEP 02 (2018) 114 [1711.02716].
- [190] I. Bena and D. R. Mayerson, Multipole Ratios: A New Window into Black Holes, Phys. Rev. Lett. 125 (2020) 221602 [2006.10750].
- [191] M. Bianchi, D. Consoli, A. Grillo, J. F. Morales, P. Pani and G. Raposo, Distinguishing fuzzballs from black holes through their multipolar structure, Phys. Rev. Lett. 125 (2020) 221601 [2007.01743].
- [192] M. Bianchi, M. Casolino and G. Rizzo, Accelerating strangelets via Penrose process in non-BPS fuzzballs, Nucl. Phys. B 954 (2020) 115010 [1904.01097].
- [193] E. J. Martinec, S. Massai and D. Turton, On the BPS sector in AdS\_3/CFT\_2 Holography, 2211.12476.
- [194] G. Giribet, C. Hull, M. Kleban, M. Porrati and E. Rabinovici, *Superstrings on*  $AdS_3 at \parallel = 1$ , *JHEP* **08** (2018) 204 [1803.04420].
- [195] S. Iguri, N. Kovensky and J. H. Toro, *Spectral flow and string correlators in*  $AdS_3 \times S^3 \times T^4$ , 2211.02521.
- [196] A. B. Zamolodchikov and A. B. Zamolodchikov, Structure constants and conformal bootstrap in Liouville field theory, Nucl. Phys. B 477 (1996) 577 [hep-th/9506136].
- [197] P. Minces, C. A. Nunez and E. Herscovich, Winding strings in AdS(3), JHEP 06 (2006) 047 [hep-th/0512196].
- [198] Y. Cagnacci and S. M. Iguri, Spectral flow and conformal blocks in AdS<sub>3</sub>, JHEP 09 (2015) 036 [1504.07492].